Commission Regulation (EC) No 794/2004 of 21 April 2004 implementing Council Regulation (EU) 2015/1589 laying down detailed rules for the application of Article 108 of the Treaty on the Functioning of the European Union

[F1COMMISSION REGULATION (EC) No 794/2004

of 21 April 2004

implementing Council Regulation (EU) 2015/1589 laying down detailed rules for the application of Article 108 of the Treaty on the Functioning of the European Union

THE COMMISSION OF THE EUROPEAN COMMUNITIES,

Having regard to the Treaty establishing the European Community,

Having regard to Council Regulation (EC) No 659/1999 of 22 March 1999 laying down detailed rules for the application of Article 93 of the EC Treaty⁽¹⁾, and in particular Article 27 thereof,

After consulting the Advisory Committee on State Aid,

Whereas:

- (1) In order to facilitate the preparation of State aid notifications by Member States, and their assessment by the Commission, it is desirable to establish a compulsory notification form. That form should be as comprehensive as possible.
- (2) The standard notification form as well as the summary information sheet and the supplementary information sheets should cover all existing guidelines and frameworks in the state aid field. They should be subject to modification or replacement in accordance with the further development of those texts.
- (3) Provision should be made for a simplified system of notification for certain alterations to existing aid. Such simplified arrangements should only be accepted if the Commission has been regularly informed on the implementation of the existing aid concerned.
- (4) In the interests of legal certainty it is appropriate to make it clear that small increases of up to 20 % of the original budget of an aid scheme, in particular to take account of the effects of inflation, should not need to be notified to the Commission as they are unlikely to affect the Commission's original assessment of the compatibility of the scheme, provided that the other conditions of the aid scheme remain unchanged.
- (5) Article 21 of Regulation (EC) No 659/1999 requires Member States to submit annual reports to the Commission on all existing aid schemes or individual aid granted outside an approved aid scheme in respect of which no specific reporting obligations have been imposed in a conditional decision.
- (6) For the Commission to be able to discharge its responsibilities for the monitoring of aid, it needs to receive accurate information from Member States about the types and amounts of aid being granted by them under existing aid schemes. It is possible to simplify and improve the arrangements for the reporting of State aid to the Commission which are currently described in the joint procedure for reporting and notification under

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

the EC Treaty and under the World Trade Organisation (WTO) Agreement set out in the Commission's letter to Member States of 2 August 1995. The part of that joint procedure relating to Member States reporting obligations for subsidy notifications under Article 25 of the WTO Agreement on Subsidies and Countervailing measures and under Article XVI of GATT 1994, adopted on 21 July 1995 is not covered by this Regulation.

- (7) The information required in the annual reports is intended to enable the Commission to monitor overall aid levels and to form a general view of the effects of different types of aid on competition. To this end, the Commission may also request Member States to provide, on an ad hoc basis, additional data for selected topics. The choice of subject matter should be discussed in advance with Member States.
- (8) The annual reporting exercise does not cover the information, which may be necessary in order to verify that particular aid measures respect Community law. The Commission should therefore retain the right to seek undertakings from Member States, or to attach to decisions conditions requiring the provision of additional information.
- (9) It should be specified that time-limits for the purposes of Regulation (EC) No 659/1999 should be calculated in accordance with Regulation (EEC, Euratom) No 1182/71 of the Council of 3 June 1971 determining the rules applicable to periods, dates and time limits⁽²⁾, as supplemented by the specific rules set out in this Regulation. In particular, it is necessary to identify the events, which determine the starting point for time-limits applicable in State aid procedures. The rules set out in this Regulation should apply to pre-existing time-limits which will continue to run after the entry into force of this Regulation.
- (10) The purpose of recovery is to re-establish the situation existing before aid was unlawfully granted. To ensure equal treatment, the advantage should be measured objectively from the moment when the aid is available to the beneficiary undertaking, independently of the outcome of any commercial decisions subsequently made by that undertaking.
- (11) In accordance with general financial practice it is appropriate to fix the recovery interest rate as an annual percentage rate.
- (12) The volume and frequency of transactions between banks results in an interest rate that is consistently measurable and statistically significant, and should therefore form the basis of the recovery interest rate. The inter-bank swap rate should, however, be adjusted in order to reflect general levels of increased commercial risk outside the banking sector. On the basis of the information on inter-bank swap rates the Commission should establish a single recovery interest rate for each Member State. In the interest of legal certainty and equal treatment, it is appropriate to fix the precise method by which the interest rate should be calculated, and to provide for the publication of the recovery interest rate applicable at any given moment, as well as relevant previously applicable rates.
- (13) A State aid grant may be deemed to reduce a beneficiary undertaking's medium-term financing requirements. For these purposes, and in line with general financial practice,

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- the medium-term may be defined as five years. The recovery interest rate should therefore correspond to an annual percentage rate fixed for five years.
- (14) Given the objective of restoring the situation existing before the aid was unlawfully granted, and in accordance with general financial practice, the recovery interest rate to be fixed by the Commission should be annually compounded. For the same reasons, the recovery interest rate applicable in the first year of the recovery period should be applied for the first five years of the recovery period, and the recovery interest rate applicable in the sixth year of the recovery period for the following five years.
- (15) This Regulation should apply to recovery decisions notified after the date of entry into force of this Regulation,

HAS ADOPTED THIS REGULATION:

Textual Amendments

F1 Substituted by Commission Regulation (EU) 2015/2282 of 27 November 2015 amending Regulation (EC) No 794/2004 as regards the notification forms and information sheets (Text with EEA relevance).

CHAPTER I

SUBJECT MATTER AND SCOPE

Article 1

Subject matter and scope

- This Regulation sets out detailed provisions concerning the form, content and other details of notifications and annual reports referred to in Regulation (EC) No 659/1999. It also sets out provisions for the calculation of time limits in all procedures concerning State aid and of the interest rate for the recovery of unlawful aid.
- 2 This Regulation shall apply to aid in all sectors.

CHAPTER II

NOTIFICATIONS

Article 2

Notification forms

Without prejudice to Member States' obligations to notify state aids in the coal sector under Commission Decision 2002/871/CE⁽³⁾, notifications of new aid pursuant to Article 2(1) of Regulation (EC) No 659/1999, other than those referred to in Article 4(2), shall be made on the notification form set out in Part I of Annex I to this Regulation.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Supplementary information needed for the assessment of the measure in accordance with regulations, guidelines, frameworks and other texts applicable to State aid shall be provided on the supplementary information sheets set out in Part III of Annex I.

Whenever the relevant guidelines or frameworks are modified or replaced, the Commission shall adapt the corresponding forms and information sheets.

I^{F2}Article 3

Transmission of notifications

- 1 The notification shall be transmitted to the Commission by means of the electronic validation carried out by the person designated by the Member State. Such validated notification shall be considered as sent by the Permanent Representative.
- The Commission shall address its correspondence to the Permanent Representative of the Member State concerned, or to any other address designated by that Member State.
- 3 As from 1 July 2008, notifications shall be transmitted electronically via the web application State Aid Notification Interactive (SANI).

All correspondence in connection with a notification shall be transmitted electronically via the secured e-mail system Public Key Infrastructure (PKI).

4 In exceptional circumstances and upon the agreement of the Commission and the Member State concerned, an agreed communication channel other than those referred to in paragraph 3 may be used for submission of a notification or any correspondence in connection with a notification.

In the absence of such an agreement, any notification or correspondence in connection with a notification sent to the Commission by a Member State through a communication channel other than those referred to in paragraph 3 shall not be considered as submitted to the Commission.

- Where the notification or correspondence in connection with a notification contains confidential information, the Member State concerned shall clearly identify such information and give reasons for its classification as confidential.
- The Member States shall refer to the State aid identification number allocated to an aid scheme by the Commission in each grant of aid to a final beneficiary.

The first subparagraph shall not apply to aid granted through fiscal measures.]

Textual Amendments

F2 Substituted by Commission Regulation (EC) No 271/2008 of 30 January 2008 amending Regulation (EC) No 794/2004 implementing Council Regulation (EC) No 659/1999 laying down detailed rules for the application of Article 93 of the EC Treaty.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Article 4

Simplified notification procedure for certain alterations to existing aid

- For the purposes of Article 1(c) of Regulation (EC) No 659/1999, an alteration to existing aid shall mean any change, other than modifications of a purely formal or administrative nature which cannot affect the evaluation of the compatibility of the aid measure with the common market. However an increase in the original budget of an existing aid scheme by up to 20 % shall not be considered an alteration to existing aid.
- 2 The following alterations to existing aid shall be notified on the simplified notification form set out in Annex II:
 - a increases in the budget of an authorised aid scheme exceeding 20 %;
 - b prolongation of an existing authorised aid scheme by up to six years, with or without an increase in the budget;
 - c tightening of the criteria for the application of an authorised aid scheme, a reduction of aid intensity or a reduction of eligible expenses;

The Commission shall use its best endeavours to take a decision on any aid notified on the simplified notification form within a period of one month.

3 The simplified notification procedure shall not be used to notify alterations to aid schemes in respect of which Member States have not submitted annual reports in accordance with Article 5, 6, and 7, unless the annual reports for the years in which the aid has been granted are submitted at the same time as the notification.

CHAPTER III

ANNUAL REPORTS

Article 5

Form and content of annual reports

Without prejudice to the second and third subparagraphs of this Article and to any additional specific reporting requirements laid down in a conditional decision adopted pursuant to Article 7(4) of Regulation (EC) No 659/1999, or to the observance of any undertakings provided by the Member State concerned in connection with a decision to approve aid, Member States shall compile the annual reports on existing aid schemes referred to in Article 21(1) of Regulation (EC) No 659/1999 in respect of each whole or part calendar year during which the scheme applies in accordance with the standardised reporting format set out in Annex IIIA.

Annex IIIB sets out the format for annual reports on existing aid schemes relating to the production, processing and marketing of agricultural products listed in Annex I of the Treaty.

Annex IIIC sets out the format for annual reports on existing aid schemes for state aid relating to the production, processing or marketing of fisheries products listed in Annex I of the Treaty.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2 The Commission may ask Member States to provide additional data for selected topics, to be discussed in advance with Member States.

Article 6

Transmission and publication of annual reports

1 Each Member State shall transmit its annual reports to the Commission in electronic form no later than 30 June of the year following the year to which the report relates.

In justified cases Member States may submit estimates, provided that the actual figures are transmitted at the very latest with the following year's data.

2 Each year the Commission shall publish a State aid synopsis containing a synthesis of the information contained in the annual reports submitted during the previous year.

Article 7

Status of annual reports

The transmission of annual reports shall not be considered to constitute compliance with the obligation to notify aid measures before they are put into effect pursuant to Article 88(3) of the Treaty, nor shall such transmission in any way prejudice the outcome of an investigation into allegedly unlawful aid in accordance with the procedure laid down in Chapter III of Regulation (EC) No 659/1999.

CHAPTER IV

TIME-LIMITS

Article 8

Calculation of time-limits

- [F31 Time-limits provided for in Regulation (EC) No 659/1999 and in this Regulation or fixed by the Commission pursuant to Article 108 of the Treaty shall be calculated in accordance with Regulation (EEC, Euratom) No 1182/71, and the specific rules set out in paragraphs 2 to 5b of this Article. In case of conflict, the provisions of this Regulation shall prevail.]
- 2 Time limits shall be specified in months or in working days.
- [F23] With regard to timelimits for action by the Commission, the receipt of the notification or subsequent correspondence in accordance with Article 3(1) and Article 3(3) of this Regulation shall be the relevant event for the purpose of Article 3(1) of Regulation (EEC, Euratom) No 1182/71.
- With regard to timelimits for action by Member States, the receipt of the relevant notification or correspondence from the Commission in accordance with Article 3(2) of this Regulation shall be the relevant event for the purposes of Article 3(1) of Regulation (EEC, Euratom) No 1182/71.]

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- With regard to the time-limit for the submission of comments following initiation of the formal investigation procedure referred to in Art. 6(1) of Regulation (EC) No 659/1999 by third parties and those Member States which are not directly concerned by the procedure, the publication of the notice of initiation in the *Official Journal of the European Union* shall be the relevant event for the purposes of Article 3(1) of Regulation (EEC, Euratom) No 1182/71.
- [F45a With regard to the time-limit for the submission of the information requested from third parties pursuant to Article 6a(6) of Regulation (EC) No 659/1999, the receipt of the request for information shall be the relevant event for the purposes of Article 3(1) of Regulation (EEC, Euratom) No 1182/71.
- With regard to the time-limit for the submission of the information requested from third parties pursuant to Article 6a(7) of Regulation (EC) No 659/1999, the notification of the decision shall be the relevant event for the purposes of Article 3(1) of Regulation (EEC, Euratom) No 1182/71.]
- Any request for the extension of a time-limit shall be duly substantiated, and shall be submitted in writing to the address designated by the party fixing the time-limit at least two working days before expiry.

Textual Amendments

- **F2** Substituted by Commission Regulation (EC) No 271/2008 of 30 January 2008 amending Regulation (EC) No 794/2004 implementing Council Regulation (EC) No 659/1999 laying down detailed rules for the application of Article 93 of the EC Treaty.
- **F3** Substituted by Commission Regulation (EU) No 372/2014 of 9 April 2014 amending Regulation (EC) No 794/2004 as regards the calculation of certain time limits, the handling of complaints, and the identification and protection of confidential information (Text with EEA relevance).
- **F4** Inserted by Commission Regulation (EU) No 372/2014 of 9 April 2014 amending Regulation (EC) No 794/2004 as regards the calculation of certain time limits, the handling of complaints, and the identification and protection of confidential information (Text with EEA relevance).

CHAPTER V

INTEREST RATE FOR THE RECOVERY OF UNLAWFUL AID

I^{F2}Article 9

Method for fixing the interest rate

- 1 Unless otherwise provided for in a specific decision, the interest rate to be used for recovering State aid granted in breach of Article 88(3) of the Treaty shall be an annual percentage rate which is fixed by the Commission in advance of each calendar year.
- The interest rate shall be calculated by adding 100 basis points to the one-year money market rate. Where those rates are not available, the three-month money market rate will be used, or in the absence thereof, the yield on State bonds will be used.
- In the absence of reliable money market or yield on stock bonds or equivalent data or in exceptional circumstances the Commission may, in close co-operation with the Member State(s) concerned, fix a recovery rate on the basis of a different method and on the basis of the information available to it.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 4 The recovery rate will be revised once a year. The base rate will be calculated on the basis of the one-year money market recorded in September, October and November of the year in question. The rate thus calculated will apply throughout the following year.
- In addition, to take account of significant and sudden variations, an update will be made each time the average rate, calculated over the three previous months, deviates more than 15 % from the rate in force. This new rate will enter into force on the first day of the second month following the months used for the calculation.]

Textual Amendments

F2 Substituted by Commission Regulation (EC) No 271/2008 of 30 January 2008 amending Regulation (EC) No 794/2004 implementing Council Regulation (EC) No 659/1999 laying down detailed rules for the application of Article 93 of the EC Treaty.

Article 10

Publication

The Commission shall publish current and relevant historical State aid recovery interest rates in the *Official Journal of the European Union* and for information on the Internet.

Article 11

Method for applying interest

- 1 The interest rate to be applied shall be the rate applicable on the date on which unlawful aid was first put at the disposal of the beneficiary.
- The interest rate shall be applied on a compound basis until the date of the recovery of the aid. The interest accruing in the previous year shall be subject to interest in each subsequent year.
- [F23] The interest rate referred to in paragraph 1 shall be applied throughout the whole period until the date of recovery. However, if more than one year has elapsed between the date on which the unlawful aid was first put at the disposal of the beneficiary and the date of the recovery of the aid, the interest rate shall be recalculated at yearly intervals, taking as a basis the rate in force at the time of recalculation.]

Textual Amendments

F2 Substituted by Commission Regulation (EC) No 271/2008 of 30 January 2008 amending Regulation (EC) No 794/2004 implementing Council Regulation (EC) No 659/1999 laying down detailed rules for the application of Article 93 of the EC Treaty.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

[F4CHAPTER Va

HANDLING OF COMPLAINTS

Article 11a

Admissibility of complaints

- Any person submitting a complaint pursuant to Articles 10(1) and 20(2) of Regulation (EC) No 659/1999 shall demonstrate that it is an interested party within the meaning of Article 1(h) of that Regulation.
- Interested parties shall duly complete the form set out in Annex IV and provide all the mandatory information requested therein. On a reasoned request by an interested party, the Commission may dispense with the obligation to provide some of the information required by the form.
- 3 Complaints shall be submitted in one of the official languages of the Union.

CHAPTER Vb

IDENTIFICATION AND PROTECTION OF CONFIDENTIAL INFORMATION

Article 11h

Protection of business secrets and other confidential information

Any person submitting information pursuant to Regulation (EC) No 659/1999 shall clearly indicate which information it considers to be confidential, stating the reasons for such confidentiality, and provide the Commission with a separate non-confidential version of the submission. When information must be provided by a certain deadline, the same deadline shall apply for providing the non-confidential version.]

CHAPTER VI

FINAL PROVISIONS

Article 12

Review

The Commission shall in consultation with the Member States, review the application of this Regulation within four years after its entry into force.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Article 13

Entry into force

This Regulation shall enter into force on the twentieth day following that of its publication in the *Official Journal of the European Union*.

Chapter II shall apply only to those notifications transmitted to the Commission more than five months after the entry into force of this Regulation.

Chapter III shall apply to annual reports covering aid granted from 1 January 2003 onwards.

Chapter IV shall apply to any time limit, which has been fixed but which has not yet expired on the date of entry into force of this Regulation.

Articles 9 and 11 shall apply in relation to any recovery decision notified after the date of entry into force of this Regulation.

This Regulation shall be binding in its entirety and be directly applicable in all Member States.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

ANNEX I

STANDARD FORM FOR NOTIFICATION OF STATE AIDS PURSUANT TO ARTICLE 88 (3) EC TREATY AND FOR THE PROVISION OF INFORMATION ON UNLAWFUL AID

This form shall be used by Member States for the notification pursuant to Article 88(3) EC Treaty of new aid schemes and individual aid It shall also be used when a non-aid measure is notified to the Commission for reasons of legal certainty.

Member States are also requested to use this form when the Commission requests comprehensive information on alleged unlawful aid.

The present form consists of three parts:

- I. **General Information:** to be completed in all cases
- II. Summary Information for publication in the Official Journal
- III. Supplementary Information Sheet depending on the type of aid

Please note that failure to complete this form correctly may result in the notification being returned as incomplete. The completed form shall be transmitted on paper to the Commission by the Permanent Representative of the Member State concerned. It shall be addressed to the Secretary General of the Commission.

If the Member State intends to avail itself of a specific procedure laid down in any regulations, guidelines, frameworks and other texts applicable to State aid, a copy of the notification shall be as well addressed to the Director General of the Commission department responsible.

IF1PART I

GENERAL INFORMATION

1. Status of the notification

Does the information transmitted on this form concern:

- (a) a pre-notification? If so, you may not need to complete the entire form at this stage but to agree with the Commission services which information is required for a preliminary assessment of the proposed measure.
- (b)□ a notification pursuant to Article 108(3) of the Treaty on the Functioning of the European Union (TFEU)?
- (c) a simplified notification under Article 4(2) of Regulation (EC) No 794/2004⁽⁴⁾? If so, please complete only the Simplified Notification Form in Annex II.
- (d) a measure which does not constitute State aid within the meaning of Article 107(1) TFEU but is notified to the Commission for reasons of legal certainty?

If you have selected point (d) above, please indicate below why the notifying Member State considers that the measure does not constitute State aid within the meaning of Article 107(1) TFEU. Please provide a full assessment of the measure in light of each of following four criteria, stressing in particular the criteria that you consider not to be met in the planned measure:

Does the notified measure imply a transfer of public resources or is it imputable to the State?

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Does the notified measure confer an advantage upon undertakings?

Is the measure discretionary, available only to a limited number of undertakings, in a limited number of sectors of the economy or does it entail any territorial restrictions?

Does the measure affect competition on the internal market or threaten to distort intra-Union trade?

2. Identification of the aid grantor

Member State concerned:

Region(s) of the Member State concerned (at NUTS level 2); include information on their regional aid status:

Contact person(s):
Name:
Address:
Phone(s):

Please indicate the name, the address (including web address) and the e-mail contact of the granting authority:

Name: Address: Web address:

E-mail:

E-mail(s):

Contact person at the Permanent Representation

Name: Phone(s): E-mail:

If you would like a copy of the official correspondence sent by the Commission to the Member State to be forwarded to other national authorities, please indicate here their name, address (including their web address) and e-mail contact:

Name:

Address:

Web address:

E-mail:

3. Beneficiaries

- 3.1. Location of the beneficiary(ies)
- (a) \Box in (an) unassisted region(s):
- (b)□ in region(s) eligible for assistance under Article 107(3)(a) TFEU (specify the region(s) at NUTS level 2):
- in region(s) eligible for assistance under Article 107(3)(c) TFEU (specify the region(s) at NUTS level 3 or lower):
- 3.2. If applicable, location of the project(s)

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (a) \Box in (an) unassisted region(s):
- (b)□ in region(s) eligible for assistance under Article 107(3)(a) TFEU (specify the region(s) at NUTS level 2):
- in region(s) eligible for assistance under Article 107(3)(c) TFEU (specify the region(s) at NUTS level 3 or lower):
- 3.3. Sector(s) affected by the aid measure (i.e. in which the aid beneficiaries are active):
- (a)□ Open to all sectors
- (b) Sector specific. If so, please specify the sector(s) at NACE group level⁽⁵⁾:
- 3.4. In the case of an aid scheme, please specify:
- 3.4.1. Type of beneficiaries:
- (a)□ large enterprises
- (b)□ small and medium-sized enterprises (SMEs)
- (c)□ medium-sized enterprises
- (d)□ small enterprises
- (e)□ micro enterprises
- 3.4.2. Estimated number of beneficiaries:
- (a) \square under 10
- (b)□ from 11 to 50
- (c) \square from 51 to 100
- (d)□ from 101 to 500
- (e) \Box from 501 to 1 000
- (f)□ over 1 000
- 3.5. In the case of individual aid, either granted within the scope of a scheme or as ad hoc aid, please specify:
- 3.5.1. Name of the beneficiary(ies):
- 3.5.2. Type of beneficiary(ies):
- □ SME

Number of employees:

Annual turnover (full amount in national currency, in the last financial year):

Annual balance-sheet total (full amount in national currency, in the last financial year):

Existence of linked enterprise or partner enterprises (please attach a declaration according to Article 3(5) of the Commission Recommendation on SMEs⁽⁶⁾ attesting to either the autonomous, linked or partner status of the beneficiary undertaking⁽⁷⁾):

□ Large enterprise

Status: Point in time view as at 31/01/2020. Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

3.6.	Is the beneficiary(ies) an undertaking in difficulty ⁽⁸⁾ ?		
□ yes		□ no	
3.7.	Outstanding recovery orders		
3.7.1.	In the case of individual aid:		
aid if th by a Co incomp	e beneficiary still has at its disposal ear ommission Decision (either as individua	suspend the award and/or payment of the notified lier unlawful aid that was declared incompatible all aid or aid under an aid scheme being declared bursed or paid into a blocked account the total he corresponding recovery interest.	
□ yes		□ no	
Please p	provide the reference to the national leg	al basis concerning this point:	
3.7.2.	In the case of aid schemes:		
under the declared an aid so a block	he notified aid scheme to any undertak d incompatible by a Commission Decis cheme being declared incompatible), un	o suspend the award and/or payment of any aid ing that has benefited from earlier unlawful aid sion (either as an individual aid or an aid under ntil that undertaking has reimbursed or paid into ful and incompatible aid and the corresponding	
□ yes		□ no	
Please p	provide the reference to the national leg	al basis concerning this point:	
4.	National Legal Basis		
4.1.	Please provide the national legal basis provisions and their respective source	s of the aid measure including the implementing es:	
Nationa	al legal basis:		
Implem	enting provisions (where applicable):		

References (where applicable):

- 4.2. Please enclose with this notification one of the following:
- a copy of the relevant extracts of the final text(s) of the legal basis (together with a (a)□ web address which provides direct access to it, if available)
- a copy of the relevant extracts of the draft text(s) of the legal basis (together with a (b)□ web address which provides direct access, if available)
- 4.3. In case of a final text, does the final text contain a stand-still clause whereby the aid granting body can only grant the aid after the Commission has authorised the aid?
- Yes

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	No: has	s a provisi	on been in	cluded to that effect in the draft text?
		Yes		
		No: ple legal ba		n why such a provision was not included in the text of the
4.4.				basis contains a stand-still clause, please indicate whether aid will be:
	the date	e of the ap	proval by	the Commission
		e of the co al of the C		t of the national authorities to grant the aid, subject to the
5.	Identif	ication of	the Aid,	Objective and Duration
5.1.	Title of	f the aid m	easure (or	name of the beneficiary of the individual aid)
5.2.	Brief d	escription	of the obj	ective of the aid
5.3.	Does the measure concern the national co-financing of an European Fund for Strategic investment (EFSI) ⁽⁹⁾ project?			
	No			
	Yes: please attach the European Investment Bank application form to the notification form			
5.4.	Type o	f aid		
5.4.1.	Does the notification relate to an aid scheme?			
	No	No		
	Yes: De	oes the sch	neme amei	nd an existing aid scheme?
		No		
				litions laid down for the simplified notification procedure e 4(2) of Regulation (EC) No 794/2004 fulfilled?
			Yes: Ple Annex I	ase use and complete the Simplified Notification Form (see I).
				ntinue with this form, and specify whether the original which is being amended was notified to the Commission
				Yes: please specify:
				Aid number ⁽¹⁰⁾ :
				Date of Commission approval (reference of the letter of the Commission) if relevant or exemption number://;
				Duration of the original scheme:
				Please specify which conditions are being amended in relation to the original scheme and why:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

		□ No: please specify who	en the scheme was implemented:	
5.4.2.	Does the notification	on relate to individual aid (11)?		
	No			
	Yes: please indicate	e whether:		
	individua or to the e	Ily notified. Please provide the rexempted scheme: Title:	tempted scheme which should be reference to the approved scheme	
		Aid number ⁽¹⁰⁾ : Letter of Commission approval	(where applicable)	
		l aid is not based on a scheme	(mass uppromore).	
5.4.3.	Does the system of	financing form an integral part	of the aid measure (for example, ecessary funds to allow for the aid	
	No			
	Yes: if yes, the syst	em of financing should be notifi	ed as well.	
5.5.	Duration			
	Scheme			
the dur		please indicate why a longer per	y be granted under the scheme. It iod is indispensable to achieve the	
	Individual aid			
Indicat	e the planned date wh	en the aid will be granted ⁽¹²⁾ :		
If the a	id will be paid out in	instalments, indicate the planned	l date(s) of each instalment	
6.	Compatibility of t	he aid		
	nept _{Ctors} (13)	6.7 do not apply to aid to the ag	riculture, fishery and aquaculture	
6.1.		ne primary objective and, we man interest to which the aid co	here applicable, the secondary ontributes:	
		Primary objective(please tick only one)	Secondary objective ^a	
	culture; Forestry; l areas			
exa and	ample, a scheme for which the d medium-sized enterprises (S		ent may have as a secondary objective small SMEs. The secondary objective may also be	

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Broadband infrastructures	
Closure aid	
Compensation of damage caused by natural disasters or exceptional occurrences	
Culture	
Aid or disadvantaged workers and/or workers with disabilities	
Energy infrastructures	
Energy efficiency	
Environmental protection	
Execution of an important project of common European interest	
Fisheries and aquaculture	
Heritage conservation	
Promotion of export and internationalisation	
Regional development (including territorial cooperation)	
Remedy for a serious disturbance in the economy	
Renewable energy	
Rescuing undertakings in difficulty	
Research, development and innovation	
Restructuring undertakings in difficulty	
Risk finance	
Sectorial development	
Services of general economic interest (SGEI)	
SMEs	
Social support to individual consumers	

a A secondary objective is one for which, in addition to the primary objective, the aid will be exclusively earmarked. For example, a scheme for which the primary objective is research and development may have as a secondary objective small and medium-sized enterprises (SMEs) if the aid is earmarked exclusively for SMEs. The secondary objective may also be sectorial, in the case for example of a research and development scheme in the steel sector.

□ yes

Document Generated: 2024-06-28

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Sport and multifunctional recreational infrastructures		
Training		
Airport infrastructure or equipment		
Airport operation		
Start-up aid to airlines for the development of new routes		
Coordination of transport		
and medium-sized enterprises (SME sectorial, in the case for example of a feet towards a situation who delivered by the mark common interest as de	mary objective is research and developmen s) if the aid is earmarked exclusively for SN a research and development scheme in the standard development scheme	MES. The secondary objective may also be steel sector. Those that the aid must be targeted all improvement which cannot be efined market failure. The secondary objective may also be steel sector.
the behaviour of an u	er the aid has an incentive effect undertaking leading it to engage ed in without the aid or would or rent manner).	e in additional activity which is
□ yes	□ no	
Please indicate whether activition are eligible.	es which started before the subr	mission of an application for aic

If they are eligible, please explain how the incentive effect requirement is complied with.

6.5. Please indicate why the aid granted is proportionate insofar as it amounts to the minimum needed to induce investment or activity.

□ no

- Please indicate the possible negative effects of the aid on competition and trade and specify the extent to which they are outweighed by the positive effects.
- 6.7. In accordance with the Transparency Communication⁽¹⁴⁾, please indicate whether the following information will be published on a single national or regional website: the full text of the approved aid scheme or the individual aid granting decision and its implementing provisions, or a link to it; the identity of the granting authority/(ies); the identity of the individual beneficiary(ies), the aid instrument⁽¹⁵⁾ and amount of aid granted to each beneficiary(ies); the objective of the aid, the date of granting, the type

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

of undertaking (for example SME, large company); the Commission's aid measure reference number; the region where the beneficiary is located (at NUTS level 2) and the principal economic sector of the beneficiary(ies) (at NACE group level)⁽¹⁶⁾.

□ yes	□ no

- 6.7.1. Please provide the address(es) of the website(s) on which the information will be made available:
- 6.7.2. If applicable, please provide the address(es) of the central website retrieving information from the regional website(s):
- 6.7.3. If the address(es) of the website referred to in point 6.7.2 are not known at the time of the submission of the notification, the Member State must undertake to inform the Commission once those websites are created and the addresses are known.

7. Aid instrument, Aid amount, Aid intensity and Means of funding

7.1. Aid instrument and aid amount

Specify the form of the aid and the aid amount⁽¹⁷⁾ made available to the beneficiary(ies) (where appropriate, for each measure):

Aid instrument		Aid amount or budget allocation ^a		
		Overall	Annual	
	Grants (or similar in effect)			
(a)□	Direct grant			
(b)□ (c)□	Interest rate subsidy Debt write-off			
	Loans (or similar in effect)			
(a)□	Soft loan (including details of how the loan is secured and its duration)			
(b)□ (c)□	Repayable advances Tax deferral			
Guarantee Where appropriate, provide a reference to the Commission decision approving the methodology to calculate the gross grant equivalent and information on the loan or other financial transaction covered by the guarantee,				

a For information on aid amounts or budget at any chapter of this form and supplementary forms, give the full amount in national currency.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

the security required and the premium to be paid, the duration, etc.		
	Any form of equity or quasi-equity intervention	
	Tax advantage or tax exemption	
(a)□	Tax allowance	
(b)□	Tax base reduction	
(c)□	Tax rate reduction	
(d)□	Reduction of social security contributions	
(e)□	Other (please specify)	
	•••	
	Other (please specify)	
Please indicate the instruments to which it would broadly match as regards its effect		

a For information on aid amounts or budget at any chapter of this form and supplementary forms, give the full amount in national currency.

For guarantees, please indicate the maximum amount of loans guaranteed:

For loans, please indicate the maximum (nominal) amount of the loan granted:

7.2. Description of the aid instrument

For each aid instrument chosen from the list in point 7.1, please describe the conditions of application of the aid (such as the tax treatment, whether the aid is awarded automatically based on certain objective criteria or whether there is an element of discretion by the awarding authorities):

- 7.3. Source of funding
- 7.3.1. Specify the financing of the aid:
- (a)□ General budget of the State/region/local
- (b) Through parafiscal charges or taxes affected to a beneficiary. Please provide full details of the charges and the products/activities on which they are levied (specify in particular whether products imported from other Member States are liable to the charges). If applicable, please annex a copy of the legal basis of the financing.
- (c)□ Accumulated reserves

Public enterprises

(d)□

8.2.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

(e)□	Structural fund co-financing			
(f)□	Other (please specify)			
7.3.2.	Is the budget adopted annually?			
	Yes			
	No. Please specify what period it covers:			
7.3.3.	If the notification concerns changes to an existing scheme, please indicate the budgetary effects for each of the aid instruments of the notified changes to the scheme on the:			
Overall b	oudget			
Annual b	oudget ⁽¹⁸⁾			
7.4.	Cumulation			
	aid be cumulated with aid or <i>de minimis</i> aid ⁽¹⁹⁾ received from other local, regional or aid ⁽²⁰⁾ to cover the same eligible costs?			
	Yes. If available, please provide the name, purpose and objective of the aid			
	Please explain the mechanisms put in place in order to insure that the cumulation rules are respected:			
	No			
8.	Evaluation			
8.1.	Is the scheme considered for evaluation ⁽²¹⁾ ?			
	No			
	If the scheme is not considered for evaluation, please explain why you consider the criteria for evaluation not to be fulfilled.			
	Yes			
Accordin	ng to which criteria is the scheme considered for ex post evaluation:			
(a)□	A scheme with large aid budget;			
(b)□	A scheme containing novel characteristics;			
(c)□	$\label{thm:continuous} A \ scheme \ where \ significant \ market, \ technology \ or \ regulatory \ changes \ can \ be \ foreseen;$			
$(d)\Box$	A scheme that you plan for evaluation even if the other criteria referred to in this point do not apply.			
and comp	the criteria referred to in this point are fulfilled, please indicate the period of evaluation plete the supplementary information sheet for the notification of an evaluation plan in Part III.8 ⁽²²⁾ .			

Please indicate whether any ex-post evaluation has already been carried out for a

similar scheme (where relevant, with a reference and a link to any relevant websites)

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

9. **Reporting and Monitoring**

In order to enable the Commission to monitor the aid scheme and individual aid, the notifying Member State undertakes to:

- Annually submit to the Commission the reports provided for by Article 26 of Council Regulation (EU) 2015/1589⁽²³⁾.
- Maintain for at least 10 years from the date of award of the aid (individual aid and aid granted under the scheme) detailed records containing the information and supporting documentation necessary to establish that all compatibility conditions are met, and provide them, on a written request, to the Commission within a period of 20 working days or such longer period as may be fixed in the request.

For fiscal aid schemes:

In case of schemes under which fiscal aid is granted automatically based on tax declarations of the beneficiaries, and where there is no *ex ante* control that all compatibility conditions are met for each beneficiary, the Member State undertakes to put in place an appropriate control mechanism, by which it regularly verifies (for example once per fiscal year), at least *ex post* and on a sample basis, that all compatibility conditions are met, and to impose sanctions in case of fraud. In order to enable the Commission to monitor fiscal aid schemes, the notifying Member State undertakes to maintain detailed records of the controls for at least 10 years from the date of the controls, and provide them, on a written request, to the Commission within a period of 20 working days or such longer period as may be fixed in the request.

10. **Confidentiality**

Does the notification contain confidential information⁽²⁴⁾ which should not be disclosed to third parties?

- Yes. Please indicate which of the form are confidential and provide reasons for confidentiality.
- □ No

11. **Other Information**

Where applicable, please indicate any other information relevant for the assessment of the aid.

12. Attachments

Please list all documents which are attached to the notification and provide paper copies or internet addresses which allow access to the documents concerned.

13 **Declaration**

I certify that to the best of my knowledge the information provided on this form, annexes and attachments is accurate and complete.

Date and place of signature

Signature:

Name and position of person signing

14. Supplementary Information Sheet

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

14.1.	Based on the information submitted in the general information form, please select the applicable supplementary information sheet to be completed:		
(a)□	Supple	ementary information sheets on regional aid	
	1.□	investment aid	
	2.□	operating aid	
	3.□	individual aid	
(b)□	Supple	ementary information sheet on research, development and innovation aid	
(c)□	Supple in diffi	ementary information sheets on aid for restructuring and rescuing undertakings culty	
	1.□	rescue aid	
	2.□	restructuring aid	
	3.□	aid schemes	
(d)□	Supple	ementary information sheet on audiovisual production aid	
(e)□	Supple	ementary information sheet on aid for broadband aid	
(f)□	Supple	ementary information sheet on environment and energy aid	
(g)□	Supple	ementary information sheet on risk finance aid	
(h)□	Supple	ementary information sheet on aid to the transport sector:	
	1.□	investment aid to airports	
	2.□	operating aid to airports	
	3.□	start-up aid to airlines	
	4.□	aid of a social character under Article 107(2)(a) TFEU	
	5.□	aid for maritime transport	
(i)□	Supple	ementary information sheet for the notification of an evaluation plan	
(j)□	Supplementary information sheets on aid in the agriculture and forestry sectors are in rural areas		
(k)□	Supple	ementary information sheet on aid to the fishery and aquaculture sector	
14.2.	For aid which is not covered by any supplementary information sheet, please selecthe relevant provision of the TFEU, guideline or another text applicable to State aid:		
(a)□	Short term export credit ⁽²⁵⁾		
(b)□	Emissi	ons Trading Systems ⁽²⁶⁾	
(c)□	Bankir	ng Communication ⁽²⁷⁾	
(d)□	Comm	unication on important projects of common European interest ⁽²⁸⁾	
(e)□	Services of General Economic Interest (Article 106(2) TFEU) ⁽²⁹⁾		

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (f)□ Article 93 TFEU
- (g)□ Article 107(2)(a) TFEU
- (h)□ Article 107(2)(b) TFEU
- (i)□ Article 107(3)(a) TFEU
- (j)□ Article 107(3)(b) TFEU
- (k)□ Article 107(3)(c) TFEU
- (l)□ Article 107(3)(d) TFEU
- (m)□ Other(s), please specify

Please provide a justification for the compatibility of the aid falling in the categories selected in this point:

For practical reasons, it is recommended to number the documents provided as annexes and to refer to those document numbers in the relevant sections of the supplementary information sheets.]

[F5]

Textual Amendments

F5 Deleted by Commission Regulation (EC) No 271/2008 of 30 January 2008 amending Regulation (EC) No 794/2004 implementing Council Regulation (EC) No 659/1999 laying down detailed rules for the application of Article 93 of the EC Treaty.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

SUPPLEMENTARY INFORMATION SHEETS

To be completed as necessary depending on the type of aid concerned:

- SME aid
- 2. Training aid
- 3. Employment aid
- 4. Regional aid
- Aid coming under the multisectoral framework
- Research and development aid
 - a) in the case of a scheme
 - b) in the case of individual aid
- Aid for rescuing firms in difficulty
 - a) in the case of a scheme
 - b) in the case of individual aid
- 8. Aid for restructuring firms in difficulty
 - a) in the case of a scheme
 - b) in the case of individual aid
- 9. Aid for audio-visual production
- 10. Environmental protection aid
- 11. Risk capital aid
- 12. Aid in the agricultural sector
 - a) Aid for agriculture
 - i. Aid for investment in agricultural holdings
 - ii. Aid for investments in connection with the processing and marketing of agricultural products
 - b) Agri-environmental aid
 - c) Aid to compensate for handicaps in the less favoured areas
 - d) Aid for the setting up of young farmers
 - e) Aid for early retirement or for the cessation of farming activities
 - f) Aid for closing production, processing and marketing capacity
 - g) Aid for producer groups
 - h) Aid to compensate for damage to agricultural production or the means of agricultural production
 - i) Aid for land reparcelling
 - j) Aid for the production and marketing of quality agricultural products
 - k) Aid for the provision of technical support in the agricultural sector
 - Aid for the livestock sector
 - m) Aid for the outermost regions and the Aegean Islands
 - n) Aid in the form of subsidised short-term loans
 - Aid for the promotion and advertising of agricultural and certain non-agricultural products
 - p) Aid for rescue and restructuring firms in difficulty
 - q) Aid for TSE tests; fallen stock and slaughterhouse waste
- 13. Aid in the transport sector
 - a) Individual aid for restructuring firms in difficulty in the aviation sector
 - b) Aid for transport infrastructure
 - c) Aid for maritime transport
 - d) Aid for combined transport
- ▶⁽¹⁾14. Aid to the fisheries sector **◄**

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

[F1PART III

SUPPLEMENTARY INFORMATION SHEETS

PART III.1.A

Supplementary Information Sheet on individual regional investment aid

This supplementary information sheet must be used for the notification of any individual investment aid covered by the Guidelines on regional State aid $2014-2020^{(30)}$ (the 'RAG'). Where several beneficiaries are involved in an individual aid measure, the relevant information must be provided for each of them.

1. Scope

- 1.1. Reasons for notifying the measure:
- a) The notification relates to individual aid granted under a scheme and the aid from all sources exceeds the notification threshold. Please provide State aid references of the approved or block-exempted aid schemes concerned.
- b) The notification relates to individual aid granted outside a scheme (ad hoc aid).
- c) The notification relates to aid granted to a beneficiary that has closed down the same or similar activity in the EEA two years preceding the date of applying for aid or at the moment when submitting the aid application the beneficiary has the intention of closing down such an activity within a period of two years after the investment to be subsidised is completed.
- d) The notification relates to an investment by a large undertaking to diversify an existing establishment in a 'c' area into new products and/or new process innovation.
- e)□ Other. Please explain:
- 1.2. Scope of the notified aid measure
- 1.2.1. Please confirm that the beneficiary is not an undertaking in difficulty⁽³²⁾
- 1.2.2. If the measure covers investment aid to broadband networks, please explain how the aid granting authority will ensure that the following conditions are complied with and provide the reference to the relevant provisions in the legal basis and/or supporting documents:
- aid is granted only to areas where there is no network of the same category (either basic broadband or next generation access networks) and where none is likely to be developed in the near future
- the subsidised network operator offers active and passive wholesale access under fair and non-discriminatory conditions with the possibility of effective and full unbundling
- aid was or will be allocated on the basis of a competitive selection process in accordance with paragraphs 78(c) and (d) of the EU Guidelines for the application of State aid rules in relation to the rapid deployment of broadband networks⁽³³⁾.
- 1.2.3. If the measure covers aid to research infrastructures⁽³⁴⁾, please confirm that the aid is conditional on giving transparent and non-discriminatory access to this infrastructure

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- and provide supporting documents to that effect and/or provide reference(s) to the relevant parts of the legal basis (paragraph 13 RAG).
- 1.2.4. Please provide a copy of the application form and the (draft) aid granting agreement.
- 2. Additional information on the beneficiary, investment project and aid
- 2.1. Beneficiary
- 2.1.1. Identity of aid recipient(s):
- 2.1.2. If the legal identity of the aid recipient is different from that of the undertaking(s) that finance(s) the project or the actual beneficiary or beneficiaries of the aid, describe those differences.
- 2.1.3. Please give a clear description of the relationship between the beneficiary, the group of enterprises it belongs to and other associated enterprises, including joint ventures.
- 2.2. The investment project
- 2.2.1. Please provide the following information on the notified investment project:

Date of application for the aid:	
(Planned) date of start of works on the investment project:	
Planned date of the start of production:	
Planned date on which the full production capacity is reached:	
Planned end date of the investment project:	

- 2.2.2. If the notification covers an investment in an 'a' area or an investment by an one or more SMEs⁽³⁵⁾ in a 'c' area, please specify the category or categories of initial investment concerned by the notification (paragraph 20(h) RAG):
- (a) □ the setting-up of a new establishment
- (b)□ the extension of the capacity of an existing establishment
- (c) the diversification of the output of an establishment into products not previously produced in the establishment
- (d)□ a fundamental change in the overall production process of an existing establishment
- (e) an acquisition of assets directly linked to an establishment provided the establishment has closed down or would have closed down had it not been purchased, and is bought by an investor unrelated to the seller (36).
- 2.2.3. If the notification covers an investment in a 'c' area by a Large Enterprise, please specify the category or categories of initial investment concerned by the notification (paragraph 15 and paragraph 20(i) RAG):
- (a) □ the setting-up of a new establishment

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (b)□ the diversification of the activity of an establishment, under the condition that the new activity is not the same or a similar activity (37) to the activity previously performed in the establishment
- (c)□ diversification of an existing establishment into new products
- (d)□ new process innovation at an existing establishment
- (e) the acquisition of the assets belonging to an establishment that has closed or would have closed had it not been purchased, and is bought by an investor unrelated to the seller, under the condition that the new activity to be performed using the acquired assets is not the same or a similar activity to the activity performed in the establishment prior to the acquisition.
- 2.2.4. Please describe briefly the investment explaining how the project concerned falls within one or more of the categories of initial investment indicated above:
- 2.3. Eligible costs calculated on the basis of investment costs
- 2.3.1. Please provide the following breakdown of the total eligible investment costs in nominal and discounted value:

	Total eligible cost (nominal) ^a	Total eligible cost (discounted) ^a
Preparatory studies or consultancy costs linked to the investment (SMEs only)		
Land		
Buildings		
Plant/machinery/ equipment ^b		
Intangible assets		
Total eligible cost		

a In national currency (see also point 2.5 below)

2.3.2.	Please confirm	that the acquired	l assets are new	(paragraph 94 RAG) ⁽³⁸⁾ .

□ yes □ no	
------------	--

- 2.3.3. Please provide evidence to show that in the case of SMEs not more than 50 % of the costs of preparatory studies or consultancy costs linked to the investment is included in the eligible cost (paragraph 95 RAG).
- 2.3.4. Please provide evidence to show that for aid awarded for a fundamental change in the production process, the eligible costs exceed the depreciation of the assets linked to the activity to be modernised in the course of the three preceding fiscal years (paragraph 96 RAG).

b In the transport sector, expenditure on the purchase of transport equipment cannot be included in the uniform set of items of expenditure. Such expenditure is not eligible for initial investment.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 2.3.5. Please provide a reference to the legal basis or explain how it is ensured that for aid awarded for a diversification of an existing establishment, the eligible costs exceed by at least 200 % the book value of the assets that are reused, as registered in the fiscal year preceding the start of works (paragraph 97 RAG). If appropriate, provide documentation containing the relevant quantitative data.
- 2.3.6. In cases involving the lease of tangible assets, please provide a reference to the relevant provisions in the legal basis where it is stipulated that the following conditions are respected (paragraph 98 RAG) or explain how respect for those conditions is otherwise achieved.
- for land and buildings, the lease must continue for at least five years after the expected date of completion of the investment for large companies, and three years for SMEs;
- for plant or machinery, the lease must take the form of financial leasing and must contain an obligation for the beneficiary of the aid to purchase the asset at the expiry of the term of the lease.
- 2.3.7. Paragraph 99 RAG provides that 'In the case of acquisition of an establishment only the costs of buying the assets from third parties unrelated to the buyer should be taken into consideration. The transaction must take place under market conditions. Where aid has already been granted for the acquisition of assets prior to their purchase, the costs of those assets should be deducted from the eligible costs related to the acquisition of an establishment. If the acquisition of an establishment is accompanied by an additional investment eligible for aid, the eligible costs of this latter investment should be added to the costs of purchase of the assets of the establishment'.

If relevant in the notified case, please explain how those conditions have been complied with, providing relevant supporting documentation.

- 2.3.8. If the eligible expenditure for the investment project includes intangible assets, please explain how it is ensured that the conditions set out in paragraphs 101 to 102 RAG⁽³⁹⁾ will be respected. In such cases please provide a precise reference to the relevant provision in the legal basis.
- 2.4. Eligible costs calculated on the basis of wage costs

Please:

- explain how eligible costs calculated on the basis of wage costs have been established (paragraph 103 RAG);
- explain how the number of jobs created has been calculated with reference to paragraph 20(k) RAG;
- explain how the wage costs of the persons hired have been established with reference to paragraph 20(z) RAG; and
- provide the relevant calculations and documentation supporting the figures.
- 2.5. Calculation of discounted eligible costs and the amount of aid
- 2.5.1. Please fill in the table in this point with the details of the eligible costs by category of eligible cost to be defrayed over the whole duration of the investment project:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	Nominal/ discounte	N+1 ^a	N+2ª	N+3ª	N+X ^a	Total ^a
Preparato	T Yominal					
studies etc. (SME only)	_s Discounted					
Land	Nominal					
	Discounted					
Buildings	Nominal					
	Discounted					
Plant/	Nominal					
machiner; equipmen	t ^{Discounted}					
Intangible	Nominal					
assets	Discounted					
Wage	Nominal					
costs	Discounted					
Other	Nominal					
(please specify)	Discounted					
Total	Nominal					
	Discounted					

a In national currency

Please indicate the date to which the amounts were discounted, as well as the discount rate used⁽⁴⁰⁾:

2.5.2. Please complete the table in this point with the details of the notified aid (to be) granted for the investment project by reference to the applicable form of aid:

	Nominal/ discounte	 N+1ª	N+2 ^a	N+3 ^a	N+X ^a	Totala
Grants	Nominal					
	Discounted					
Soft	Nominal					
loan	Discounted					
Guarante	eNominal					
	Discounted					
Tax reduction	Nominal					

a In national currency

b In the transport sector, expenditure on the purchase of transport equipment cannot be included in the uniform set of items of expenditure. Such expenditure is not eligible for initial investment.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	Discounted
•••	Nominal
	Discounted
•••	Nominal
	Discounted
Total	Nominal
	Discounted

a In national currency

Please indicate the date to which the amounts were discounted, as well as the discount rate used:

For each form of aid mentioned in the table in point 2.5.2 please indicate how the grant equivalent is calculated:

Soft loan:

Guarantee:

Tax reduction:

Other:

- 2.5.3. Please specify whether any of the aid measures to be granted to the project are not yet defined, and explain how the granting authority will ensure that the applicable maximum aid intensity is respected (paragraphs 82 and 83 RAG):
- 2.5.4. Is the project co-financed by the ESI Funds? If so, please explain under which operational programme the ESI Funds finance will be obtained. Please also indicate the amount of ESI Fund finance that will be involved.
- 2.5.5. If the beneficiary (at group level) has received aid for one (or more) initial investment(s) started in the same NUTS 3 region over a period of three years from the date of start of works of the notified investment project (paragraph 20(t) RAG), please provide details of the aid measures for each of the previous aided initial investments (including a short description of the investment project, the date of aid application, the date of the aid award, the date of start of works, the aid amount(s) and the eligible costs⁽⁴¹⁾).

	Eligible investm. cost ^a	Aid amount granted ^a	Date of application	Date of aid granting	Date of start of work	Short description	Aid onreference(s)
Initial invest. project							
Initial invest. Project 2							

a In national currency

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Initial invest. project 3				
•••				

- a In national currency
- 2.5.6. Please confirm that the total amount of aid that will be granted to the initial investment project does not exceed the 'maximum aid intensity' (as defined in paragraph 20(m) RAG), taking into account the increased aid intensity for SMEs (as determined in paragraph 177 RAG) and the 'adjusted aid amount' (as defined in paragraph 20(c) RAG), where applicable. Please provide the relevant supporting documentation and calculations.
- 2.5.7. Where the aid to be granted to the investment project is to be awarded under several regional aid schemes or cumulated with ad hoc aid, please confirm that the maximum permissible aid intensity that may be granted to the project has been calculated in advance by the first granting authority and specify the amount of that maximum aid intensity. Please explain how the aid granting authorities will ensure that that maximum aid intensity will be respected (paragraph 92 RAG).
- 2.5.8. If the initial investment is linked to a European Territorial Cooperation (ETC) project please explain with reference to the provisions set out in paragraph 93 RAG how the maximum aid intensity applicable to the project and the different beneficiaries involved are established.

3. Compatibility assessment of the measure

- 3.1. Contribution to regional objectives and need for State intervention
- 3.1.1. Please:
- indicate the exact location of the aided project (that is to say the locality and the NUTS 2 or 3 region to which the locality belongs);
- provide details on its regional aid status in the current regional aid map (that is to say is it in an area eligible for regional aid under Article 107(3)(a) or (c) TFEU?); and
 provide the maximum aid intensity applicable to large enterprises.
- 3.1.2. Please explain how the aid will contribute to regional development⁽⁴²⁾.
- 3.1.3. If the notification concerns an individual application for aid under a scheme, please explain how the project contributes towards the objective of the scheme and provide relevant supporting documents (paragraph 35 RAG).
- 3.1.4. If the notification concerns ad hoc aid, please explain how the project contributes towards the development strategy of the area concerned and provide relevant supporting documents (paragraph 42 RAG).
- 3.1.5. Please explain how the provision requiring that the investment will be maintained in the area concerned for at least five years (three years for SMEs) after its completion is implemented (paragraph 36 RAG). Please provide the reference to the relevant provision in the legal basis (e.g. the aid granting contract).
- 3.1.6. In cases where the aid is calculated on the basis of wage costs, please explain how the method of implementation of the provision requiring that jobs are created within three

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

years of the investment completion and that each job created by the investment will be maintained within the area for a period of five years (three years for SMEs) from the date the post is first filled is implemented (paragraph 37 RAG). Please provide the reference to the relevant provision in the legal basis (e.g. the aid granting contract).

- 3.1.7. Please provide a reference to the legal basis or demonstrate that the recipient(s) has to make a contribution of at least 25 % of the eligible costs, through its own resources or by external financing, in a form that is exempted of any public financial support (paragraph 38 RAG).
- 3.1.8. Did you or do you commit to carry out an Environmental Impact Assessment ('EIA') for the investment (paragraph 39 RAG).

□ yes	□ no

If no, please explain why an EIA is not required for this project.

- 3.2. Appropriateness of the measure
- 3.2.1. If the notification concerns ad hoc aid, please demonstrate how the development of the area concerned is better ensured by such aid than by aid under a scheme or other types of measures (paragraph 55 RAG):
- 3.2.2. If the aid is granted in forms that provide a direct pecuniary advantage⁽⁴⁴⁾, please demonstrate why other potentially less distortive forms of aid such as repayable advances or forms of aid that are based on debt or equity instruments⁽⁴⁵⁾ are not appropriate (paragraph 57 RAG):
- 3.3. Incentive effect and proportionality of the measure
- 3.3.1. Please confirm that works on the notified individual investment started only after submission of the application for the aid (paragraph 64 RAG). Please provide a copy of the aid application sent to the granting authority by the beneficiary and documentary evidence of the date of the start of works.
- 3.3.2. Please explain the incentive effect of the aid by describing the counterfactual scenario with reference to one of the two possible scenarios set out in paragraph 61 RAG.
- 3.3.3. In Scenario 1 cases (that is to say, investment decisions under paragraph 61 RAG), please provide the following information (or refer to the relevant parts of the submitted counterfactual scenario) (paragraph 104 RAG):
- the calculation of the investment's Internal Rate of Return ('IRR') with and without the aid (46):
- information on the relevant benchmarks for the company (e.g. normal rates of return required by the beneficiary to undertake similar projects, cost of capital of the company as a whole, relevant industry benchmarks):
- an explanation of why, on the basis of the criteria in this point, the aid is the minimum necessary to render the project sufficiently profitable (cf. paragraph 79 RAG):
- 3.3.4. In Scenario 2 cases (that is to say, location decisions under paragraph 61 RAG), please provide the following information (or refer to the relevant parts of the submitted counterfactual scenario) (paragraph 105 RAG):
- the calculation of the difference of the Net Present Value ('NPV') of the investment in the target area and the NPV of the investment in the alternative location (47):

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- all the parameters used for the calculation of the NPV of the investment in the target area and the NPV of the investment in the alternative location (among others the time frame concerned, discount rate used, etc.):
- an explanation of why on the basis of the information referred to in the first two subpoints the aid does not exceed the difference between the NPV of investment in the target area and the NPV in the alternative location (cf. paragraph 80 RAG):
- 3.3.5. If the regional aid is awarded through ESI Funds in 'a' regions to investments necessary to achieve standards set by Union law, please explain the following (and provide supporting documentation):
- What is the standard concerned?
- Why is the investment necessary to achieve the standard?
- Why it would not be sufficiently profitable for the beneficiary to make the investment in the area concerned which would lead to the closure of an existing establishment in the area in the absence of the aid (paragraph 63 RAG)?
- 3.4. Avoidance of negative effects on competition and trade

For scenario 1 cases

Definition of the relevant market

- 3.4.1. Please provide the information specified in this point to identify the relevant product market(s) (that is to say, products affected by the change in behaviour of the aid beneficiary) and to identify the competitors and customers/consumers affected (paragraphs 129 and 130 RAG):
- Specify all the product(s) that will be produced in the aided facility upon the completion of the investment and indicate, where appropriate, the NACE code or Prodcom code or CPA nomenclature⁽⁴⁸⁾ for projects in the service sectors.
- Will the products envisaged by the project replace any other products produced by the beneficiary (at group level)? What product(s) will it replace? If these replaced products are not produced at the location of the project, indicate where they are currently produced. Please provide a description of the link between the replaced production and the current investment and provide a time schedule for the replacement.
- What other product(s) can be produced with the same new facilities (through flexibility of the production installations of the beneficiary) at little or no additional cost?
- Explain whether the project concerns an intermediate product and whether a significant part of the output is sold otherwise than on the market (under market conditions). Based on that explanation, for the purpose of calculating the market share and capacity increase in the remainder of this section, please indicate whether the product concerned is the product envisaged by the project or whether it is a downstream product.
- Please indicate the relevant product market(s) of the product(s) concerned and, if possible, provide evidence in support, from an independent third party. A relevant product market includes the product concerned and its demand-side substitutes, that is to say products considered to be such by the consumer (by reason of the product's characteristics, prices and intended use) and its supply-side substitutes, that is to say products considered as such by producers (through flexibility of the production installations of the beneficiary and its competitors).
- 3.4.2. Please provide information, and supporting evidence, on the relevant geographic market of the beneficiary:

Market Power (paragraph 115 and paragraph 132(a)RAG)

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 3.4.3. Please provide the following information on the market position of the beneficiary (over a period of time before receiving the aid and the expected market position after finalising the investment):
- an estimate of all sales (in value and volume terms) of the beneficiary in the relevant market (at group level).
- an estimate of the overall sales of all producers in the relevant market (in value and volume terms). If available, include statistics prepared by public and/or independent sources.
- 3.4.4. Provide an assessment of the structure of the relevant market including, for example, the level of concentration in the market, possible barriers to entry, buyer power and barriers to expansion or exit. Please provide evidence, from an independent third party if possible, to support the conclusions on this point.

Capacity (paragraph 132(a) RAG)

3.4.5. Provide an estimate of the additional production capacity created by the investment (in volume and value terms):

For all cases

Manifest negative effects

- 3.4.6. In Scenario 1 cases, please provide the following information, and supporting evidence, on the relevant product market⁽⁴⁹⁾:
- From a long-term perspective, is the relevant market structurally in absolute decline (that is to say, showing a negative growth rate)? (paragraph 135 RAG)?
- Is the relevant market in relative decline (that is to say, showing a positive growth rate, but not exceeding a benchmark growth rate)? (paragraph 135 RAG)

In Scenario 2 cases, please indicate whether without aid the investment would have been located in a region with a regional aid intensity which is higher or the same as the target region (paragraph 139 RAG). Please provide supporting evidence.

3.4.7. Please confirm whether the beneficiary has submitted a statement in which it confirms that, at group level, it has not closed down the same or a similar activity in the EEA in the two years preceding the aid application and does not intend to close down the same or a similar activity elsewhere in the EEA in the two years after the completion of investment (paragraph 23 RAG).

If such a statement was provided, please annex a copy of it to the notification, otherwise, please explain why it was not provided.

- 3.4.8. If the beneficiary at group level has closed down the same or a similar activity in another area in the EEA in the two years preceding the aid application, or intends to do so in the two years after the completion of investment, and has relocated that activity to the target area, or intends to do so, please explain why it considers that there is no causal link between the aid and the relocation (paragraph 122 RAG).
- 3.4.9. Please explain whether the State aid would directly result in a substantial loss of jobs in existing locations within the EEA. If the State aid would result in a substantial loss of jobs in existing locations within the EEA, please indicate their number and the proportion compared to the total employment of the location(s) concerned.

4. **Other information**

Please provide any other relevant information to assess the notified aid measure under the RAG:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

PART III.1.B

Supplementary Information Sheet on regional investment aid schemes

This supplementary information sheet must be used for the notification of any investment aid schemes covered by the Guidelines on regional State aid $2014-2020^{(50)}$ (the 'RAG').

1. Scope

- 1.1. Reasons for notifying the scheme instead of putting it in place under the General Block Exemption Regulation ('GBER')⁽⁵¹⁾ or *de minimis* Regulation⁽⁵²⁾:
- (a) The notification relates to a sectoral scheme. In that case please indicate the sector covered by the scheme (NACE code):
- (b) The notification relates to a general scheme also covering the shipbuilding sector.
- (c)□ Other. Please specify:
- 1.2. Scope of the notified scheme
- 1.2.1. I confirm that the legal basis for the notified scheme foresees an obligation to notify to the Commission individual aid to beneficiaries that have closed down the same or a similar activity⁽⁵³⁾ in the EEA during the two years preceding the date of applying for aid or at the moment of the aid application have the intention to close down such an activity within a period of two years after the investment to be subsidised is completed (paragraph 23 RAG).

Please provide the reference to the relevant provision of the legal basis:

1.2.2. I confirm that the notified aid scheme provides that no regional investment aid will be granted to categories of companies and sectors listed below. In each case, please list the relevant provision in the legal basis of the scheme.

Excluded categories of companies and sectors	Relevant provision in the legal basis of the scheme
Undertakings in difficulty ^a	
The steel sector ^b	
Synthetic fibres sector ^b	
Production of agricultural products listed in Annex I to the TFEU	

- a As defined in the Communication from the Commission on Guidelines on State aid for rescuing and restructuring non-financial undertakings in difficulty (OJ C 249, 31.7.2014, p. 1).
- **b** As defined in Annex IV to Guidelines on regional State aid for 2014–2020.
- The Regional Aid Guidelines apply to aid schemes supporting activities outside the scope of Article 42 TFEU but covered by Regulation (EU) No 1305/2013 of the European Parliament and of the Council of 17 December 2013 on support for rural development by the European Agricultural Fund for Rural Development (EAFRD) and repealing Council Regulation (EC) No 1698/2005 (OJ L 347, 20.12.2013, p. 487) and which are either co-financed by the European Agriculture Fund for Rural Development or are granted as additional national financing to such co-financed schemes, unless sectoral rules provide otherwise.
- d As defined in footnote 12 of the Regional Aid Guidelines.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- **a** As defined in the Communication from the Commission on Guidelines on State aid for rescuing and restructuring non-financial undertakings in difficulty (OJ C 249, 31.7.2014, p. 1).
- **b** As defined in Annex IV to Guidelines on regional State aid for 2014–2020.
- c The Regional Aid Guidelines apply to aid schemes supporting activities outside the scope of Article 42 TFEU but covered by Regulation (EU) No 1305/2013 of the European Parliament and of the Council of 17 December 2013 on support for rural development by the European Agricultural Fund for Rural Development (EAFRD) and repealing Council Regulation (EC) No 1698/2005 (OJ L 347, 20.12.2013, p. 487) and which are either co-financed by the European Agriculture Fund for Rural Development or are granted as additional national financing to such co-financed schemes, unless sectoral rules provide otherwise.
- d As defined in footnote 12 of the Regional Aid Guidelines.
- 1.2.3. If the scheme covers investment aid to broadband networks please specify whether each of the following conditions are complied with:
- (a) aid is granted only to areas where there is no network of the same category (either basic broadband or new generation access networks) and where none is likely to be developed in the near future;
- (b) the subsidised network operator offers active and passive wholesale access under fair and non-discriminatory conditions with the possibility of effective and full unbundling;
- (c) aid should be allocated on the basis of a competitive selection process in accordance with paragraphs 78(c) and (d) of the Broadband guidelines (54);

Please provide the reference to the relevant provision of the legal basis:

- 1.2.4. If the scheme covers aid to research infrastructures, is the aid made conditional on giving transparent and non-discriminatory access to this infrastructure?
- (a)□ No
- (b) Yes. Please provide the reference to the relevant provision of the legal basis:
- 2. Initial investment, eligible costs and aid
- 2.1. Types of initial investments covered by the scheme
- 2.1.1. Where the scheme covers investments in 'a' areas by SMEs or large enterprises or investments by SME(s) in 'c' areas (paragraph 34 RAG), please specify the category or categories of initial investment concerned by the notification (paragraph 20(h) RAG):
- (a)□ the setting-up of a new establishment?
- (b) \Box the extension of the capacity of an existing establishment

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (c) the diversification of the output of an establishment into products not previously produced in the establishment
- (d)□ a fundamental change in the overall production process of an existing establishment
- (e) an acquisition of assets directly linked to an establishment provided the establishment has closed or would have closed if it had not been purchased, and is bought by an investor unrelated to the seller (56)
- 2.1.2. In case the scheme covers investments in 'c' areas by large enterprises, please specify the category or categories of initial investment concerned by the notification (paragraph 15and paragraph 20(i) RAG):
- (a)□ the setting-up of a new establishment
- (b)□ the diversification of the activity of an establishment, under the condition that the new activity is not the same or a similar activity (57) to the activity previously performed in the establishment
- (c)□ diversification of an existing establishment into new products
- (d)□ new process innovation at an existing establishment
- (e) the acquisition of the assets belonging to an establishment that has closed or would have closed if it had not been purchased, and is bought by an investor unrelated to the seller, under the condition that the new activity to be performed using the acquired assets is not the same or a similar activity to the activity performed in the establishment prior to the acquisition
- 2.1.3. Please provide the reference to the relevant provisions of the legal basis where it is stipulated that the Commission will be notified of any individual aid to be granted under the legal basis of the scheme to large enterprises in 'c' areas for (paragraphs 24 and 34 RAG):
- (a) diversification of an existing establishment into new products;
- (b) new process innovation at an existing establishment.
- 2.1.4. Please provide the reference to the relevant provisions of the legal basis where it is stipulated that the Commission will be notified of any individual aid to be granted under the legal basis of the scheme that would lead to the notification threshold being exceeded⁽⁵⁸⁾ (paragraph 23 RAG).
- 2.2. Eligible costs calculated on the basis of investment costs
- 2.2.1. Where the eligible expenditure (paragraph 20(x) RAG) under the scheme relates to tangible assets, is the value of the investment established as a percentage on the basis of land, buildings and plant, machinery and equipment?⁽⁵⁹⁾:
- $(a)\Box$ land
- (b)□ buildings
- (c)□ plant/machinery/equipment?

Please provide the reference to the relevant parts of the legal basis:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 2.2.2. Please provide a reference to the relevant provision of the legal basis which stipulate that the acquired assets should be new⁽⁶⁰⁾ (paragraph 94 RAG)
- 2.2.3. Please provide a reference to the relevant provision in the legal basis which stipulate that in the case of SMEs not more than 50 % of the costs of preparatory studies or consultancy costs linked to the investment can be considered as eligible costs (paragraph 95 RAG).
- 2.2.4. Please provide a reference to the relevant provision of the legal basis which stipulates that for aid awarded for a fundamental change in the production process, the eligible costs exceed the depreciation of the assets linked to the activity to be modernised in the course of the preceding three fiscal years (paragraph 96 RAG).
- 2.2.5. Please provide a reference to the relevant provision of the legal basis which stipulates that for aid awarded for a diversification of an existing establishment, the eligible costs exceed by at least 200 % the book value of the assets that are reused, as registered in the fiscal year preceding the start of works (paragraph 97 RAG).
- 2.2.6. In cases involving the lease of tangible assets, please provide a reference to the relevant provisions of the legal basis which stipulate that the following conditions should be respected (paragraph 98 RAG):
- for land and buildings, the lease must continue for at least five years after the expected date of completion of the investment for large companies, and three years for SMEs;
- for plant or machinery, the lease must take the form of financial leasing and must contain an obligation for the beneficiary of the aid to purchase the asset at the expiry of the term of the lease.
- 2.2.7. Paragraph 99 RAG provides that 'In the case of acquisition of an establishment only the costs of buying the assets from third parties unrelated to the buyer should be taken into consideration. The transaction must take place under market conditions. Where aid has already been granted for the acquisition of assets prior to their purchase, the costs of those assets should be deducted from the eligible costs related to the acquisition of an establishment. If the acquisition of an establishment is accompanied by an additional investment eligible for aid, the eligible costs of this latter investment should be added to the costs of purchase of the assets of the establishment'.

If relevant for the notified scheme, please provide a reference to the provisions of the legal basis which stipulate that the conditions in this point should be respected.

- 2.2.8. Where the eligible expenditure under the scheme relates to intangible assets (paragraph 20(j) RAG), is the value of the investment established on the basis of expenditure entailed by the transfer of technology through the acquisition of patent rights, licences, know-how or unpatented technical knowledge?
- (a)□ patent rights
- (b)□ licences
- (c)□ know-how
- (d)□ unpatented technical knowledge

Please provide the reference to the relevant parts of the legal basis:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 2.2.9. Please provide a reference to the relevant provisions of the legal basis which stipulate that for large undertakings the expenditure on eligible intangible investment must not exceed 50 % of the total eligible investment costs of the project (paragraph 100 RAG).
- 2.2.10. Please provide a reference to the relevant provisions of the legal basis which stipulate that the conditions set out in paragraphs 101 and 102 RAG⁽⁶¹⁾ should be respected.
- 2.3. Eligible costs calculated on the basis of wage costs

Please provide a reference to the relevant provisions of the legal basis where it is stipulated how the eligible costs calculated on the basis of wage costs should be established (paragraph 103 RAG), how the number of jobs created should be calculated with reference to paragraph 20(k) RAG and how the wage costs of the persons hired should be established with reference to paragraph 20(z) RAG.

- 2.4. Calculation of discounted eligible costs
- 2.4.1. Please indicate which forms of aid are allowed under the scheme:
- (a) grants. Please provide the reference to the relevant provisions of the legal basis:
- (b) soft loans. Please indicate how the grant equivalent will be calculated and provide the reference to the relevant provisions of the legal basis:
- (c) guarantees. Please indicate how the grant equivalent will be calculated and provide the reference to the relevant provisions of the legal basis:
- (d) tax measures. Please specify which type of measures and indicate how the grant equivalent will be calculated. Please also provide the reference to the relevant provisions of the legal basis:
- (e) other. Please specify and indicate how the grant equivalent will be calculated. Please also provide the reference to the relevant provisions of the legal basis:
- 2.4.2. Is the aid scheme eligible to be co-financed by the ESI Funds? If so, please mention under which operational programmes the ESI Funds finance might be obtained. Please also indicate the amount of ESI Funds finance that will be involved, if known at this stage.
- 2.4.3. Please provide the reference to the relevant provisions of the legal basis which stipulate that the aid granting authority should establish before granting individual aid under the notified scheme whether the beneficiary (at group level) received aid for one (or more) initial investment(s) started in the same NUTS 3 region in a period of three years from the date of start of works on the investment project.
- 2.4.4. Please provide the reference to the relevant provisions of the legal basis which stipulate that the total amount of aid that will be granted under the scheme to any initial investment project does not exceed the 'maximum aid intensity' (as defined in paragraph 20(m) RAG), taking into account the increased aid intensity for SMEs (as determined in paragraph 177 RAG) or the 'adjusted aid amount' (as defined in paragraph 20(c) RAG), where applicable.
- 2.4.5. Please provide the reference to the relevant provisions of the legal basis which stipulate that where individual aid is awarded under several regional aid schemes or cumulated with ad hoc aid, the maximum permissible aid intensity that may be granted to the project will be calculated in advance by the first granting authority (paragraph 92 RAG)

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2.4.6. Where the aid scheme allows for aid to initial investment linked to European Territorial Cooperation (ETC) projects please provide the reference to the relevant provisions of the legal basis which stipulate (with reference to the provisions set out in paragraph 93 RAG) how the maximum aid intensity applicable to the project and the different beneficiaries involved will be established.

3. Compatibility assessment of the aid scheme

3.1. Contribution to regional objective and need for State intervention

Is the scheme part of an Operational Programme⁽⁶²⁾ (paragraph 32 RAG)?

- Yes. Please provide the reference(s) to the Operational Programme(s) concerned:
- No. Please explain how the scheme is consistent and contributes to the development strategy of the area concerned (paragraph 33 RAG):
- 3.1.1. Please provide the reference to the relevant provisions of the legal basis containing the requirement to carry out an Environmental Impact Assessment ('EIA') for the investments concerned before granting aid to individual projects, when so required by law (paragraph 39 RAG).
- 3.1.2. Please explain how the granting authorities will prioritise and select the investment projects according to the objectives of the scheme (for example, on the basis of a formal scoring approach) (paragraph 33 RAG). Please also provide the reference to the relevant provisions of the legal basis or other related administrative acts.
- 3.1.3. Please explain how, when awarding aid to individual investment projects under the notified scheme, the aid granting authority will establish that the selected project(s) contributes towards the objective of the scheme and thus towards the development strategy of the area concerned (paragraph 35 RAG).
- 3.1.4. Please explain how the provision requiring that any investment supported under the notified scheme will be maintained in the area concerned for at least five years (three years for SMEs) after its completion is implemented (paragraph 36 RAG). Please provide the reference to the relevant provisions in the legal basis.
- 3.1.5. In cases where the aid granted under the notified scheme is calculated on the basis of wage costs, please explain the method of implementation of the provision requiring that jobs are created within three years of the investment completion and that each job created by the investment concerned is maintained within the area for a period of five years (three years for SMEs) from the date the post is first filled (paragraph 37 RAG). Please provide the reference to the relevant provision in the legal basis.
- 3.1.6. Please provide a reference to the relevant provisions of the legal basis where it is stipulated that recipients have to make a contribution of at least 25 % of the eligible costs, through their own resources or by external financing, in a form that is exempted of any public financial support⁽⁶³⁾ (paragraph 38 RAG).
- 3.1.7. Please provide a reference to the relevant provisions of the legal basis showing that the scheme should respect the ceilings of the regional aid map applicable at the time of awarding the aid (paragraph 81 RAG). Please also provide the reference to the Commission decision approving the regional aid map concerned.
- 3.2. Appropriateness of the scheme

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 3.2.1. If the scheme is not eligible under an Operational Programme, please explain why regional aid is an appropriate instrument to tackle the common objective of equity or cohesion⁽⁶⁴⁾ (paragraph 52 RAG):
- 3.2.2. Where the scheme is sector-specific and is not eligible for co-financing by Structural Funds, please demonstrate the advantages of such an instrument compared to a multi-sectoral scheme or other policy options (paragraph 53 RAG):
- 3.2.3. Will individual aid under the notified scheme be granted:
- automatically, should the conditions of the scheme be fulfilled or
- on a discretionary basis, following a decision of the authorities?

Please provide the reference to the relevant provision of the legal basis:

If aid is to be granted on a discretionary basis, please provide a short description of the criteria used and attach a copy of the internal administrative provisions of the granting authority applicable for the awarding of aid:

- 3.2.4. If the aid under the scheme is granted in forms that provide a direct pecuniary advantage⁽⁶⁵⁾, please demonstrate why other potentially less distortive forms of aid such as repayable advances or forms of aid that are based on debt or equity instruments⁽⁶⁶⁾ are not appropriate (paragraph 57 RAG):
- 3.3. Incentive effect and proportionality of the scheme
- 3.3.1. Please provide the reference to the relevant provisions of the legal basis stipulating that any application for aid must be submitted before work is started on the investment project concerned (paragraph 64 RAG):
- 3.3.2. Please provide the reference to the relevant provisions of the legal basis which stipulate that those applying for aid under the notified scheme will be obliged to submit a standard application form provided by the aid granting authority in which they must explain counterfactually what would happen if they do not receive the aid and indicate which of the scenarios (*Scenario 1* investment decision or *Scenario 2* location decision) applies (paragraph 66 and paragraph 61 RAG). If that application form differs from the example provided in Annex V to the RAG, please provide a copy of that application.
- 3.3.3. Please provide the reference to the relevant provisions of the legal basis which stipulate that large enterprises who apply for aid under the notified scheme are required to provide documentary evidence in support of the counterfactual described. (paragraph 67 RAG). Please also explain what type of documents will be required.
- 3.3.4. Please provide the reference to the relevant provisions of the legal basis which stipulate that when assessing individual aid applications the aid granting authority is obliged to carry out a credibility check of the counterfactual provided and to verify that regional aid has the required incentive effect corresponding to *Scenario 1* or *Scenario 2* (paragraph 68 RAG).
- 3.3.5. Please provide the reference to the relevant provisions of the legal basis which stipulate that individual aid granted to large enterprises under the notified scheme will be limited to the net extra costs of implementing the investment in the area concerned compared to the counterfactual in the absence of aid, using the method explained in paragraph 79 and paragraph 80 RAG (paragraph 88 RAG).

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 3.4. Avoidance of undue negative effects on competition and trade
- 3.4.1. Please explain how the distortions of competition and trade caused by the notified aid scheme will be limited to the minimum (paragraph 125 RAG)⁽⁶⁸⁾:
- 3.4.2. Please provide references to the relevant provisions in the legal basis stipulating that when awarding aid under the scheme to individual projects, the granting authority must verify and confirm that without aid the investment would not have been located in a region with a regional aid intensity which is higher or the same as the target region (paragraph 126 RAG).
- 3.4.3. Please provide references to the relevant provisions in the legal basis which stipulate that when awarding aid under the scheme to individual projects, the granting authority must notify individual aid grants in cases where the beneficiary has closed down the same or a similar activity in another area in the EEA in the two years preceding the date of applying for aid or at the moment of the aid application has the intention to close down such an activity within a period of two years after the investment to be subsidised is completed (paragraph 122 RAG).

4. **Other information**

Please provide any other information that is of relevance to assess the notified aid measure under the RAG:

PART III.1.C

Supplementary Information Sheet on regional operating aid schemes

This supplementary information sheet must be used for the notification of any operating aid schemes covered by the Guidelines on regional State aid $2014-2020^{(69)}$ (the 'RAG').

1. Scope

- (a) Please specify what kind of operating aid will be granted:
 - (i.)□ Operating aid to reduce certain specific difficulties faced by SMEs in 'a' areas
 - (ii.)□ Operating aid to compensate additional costs in the outermost regions
 - (iii.)□ Operating aid to reduce depopulation in very sparsely populated areas
 - (iv.)□ Other. Please specify:
- (b) Does the notified aid scheme provide that no operating aid will be granted to the following categories of companies and sectors? In each case, please list the relevant provision in the legal basis of the scheme.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Excluded categories of companies and sectors	Undertakings in difficulty ^a	Relevant provision in the legal basis of the scheme
The steel sector ^b	□Yes	
Synthetic fibres sector ^b	□Yes	
Production of agricultural products listed in Annex I to the TFEU	Yes	
Processing and/or marketing of agricultural products listed in Annex I to the TFEU ^c into products listed in that Annex I	□Yes	
Production, processing and/ or marketing of fishery and aquaculture products listed in Annex I to the TFEU	□Yes	
The transport sector	□Yes	
The energy sector	□Yes	
Section K 'Financial and insurance activities' of the NACE Rev.2 statistical classification of economic activities	□Yes	
NACE 70.10 'Activities of head offices' and NACE 70.22 'Business and other management consultancy activities'	□Yes	

- a As defined in the Guidelines on State aid for rescuing and restructuring non-financial undertakings in difficulty (OJ C 249, 31.7.2014, p. 1).
- **b** As defined in Annex IV to Guidelines on regional State aid for 2014–2020.
- c The Regional Aid Guidelines apply to aid schemes supporting activities outside the scope of Article 42 TFEU but covered by the Rural Development Regulation (Regulation (EU) No 1305/2013 of the European Parliament and of the Council of 17 December 2013 on support for rural development by the European Agricultural Fund for Rural Development (EAFRD) and repealing Council Regulation (EC) No 1698/2005 (OJ L 347, 20.12.2013, p. 487)) and which are either cofinanced by the European Agriculture Fund for Rural Development or are being granted as additional national financing to such co-financed schemes, unless sectoral rules provide for otherwise.

2. **Basic elements of the scheme**

- 2.1. Please provide a description of the main elements of the scheme and its objectives:
- 2.2. Please indicate which forms of aid are allowed under the scheme:
- (a) Grants. Please provide the reference to the relevant provisions of the legal basis:
- (b) Soft loans. Please indicate how the grant equivalent will be calculated and provide the reference to the relevant provisions of the legal basis:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (c) Guarantees. Please indicate how the grant equivalent will be calculated and provide the reference to the relevant provisions of the legal basis:
- (d) Tax measures. Please specify which ones and indicate how the grant equivalent will be calculated. Please also provide the reference to the relevant provisions of the legal basis:
- (e) Other. Please specify and indicate how the grant equivalent will be calculated. Please also provide the reference to the relevant provisions of the legal basis:
- 2.3. The individual aid under the notified scheme will be granted:
- (a) automatically, should the conditions of the scheme be fulfilled
- (b)□ on a discretionary basis, following a decision of the authorities.

Should the aid be granted on a case by case basis, please provide a short description of the criteria that will be applied. If administrative guidance for the assessment of the aid application exists, please attach a copy:

2.4. Will the aid scheme be co-financed by the ESI Funds? If so, please explain under which operational programmes ESI Funds finance will be obtained. Please also indicate the amount of ESI Funds finance that will be involved.

3. Compatibility of the aid

3.1. Contribution to regional objective and incentive effect:

For aid to reduce certain specific difficulties faced by SMEs⁽⁷⁰⁾in 'a' areas

- 3.1.1. Please list the specific difficulties faced by SMEs in the region concerned to be addressed by the scheme (paragraph 43 RAG) and demonstrate the existence and importance of those difficulties (paragraph 44 RAG).
- 3.1.2. Please explain why the difficulties listed in point 3.1.1 cannot be overcome by investment aid and thus the notified operating aid scheme is needed (paragraph 44 RAG):

For aid to compensate certain additional costs in the outermost regions

3.1.3. Please identify the specific additional costs⁽⁷¹⁾ that will be compensated under the scheme and demonstrate how those costs are related to the permanent handicaps set out in Article 349 TFEU (paragraph 45 RAG):

For aid to reduce depopulation in very sparsely populated areas

- 3.1.4. Please demonstrate the risk of depopulation of the relevant area in the absence of operating aid (paragraph 46 RAG):
- 3.2. Appropriateness of the scheme

Please indicate why the aid proposed is considered appropriate to achieve the objective of the scheme. Please explain in particular why other less distortive policy instruments and other less distortive types of aid instrument are not appropriate to achieve the same positive contribution to regional development (paragraphs 50, 56, 57 and 58 RAG):

3.3. Proportionality of the scheme

For all types of operating aid

3.3.1. Please determine the eligible costs that are fully attributable to the problems the aid is intended to address (paragraph 109 RAG):

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 3.3.2. Please confirm that depreciation charges and the costs of financing included in the eligible costs relevant to regional investment aid will not be included in the eligible costs for operating aid (paragraph 109 RAG), and provide the reference to the relevant provision of the legal basis:
- 3.3.3. Please describe the compensation model (paragraph 56 RAG) that will be adopted and how that model will allow an appropriate calculation of the aid amount, ensuring that there is no overcompensation, as defined in paragraph 109 RAG:
- 3.3.4. Please indicate whether operating aid is also granted through other operating aid schemes in the region, by mentioning the relevant State aid reference of those schemes.
- 3.3.5. In case other operating aid schemes are applicable in the same region, please explain how it is ensured that operating aid granted under different operating aid schemes does not lead to overcompensation:

For operating aid in outermost regions only

3.3.6. Please demonstrate that the additional costs to be compensated under the notified scheme will be quantified in relation to the level of costs incurred by similar undertakings established in other regions of the Member State concerned (paragraph 110 RAG):

For operating aid to reduce certain specific difficulties faced by SMEs in certain 'a' areas only

- 3.3.7. Please explain how the level of aid will be progressively reduced over the duration of the scheme (paragraph 111 RAG) and provide the reference to the relevant provision of the legal basis:
- 3.4. Avoidance of undue negative effects on competition and trade

Please explain why it is unlikely that the aid granted under the scheme will create very significant distortions of competition in the market (paragraph 140 RAG):

4. Other information

Please provide any other information that is of relevance to assess the notified aid measure under the RAG:

PART III.2

Supplementary information sheet for research and development and innovation aid

This supplementary information sheet must be used for the notification of any aid measures (aid schemes and individual aid) covered by the Framework for state aid for research and development and innovation ('the R & D&I Framework')⁽⁷²⁾.

In case there are several beneficiaries involved in an individual aid measure, the relevant information must be provided for each of them.

1. Characteristics of the notified aid measure

- 1.1. Aid schemes
- A) Reasons for notifying the scheme:
 - (a)□ the scheme includes aid that is not transparent in the sense of Article 5 of the General Block Exemption Regulation ('GBER')⁽⁷³⁾;

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	(b)□	other reasons.				
	Please	specify:				
B)	Sectora	ectoral scope of the notified scheme:				
C)	Please tick the box below to confirm that any aid awarded under the notified scher will be notified individually if it exceeds the thresholds laid down in Article 4 of t GBER:					
1.2.	Individ	dividual aid				
A)	that sc	the aid is based on an approved scheme, please provide information concerning theme, including its publication reference (internet address) and state aid tion number:				
B)		cable, please provide the exchange rate which has been used for the purpose otification:				
1.3.	Genera	linformation				
A)	Please	specify the type of aid:				
	(a)□	aid for R & D projects;				
	(b)□	aid for feasibility studies;				
	(c)□	aid for the construction and upgrade of research infrastructures;				
	(d)□	innovation aid for SMEs;				
	(e)□	aid for process and organisational innovation;				
	(f)□	aid for innovation clusters.				
B)	institut	he notified measure involve Union funding centrally managed by the ons, agencies, joint undertakings or other bodies of the Union that is not or indirectly under the control of Member States?				
	□yes	□no				
	If so, p	lease specify:				
C)	Guideli	Does the notified measure involve undertakings in difficulty, as defined by the Guidelines on State aid for rescuing and restructuring non-financial undertakings in difficulty ⁽⁷⁴⁾ ?				
	□yes	□no				
	If so, p	lease specify:				
D)	recover	ne notified measure involve undertakings that are subject to an outstanding y order following a previous decision declaring aid illegal and incompatible internal market?				

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□yes	□no
If so, please specify and indi	icate the amounts still to be recovered:
	re involve research and knowledge dissemination ganisations') or research infrastructures, as defined in R & D&I Framework?
□yes	□no
If so, please specify:	
Does the notified measure in services?	nvolve public procurement of research and development
□yes	□по
If so, please specify:	
Can the aid awarded under t	he notified measure be combined with other aid?
□yes	□по
If so, please specify:	
	box below to confirm that the beneficiaries comply with down in Annex I to the GBER and, for individual aid, and evidence:
Research organisations and	d research infrastructures
	or research infrastructures concerned by the notified aid comic activity consisting of offering goods or services on
□yes	□no
Please provide details:	
	out activities of both an economic and a non-economic activities and their costs, funding and revenues be clearly

is the amount of public funding allocated to it for a specific accounting period limited

to the costs of non-economic activities incurred in the same period?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

DVAS	□no
□yes	
If so, please specify:	
does the economic use remain p activity which is directly related	vities of both an economic and a non-economic nature, burely ancillary, that is to say does it correspond to an d to and necessary for the operation of the research acture or intrinsically linked to its main non-economic pe?
□yes	□no
If so, please specify and indicate estimated to be used for such eco	e the proportion of the overall capacity that is used or onomic activities each year:
organisations or research infrastrand any advantage acquired the	for non-ancillary economic activities of research ructures, can it be shown that both the public funding rough it are fully passed on to the final recipients, ices, and that no further advantage is awarded to the
□yes	□по
If so, please specify:	
· · ·	ings through research organisations and research
Research on behalf of undertaking	ngs
	esearch infrastructures concerned by the notified aid rich or provide research services to undertakings?
□yes	□по
If so, please specify:	
	search infrastructures perform contract research or dertakings, do they provide such services at market
□yes	□по
If so, please specify:	
If research examinations or re-	search infrastructures perform contract research or

C) If research organisations or research infrastructures perform contract research or provide research services to undertakings and there is no market price, do they provide such services at a price which reflects the full costs of the services and generally includes a margin established by reference to those commonly applied by undertakings active in the sector concerned, or is the result of arm's length negotiations where research organisations or research infrastructures negotiate in order to obtain the

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

at least their marginal costs?

maximum economic benefit at the moment when the contract is concluded and cover

If so 1	please specify:	,	
-	poration with undertakings		
Do res	search organisations or research		uctures concerned by the notified angs with a view to jointly carrying of
□yes		□nc)
If so, j	please specify:		
			ructures effectively collaborate was following conditions are fulfilled:
(a)	the participating undertakings	bear th	e full cost of the projects
	□yes		□no
(b)	rights (IPR) may be widely d	lissemii	do not give rise to intellectual proper nated and any IPR resulting from to or research infrastructures are full
	□yes		□no
(c)	are allocated to the different	t collab	ct, as well as related access rigoration partners in a manner whickages, contributions and respecti
	□yes		□no
If the	answer to any of the above question	ons is y	ves, please provide details:
undert		o questi	ructures effectively collaborate with ion (B) is yes, please indicate whether
(a)		establi	infrastructures receive a compensation shed by means of an open, transpare tale procedure:
	□yes		□no
(b)		ned to l	infrastructures receive a compensati be at least equal to the market price

Status: Point in time view as at 31/01/2020. Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

		Πno.
	□yes	□no
(c)	they effectively negoti	ons or research infrastructures can demonstrated the compensation, at arm's length, in conomic benefit at the moment when the con
	□yes	□no
(d)	undertakings with a rigi organisations or researce to solicit more econom	ollaboration agreement provides the collaboration agreement provides the collaboration of first refusal as regards IPR generated by respectively in the infrastructures, the latter exercise a reciproducally advantageous offers from third parties takings will have to match their offer according
	□yes	□по
If the a	nswer to any of the quest	ons in this section is yes, please provide deta
	•	and development services
	s from undertakings, are	public procurement of research and devel the providers selected by means of an oper ace with the applicable directives ⁽⁷⁵⁾ ?
	s from undertakings, are	the providers selected by means of an oper
procedi □yes	s from undertakings, are	the providers selected by means of an operace with the applicable directives (75)?
□yes If yes, j In all or and dev	s from undertakings, are ure carried out in accorda please specify: ther cases where the notify elopment services from u	the providers selected by means of an operace with the applicable directives (75)?
□yes If yes, j In all or and developlease i	please specify: ther cases where the notify elopment services from undicate whether the following the selection procedure.	the providers selected by means of an operace with the applicable directives (75)? □no ed measure involves public procurement of indertakings, including pre-commercial procu
□yes If yes, j In all or and dev	please specify: ther cases where the notify relopment services from undicate whether the folloop the selection procedure based on objective selections.	the providers selected by means of an operace with the applicable directives (75)? □no ed measure involves public procurement of redertakings, including pre-commercial procure wing conditions are fulfilled: e is open, transparent and non-discriminator
□yes If yes, j In all or and developlease i	please specify: ther cases where the notify velopment services from undicate whether the folloon the selection procedure based on objective selection procedure □yes If no, please provide of and non-discriminatory	ed measure involves public procurement of redertakings, including pre-commercial procure ving conditions are fulfilled: e is open, transparent and non-discriminate ction and award criteria specified in advance that indicating whether a competitive, transprocedure in line with the applicable directic competitive procedure with negotiations, inneredections, inneredections, inneredections.
□yes If yes, j In all or and dev please j (a)	please specify: ther cases where the notify velopment services from undicate whether the following the selection procedure based on objective selection procedure ———————————————————————————————————	ed measure involves public procurement of redertakings, including pre-commercial procure ving conditions are fulfilled: e is open, transparent and non-discriminate ction and award criteria specified in advance that indicating whether a competitive, transprocedure in line with the applicable directic competitive procedure with negotiations, inneredections, inneredections, inneredections.
□yes If yes, j In all or and dev please i	please specify: ther cases where the notify velopment services from undicate whether the following the selection procedure based on objective selection procedure ———————————————————————————————————	the providers selected by means of an operace with the applicable directives (75)? Ino ed measure involves public procurement of redertakings, including pre-commercial procuring conditions are fulfilled: e is open, transparent and non-discriminate etion and award criteria specified in advance to the applicable direction and award enterior with the applicable directic competitive procedure with negotiations, innertive dialogue): al arrangements describing all rights and obling with regard to IPR, are made available.

Please provide details:

Status: Point in time view as at 31/01/2020. Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (c) the procurement does not give any of the participant providers any preferential treatment in the supply of commercial volumes of the final products or services to a public purchaser in the Member State concerned⁽⁷⁶⁾, and one of the following conditions is met:
 - all results which do not give rise to IPR may be widely disseminated in a way that allows other undertakings to reproduce them, and any IPR are fully allocated to the public purchaser
 - any service provider to which results giving rise to IPR are allocated is required to grant the public purchaser unlimited access to those results free of charge, and to grant access to third parties under market conditions.

□yes	□no

Please provide details:

5. Description of the notified aid measure

- 5.1. Aid for R & D projects
- Please indicate which R & D stages are supported under the notified measure: A)
 - fundamental research;
 - (b)□ industrial research;
 - (c)□ experimental development.
- B) For individual aid, if the project encompasses different research categories, please list and qualify the different tasks as falling under the categories of fundamental research, industrial research or experimental development:
- Please specify the eligible costs and, for individual aid, indicate their amount: C)

	Fundamental research	Industrial research	Experimental development
Personnel costs			
Costs of instruments and equipment			
Costs of buildings and land			
Cost of contractual research, knowledge and patents bought or licensed from outside sources in arm's length transactions			

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Additional overheads incurred directly as a result of the project		
Other operating expenses		
- inpenses		

D) Please specify the applicable maximum aid intensities:

	Small enterprise	Medium-sized enterprise	Large enterprise
Fundamental research		-	
Industrial research			
— subject to effective collabor between undertak (for larg enterprise crossborder of with at least one SME) of between an undertak and a research organisation or subject to wide dissemin of result	e ation kings e ses or e transition;		
Experimental development		1	
— subject to effective collabor between undertak (for larg enterprise crossborder of with at least one	e ation kings e ses		

5.2.

A)

B)

5.3.

A)

B)C)

D)

E)

F)

 $\Box yes$

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	to the research intra natory basis?	frastructures ope	en to several users on a	a transparent and non-
Please sp	pecify:			
□yes			□по	
	e price charged and to a market pri		on or use of the res	earch infrastructures
Please p	rovide relevant int	formation and e	vidence:	
economi back me	c activities, please	tick the box be	blic funding for both low to confirm that a ensure that the appl	monitoring and claw-
			t information and evid	
please ti type of a	ck the box below	to confirm that nted for separate	th economic and non t the financing costs a ely on the basis of conciples:	and revenues of each
Please sp	pecify the applicat	ole maximum ai	d intensity:	
Please sp	pecify the eligible	costs and, for ir	ndividual aid, indicate	their amount:
Aid for t	he construction ar	nd upgrade of re	search infrastructures	
Please s			aid intensities, includ	
	•	costs and, for ir	ndividual aid, indicate	their amount:
Aid for f	easibility studies			
_	SME) or between an undertaking and a research organisation; or subject to wide dissemination of results			

 \Box no

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

In case preferential access is granted to some undertakings, please provide details and indicate the share of the investment costs borne by those undertakings:

	indicate	e the share of the investment costs l	oorne by those undertakings:			
5.4.	Innovation aid for SMEs					
A)	Please indicate which activities are supported under the notified measure:					
	(a)□	obtaining, validating and defend	ing patents and other intangible assets;			
	(b)□	(b)□ secondment of highly qualified personnel;				
	(c)□	acquiring innovation advisory an	nd support services.			
B)	Please specify the eligible costs and, for individual aid, indicate their amount:					
C)	Please specify the applicable maximum aid intensities:					
5.5.	Aid for process and organisational innovation					
A)	Please	Please indicate which activities are supported under the notified measure:				
		process innovation;				
		organisational innovation.				
B)	Please	Please specify the eligible costs and, for individual aid, indicate their amount:				
	Personnel costs					
	the ext	Costs of instruments and equipment (to the extent and for the period used for the project)				
	extent	Costs of buildings and land (to the extent and for the period used for the project)				
	Cost of contractual research, knowledge and patents bought or licensed from outside sources in arm's length transactions					
	Additional overheads and other operating costs incurred directly as a result of the research project					
C)	Please	specify the applicable maximum ai	d intensities:			
D)	they ef	Where aid is awarded to large enterprises, please tick the box below to confirm that they effectively collaborate with SMEs in the aided activity and the collaborating SMEs incur at least 30 % of the total eligible costs:				
	For ind	For individual aid, please provide relevant information and evidence:				
5.6.	Aid for innovation clusters					

6.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

A)	Please tick the box below to confirm that the aid is awarded exclusively to the legal entity operating the innovation cluster:					
	For inc	dividual aid, please provide details:				
B)		Do the fees charged for using the clusters' facilities and participating in the clusters activities correspond to the market price or reflect their costs?				
	□yes	□no				
	Please	e specify:				
C)		Is access to the clusters' premises, facilities and activities open to several users on a transparent and non-discriminatory basis?				
	□yes	□no				
		In case preferential access is granted to some undertakings, please provide details and indicate the share of the investment costs borne by those undertakings:				
D)	For individual aid, please provide information on the planned or expected specialisation of the innovation cluster, existing regional potential and presence of clusters in the Union with similar purposes:					
5.6.1.	Investment aid					
A)	Please specify the eligible costs and, for individual aid, indicate their amount:					
B)	Please specify the applicable maximum aid intensities, including any increase ('bonuses') for clusters located in assisted regions fulfilling the conditions of Article 107(3)(a) or 107(3)(c) TFEU:					
5.6.2.	Operat	ting aid				
A)	Please indicate which activities are supported under the notified measure:					
	(a)□	animation of the cluster;				
	(b)□	marketing of the cluster;				
	(c)□	management of the cluster's facilities;				
	$(d)\Box$	organisation of training programmes, workshops and conferences.				
B)	Please	Please specify the eligible costs and, for individual aid, indicate their amount:				

C) Please specify the applicable maximum aid intensity and duration of the aid:

Compatibility assessment of the notified aid measure

For individual aid, please provide a comprehensive description of the aided project or activity:

- 6.1. Contribution to a well-defined objective of common interest
- A) Please define precisely the objective pursued and explain how the notified measure intends to promote R & D&I activities in the Union:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□yes			□no)	
measu		where relevant including	refere	ence to evaluations of similar p	
A)	Will th	e project size be increase	d due	to the notified measure?	
	□yes			□no	
	If so, please specify the type of increase and provide relevant evidence				
	increase in the total project costs (without a decrease by the aid beneficiary when compared to the situation				
	(b)□	increase in the numbe	r of pe	eople assigned to R & D&I activ	
	(c)□	other type of increase			
B)	Will th	ue to the notified measure?			
	□yes			□no	
	If so, please specify the type of increase and provide relevant eviden				
	(a)□	increase in the numbe	r of th	e expected deliverables of the p	
	(b)□	increase in the level of ambition of the project evidence a higher number of partners involved, a higher probabilit a scientific or technological break-through or a higher ris failure (notably linked to the long-term nature of the project uncertainty about its results);			
	(c)□	other type of increase			
Will tl	ll the project speed be increased due to the notified measure?				
□yes			□no)	
If so, j	please provide relevant evidence:				
Will th	ne total am	ount spent be increased	due to	the notified measure?	
yes			□no		

(a) \square increase in total R & D&I spending by the aid beneficiary, in absolute terms or as a proportion of turnover;

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (b)□ changes in the committed budget for the project (without a corresponding decrease in the budget allocated to other projects);
- (c) \Box other type of increase.
- E) Will the notified measure be subject to a publicly available *ex post* evaluation of its contribution to the common interest?

DVAC	
□yes	□no

If so, please specify:

- 6.2. Need for State intervention
- A) Please identify the market failures hampering R & D&I activities in the present case and justifying the need for state aid and provide relevant evidence:
 - (a)□ positive externalities/knowledge spillovers;
 - (b)□ imperfect and asymmetric information;
 - (c)□ coordination and network failures.
- B) Please explain how the notified measure can effectively mitigate the market failures associated with reaching the objective of common interest without aid
- 6.2.1. Individual aid
- A) Please explain whether the aid addresses a general market failure regarding R & D&I activities in the Union, or a specific market failure regarding, for example, a particular sector or line of business:
- B) If available, please provide any sectoral comparisons and other studies that can substantiate the analysis of the alleged market failures:
- C) If available, please provide any information regarding R & D&I projects or activities undertaken within the Union which, with respect to their technological content, level of risk and size, are similar to those concerned by the notified measure and explain why the aid is needed in the case concerned:
- 6.3. Appropriateness of the aid measure
- A) Please explain how the advantages of using a selective policy instrument such as state aid in order to increase R & D&I activities have been established, and provide any related impact assessment and supporting documents:
- B) If aid is awarded in forms that provide a direct pecuniary advantage (such as direct grants, exemptions or reductions in taxes or other compulsory charges, or the supply of land, products or services at favourable prices), please provide an analysis of other options and explain why or how other forms of aid are less appropriate to address the identified market failures:
- 6.4. Incentive effect
- A) Please tick the box below to confirm that when awarding the aid under the notified measure, it will be ensured that work on the relevant R & D&I activities has not started

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

before the aid application by the beneficiary to the national authorities⁽⁷⁷⁾ and, for individual aid, specify the relevant dates:

B) Please tick the box below to confirm that the aid applications include at least the applicant's name and size of the undertaking, a description of the project, including its location and start and end dates, the amount of public support needed to carry it out, and a list of eligible costs:

. . .

. . .

- C) If the aid is awarded in the form of a fiscal measure, please provide details and, for non-incremental measures, any evaluation studies establishing its incentive effect:
- 6.4.1. Individual aid
- A) Please provide a description, by means of counterfactual analysis, of the behaviour of the beneficiary in the absence of aid and specify the intended change:
- B) Please specify the elements that are relevant for the notified measure and provide supporting evidence, such as board documents, risk assessments, financial reports, internal business plans, expert opinions and other studies related to the project under assessment:
 - (a)□ level of profitability;
 - (b)□ amount of investment and the timeframe of cash flows;
 - (c)□ level of risk involved.
- C) If available, please provide industry-specific data demonstrating that the beneficiary's counterfactual scenario, its required level of profitability and its expected cash-flows are reasonable:
- 6.5. Proportionality of the aid
- A) If the aid is awarded in the form of a repayable advance expressed as gross grant equivalent, please provide details on the methodology applied in order to calculate such gross grant equivalent, including underlying verifiable data or, for individual aid, specify on the basis of which approved aid scheme the aid is awarded:

If the aid is awarded in the form of a repayable advance expressed as a percentage of the eligible costs and exceeds, by up to 10 percentage points, the maximum aid intensities laid down in the R & D&I Framework, please confirm that:

- (a) in the case of a successful outcome, the notified measure provides that the advance is to be repaid with an interest rate not less than the discount rate resulting from the application of the Communication from the Commission on the revision of the method for setting the reference and discount rates (78);
- (b)□ in the case of a success exceeding the outcome defined as successful, the Member State concerned requests payments beyond repayment of the advance amount including interest according to the applicable discount rate;

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

in the case of failure or partial success, the repayment is proportional to the degree of success achieved.

Please provide details on the repayment of the advance and clearly define what is considered as a successful outcome of the aided activities, on the basis of a reasonable and prudent hypothesis:

- B) If the aid is awarded in the form of a fiscal measure, please specify how the aid intensities are calculated and provide any relevant details:
 - on the basis of individual projects;
 - at the level of the undertaking, as the ratio between the overall tax relief and the sum of all eligible R & D&I costs incurred in a period not exceeding three consecutive fiscal years.

6.5.1. Individual aid

A) Please provide a comprehensive business plan for the aided project (with and without aid), including all relevant expected costs and benefits:

If the aid beneficiary faces a clear choice between carrying out either the aided project or an alternative one without aid, please provide also a comprehensive business plan for the counterfactual project:

- B) In the absence of an alternative project, please explain why the aid is limited to the minimum necessary for the aided project to be sufficiently profitable, for example by making it possible to achieve an internal rate of return ('IRR') corresponding to the sector or firm specific benchmark or hurdle rate:
- C) If the aid beneficiary faces a clear choice between carrying out either the aided project or an alternative one without aid, please explain why the aid is limited to the minimum necessary to cover the net extra costs of the aided project compared to the counterfactual project, if relevant by taking account of the probabilities of different business scenarios occurring:

Please provide any supporting documents, such as internal company documents, showing that the counterfactual project is a clearly defined and sufficiently predictable alternative project that has been considered by the beneficiary in its internal decision making:

- D) Please explain how the aid amount has been established and provide any supporting documents:
- E) If there were multiple potential candidates for carrying out the aided activity, is the aid awarded on the basis of transparent, objective and non-discriminatory criteria?

□yes	□no

Please provide details:

F) If the aid is meant to address actual or potential direct or indirect distortions of international trade, please provide any available evidence indicating that, directly or indirectly, competitors located outside the Union have received (normally in the last three years) or are going to receive aid of an equivalent intensity for similar projects:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

If available, please provide also sufficient information to assess the need to take account of the competitive advantage enjoyed by a third country competitor:

6.6. Avoidance of undue negative effects on competition and trade

Please indicate whether:

□yes	□no	
the award of aid is subject to the obligation for the beneficiary to use national products or services:		
□yes	□no	
the aid measure restricts the possibility for the beneficiary to exploit the R & D& results in other Member States:		
□yes	□no	
the aid measure imposes a	ny other obligation on the beneficiary:	

If the answer to any of the questions in this section is yes, please provide details:

6.6.1. Aid scheme

For aid schemes, please indicate how it will be ensured that any negative effects will be limited to the minimum (taking account, for example, of the size of the projects concerned, the individual and cumulative aid amounts, the number of expected beneficiaries and the characteristics of the targeted sectors) and provide any impact assessment or *ex-post* evaluations carried out for similar predecessor schemes:

6.6.2. Individual aid

- A) If applicable, please describe the likely impact of the aid on competition in the innovation process:
- B) Please identify the product markets on which the aid is likely to have an impact and provide the current market share of the beneficiary in each of the markets concerned, as well as any changes in those market shares that would result from the aided activities:
- C) For each of the product markets concerned, please identify the main competitors of the aid beneficiary and provide their market shares:
 - If available, please provide the associated Herfindahl-Hirschman Index ('HHI'):
- D) For each of the product markets concerned, please provide information on the customers or consumers affected by the aided activities:
- E) Please describe the structure and dynamics of the relevant markets with regard to the following aspects:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

(a)	recent developments and future	growth prospects:		
(b) amount spent by the main players on project(c) levels of entry and exit barriers:(d) existence of countervailing buyer power:		rs on projects of a similar kind:		
		er power:		
(e)	incentives to compete for future	markets:		
(f)	product differentiation and inten	sity of competition:		
(g)	other features likely to affect con	mpetitors, customers or consumers:		
Has the aid beneficiary any influence in the selection process, for example by having the right to recommend undertakings or influencing the research path?				
□yes		□no		
If so, please provide details:				
Is the aid awarded in markets featuring overcapacity or in declining industries?				
□yes		□по		
If so, j	please provide details:			
Has th	ne beneficiary considered any alterna	ative locations for the aided activities?		
		по		

Please provide details:

7. **Other information**

Please provide any other information that would be of relevance to assess the notified aid measure under the R & D&I Framework:

PART III.3.A

Supplementary information sheet on aid for rescuing non-financial undertakings in difficulty: individual aid

This supplementary information sheet must be filled in for the notification of individual rescue aid covered by the Guidelines on State aid for rescuing and restructuring non-financial undertakings in difficulty⁽⁷⁹⁾ ('Guidelines').

1. Eligibility

- 1.1. Undertaking in difficulty
- A) Is the undertaking a limited liability company⁽⁸⁰⁾, where more than half of its subscribed share capital has disappeared as a result of accumulated losses⁽⁸¹⁾?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	□yes		□no	
	Is the undertaking a company where at least some members have unlimited liability for the debt of the company ⁽⁸²⁾ , and where more than half of its capital as shown in the company accounts has disappeared as a result of accumulated losses?			
	□yes		по	
	Is the undertaking subject to collective insolvency proceedings or does it fulfil the criteria under its domestic law for being placed in collective insolvency proceedings at the request of its creditors?			
	□yes		□no	
In the case of an undertaking that is not an SME: — has the undertaking's book debt to equity ratio been greater the past two years and — has the undertaking's EBITDA interest coverage ratio been bell past two years?		to equity ratio been greater than 7,5 for the		
	□yes		□по	
	If you have answered yes to any of the questions in sections A to D, please substantiate the answer, including references in the answer to the supporting evidence or documents provided in annex (latest profit and loss account statements with balance sheets, or court decision opening collective insolvency proceedings on the company or evidence that the criteria for being placed under insolvency proceedings at the request of creditors under national company law are met, etc.).			
	An undertaking facing acute liquidit	An undertaking facing acute liquidity needs:		
ıl F	king in difficulty, please explain why y	ou coi	cue aid even though it does not qualify as an asider that it faces acute liquidity needs due clude reference to the supporting evidence	
	Newly created undertaking/larger but	isiness	s group:	
	When was the undertaking created?	When was the undertaking created?		
	Since when has the undertaking been	Since when has the undertaking been operating?		
	Does the company belong to a larger business group?			
	□yes		□no	
			I	

D) If the answer to point C is 'yes', please submit full details about the group (organisation chart, showing the links between the group's members with details on capital and voting rights) and demonstrate that the company's difficulties are intrinsic and are not the result of an arbitrary allocation of costs within the group, and that the difficulties are too serious to be dealt with by the group itself.

E)

undertaking into bankruptcy?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.4.	Sectoral scope:			
Is th	e undertaking active in:			
A)	the coal sector ^a :	□yes	□no	
B)	the steel sector ^b :	□yes	□no	
C)	sectors covered by specific rules for financial institutions ^c :	□yes	□no	
a	As defined in Decision 2010/787/E	U.		
b	As defined in Annex IV to the Com C 209, 23.7.2013, p. 1).	munication from the Commi	ssion: Guidelines on regional sta	ate aid for 2014-2020 (OJ
c	Communication from the Commission favour of banks in the context of			
2. 2. 1. A)	unemployment rate is — higher than creating new higher than in creating is — wes — listhere a risk of disru	pjective of common in ocated in a region of some of the following the Union average, power employment in the the national average new employment in the ption to an important for any competitor significant.	or regions (at NUTS	nied by difficulty ir erned panied by difficulty? replicate and where
	□yes		□no	
C)		any potential negative	ystemic role in a particu e consequences (for ex-	
	□yes		□no	
D)	Is there a risk of inter	ruption to the continu	uity of provision of an S	SGEI?
	□yes		□no	

Would the failure or adverse incentives of credit markets push an otherwise viable

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□yes	□no			
Would the exit from the market of the undertaking concerned lead to an irremediabl loss of important technical knowledge or expertise?				
□yes	□no			
Would the failure of the beneficiary involve any similar situation of severe hardship not listed above?				
□yes	□no			
If you have answered yes to any of the questions in points A to G, please fully substantiate the answer(s), including reference to the further supporting evidence or documents provided in annex.				
Appropriateness/Form of aid				
Is the aid in the form of loan	guarantees or loans?			
□yes	□по			
If you have answered yes, please describe the terms of the loan or the guarantee and attach the relevant documents (e.g. draft loan agreement, draft guarantee).				
is the interest rate of the loan (or, where relevant, the total financial cost of the guaranteed loan, including the interest rate of the loan and the guarantee premium set at a rate not lower than the reference rate set out by the Commission in its Reference Rate Communication ⁽⁸³⁾ for weak undertakings offering normal levels collateralisation?				
□yes	□no			
Please explain the use to which the rescue aid will be put: will the rescue aid be used to finance structural measures, such as acquisition of significant businesses or asset other than those required during the rescue period for the survival of the beneficiary.				
□yes	□по			
□yes If so, please explain.	□по			
If so, please explain. Will the loan be reimbursed o	r will the loan guarantee come to an end within a period of the disbursement of the first instalment to the beneficiary			
If so, please explain. Will the loan be reimbursed o	r will the loan guarantee come to an end within a period of			

a restructuring plan;

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the

Commission Regulation (EC) No 794/2004. (See end of Document for details)

a liquidation plan setting out in a substantiated way the steps leading to the liquidation of the beneficiary within a reasonable time frame without further aid?

□yes	□no

2.3. Proportionality of the aid/aid limited to the minimum

Is the amount of the rescue aid determined in accordance with the formula set out in Annex I to the Guidelines?

FVAS	Πno.
□yes	
3	

If so, please provide the calculation of the amount of the rescue aid in accordance with the formula.

If the amount of the rescue aid exceeds the result of calculations on the basis of the formula set out in Annex I to the Guidelines, please provide a duly justified liquidity plan setting out the beneficiary's liquidity needs for the coming six months.

2.4. Negative effects — 'One time, last time' principle

Has the undertaking (or the group to which it belongs) already received in the past any rescue aid, restructuring aid or temporary restructuring support⁽⁸⁴⁾ and/or any non-notified aid?

□yes	□no
------	-----

If so, please provide full details (date, amount, reference to previous Commission decision if applicable, etc.)⁽⁸⁵⁾.

3. Other information

Please indicate here any other information you consider relevant for the assessment of the measure(s) concerned under the Guidelines:

...

PART III.3.B

Supplementary information sheet on aid for restructuring non-financial undertakings in difficulty: individual aid

This supplementary information sheet must be filled in for the notification of individual restructuring aid covered by the Guidelines on State aid for rescuing and restructuring non-financial undertakings in difficulty (86) ('Guidelines').

1. Eligibility

- 1.1. Undertaking in difficulty
- A) Is the undertaking a limited liability company⁽⁸⁷⁾, where more than half of its subscribed share capital has disappeared as a result of accumulated losses⁽⁸⁸⁾?

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	ere at least some members have unlimited liability	
	nd where more than half of its capital listed in the das a result of accumulated losses?	
□yes	□no	
	ective insolvency proceedings or does it fulfil ing placed in collective insolvency proceedings	
□yes	□no	
In the case of an undertaking that is not an SME: has the undertaking's book debt to equity ratio been greater than 7,5 for the past two years		
andhas the undertaking's EBI past two years?	ITDA interest coverage ratio been below 1,0 for t	
□yes	□no	
substantiate the answer, including to or documents provided in annex (la sheets, or court decision opening co	my of the questions in points A to D, pleareferences in the answer to the supporting evider atest profit and loss account statements with balar ollective insolvency proceedings on the company placed under insolvency proceedings at the requiry law are met, etc.).	
Newly created undertaking or large	er business group	
When was the undertaking created?		
Since when has the undertaking been operating?		
Does the company belong to a larger business group?		
Does the company belong to a larg		

1.3. Sectoral scope

Is the undertaking active in:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

		T		
A)	the coal sector ^a :	□yes		□no
B)	the steel sector ^b :	□yes		□no
C)	sectors covered by specific rules for financial institutions ^c :	□yes		по
a	As defined in Decision 2010/787/E	U.		
b	As defined in Annex IV to the Com C 209, 23.7.2013, p. 1).	munication from the Commi	ssion: Guidelin	es on regional state aid for 2014-2020 (OJ
c				of the State aid rules to support measures on') (OJ C 216, 30.7.2013, p. 1).
1.4. A)	SGEI provider: Is the undertaking pro	oviding services of g	eneral econ	omic interest?
	□yes		□no	
B)		If the answer to point A is yes, please describe the service(s) of general economic interest and provide a copy of the entrustment act(s).		
C)	compensation the un of compensation and	If the answer to point A is yes, please indicate the amount of the public service compensation the undertaking is receiving, describe the methodology of calculation of compensation and provide the relevant legal basis establishing the methodology of calculation of the public service compensation.		
2.	Compatibility with	Compatibility with the internal market		
2.1.	Contribution to an ob-	Contribution to an objective of common interest:		
A)	unemployment rate is — higher than creating new — higher than	Is the beneficiary located in a region or regions (at NUTS level 2) where the unemployment rate is one of the following: — higher than the Union average, persistent and accompanied by difficulty is creating new employment in the region or regions concerned — higher than the national average, persistent and accompanied by difficult in creating new employment in the region(s) concerned.		
	□yes		□no	
B)		for any competitor		which is hard to replicate and tep in (for example, a national
	□yes		□no	
C)		Does the undertaking have any important systemic role in a particular region or sector Would its exit have any potential negative consequences (for example as a supplie		

of an important input)?

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□no
tinuity of provision of an SGEI?
□no
s of credit markets push an otherwise viable
□no
indertaking concerned lead to an irremediable or expertise?
□no
volve any similar situation of severe hardship
ı I

- H) If you have answered yes to any of the questions in points A to G, please fully substantiate the answer(s), including a reference to the further supporting evidence or documents provided in annex.
- 2.2. Restructuring plan and return to long-term viability

Please provide the restructuring plan⁽⁹⁰⁾ aiming at restoring the long-term viability⁽⁹¹⁾ of the beneficiary within a reasonable timescale together with a market survey and a sensitivity analysis identifying the driving parameters of the beneficiary's performance and the main risk factors going forward (please follow as much as possible the indicative restructuring plan set out in Annex II to the Guidelines).

3. Need for State intervention/incentive effect

- 3.1. Please provide a comparison between the measures set out in the restructuring plan and a credible alternative scenario not involving State aid⁽⁹²⁾ demonstrating that in such an alternative scenario the relevant objective or objectives that you have identified in section 2.1 would not be attained, or would be attained to a lesser degree.
- 3.2. Please demonstrate that in the absence of the aid, the beneficiary would have been restructured, sold or wound up in a way that would not have achieved the objective of common interest identified in section 2.1.

4. **Appropriateness**

- 4.1. Please provide a short description of the State aid instruments chosen, including the form, amount and remuneration⁽⁹³⁾:
- 4.2. Please explain whether the problems of the beneficiary are caused by liquidity and/ or solvency issues, or by both:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

4.3.	Please demonstrate that the State aid instruments chosen are appropriate to address the problems identified in point 4.2 (that is to say liquidity or solvency issues). Proportionality of the aid/aid limited to the minimum			
5.				
5.1.	Own contribution			
A)	Does the own contribution of the beneficiary amount to at least 50 % of the restructuring costs ⁽⁹⁴⁾ ?			

□yes □no

- B) Please describe and quantify each category of restructuring costs to be incurred, provide for their total amount and indicate which percentage of the restructuring costs will be covered by the own contribution:
- C) Please describe and quantify the own contribution to the restructuring costs to be provided from the own resources of the beneficiary, its shareholders or creditors, or the business group to which it belongs, or from new investors:
- D) Please explain why you consider this own contribution to be real and aid-free:
- E) Please demonstrate that the own contribution is comparable to the aid granted in terms of effects on the solvency or liquidity position of the beneficiary⁽⁹⁵⁾ and, if not, explain why, with reference, if appropriate, to supporting documents (e.g. balance sheet, cash flow statements):

• • •

5.2. Burden-sharing

To be completed if the State aid is granted in a form that enhances the beneficiary's equity position⁽⁹⁶⁾

A) Have the beneficiary's losses been fully accounted for, attributed to and absorbed by existing shareholders and/or subordinated creditors?

□yes □no

- B) If the answer to point A is yes, please provide evidence, on the basis of an up-to-date analysis of the beneficiary's balance sheet situation.
- C) Will cash outflows from the beneficiary to holders of equity and/or subordinated debt be prevented during the restructuring period?

□yes	□no

- D) If the answer to point C is no, please explain the reason(s).
- E) Will the State receive a reasonable share of the future gains in value of the beneficiary, in view of the amount of State equity injected in comparison with the remaining equity of the company after losses have been accounted for?

□yes	□no

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- F) If the answer to point E is yes, please substantiate the answer and provide evidence accordingly.
- G) If the beneficiary's losses are not absorbed in full (see point A above) and/or cash outflows from the beneficiary to holders of equity and/or subordinated debt are not prevented during the restructuring period (see point C above), please explain the reasons, in particular why full implementation of those conditions would lead to disproportionate results:
- H) Are senior debt holders contributing to restoring the beneficiary's equity position?

	¬no
□ycs	

I) If the answer to point H is yes, please explain in which way senior debt holders will contribute.

6. **Negative effects**

6.1. 'One time, last time' principle

Has the undertaking (or the group to which it belongs) already received any rescue aid, restructuring aid or temporary restructuring support⁽⁹⁷⁾ and/or any non-notified aid?

□yes	□no

If yes, please provide full details (date, amount, reference to previous Commission decision if applicable, etc.)⁽⁹⁸⁾:

- 6.2. Measures to limit distortions of competition: Structural measures divestments and reduction of business activities
- A) Please describe the divestments of assets, reduction of capacity or market presence committed to be undertaken. Please show that divestments, write-offs and closure of loss-making activities included in the commitments are not necessary to restore long-term viability of the beneficiary. Please indicate in addition the relevant markets in which those divestments will take place and their timing⁽⁹⁹⁾. Please indicate as well whether the beneficiary will facilitate divestitures, for example through ring-fencing of activities and by agreeing not to solicit clients of the divested business.
- B) If structural measures exceptionally take the form of divestment of assets alone and do not involve the creation of a viable entity able to compete in the market, please demonstrate that no other form of structural measures would be feasible or that other structural measures would seriously jeopardise the economic viability of the undertaking:

Behavioural measures

C) Does the beneficiary agree to refrain from acquiring shares in any company during the restructuring period, except where indispensable to ensure its long-term viability and subject, in that case, to notification to and approval by the Commission?

□yes	□no

D) Does the beneficiary agree to refrain from publicising State support as a competitive advantage when marketing its products and services?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	□yes		□no		
E)	Are there any other b	pehavioural measures	foreseen?		
	□yes		□no		
Marke	t opening measures				
F)		g more open, sound		rities or by the beneficiary with etitive markets, for instance by	
	□yes		□no		
G)	If the answer to point F is yes, please describe which measures and in which market indicating how the measures are directly or indirectly linked to the beneficiary's activities:				
Calibr	ation of measures to lim	it distortions of comp	petition		
H)	Is any of the aid aim	ed at covering the soc	cial costs of	restructuring ⁽¹⁰¹⁾ ?	
	□yes		□no		
I)	If the answer to poin	If the answer to point H is yes, please specify:			
7.	Other information				
7.1.	Please indicate here any other information you consider relevant for the assessment of the measure(s) concerned under the Guidelines (e.g. as regards measures to increas employability of redundant workers or assistance with finding new employment):				
		PART III.3	. <i>C</i>		
		information sheet omporary restructur			
aid an		ng support schemes	covered by	ard to rescue aid, restructuring the Guidelines on State aid for ulty ⁽¹⁰²⁾ ('Guidelines').	
1.	Scope of the scheme	2			
1.1.	Does the scheme cor	Does the scheme concern provision of:			
(a)	rescue aid:	□yes		по	
(b)	restructuring aid:	□yes		□no	
(c)	temporary restructuring support:	□yes		□no	

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2.	Eligibility		
2.1.	Is the scheme limited to SMEs ⁽¹⁰³⁾ in difficulty or smaller State-owned undertain difficulty ⁽¹⁰⁴⁾ (unless clearly indicated otherwise hereafter together refere 'SMEs')?		
□yes		□no)
2.2.	Is the scheme limited to SMEs that f	fulfil o	one of the following eligibility criteria:
(a)	In relation to SMEs that are limited liability companies ⁽¹⁰⁵⁾ : more than half subscribed share capital has disappeared as a result of accumulated losses ⁽¹⁰⁶⁾		
	□yes		□по
(b)	In relation to SMEs which are companies where at least some members have unlimit liability for the debt of the company ⁽¹⁰⁷⁾ : more than half of the capital shown in to company accounts has disappeared as a result of accumulated losses?		
	□yes		□no
(c)	SMEs that are subject to collective insolvency proceedings or fulfil the criteria unde their domestic law for being placed in collective insolvency proceedings at the reques of their creditors?		
	□yes		□по
(d)	In relation to smaller State-owned undertakings: the undertaking's book debt to eq ratio has been greater than 7,5 and its EBITDA interest coverage ratio has been be 1,0 for the past two years?		
	□yes		□no
2.3.	Does the scheme foresee that rescue aid or temporary restructuring support, or both can be granted to SMEs that do not qualify as undertakings in difficulty but are merely facing acute liquidity needs due to exceptional and unforeseen circumstances?		
□yes		□no)
2.4.	If the answer to point 2.3 is yes, please explain how it will be assessed whether at SME faces acute liquidity needs and which kinds of circumstances are qualified a exceptional and unforeseen.		
2.5.	Does the scheme apply to newly-cre	ated S	SMEs?
□yes		□no)

Does the scheme apply to SMEs active in:

2.6.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

(a)	the coal sector ^a :	□yes	□no		
(b)	the steel sector ^b :	□yes	□по		
(c)	sectors covered by specific rules for financial institutions ^c :	□yes	по		
a	As defined in Decision 2010/787/E	U.			
b	As defined in Annex IV to the Con C 209, 23.7.2013, p. 1).	nmunication from the Commission	: Guidelines on regional state aid for 2014-2020 (OJ		
c			gust 2013, of the State aid rules to support measures immunication?) (OJ C 216, 30.7.2013, p. 1).		
3.	Maximum amount	of aid			
			A LA COME A AL A		
3.1.			nted to any one SME under the scheme including any aid obtained from other		
	sources or under other		including any aid obtained from other		
□ye	es 	□no			
3.2.	Please indicate the m	aximum amount of aid to	be granted to an SME under the scheme:		
4		the internal market			
	ne case of rescue, restructu		estructuring support		
4.1.	Contribution to an ol	ojective of common interes	est:		
a) Does the scheme apply only in cases where the failure of the benefic likely to involve social hardship or a market failure, in particular:					
		an innovative SME or an tial negative consequence	SME with high growth potential would es:		
	□yes		□no		
		— the exit of an SME with extensive links to other local or regional SMEs would have potential negative consequences:			
	□yes		□no		
		the failure or adverse incentives of credit markets would push an otherwise viable SME into bankruptcy:			
	□yes		□no		
	— similar situ	similar situations of hardship duly substantiated by the beneficiary:			
	□yes		□no		

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

b) If the answer to any of the questions in point (a) is yes, please fully substantiate the answer(s) and explain the criteria on the basis of which the national authorities are going to assess the contribution to the objectives of common interest.

In the case of restructuring aid

4.2. Restructuring plan and return to long-term viability

In relation to the grant of restructuring aid, does the scheme require the provision of a restructuring plan⁽¹⁰⁸⁾ aiming at restoring the long-term viability⁽¹⁰⁹⁾ of the beneficiary within a reasonable timescale (please see an indicative restructuring plan in Annex II to the Guidelines)?

	•	<i>7</i>
□yes	ום	10
5.	Need for State intervention and incent	ive effect
5.1.	national authorities compare the measu credible alternative scenario not involv	ng aid, does the scheme require that the res set out in the restructuring plan with a ring State aid (110) demonstrating that in such

national authorities compare the measures set out in the restructuring plan with a credible alternative scenario not involving State aid⁽¹¹⁰⁾ demonstrating that in such an alternative scenario the relevant objective or objectives of common interest, as described in section 4.1 above, would not be attained, or would be attained to a lesser degree? In particular, does the scheme require a demonstration that in the absence of the aid, the beneficiary would have been restructured, sold or wound up in a way that would not have achieved the objective or objectives of common interest, as described in section 4.1 above?

□yes	□no

5.2. If the answer to point 5.1 is yes, please explain according to which criteria the national authorities will assess it.

6. **Appropriateness**

In the case of rescue aid

6.1. Is the aid granted under the scheme restricted to loan guarantees or loans?

□yes	□no
------	-----

6.2. Does the scheme require that the financial cost of the loan (or, in the case of loan guarantees, the total financial cost of the guaranteed loan, including the interest rate of the loan and the guarantee premium) is set at a rate not lower than the reference rate set out by the Commission in its Reference Rate Communication⁽¹¹¹⁾ for weak undertakings offering normal levels of collateralisation?

□yes	□no

6.3. Does the scheme provide that rescue aid is granted for no longer than six months, during which an analysis must be made of the beneficiary's position?

□yes □no

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

6.4.	Does the scheme provide that, within six months after granting the rescue a loan will be reimbursed or the guarantee terminated, unless before that time eith the national authorities have approved a restructuring plan, or liquidation plan, the beneficiary has submitted a simplified restructuring plan ⁽¹¹²⁾ (in case of tem restructuring support)?	
□yes		□no
6.5.	Does the scheme provide that the rescue aid must not be used to finance structura measures, such as acquisition of significant businesses or assets, other than thos required during the rescue period for the survival of the beneficiary?	
□yes		□no
In the c	ease of restructuring aid	
6.6. In the c	extent the problems of the beneficiar	teria the national authorities will assess to what y relate to liquidity or solvency or both, and how nents which will address the problems identified
6.7.	Is the temporary restructuring suppolean guarantees or loans?	ort to be granted under the scheme restricted to
□yes		□no
6.8.	Does the scheme require that the financial cost of the loan (or, in the case of loar guarantees, the total financial cost of the guaranteed loan, including the interest rate of the loan and the guarantee premium) be set at a rate not lower than the reference rate set out by the Commission in its Reference Rate Communication for weak undertakings offering normal levels of collateralisation?	
□yes		□no
6.9.	Does the scheme require that the remuneration for the temporary restructuring support be increased by at least 50 basis points 12 months after the disbursement of the first instalment to the beneficiary (less any immediately preceding period of rescue aid)?	
□yes		□no
6.10.	Does the scheme provide that tempor than 18 months, less any immediatel	ary restructuring support be granted for no longer y preceding period of rescue aid?
□yes		□no

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

6.11.	Does the scheme provide that not later than six months after disbursement of the
	first instalment of the temporary restructuring support to the beneficiary, less any
	immediately preceding period of rescue aid, the national authorities will approve a
	simplified restructuring plan?

□yes	□no

6.12. Does the scheme provide that within 18 months from the date of granting of the temporary restructuring support, less any immediately preceding period of rescue aid, the loan will be reimbursed or the guarantee terminated, unless before that time a restructuring plan or liquidation plan of the beneficiary has been approved by the national authorities?

□yes	□no

7. Proportionality of the aid or aid limited to the minimum

In the case of rescue aid and temporary restructuring support

- 7.1. Amount of aid
- (a) Does the scheme provide that the amount of aid will not exceed the result of the calculation on the basis of the formula set out in Annex I to the Guidelines?

□yes	□no
------	-----

(b) If the answer to point (a) is no, does the scheme require the preparation of a liquidity plan setting out the beneficiary's liquidity needs for the next six months (18 months in the case of temporary restructuring support)?

□yes	□по
------	-----

(c) If the answer to point (b) is yes, please explain on which basis and according to which information the national authorities will examine whether the liquidity plan setting out the beneficiary's liquidity needs for the next six months (18 months in the case of temporary restructuring support) is duly justified:

In the case of restructuring aid

- 7.2. Own contribution
- (a) Does the scheme require provision of a real and aid-free contribution to the restructuring costs from the own resources of the beneficiary, its shareholders, creditors, the business group to which it belongs or from new investors amounting to at least 40 % of the restructuring costs in the case of medium-sized enterprises or 25 % of the restructuring costs in the case of small enterprises?

□yes	$ \Box no$
J	

(b) If the answer to point (a) is yes, please explain which elements the national authorities will take into account to assess whether the own contribution is real and aid-free:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

(c)	Does the scheme require that the own contribution should be compara granted in terms of effects on the solvency or liquidity position of the be			•			
	□yes			□по			
(d)	If the artit:	If the answer to point (c) is yes, please explain how the national authorities will assess it:					
7.3.	Burden	-sharing					
		if the scheme provides that tequity position ⁽¹¹⁴⁾ :	the Sta	te aid can be granted in a form enhancing			
(a)	have b			rvention may only take place after losses outed to the existing shareholders and/or			
	□yes			по			
(b)	Will cash outflows from the beneficiary to holders of equity and subordinated debt be prevented during the restructuring period to the extent legally possible?						
	□yes			□no			
(c)	If the answer to point (b) is no, please explain on the basis of which criteria the national authorities will assess whether such cash outflows would not disproportionately affect those that have injected fresh equity.						
(d)		e national authorities allow (a) and (b) above?	any ex	ceptions from the conditions described in			
		no					
		yes. If so, please explain.					
(e)	gains in com	n value of the beneficiary,	in vie	ill receive a reasonable share of the future w of the amount of State equity injected y of the company after losses have been			
	□yes			□no			
8.	Negati	ve effects					
8.1.	'One ti	me, last time' principle:					
				o any SME which has received rescue aid, in the past ⁽¹¹⁶⁾ and/or any non-notified aid?			
□yes			□nc)			
In the c		itus otravino oi d					

In the case of restructuring aid

8.2. Measures to limit distortions of competition⁽¹¹⁷⁾:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	Structural	measures —	divestments	and	reduction	of	business	activitie
--	------------	------------	-------------	-----	-----------	----	----------	-----------

(a)	Does the scheme require from beneficiaries divestments of assets, reduction of capacity or market presence ⁽¹¹⁸⁾ together with indication of the relevant markets or which those divestments will take place ⁽¹¹⁹⁾ and their timing ⁽¹²⁰⁾ ?				
	□yes	□по			
(b)	divestments on a going-concer	structural measures should normally take the form of in basis of viable stand-alone businesses that, if operated compete effectively in the long term?			
	□yes	□по			
(c)	If the answer to point (b) is no, does the scheme provide that, where there of such entities, the beneficiary might carve out and subsequently divest a and appropriately funded activity, creating a new and viable entity that show to compete in the market?				
	□yes	□по			
(d)	tural measures in the form of divestment of assets alone, a of a viable entity able to compete in the market, does it the beneficiary that no other form of structural measures are structural measures would seriously jeopardise the pertaking?				
	□yes	□по			
Behav	iou ral measures				
(e)		t the beneficiary refrain from acquiring shares in any aring period, except where indispensable to ensure the			
	□yes	□по			
Does the scheme require that the beneficiary refrain from publicising S a competitive advantage when marketing their products and services?					
	□yes	□по			
(g)	Does the scheme provide for the beneficiary to refrain from engaging in commerce behaviour aimed at a rapid expansion of its market share relating to specific productor geographic markets by offering terms (for example as regards prices and other commercial conditions) which cannot be matched by competitors that are not in received State aid?				
	□yes	□по			
(h)	If the answer to point (g) is yes, under what circumstances? Please explain.				

- (i) Does the scheme foresee any other behavioural measures?

Status: Point in time view as at 31/01/2020. Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details) П no yes. If so, please describe. П Market opening measures (j) Does the scheme allow for any measures to be adopted by the national authorities or by the beneficiary that are aimed at promoting more open, sound and competitive markets, for instance by favouring entry and exit(121)? □yes If the answer to point (i) is yes, please describe: (k) 9. 9.1. Does the scheme apply to SMEs in an assisted area? □no □yes 9.2. Do specific provisions apply to SMEs in assisted areas under the scheme? □yes □no 9.3. If the answer to point 9.2 is yes, please explain which specific provisions apply and why they are justified. Do the national authorities intend to accept a contribution which constitutes less than 9.4.

- 40 % of the restructuring costs in the case of medium-sized enterprises or less than 25 % of the restructuring costs in the case of small enterprises?

□yes	□no
------	-----

9.5. If the answer to point 9.4 is yes, please explain how the national authorities will apply the requirements concerning the measures to limit distortions of competition in order to limit the negative systematic impacts for the region:

10. Other information

Please provide any other information that would be of relevance to assess the notified aid measure under the Guidelines (e.g. as regards measures to increase employability of redundant workers or assistance with finding new employment):

PART III.4

Supplementary Information Sheet on aid for films and other audiovisual works

This supplementary information sheet must be used for notifications of aid covered by the Commission Communication on State aid for films and other audiovisual works (122)

1. Characteristics of the notified aid measure(s)

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 1.1. Please describe as accurately as possible the purpose of the aid, where appropriate, for each measure:
- 1.2. Please describe the scope of each measure, in terms of:
- 1.2.1. the type of activities covered (e.g. development, production, distribution):
- 1.2.2. the type of works covered (e.g. cinematographic works, television series, transmedia projects):
- If the aid scheme includes a support measure for transmedia projects, do the aided 1.3. activities directly relate to the film production component of the work?

□yes	□по

- 1.4. Please indicate what provisions exist to guarantee the cultural objective of the aid:
- 2. **Conditions for eligibility**
- 2.1. Please indicate the conditions for eligibility of activities or works under the planned aid measure:
- 2.2. Please indicate the conditions for eligibility of beneficiaries under the planned aid measure:
- Does the scheme differentiate on grounds of nationality or place of residence?
- Are beneficiaries obliged to fulfil any conditions other than that of being represented by a permanent agency at the moment of the payment of the aid?
 - If the aid has a tax component, must the beneficiary fulfil any obligations or conditions other than that of having taxable revenue in the territory of the Member State?
 - Other conditions:

3. **Territorial spending obligations**

- Please indicate whether the measure includes provisions requiring the producer to 3.1. spend the production budget, or parts of it, in the territory of the Member State or in one of its subdivisions:
- 3.1.1. In order to be eligible for aid?

□yes		□no
3.1.2.	As a condition attached to the granting	g of the aid?
□yes		□no

- 3.2. Do the conditions of territorial spending apply to certain specific items of the production budget?
- 3.3. If it is necessary to comply with a minimum degree of territorial spending in order to be eligible for the aid, please describe the nature of the requirements:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 3.3.1. Of implicit nature (e.g. a minimum number of days shooting the production in the territory):
- 3.3.2. Of explicit nature (e.g. a minimum amount or percentage of expenditure):
- 3.4. In case there are territorial conditions attached to the granting of the aid, please explain:
- 3.4.1. Is the aid calculated as a percentage of territorial expenditure?
- 3.4.2. Is the required territorial spending calculated with regard to the overall budget of the film?
- 3.4.3. Is the required territorial spending calculated with regard to the amount of aid granted?

4. Eligible costs

Please specify the costs which may be taken into account to determine the amount of aid.

5. **Aid intensity**

5.1. Please indicate whether the scheme allows for aid intensities exceeding 50 % of the production budget? If so, please clarify the types of works concerned and the aid intensity limits set.

• • •

- 5.2. If the concept of 'difficult audiovisual works' is used, please indicate the categories of works covered by this concept (that is to say, please indicate the definition used).
- 5.3. If scriptwriting or development is aided under the scheme: are the costs of scriptwriting and development included in the production budget and taken into account for calculating the aid intensity of the audiovisual work?
- 5.4. If distribution and promotion activities are supported under the scheme: what are the aid intensities allowed under the scheme?

6. Film heritage

If applicable, please provide information on the measures taken with regard to film heritage.

7. **Compatibility**

- 7.1. Please provide a reasoned justification in support of compatibility of the aid in the light of the principles set out in the Commission Communication on State aid for films and other audiovisual works.
- 7.2. If the scheme concerns aid to cinemas, please provide a reasoned justification in support of the compatibility of the aid as aid to promote culture within the meaning of Article 107(3)(d) TFEU, in particular the necessity, adequacy and proportionality of the aid.

8. **Other Information**

Please indicate here any other information you consider relevant to the assessment of the measure(s) concerned under the Communication on State aid for films and other audiovisual works.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

PART III.5

Supplementary Information Sheet on State aid to broadband

This supplementary information sheet should be used for notifications of aid covered by the EU Guidelines for the application of State aid rules in relation to rapid deployment of broadband networks⁽¹²³⁾('Broadband Guidelines').

	s ⁽¹²³⁾ ('Broadband Guide	v	id futes in retation to	парна аср	юутені ој огошошни	
1.	Characteristics of the notified aid measure					
1.1.	Please describe the objective of the aid measure:					
1.2.	Please explain how the aid measure fits with the national broadband strategy and the Union objectives (including for the EU 2020 strategy and the Digital Agenda ⁽¹²⁴⁾).					
1.3.	Please present the rationale for public intervention and explain the anticipated benefits of the aid measure (e.g. economic and social benefits, increased broadband coverage and internet penetration rates, etc.).					
1.4.	What category of netv	vork does	the aid measure aim t	o support	?	
	backbone networks (o	r trunk ne	etworks);			
	backhaul networks (or regional, or middle-mile networks);					
	access networks (or last mile networks).					
1.5.	What elements of the network does the aid measure aim to support?					
	Passive infrastructure elements;					
	Active infrastructure e	equipmen	t.			
1.6.	What type of broadba	nd netwo	rks does the aid measu	ire aim to	support?	
	basic broadband networks (providing at least 2Mbps download speed);					
	next generation access networks ('NGA') ⁽¹²⁵⁾ ;					
	ultrafast broadband networks ⁽¹²⁶⁾ .					
1.7.	What types of areas a respect to the different classification on the b	it segmen				
	Basic White		Basic Grey		Basic Black	
	NGA White		NGA Grey		NGA Black	
	Ultrafast White		Ultrafast Grey		Ultrafast Black	

- 1.8. Please provide any other relevant information useful to clarify the general context of the aid measure:
- 1.9. What type of investment and business model will be adopted 427?

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2. Process and granting of the aid measure

Mapping, coverage analysis

- 2.1. What is the scope of the aid measure in terms of territorial coverage?
- 2.2. Please provide information, including the date, and submit the results of the detailed mapping and coverage analysis undertaken to clearly identify the target areas:

Public consultation

2.3. Please describe the process and the results of the open transparent public consultation giving the opportunity to all interested stakeholders to comment on the planned aid measure. Please submit the relevant internet links on which the information about the measure has been published:

Competitive selection procedure

- 2.4. In case of deployment and/or operation of the subsidised network by a third-party operator, please confirm that a competitive selection procedure will be conducted in line with the EU Public Procurement Directives⁽¹²⁸⁾. Please provide all relevant information in that respect:
- 2.5. Please provide information on how the most economically advantageous offer is to be selected (including the award criteria and the relative weighting which is given to each of the criteria chosen) taking into account qualitative criteria (for instance, coverage, sustainability of the technological approach or the impact of the solution on competition) and price:
- 2.6. Are there are any minimum service requirements that the subsidised network has to comply with (such as minimum bandwidth, sustained services, minimum geographical coverage, etc.)?

□yes		□no
	If yes, please describe: ogical neutrality	
2.8.	Is the aid measure technologically neu	utral?
□yes		□по

- 2.9. If the answer to point 2.8 is yes, please describe how that principle is ensured: *Use of existing infrastructure*
- 2.10. Please submit a map depicting the existing infrastructures in place in the country or region concerned, including also any new infrastructure(s) planned by commercial operators in the near future, that is to say within a three year period:
- 2.11. Please explain how it is ensured that the operators wishing to participate in the selection procedure provide all relevant information on any existing infrastructure they own or control in the target area:

Wholesale access

2.12. Please explain what type of 'wholesale access' obligations will be imposed on the subsidised network (including access to passive and active infrastructure, the right to

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

use ducts and poles, dark fibre and street cabinets), and how long the access obligations will be maintained:

Wholesale access pricing

- 2.13. Please explain how access prices will be benchmarked: *Claw-back and monitoring mechanism*
- 2.14. Will a claw-back mechanism apply to the aid measure?

□yes	□no

- 2.15. If the answer to point 2.14 is yes, please describe this mechanism, its characteristics and duration:
- 2.16. Please explain how the monitoring of the aid measure will be organised:
- which public authority will undertake the monitoring?
- what aspects of the measure will be subject to monitoring?
- which performance criteria will be analysed?
- what will be the timing of the monitoring?

National regulatory authority ('NRA')

- 2.17. Please describe the role of the NRA in particular regarding: identification of target areas, wholesale access pricing including benchmarking, dispute resolution, etc.:
- 2.18. Please report the position of the NRA on the proposed aid intervention and, where available, the opinion of the national competition authority:

3. Compatibility criteria

Please explain how the notified aid measure complies with the conditions specified in Section 2.5 of the Broadband Guidelines, in particular with regard to: *Objective of common interest*

(a) Is the aid measure aimed to address well-defined objectives of common interest?

□yes	□no

(b) If the answer to (a) is yes, please give a description of the objectives of common interest pursued by the aid measure:

Market failure

(c) Does the aid measure address a market failure or important inequalities hampering the availability of broadband services?

□yes	□no

(d) If the answer to (c) is yes, please describe this market failure and provide a general overview of the broadband market in the country or region concerned and of the areas targeted by the aid measure.

That overview should contain information on the current level of broadband coverage, internet penetration rate (number of connected households, enterprises), breakdown of services availability by technology, major trends on the (national or regional) broadband market, rural/urban divide in terms of broadband coverage, comparison of

(e)

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

retail prices with those charged for the same services offered in more competitive but otherwise comparable areas or regions of the country, available technological solutions for broadband development and provision of connectivity services, competitive situation in the electronic communication markets (structure and dynamics of the markets), overview of the national regulatory framework and of existing regulatory obligations for electronic communication operators:

Appropriate instrument

(f)	Have any other alternative (less distortive) instruments already been implemented
	4 11 4 1 61 11 1 1 1 4 1 1 61 11

Please demonstrate the appropriateness of the aid measure:

to enable the supply of broadband services and to overcome the lack of broadband connectivity, including *ex ante* regulation or demand stimulation?

□yes	□no

- (g) If the answer to (f) is yes, please describe those initiatives and explain why they were deemed insufficient to achieve the desired broadband development objectives:
- (h) Have the existing network operators undertaken any private investment in the target areas over the last three years?

□yes	□no
,	

(i) If the answer to (h) is yes, please describe those investments and explain why existing broadband infrastructure is not sufficient to satisfy the needs of citizens and business users and why State aid is necessary:

Incentive effect

(j) Please describe the positive effects that the aid measure is expected to generate: *Step Change*

(k) Will the aid measure ensure a 'step change' in terms of broadband availability?

□yes	□no

- (l) If the answer to (k) is yes, please demonstrate this by means of comparison to the existing and planned network deployment (that is to say before and after the planned intervention). In particular, please describe in detail whether and to what extent significant new investment in the broadband network will be undertaken and what new capabilities will be brought to the market in terms of broadband service availability and quality:
- (m) Please explain the expected changes in the behaviour of the beneficiary(s) of the aid measure:
- (n) Please demonstrate that similar broadband network investment would not have been undertaken within the same time frame and/or under the same terms and conditions in the absence of State aid:

Proportionality

(o) Please explain how it is ensured that the aid measure is limited to the minimum necessary:

Distortion of competition and effect on trade

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

(p) Please explain what potential negative effects on competition and trade the aid measure may have (for example: potential to cause crowding out of private investments or reinforcement of a dominant position) and what elements in the design of the measure (129) could minimize those risks:

4. **Other information**

Please indicate here any other information you consider relevant to the assessment of the measure concerned under the Broadband Guidelines or any other information that is relevant from the point of view of Union competition and internal market rules⁽¹³⁰⁾.

PART III.6

Supplementary information sheet for State aid for environmental protection and energy

This supplementary information sheet must be used for the notification of any aid covered by the Guidelines on State aid for environmental protection and energy 2014–2020 (hereinafter 'the EEAG')⁽¹³¹⁾.

All documents provided by Member States as annexes to the notification form must be numbered and document numbers must be indicated in the relevant sections of this supplementary information sheet.

If there are several beneficiaries involved in an individual aid measure, the relevant information must be provided for each of them.

This supplementary information form is to be filled in in addition to the 'Part I. General information' form.

Scope

General Block Exemption Regulation

Before completing this notification form you should consider whether this measure could be implemented under Commission Regulation (EU) No 651/2014 (the General Block Exemption Regulation ('the GBER')⁽¹³²⁾, and in particular section 7 of chapter III thereof (Aid for environmental protection).

Could the aid be implemented under the GBER?

□yes	по

If the aid could be implemented under the GBER, please explain why the measure is notified. *De minimis Regulation*

Before completing this notification form you should consider whether this measure could be implemented under the *de minimis* Regulation⁽¹³³⁾:

Could the aid be implemented under the *de minimis* Regulation?

□yes	□no

If the aid could be implemented under the *de minimis* Regulation, please explain why the measure is notified.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

This form should only be completed for notification of State aid which is intended to be implemented under the EEAG. Please fill in the relevant parts of the notification form corresponding to the character of the notified measure.

Section A:

General information on environmental and energy aid measures

1. Please specify the type of aid below and then complete the appropriate part of section B of this supplementary information sheet ('General compatibility assessment'). If the scheme provides more than one type of aid in the list below you should complete section B for each box ticked.

If the measure is aid in the form of reductions in, or exemptions from, environmental taxes, or from funding the support for energy from renewable energy sources⁽¹³⁴⁾, please fill in Section C of the form ('Compatibility assessment for aid in the form of reductions in or exemptions from environmental taxes and in the form of reductions in funding support for energy from renewable sources').

- (a) Aid for undertakings which go beyond Union standards or which increase the level of environmental protection in the absence of Union standards;
- (b) Aid for the acquisition of new transport vehicles which go beyond Union standards or which increase the level of environmental protection in the absence of Union standards;
- (c)□ Aid for early adaptation to future Union standards;
- (d)□ Aid for renewable energy sources;
 - Please confirm that the measure only supports energy from renewable energy sources as defined in point 19(5) and 19(11) of the EEAG.
 - For biofuels, please confirm that the measure only supports sustainable biofuels as defined in point 19(9) of the EEAG.
- (e)□ Aid for energy efficieny measures;
 - Please confirm that the measure only supports efficiency measures as defined in point 19(2) of the EEAG.
- (f)□ Aid for energy efficient district heating or cooling measures;
 - Please confirm that the measure only supports energy efficient district heating or cooling measures as defined in point 19(14) of the EEAG.
- $(g)\square$ Aid for high efficient cogeneration;
 - Please confirm that the measure only supports high efficient cogeneration as defined in point 19(13) of the EEAG.
- (h)□ Aid for waste management;
- (i)□ Aid for environmental studies;
- (j)□ Aid for the remediation of contaminated sites;
- (k)□ Aid for the relocation of undertakings;
- (l)□ Aid involved in tradable permits schemes;

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

(m)□	Aid for energy infrastructure;	
(n)□	Aid for CO ₂ capture, transport and storage (CCS);	
(o)□	Aid for generation adequacy measures;	
(p)□	Aid in the form of reductions in or exemptions from environmental taxes;	
(q)□	Aid in the form of reductions in funding support for energy from renewable sources.	
2.	Please provide a detailed description of the main characteristics of the notified measure (objective, likely effects of the aid, aid instrument, aid intensity, beneficiaries, budget, conditions, etc.).	
3.	Can the aid be combined with other aid?	
□yes		□по
	answer is yes, please explain and fill in of section B.	n the section on cumulation in the proportionality
4.	In case the notified measure concerns a scheme, please tick the box in this point to confirm that any aid awarded under the notified scheme will be notified individually if it exceeds the thresholds laid down in point 20 of the EEAG:	
	yes	

- 5. In case the notified individual aid is based on an approved scheme, please provide details concerning that scheme (case number, title of the scheme, date of Commission approval):
- 6. If applicable, please indicate the exchange rate which has been used for the purposes of the notification:
- 7. Please indicate whether such conditions are attached to the measure, including its financing method when it forms an integral part of it, that can entail a non-severable violation of Union law (point 29 of the EEAG).

□yes	□no
------	-----

If the answer is yes, please explain how compliance with Union law is ensured.

Section B:

General compatibility assessment

1. Contribution to an objective of common interest

To answer the questions in this section, please refer to section 3.2.1 of the EEAG and where further specified to sections 3.2-3.6 and 3.8-3.10 of the EEAG. State aid schemes

1. In the light of the objectives of common interest addressed by the EEAG, please indicate the environmental or energy objectives pursued by the notified measure⁽¹³⁵⁾.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Please provide a detailed description of each distinct type of aid to be granted under the notified measure:

2. Has the notified measure already been applied in the past?

	T
□yes	□no

If yes, please indicate its results in terms of environmental protection or energy system improvements, the relevant case number and date of Commission approval and, if possible, attach national evaluation reports on the measure:

- 3. If the measure is new, please indicate the expected results and the period over which they will be achieved, as well as how those results will contribute to the achievement of the objective pursued:
- 4. Please explain how you ensure that the generation adequacy investment will not contradict the objective of phasing out environmentally harmful subsidies including for fossil fuels, in line with point 220 of EEAG. How are for example demandside management and interconnection capacity taken into account? Is there for example a preference for low carbon capacity providers in case of equivalent technical performance?
- 5. In the case of generation adequacy, please explain and clearly define the generation adequacy problem that is expected to arise and the consistency with the generation adequacy analysis regularly carried out by ENTSO-E⁽¹³⁶⁾. Please provide in the answer details in view of the considerations set out in point 221 of the EEAG.
- 6. In the case of waste management, please confirm that the following conditions are met:
- (a) the waste hierarchy principle is respected (see point 118 of the EEAG)
- (b)□ the investment is aimed at reducing pollution generated by undertakings ('polluters') and does not extend to pollution generated by the beneficiary of the aid (see point 158(a) of the EEAG)
- (c) the aid does not indirectly relieve the polluters from a burden that should be borne by them under Union law, or from a burden that should be considered as a normal company cost for the polluters (see point 158(b) of the EEAG)
- (d) the investment goes beyond the 'state of the art' (see point 158(c) of the EEAG)
- (e) the treated materials would otherwise be disposed of, or be treated in a less environmentally friendly manner (see point 158(d) of the EEAG)
- (f) the investment does not merely increase demand for the materials to be recycled without increasing collection of those materials (see point 158 of the EEAG)

Furthermore, please provide details and evidence of compliance with the conditions mentioned in this point:

7. In the case of tradable permits, please describe in detail the tradable permit scheme, including, inter alia, the objectives, the granting methodology, the authorities or entities involved, the role of the State, the beneficiaries and the procedural aspects:

Please explain how:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (a) the tradable permit scheme is set up in such a way as to achieve environmental objectives beyond those intended to be achieved on the basis of Union standards that are mandatory for the undertakings concerned:
- (b) the allocation is carried out in a transparent way and based on objective criteria and on data sources of the highest quality available:
- (c) the total amount of tradable permits or allowances granted to each undertaking for a price below their market value is not higher than its expected needs as estimated for the situation in the absence of the trading scheme:
- (d)□ the allocation methodology does not favour certain undertakings or certain sectors;

 In case the allocation methodology favours certain undertakings or certain sectors, please explain how that is justified by the environmental logic of the scheme itself or is necessary for consistency with other environmental policies:
- (e) new entrants must not in principle receive permits or allowances on more favourable conditions than existing undertakings operating on the same markets:
- (f) granting higher allocations to existing installations compared to new entrants should not result in creating undue barriers to entry:

Please provide details and evidence demonstrating compliance with the conditions mentioned in this point:

Individually notifiable aid — additional information

8. If the aid is to be made available to individual undertakings please provide as much as possible quantifiable information to demonstrate the contribution to the relevant common objective for the notified measure.

2. Need for State Aid

To answer the questions in this section, please refer to section 3.2.2 of the EEAG and where further specified to sections 3.2-3.6 and 3.8-3.10 of the EEAG. State aid schemes

- 1. Please identify the market failure which is hampering the level of environmental protection or a well-functioning, secure, affordable and sustainable energy market (see point 35 of the EEAG):
- 2. If the market failure is already being addressed by other policies or measures, please provide evidence that the notified measure is only targetted at the residual market failure (see point 36 of the EEAG):
- 3. Please provide detailed information on the nature and the reasons why the notified measure is necessary:
- 4. In the case of investment in energy infrastructure projects, please explain (see points 206 208 of the EEAG):
- (a) To what extent does the measure benefit projects of common interest as defined in Regulation (EU) No 347/2013, smart grids or infrastructure in assisted regions?
- (b) To what extent does the market failure lead to a sub-optimal provision of the necessary infrastructure?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- To what extent is the infrastructure accessible to third parties and subject to tariff (c) regulations?
- 5. In the case of generation adequacy please provide the following information set out in points 222 to 224 of the EEAG.
- Assessments of the impact of variable generation, including that originating from (a) neighbouring systems.
- Assessment of the impact of the demand-side participation, including a description of (b) measures to encourage demand-side management.
- Assessment of the actual or potential existence of interconnectors including a (c) description of projects under construction and planned projects.
- (d) Assessment of any other element which might cause or exacerbate the generation adequacy problem, such as regulatory or market failures, including for example caps on wholesale prices.

Individually notifiable aid — *additional information*

- 6. If the aid is to be made available to individual undertakings please provide clear evidence that the individual undertakings concerned are actually confronted with the market failures or the residual market failures identified above (see points 38 and 39 of the EEAG):
- 7. To the extent relevant, please provide specific information on:
- whether the market failure is already sufficiently addressed by other policy measures, (a) in particular the existence of environmental or other Union standards, the Union ETS or environmental taxes:
- whether State intervention is needed, taking into account the cost of implementation of (b) national standards for the aid beneficiary in the absence of aid compared to the costs, or absence thereof, of implementation of those standards for the main competitors of the aid beneficiary:
- in the case of coordination failures, the number of undertakings required to collaborate. (c) diverging interests between collaborating parties and practical problems to coordinate collaboration, such as linguistic issues, sensitivity of information and non-harmonised standards:

3. Appropriateness of the aid

To answer the questions in this section, please refer to section 3.2.3 of the EEAG and where further specified to sections 3.2 - 3.6 and 3.8 - 3.10 of the EEAG.

- 1. Please explain why State aid is the appropriate instrument rather than other policy instruments (non-State aid instruments) or the full implementation of the 'polluter pays priciple' (see points 41 - 44 of the EEAG).
- 2. Please explain why the State aid instrument selected is deemed to be the most appropriate State aid instrument to address the policy objective concerned and why it is likely to generate the least distortions of trade and competition in comparison with other State aid instruments (see point 45 - 48 of the EEAG).

Status: Point in time view as at 31/01/2020. Changes to legislation: There are currently no known outstanding effects for the

Commission Regulation (EC) No 794/2004. (See end of Document for details)

In the case of generation adequacy please confirm and explain that the aid is 3. remuneration purely for availability in accordance with point 225 of the EEAG. Please also describe how the measure provides adequate incentives to both existing and future generators and to operators using substitutable technologies, such as demandside response or storage solutions (such as allowing for different lead times to allow different technologies) in accordance with point 226 of the EEAG. Please explain to what extent interconnection capacity could remedy any possible problem of generation adequacy (as stated in point 226 of the EEAG).

4. **Incentive effect**

To answer the questions in this section, please refer to section 3.2.4 of the EEAG and where

project has not started before the submission of the aid application by the beneficiary to the national authorities? Is it ensured that the aid will in no case be paid, and has not been already paid, where work on the aided project had already started prior to the submission of the aid application by the beneficiary to the national authorities (138)? □yes □no		specified to sections $3.2 - 3.6$ and $3.8 - $	- 3.10 of the EEAG.
If so, please provide details of the competitive process and attach a copy of the tender notice or its draft. 2. When awarding the aid under the notified measure, is it ensured that work on the project has not started before the submission of the aid application by the beneficiary to the national authorities? Is it ensured that the aid will in no case be paid, and has not been already paid, where work on the aided project had already started prior to the submission of the aid application by the beneficiary to the national authorities (138)? □yes □no □no Do the aid applications include at least the applicant's name and in the case of an undertaking, its size, a description of the project, including its location and start and end dates, the amount of public support needed to carry it out, and a list of eligible costs?	1.	Is the aid to be awarded on the basis of a competitive bidding process?	
When awarding the aid under the notified measure, is it ensured that work on the project has not started before the submission of the aid application by the beneficiary to the national authorities? Is it ensured that the aid will in no case be paid, and has not been already paid, where work on the aided project had already started prior to the submission of the aid application by the beneficiary to the national authorities ⁽¹³⁸⁾ ? □yes □no □no □no □no □no □no □no □no □no	□yes	□no	
project has not started before the submission of the aid application by the beneficiary to the national authorities? Is it ensured that the aid will in no case be paid, and has not been already paid, where work on the aided project had already started prior to the submission of the aid application by the beneficiary to the national authorities (138)? Do the aid applications include at least the applicant's name and in the case of an undertaking, its size, a description of the project, including its location and start and end dates, the amount of public support needed to carry it out, and a list of eligible costs?			e process and attach a copy of the tender notice
3. Do the aid applications include at least the applicant's name and in the case of an undertaking, its size, a description of the project, including its location and start and end dates, the amount of public support needed to carry it out, and a list of eligible costs?	2.	When awarding the aid under the notified measure, is it ensured that work on the project has not started before the submission of the aid application by the beneficiary to the national authorities? Is it ensured that the aid will in no case be paid, and has not been already paid, where work on the aided project had already started prior to the submission of the aid application by the beneficiary to the national authorities ⁽¹³⁸⁾ ?	
undertaking, its size, a description of the project, including its location and start and end dates, the amount of public support needed to carry it out, and a list of eligible costs?	□yes		□no
□yes □no	3.	Do the aid applications include at least the applicant's name and in the case of an undertaking, its size, a description of the project, including its location and start and end dates, the amount of public support needed to carry it out, and a list of eligible costs?	
	□yes		□no

- Please describe, providing examples, the counterfactual credibility checks that will be 4. carried out to ensure that aid results in the desired incentive effects
- 5. If the purpose of the aid granted is to help with adapting to future Union standards or going beyond Union standards, please provide details of what the aid will consist of, which standards will be met and when or which standards will be exceeded (see points 53 to 55 of the EEAG).
- If the aid is granted for transport vehicles for road, railway, inland waterway 6. and maritime transport, please provide details relating to the applicability of the Union standards (including geographical scope where relevant) and in particular the retroacive application thereof (see point 54(a) or (b) of the EEAG).

Was a Union standard adopted?

□yes

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the

	Commission Regulation (EC) No 794/20	004. (See end of Document for details)
□yes		□no
7.	If the answer to question 6 is yes, plea it is not in force yet, by what date will	is econfirm whether it has entered into force. It it enter into force?
□yes		□no
8.	If the answer to question 6 is yes, will the investment be implemented and finalised a least one year before the date the Union standard concerned enters into force?	
□yes		□no
Please	explain by what date the will investment	be finalised:
9.	If the aid concerns energy audits of large enterprises, can you confirm that the aid does not pay for an energy audit that the beneficiary is required to undertake under Directive 2012/27/EU on Energy Efficiency ⁽¹³⁹⁾ ?	

Individually notifiable aid — *additional information*

10. If the aid is to be made available to individual undertakings please provide clear evidence that the aid has an incentive effect on the investment decision, that it changes the behaviour of the beneficiary leading either to an increase in the level of environmental protection or leading to a better functioning of the Union energy market (see section 3.2.4.2 of the EEAG):

□no

- 11. Please clarify what advantages the supported project will bring to the beneficiary, in particular production advantages for example an impact on capacity and product quality (see point 59 of the EEAG):
- 12. Please provide, explain and justify the counterfactual scenario for the individual undertaking involved (see point 60 of the EEAG):
- Please provide profitability for the project concerned and typical normal rates of return accepted by the undertaking for other investments of a similar kind. Please provide evidence in support of the information provided (see points 61 65 of the EEAG):
- 14. In case of a Union standard which is applicable or which will be applicable, please provide evidence, including quantitative information, that that standard does not remove the incentive effect of the aid (see points 66 68 of the EEAG):

5. **Proportionality**

To answer the questions in this section, please refer to section 3.2.5 of the EEAG and sections 3.2-3.6 and 3.8-3.10 of the EEAG.

If the measure concerns investment aid only, please fill in the first part of this section. If the aid is operational aid, please fill in the second part of this section. If the measure consists of both investment and operating aid both sections should be completed.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

5.1. State aid schemes

5.1.1. Investment aid schemes

The aid amount per beneficiary should be limited to the minimum required to achieve the level of environmental protection or energy objective aimed for. As a general principle aid will be deemed limited to the minimum necessary if the aid corresponds to the total net extra cost necessary to meet the objective compared to the counterfactual scenario (see section 3.2.5 of the EEAG, point (70).

For aid to energy efficient district heating and cooling projects, this section is only applicable to the production plant. For the infrastructure part the funding gap approach for energy infrastructure should be used (see section 5.1.2 operating aid schemes) (point 76 of the EEAG).

- 5.1.1.1. Eligible costs⁽¹⁴⁰⁾: please provide the details of the eligible costs as appropriate.
- (1) Please confirm that the eligible costs are limited to the extra investment costs necessary to achieve the objective of common interest and do not exceed an aid intensity of 100 %:
- □ yes
- (2) Please confirm the following by ticking the appropriate box and provide explanations about:
- the precise environmental protection related cost constitutes the eligible costs, if the cost of investing in environmental protection can be easily identified;

or

- the extra investment costs are established by comparing the aided investment with the counterfactual situation in the absence of aid, that is to say the reference investment (141)
- (3) The counterfactual scenario can be difficult to establish in integrated projects such as integrated energy-efficiency measures or biogas projects. Where a counterfactual scenario cannot be established the Commission may consider the total costs of the project as an alternative which may imply lower aid intensities to reflect the calculation of the eligible costs (see point 75 of the EEAG).

If you propose to use that approach please explain in detail the reason for doing so and provide the details of the calculation, in particular showing how the maximum aid intensities are appropriately adjusted:

- (4) Please provide a detailed calculation methodology, by reference to the counterfactual situation, which will be applied to all individual aid grants based on the notified scheme, and provide the relevant evidence:
- (5) What form do the eligible costs take?
- (a)□ investments in tangible assets
- (b)□ investments in intangible assets

If the investment relates to tangible assets please complete point 6; if it relates to intangible assets please complete point 7. If the investment relates to both tangible and intetangible assets both points 6 and 7 should be completed.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (6) In case of investments in tangible assets please indicate the form(s) of investments concerned:
- (a)□ investments in land which are strictly necessary in order to meet environmental objectives;
- (b) investments in buildings intended to reduce or eliminate pollution and nuisances;
- (c) investments in plant and equipment intended to reduce or eliminate pollution and nuisances;
- (d)□ investments to adapt production methods with a view to protecting the environment.
- (7) In case of investments in intangible assets (technology transfer through the acquisition of operating licenses or of patented and non-patented know how) please confirm that any such intangible asset satisfies the following conditions:
- (a)□ it is regarded as a depreciable asset;
- (b)□ it is purchased on market terms, from an undertaking over which the acquirer has no direct or indirect control;
- (c) it is included in the assets of the undertaking, and remains in the establishment of the recipient of the aid and is used there for at least five years $^{(142)}$.

Furthermore, please confirm that if the intangible asset is sold during the first five years:

- □ the yield from the sale will be deducted from the eligible costs; and
- all or part of the amount of aid will, where appropriate, be reimbursed.

If the investment relates to Union standards, please fill in point 8.

- (8) In case of measures aiming at obtaining a level of environmental protection higher than Union standards, please confirm the relevant statements⁽¹⁴³⁾ by ticking the relevant box:
- (a) if the undertaking is adapting to national standards adopted in the absence of Union standards, the eligible costs consist of the additional investment costs necessary to achieve the level of environmental protection required by the national standards;
- (b)□ if the undertaking is adapting to or goes beyond national standards which are more stringent than the relevant Union standards or goes beyond Union standards, the eligible costs consist of the additional investment costs necessary to achieve a level of environmental protection higher than the level required by the Union standards⁽¹⁴⁴⁾;
- if no standards exist, the eligible costs consist of the investment costs necessary to achieve a higher level of environmental protection than that which the undertaking or undertakings in question would achieve in the absence of any environmental aid.

5.1.1.2. Aid intensity and bonuses

The aid intensities which are allowed for the different measures can be found in Annex 1 to the EEAG.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

(9)	What is the basic aid intensity applicable to the notified measure (without any bonus see point 10 below)?		
(10) Bonus fo	Bonuses: for assisted regions		
(a)	Is a bonus envisaged due to the investment being located in an assisted area ⁽¹⁴⁵⁾ ?		
	□yes	□no	
	If so, please specify the amount of bonus applicable (as percentage of the eligible costs):		
(b)	is granted, only beneficiaries located in	lease confirm that if the aid or bonus for regions covered by Article 107(3)(a) TFEU s granted, only beneficiaries located in the regions referred to in Article 107(3)(a) FEU as defined in point 19(46) of the EEAG are eligible for such aid/bonus:	
	□ yes		
(c)	Please confirm that if the aid or bonus for regions covered by Article 107(3)(c) TFEU is granted, only beneficiaries located in Article 107(3)(c) regions as defined in point 19(46) of the EEAG are eligible for such aid/bonus:		
SME box	□ yes nus		
(d)	Is an SME bonus ⁽¹⁴⁶⁾ applied under the notified measure?		
	□yes	□no	
	If so, please specify the amount of bonus applicable (as percentage of the eligible costs):		
(e)	Please confirm that if the aid/bonus for small enterprises is granted, the beneficiaries comply with the definition for small enterprises as defined in point 19(17) of the EEAG:		
	□ yes		
(f)	Please confirm that if the aid/bonus for medium enterprises is granted, the beneficiarie comply with the definition for medium enterprises as defined in point 19(17) of the EEAG:		
Eco-inne	□ yes ovation bonus		
(g)	Is an eco-innovation bonus ⁽¹⁴⁷⁾ applied under the notified measure?		
	□yes	□no	
(h)	If so, please specify the level of bonu requirements set out in point 78(c)(i) to (s applicable and explain how the specific iii) of the EEAG are fulfilled:	
Bidding		,	

Is the aid granted in a genuinely competitive bidding process⁽¹⁴⁸⁾?

(i)

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□yes	□no

If the answer is yes, please provide details of the competitive process and evidence of compliance with point 19(43) of the EEAG. Please attach a copy of the tender notice or its draft:

- Please specify the total aid intensity of the projects supported under the notified scheme as a percentage, taking into account the basic aid intensity and the bonuses:
- 5.1.1.3. Cumulation (see Section 3.2.5.2 of the EEAG)
- (12) Is the aid granted under the notified measure combined with other aid?

□yes	□no
------	-----

- (13) If the answer to point 12 is yes, please describe the cumulation rules applicable to the notified aid measure (see points 81 and 82 of the EEAG):
- Please specify whether the process to ensure compliance with cumulation rules will be verified under the notified aid measure:
- 5.1.2. Operating aid schemes
- 5.1.2.1. Operating aid for energy from renewable sources
- 1. Please specify the type(s) of renewable energy sources supported under the notified measure and provide details.

Please note that the aid for investment or operating aid for the production of biofuels can be allowed only with regard to sustainable biofuels. No investment aid for food based biofuels can be granted, and operating aid can only be granted until 2020 unless the plant is already depreciated. Please confirm that those conditions are being met (see point 113 of the EEAG):

2. If hydro power is supported, please confirm compliance with Directive 2000/60/ EC⁽¹⁴⁹⁾.

□yes	□по

3. If waste is supported, please confirm that the waste hierarchy is not circumvented.

□yes	□no

Biofuels

- 4. Please explain whether aid is granted to food-based biofuels and, if it is granted, under what conditions such aid is granted (see point 113 of the EEAG).
- 5. If the measure promotes biofuels, please confirm that all the following conditions regarding food-based biofuels are met:
- (a) The operating aid for food-based biofuels is granted only until 2020:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	□yes	□по		
(b)		The operating aid for food-based biofuels is granted exclusively to plants that started operations before 31 December 2013:		
	□yes	□по		
	and			
(c)	The operating aid for food-based biofuels is limited to plants which are not fully depreciated:			
	□yes	□по		
6.	Please specify whether ther it is envisaged:	e is a supply or blending obligation for biofuels or whethe		
□yes		□no		
_		3. Soperation agreement is in place:		
□yes		□no		
		etails of the cooperation agreement (see point 122 of the		
If yes, EEAG):	l .		
If yes, EEAG	1. Operating aid for electricity Please confirm that the	etails of the cooperation agreement (see point 122 of the		
If yes, EEAG) 5.1.2.1	1. Operating aid for electricity Please confirm that the fully depreciated according	etails of the cooperation agreement (see point 122 of the y from renewable sources (until plant depreciation) aid will be granted only until the plant has been		
If yes, EEAG) 5.1.2.1	Please confirm that the fully depreciated accordingly:	etails of the cooperation agreement (see point 122 of the y from renewable sources (until plant depreciation) aid will be granted only until the plant has been g to normal accounting rules and provide explanation.		

In order to incentivise market integration it is important that beneficiaries sell their electricity in the market and are subject to market obligations (see points 124 and 125 of the EEAG).

10. Please confirm that aid is granted in addition to the market price charged by generators selling directly in the market (see point 124(a) of the EEAG):

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	231108		
□yes		□no	
If so, p	lease explain in detail how that condition	n is implemented in practice:	
11.	. Please confirm that beneficiaries are subject to standard balancing responsibilities:		
□yes □no		□no	
	able energy and the responsibilities appl	g responsibilities applicable for generators of icable for other generators (see point 124(b) of	
If the a	inswer is no, do you consider that there is	s no liquid intra-day market in your country?	
□yes		□по	
	answer is no, please explain in detail an able for generators of renewable energy:	d justify why balancing responsibilities are not	
12.	Please confirm that there are measures in force to ensure that generators have no incentive to generate electricity at negative prices (see point 124(c) of the EEAG):		
□yes		□no	
Please	explain in detail how this is ensured in p	practice:	
13.	Does the scheme provide aid for installations, except wind energy, with an installed electricity capacity of less than 500 kW?		
□yes		□no	
		neficiaries or only to a subgroup of beneficiaries. 24 of the EEAG apply to such installations:	
14.	Does the scheme provide aid for demonstration projects as defined in the EEAG? (see point 127 of the EEAG)		
□yes		□no	
	lease detail and provide evidence that su ion of the EEAG (point 19(45) of the EE	ach projects meet all the conditions set out in the EAG).	
		neficiaries or to a subgroup of beneficiaries only. A of the EEAG apply to demonstration projects.	
15.	Does the scheme provide aid for wind energy installations with an installed electricity capacity of less than 3 MW or 3 generation units? (see point 125 of the EEAG)		
□yes		□no	

If so, please detail the installations that are covered by this provision:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

If so, please explain whether this applies to all beneficiaries or only a subgroup of beneficiaries. Please detail whether the conditions of point 124 of the EEAG apply to such installations:

In a transitional phase covering the years 2015 and 2016, aid for at least 5 % of the planned new electricity capacity from renewable energy sources should be granted in a competitive bidding process on the basis of clear, transparent and non-discriminatory criteria (See points 124 and 125 of the EEAG).

16.	Please confirm that aid for at least 5 % of the planned new electricity capacity from
	renewable energy sources should be granted in a competitive bidding process:

□yes	□no

- 17. Please explain in detail how the condition laid down in this point is implemented in practice, including the calculation of the minimum 5 % for both 2015 and 2016:
- 18. From 1 January 2017 operating aid is in principle granted in a competitive bidding process on the basis of clear, transparent and non-discriminatory criteria, unless exceptions apply (see point 126 EEAG). Please confirm that the aid is granted in a competitive bidding process:

□yes	□no
•	

If so, please explain in detail the bidding process:

If the answer is no, please explain whether one of the following reasons is applicable:

- (a)□ very limited number of eligible projects;
- (b)□ higher support levels would arise in case of a bidding process;
- (c) low realisation rates would result (projects that would be constructed);
- 19. Please detail, substantiate and explain the reasons for not applying a competitive bidding process. Please provide quantitative and qualitative evidence that one of the reasons invoked would actually occur:

If the answer to this point is no, please fill in section 5.1.2.1.2: operating aid for energy from renewable sources other than electricity (until plant depreciation).

20. Please indicate whether the competitive bidding process is open to all generators producing electricity from renewable sources:

□yes	□по

If so, please explain the openness of the bidding process:

If the answer is no, please explain whether one of the following reasons is applicable:

- (a)□ longer term potential of new innovative technology;
- (b)□ need to achive diversification;
- (c)□ network constraints and grid stability;
- (d)□ system (integration) costs;

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (e)□ biomass: need to avoid distortions on the raw material market;
- 21. Please detail, substantiate and explain in detail the reasons for making an exception to the competitive bidding process open to all generators. Please provide quantitative and qualitative evidence that effectively there would be a situation involving one of the reasons listed in point 20:
- 22. Please detail, substantiate and explain in detail why the reasons for applying an exception cannot be addressed in the tender design (see point 124 of the EEAG):
- Does the scheme provide aid for installations, except wind energy, with an installed electricity capacity of less than 1 MW?

□yes	□no

If so, please explain whether it applies to all beneficiaries or to a subgroup of beneficiaries only. Please detail whether such installations will be granted aid in a competitive bidding process:

24. Does the scheme provide aid for demonstration projects as defined in the EEAG?

□yes	□no

If so, please explain whether it applies to all beneficiaries or to a subgroup of beneficiaries only. Please detail whether such installations will be granted aid in a competitive bidding process:

Does the scheme provide aid for wind energy installations with an installed electricity capacity of less than 6 MW or 6 generation units?

□yes	□no

If so, please provide details of the installations that are covered by this provision:

If so, please explain whether it applies to all beneficiaries or to a subgroup of beneficiaries only. Please detail whether such installations will be granted aid in a competitive bidding process:

- 5.1.2.1.2. Operating aid for energy from renewable sources other than electricity (until plant depreciation)
- 26. Please provide the following information to show that the operating aid granted does not exceed the difference between the levelised costs of producing energy (LCOE) from the technology in question and the market price of the form of energy concerned:
- detailed analysis of the cost of producing energy of the technology concerned in the form of LCOE per units of each of the renewable sources⁽¹⁵¹⁾:
- detailed analysis of the market price of the form of energy concerned:
- 27. Please provide evidence that the aid will be granted only until the plant has been fully depreciated according to normal accounting rules and provide a detailed analysis of the depreciation of each type of the investments for environmental protection (see point 131(d) of the EEAG):
- 28. For aid schemes, please specify how the compliance with the condition in point 131(d) of the EEAG will be ensured:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

29.	. When determining the amount of operating aid, please demonstrate how any investment aid granted to the undertaking in question in respect of a new plant is deducted from production costs:		
30.	Does the aid also cover a normal return on capital?		
□yes		□no	
	please provide details and information we reasons why the chosen rate is appro-	or calculations showing the rate of normal return opriate:	
31.	Are the production costs updated re	gularly and at least once a year?	
□yes		□no	
Please	Please provide details and explanations:		
5.1.2.1	.3. Operating aid for existing biomass p	plants after plant depreciation.	
if a Mo	ember State demonstrates that the ope iation are still higher than the market	tation may be compatible with the internal market erating costs borne by the beneficiary after plant price of the energy concerned (point 133 of the	
32.	Please confirm and explain that the aid is granted only on the basis of the energy produced from renewable sources:		
33.	Please provide the following inform	nation:	
	detailed analysis of the operating cost of producing energy from biomass after plant depreciation:		
	detailed analysis of the market price	e of the form of energy concerned:	
_	detailed analysis of the design of the measure showing that it is meant to compensate only the difference between the operating costs after plant depreciation and the market price of the form of energy concerned:		
34.	Is there a monitoring mechanism to verify whether the operating costs borne are still higher than the market price of energy concerned?		
□yes		□no	
Is that monitoring mechanism updated at least on an annual basis?			
yes		□no	
-			

Please describe in detail the monitoring mechanism:

Operating aid for biomass after plant depreciation may be compatible with the internal market if a Member State demonstrates that, irrespective of the market price of the energy concerned, the use of fossil fuels as an input is more economically advantageous than the use of biomass (point 134 of the EEAG).

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the

Please confirm and explain that the aid is granted only on the basis of the energy produced from renewable sources:

Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 36. Please provide the following information:
- detailed analysis of the operating costs of producing energy from biomass after plant depreciation:
- detailed analysis of the operating costs of producing energy from the fossil fuel concerned after plant depreciation:
- detailed analysis of the design of the measure showing that it is meant to compensate only the difference between the operating costs after plant depreciation between the use of biomass and the fossil fuel concerned:
- Please provide evidence that without the aid there would be a switch from the use of biomass to the use of fossil fuels within the same plant:
- 38. Is there a monitoring mechanism in place to verify whether the operating costs borne for using biomass are still higher than the operating costs borne from using the fossil fuel concerned?

□yes		□no	
39.	Is that monitoring mechanism updated at least on an annual basis?		
□yes		□по	

Please describe in detail the monitoring mechanism:

- 5.1.2.1.4. Operating aid granted by way of certificates.
- 40. Please provide a detailed description of the green certificate or tender system (including, inter alia, the information on the level of discretionary powers, the role of the administrator, the price determination mechanism, the financing mechanism, the penalty mechanism and re-distribution mechanism).
- 41. What is the duration of the notified measure (152)?
- 42. Please provide data or calculations showing that the aid is essential to ensure the viability of the renewable energy sources.
- 43. Please provide data or calculations showing that the aid does not in the aggregate result in overcompensation for renewable energy.
- 44. Please provide information or calculations showing that the aid does not dissuade renewable energy producers from becoming more competitive.
- 45. Please provide information required under section 5.1.2.1.1 operating aid for electricity from renewable sources (until plant depreciation).
- 46. If for technical reasons the conditions of points 124 and 125 of the EEAG cannot be applied, please provide information or calculations accordingly.
- 5.1.2.2. Operating aid granted for high efficient Combined heat and Power (CHP) plants.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 47. Please fill in the information required in section 5.1.2.1 to the extent that the relevant subsection is applicable:
- For support to electricity from CHP plants until plant depreciation: section 5.1.2.1.1.
- For support to heat from CHP plants until plant depreciation: section 5.1.2.1.2.
- For support to heat or electricity from CHP plants after plant depreciation: section 5.1.2.1.3.
- For support by way of certificates: section 5.1.2.1.4.
- 48. Please confirm that the operating aid for high efficiency cogeneration is granted exclusively to:
- undertakings distributing electric power and heat to the public, where the costs of producing such electric power or heat exceed its market price⁽¹⁵³⁾;
- for the industrial use of the combined production of electric power and heat where it can be shown that the production cost of one unit of energy using that technique exceeds the market price of one unit of conventional energy⁽¹⁵⁴⁾.

Please provide details and evidence that the relevant condition(s) is/are complied with:

- 5.1.2.3. Operating aid granted for energy efficiency measures.
- 49. Please provide information or calculations demonstrating that the aid is limited to compensating for net extra production costs resulting from the investment taking account of benefits resulting from energy efficiency⁽¹⁵⁵⁾.
- 50. What is the duration of the operating aid measure (156)?
- 5.1.2.4. Operating aid granted for energy infrastructure and CCS.
- 51. Please provide information and calculations demonstrating that the aid is limited to compensating for net extra production costs resulting from the investment taking account of costs and benefits of the project.

Please provide a detailed cash flow overview over the lifetime of the project.

Please provide an explanation of the discount rates and rates of return used.

Please provide details about the counterfactual scenario or justify the absence thereof.

- 52. In the case of CCS, please confrm and provide details demonstrating that the aid does not benefit the CO₂ emitting installation:
- 53. In the case of energy infrastructure, the eligible costs are therefore the funding gap. Please demonstrate that the aid does not exceed the funding gap by providing detailed calculation and justification for the data used, for example rate of return (see point 211 of the EEAG):
- 54. What is the duration of the operating aid measure?
- 5.1.2.5. Operating aid granted for generation adequacy.
- 55. Describe the built in measures to prevent windfall profits that are in place:
- Describe the mechanism by which the price paid returns to zero when the capacity supplied is expected to be adequate to meet the level of capacity demanded (see point 231 of the EEAG):

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 57. Is the scheme based on a competitive bidding process? Please provide details (see point 229 of the EEAG):
- 58. Please describe the expected rate of return of the beneficiaries under the scheme:
- 5.1.2.6. Operating aid in tradable permits. See point 235 of the EEAG
- 59. Please confirm that the scheme complies with all of the following criteria:
- (a) the choice of beneficiaries is based on objective and transparent criteria and the aid is granted in principle in the same way for all competitors in the same sector or relevant market if they are in a similar factual situation;
- (b) full auctioning leads to a substantial increase in production costs for each sector or category of individual beneficiaries;
- (c) the cost increase from the tradable permit scheme can not be passed on to customers without leading to important sales reductions;
- (d) the best performing technique in the EEA was used as a benchmark for the level of the allowance granted.

Please provide details demonstrating how the criteria in this point are applied:

- 5.2. Individually notifiable aid additional information
- 60. For individual aid measures, please provide a detailed calculation of the eligible costs of the notified investment project, by reference to the counterfactual situation, and provide relevant evidence:
- 61. Please provide detailed descriptions for each of the measures subject to indivudual notification. The information should not be general, for example sector specific, but as applicable for the individual beneficiary:

6. **Avoidance of negative effects**

To answer the questions in this section, please refer to section 3.2.6 of the EEAG and where further specified to sections 3.2 - 3.6 and 3.8 - 3.10 of the EEAG.

- 6.1. State aid schemes
- 1. Please explain how the distortions of competition and trade caused by the notified aid scheme will be limited to the minimum (see section 3.2.6 of the EEAG).
- 2. Is the measure preventing efficient environmental outcomes by more efficient and innovative producers?

□yes	□no
•	

If the answer is no please explain why.

- 3. Will the aid have distorive effects by strengthening or maintaining the market power of the beneficiary?
- 4. Was the aid designed in such a way as to allow appropriate parties access to the measure? Please explain what measures have been taken to ensure that access.

For generation adequacy

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 5. Please explain how the measure complies with point 233 of the EEAG
- 6. Please explain how the measure allows any capacity which can effectively contribute to addressing the generation adequacy problem to participate in the measure. (See point 232 of the EEAG)
- 6.2. Individually notifiable aid additional information
- 7. If the aid is to be made available to individual undertakings please provide clear evidence about the negative effects at the level of the undertaking. (See section 3.2.4.2 of the EEAG):

7. Transparency

To answer the questions in this section, please refer to section 3.2.7 of the EEAG and where further specified to sections 3.2 - 3.6 and 3.8 - 3.10 of the EEAG.

- 1. Please provide references to the relevant provisions in the legal basis stipulating that the granting authority should publish on a central website, or on a single website retrieving information from several websites (for example, regional websites), at least the following information about the notified State aid schemes: the full text of the approved aid scheme or the individual aid granting decision and its implementing provisions, or a link to it, the identity of the granting authority or authorities, the identity of the individual beneficiaries, the form and amount of aid granted to each beneficiary, the date of granting, the undertaking (SME/large company), the region in which the beneficiary is located and the principal economic sector in which the beneficiary has its activities (see section 3.2.7 of the EEAG):
- 2. Please provide the link to the single website:

Section C:

Compatibility assessment for aid in the form of reductions in or exemptions from environmental taxes and in the form of reductions in funding support for the energy from renewable sources

To answer the questions in this section, please refer to section 3.2.7 of the EEAG in particular.

Please complete section C1 for measures concerning reductions from environmental taxes and section C2, posibly combined with section C3, for measures concerning reductions in the funding of support for renewable energy sources.

Please complete the section on Transparency in section B, point 7.

Section C1:

Aid in the form of reductions in or exemptions from environmental taxes (157)

- 1. Please explain how the tax reductions or exemptions contribute indirectly to an improvement of the level of the environmental protection and explain why the tax reductions and exemptions do not undermine the general objective pursued:
- 2. For reductions in or exemptions from harmonised taxes at Union level, please confirm that:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

(a) \square the aid is granted for a maximum period of 10 years;

and

(b) the aid beneficiaries are selected on objective, transparent and non-discriminatory criteria;

and

(c) the aid is granted in principle in the same way for all competitors in the same sector if they are in a similar factual situation;

and

(d)□ the beneficiaries pay at least the Union minimum tax level set by the relevant applicable directive⁽¹⁵⁸⁾.

Please provide for each category of beneficiaries evidence regarding the payable minimum tax level (rate actually paid preferably in EUR and in the same units as the applicable Union legislation):

(e) the reductions or exemptions are compatible with the relevant applicable Union legislation and comply with the limits and conditions set out therein:

Please refer to the relevant provision(s) and provide the relevant evidence:

- 3. If the conditions in point 2 are confirmed and duly substantiated, there is no need to complete this section, unless those conditions are not fulfilled for the total measure.
- 4. For reductions in or exemptions from environmental taxes which have not been harmonized or for those which have been harmonized but where beneficiaries pay a sum lower than the Union minimum tax level, please confirm that the aid is granted for a maximum period of 10 years:

□yes	□no
5	

Furthermore, please provide the following:

- a detailed description of the exempted sector(s):
- a list of the 20 largest beneficiaries covered by the exemptions or reductions as well as a detailed description of their situation, in particular their turnover, their market shares and the size of the tax base:
- 5. Please confirm that:
- (a) the choice of beneficiaries is based on objective and transparent criteria and the aid is granted in principle in the same way to all competitors in the same sector or relevant market which are in a similar factual situation

and

(b) the environmental tax without reduction would lead to a substantial increase in production costs for each sector or category of individual beneficiaries;

and

(c) \square without the aid the substantial increase in production costs would, if passed to the customers, lead to a substantial reduction in sales⁽¹⁵⁹⁾.

Status: Point in time view as at 31/01/2020. Changes to legislation: There are currently no known outstanding effects for the

Commission Regulation (EC) No 794/2004. (See end of Document for details)

- Please provide qualitative and quantitative evidence related to the conditions in point 6.
- 7. Please explain in what form the tax reduction or exemption is given (point 174 of the EEAG):
- 8. Please specify which one of the following conditions is met:
- Are aid beneficiaries paying at least 20 % of the national tax? (a)

□yes □no	
----------	--

If the answer is no, please demonstrate how a lower rate can be justified in view of a limited distortion of competition:

Are the reductions or exemptions conditional on the conclusion of agreements between (b) the Member State and the recipient undertakings or associations of undertakings?

□yes	□no
------	-----

If so, please provide details and evidence illustrating that the undertakings or associations of undertakings commit themselves to achieve environmental protection objectives which have the same effect of the application of (i) 20 % of the national tax or (ii) the Union minimum tax level:

Please also confirm that:

- the substance of the agreements has been negotiated by the Member State and specifies (a)□ the targets and fixes a time schedule for reaching those targets;
- the Member State ensures independent and timely monitoring of the commitments (b)□ concluded in those agreements;
- those agreements will be revised periodically in the light of technological and other (c)□ developments and will stipulate effective penalty arrangements which will apply if the commitments are not met.

Specify per sector the targets and time schedule and describe the monitoring and review mechanisms (for example by whom and at what intervals) as well as the penalty mechanism:

- 9. Where a carbon tax levied on energy products used for electricity production is introduced and compensation for the electricity used is envisaged (point 179 of the EEAG), please provide the following information:
- The direct link to the ETS allowance price: (a)
- The maximum aid intensities over time and how it complies with the intensities set (b) out in the Union ETS State aid Guidelines (160):
- (c) A list of the beneficiaries and whether they are eligible under the Union ETS State aid Guidelines:
- (d) Please give details about the way the lump sum payment is made:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Section C2:

Aid in the form of reductions in funding support for energy from renewable sources

- 1. Please explain what the additional costs⁽¹⁶¹⁾ resulting from the financing of renewable energy are and how they are reflected in the electricity prices that are established. Please detail:
- 2. Please confirm that only reductions from the costs of funding renewable energy support are covered by the measure and that no other costs are included. Please detail. The additional costs cannot exceed the funding of support to energy from renewable sources

Eligibility

- 3. Please provide an overview of the beneficiaries of the measure and identify to what extent the beneficiaries are active in the sectors referred to in Annex 3 to the EEAG.
- 4. To the extent that the beneficiaries of the measure are not active in sectors that are part of Annex 3 to the EEAG, please identify to what extent the beneficiaries are active in sectors referred to in Annex 5 to the EEAG.
- 5. For the beneficiaries referred to in this section, please demonstrate for the undertakings the level of electro intensity (point 186 of the EEAG). Where available please use standard electricity consumption benchmarks.
- 6. To the extent that the beneficiaries of the measure are not active in sectors referred to in Annex 3 or Annex 5 to the EEAG:
- Please demonstrate for the undertakings the level of electro intensity. Where available please use standard electricity consumption benchmarks.
- Please demonstrate for the undertakings the level of trade intensity at Union level.
- 7. Please confirm providing relevant evidence that the choice of beneficiaries is based on objective, transparent and non-discriminatory criteria and the aid is granted in principle in the same way to all competitors in the same sector or relevant market which are in a similar factual situation.

Proportionality

- 8. Please confirm that the beneficiaries pay in full at least 15 % of the additional costs.
- 9. Please confirm whether the maximum contributions for electro intensive undertakings are capped.
- at 4 % Gross value Added (GVA)
- at 0,5 % GVA (for undertakings with an electro intensity of at least 20 %)

If so,

- (a) Please demonstrate how the maximum levels and GVA are calculated (see Annex 4 to the EEAG):
- (b) Please explain whether any adjustments are made to the calculation in order to cover all labour costs (point 191 of the EEAG):
- (c) Please explain how the cap applies to all eligible undertakings:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Section C3:

Transitional rules for reductions in funding support for energy from renewable sources

This section only applies if an adjustment plan was submitted to the Commission before 1 July 2015.

1. Please explain in detail whether reductions in or exemptions from funding support for renewable energy were granted before 1 July 2014.

If so, please show whether new beneficiaries have entered the scheme after 1 July 2014.

- 2. Please provide an overview of the beneficiaries who benefitted from the measure before 1 July 2014.
- 3. Please divide those beneficiaries in two groups: one eligible for the aid under the EEAG (group 1) and another not eligible for aid under the EEAG (group 2).
- 4. Please provide an adjustment plan that would progressively bring the aid levels in line with the application of the eligibility and proportionality criteria set out in section C.2.
- (a) Please demonstrate how the plan foresees a minimum own contribution of 15 % by 2019 for group 1 referred to in point 3.
- (b) Please demonstrate how the plan foresees a minimum own contribution of 20 % by 2019 for group 2 referred to in point 3.

PART III.7

Supplementary Information Sheet on risk finance aid

Please complete this supplementary information sheet, in addition to the 'General information' form, for the notification of any aid scheme covered by the Guidelines on State aid to promote risk finance investments ('RFG') $^{(162)}$.

Please consult paragraph 52 RFG for definitions.

1. Scope

- 1.1. Reasons for notifying the scheme:
- (a) The scheme does not comply with the General Block Exemption Regulation ('GBER')⁽¹⁶³⁾. Please identify the provisions in the legal basis of the scheme that go beyond the GBER and indicate which GBER provisions they go beyond:
- (b)□ The scheme does not comply with the *de minimis* Regulation⁽¹⁶⁴⁾. Please indicate the reasons why:
- (c) The scheme does not comply with the market economy operator test at one or more levels (at the level of the investors, the financial intermediary and its manager, and the undertakings in which the investment is made) (See section 2.1 RFG; for loans, reference is made to the Communication on the Reference Rate⁽¹⁶⁵⁾, and for guarantees to the Guarantee Notice⁽¹⁶⁶⁾). Please specify the reasons why:
- (d) The scheme does not contain aid and it is notified for legal certainty reasons.
- 1.2. Scope of the notified scheme: Please tick as appropriate to confirm:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (a) The notified scheme is deployed through financial intermediaries or alternative trading platforms, except for fiscal incentives for direct investments in eligible undertakings (paragraph 20 RFG).
 - Please provide the reference to the relevant provision of the legal basis:
- (b)□ The notified scheme excludes large companies, except for small or innovative mid-caps (paragraph 21 RFG).
 - Please provide the reference to the relevant provision of the legal basis:
- (c) The notified scheme excludes risk finance aid to companies listed on the official list of a stock exchange or a regulated market (paragraph 22 RFG).
 - Please provide the reference to the relevant provision of the legal basis:
- (d) The risk finance scheme involves private investors (paragraph 23 RFG).
 - Please provide the reference to the relevant provision of the legal basis:
- (e) The risk finance scheme provides that as regards asymmetric risk-reward sharing between the State and private investors there is a substantial risk incurred by private investors or the State receives a reward on its investment (paragraph 24 RFG).
 - Please provide the reference to the relevant provision of the legal basis:
- (f) The risk finance scheme cannot be used to support buyouts (paragraph 25 RFG).
 - Please provide the reference to the relevant provision of the legal basis:
- (g) The risk finance scheme provides that no risk finance aid will be granted to undertakings in difficulty, as defined in the RFG (Please note that under the RFG, SMEs within seven years from their first commercial sale that qualify for risk finance investments following due diligence by the selected financial intermediary will not be considered as undertakings in difficulty, unless they are subject to insolvency proceedings or fulfil the criteria under their domestic law for being placed in collective insolvency proceedings at the request of their creditors).
 - Please provide the reference to the relevant provision of the legal basis:
- (h) The risk finance scheme excludes aid to undertakings that have received illegal aid that has not yet been fully recovered (paragraph 26 RFG).
- (i) The risk finance scheme does not concern aid to export-related activities towards third countries or Member States, namely aid directly linked to the quantities exported, the establishment and operation of a distribution network or to other current costs linked to the export activity, as well as aid contingent upon the use of domestic over imported goods (paragraph 27 RFG).
- (j) The risk finance scheme does not make aid subject to the obligation to use nationally produced goods or national services, and does not violate the freedom of establishment, where the aid is subject to the obligation for financial intermediaries, their managers or final beneficiaries to have or move their headquarters in the territory of the Member State concerned.

2. **Description of the scheme**

2.1. Budget of the scheme:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- What is the total risk finance investment amount (including both the public and private part) per target undertaking over the whole investment cycle for each undertaking benefiting from the scheme (that is to say not per annum)? Please specify:
- What is the size of the annual budget of the scheme?
- What is the overall size of budget of the measure for its entire duration? Please specify:
- What is the size of the investment fund(s) set up under the scheme?
- Is the scheme to be co-financed by Union funds (European Social Fund, European Regional Development Fund, other)? Please specify:
- 2.2. Duration of the scheme:
- (a) What is the duration of the scheme? (Please specify the dates of its entry into force and its end date)
- (b) What is the envisaged duration of the investment period?
- (c) What is the envisaged duration of the holding period?
- 2.3. Target undertakings which are the final beneficiaries of the scheme:

The *ex-ante* assessment⁽¹⁶⁷⁾ demonstrates the need for the following undertakings to be targeted by the scheme as final beneficiaries (paragraphs 63 - 79 RFG) (please provide details):

- (a) Small midcaps (an undertaking (i) whose number of employees does not exceed 499, and (ii) whose annual turnover does not exceed EUR 100 million or whose annual balance sheet does not exceed EUR 86 million). With reference to the *exante* assessment, please provide a summary of its economic evidence and appropriate justification:
- (b) Innovative mid-caps (mid-caps whose number of employees does not exceed 1 500 and whose R & D and innovation costs, as defined by the GBER, represent (a) at least 15 % of its total operating costs in at least one of the three years preceding the first investment under the risk finance measure, or (b) at least 10 % per year of its total operating costs in the three years preceding the first investment under the risk finance measure). With reference to the *ex-ante* assessment, please provide a summary of its the economic evidence and appropriate justification:
- (c) Undertakings receiving the initial risk finance investment more than seven years after their first commercial sale: With reference to the *ex-ante* assessment, please provide a summary of its economic evidence and appropriate justification:
- (d) Undertakings requiring an overall risk finance investment (including public and private) of an amount exceeding the EUR 15 million cap fixed in the GBER: With reference to the *ex-ante* assessment, please provide a summary of its economic evidence and appropriate justification:
- (e) Alternative trading platforms not fulfilling the conditions of Article 23 of the GBER: With reference to the *ex-ante* assessment, please provide a summary of its economic evidence and appropriate justification:
- (f)□ Other:
 - With reference to the *ex-ante* assessment, please provide a summary of its economic evidence and appropriate justification:
- 2.4. Financial instruments: the *ex-ante* assessment demonstrates a need for the following design parameters not complying with the GBER (paragraph 80 to 86 RFG):

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (a) Independent private investors' participation below the ratios required in Article 21(10) of the GBER (paragraphs 80 to 81 RFG).
 - With reference to the *ex-ante* assessment, please provide a summary of its economic evidence and appropriate justification for having ratios below the ratios required in the GBER:
- (b) Financial instruments with design parameters above the ceilings provided for in the GBER, that is to say where the public investor takes more risk than allowed under the GBER (paragraphs 82 83 RFG).
 - With reference to the *ex-ante* assessment, please provide a summary of its economic evidence and appropriate justification for having design parameters above the ceilings provided for in the GBER:
- (c) Financial instruments other than guarantees where investors, financial intermediaries and their managers are selected by giving preference to downside protection over asymmetric profit-sharing.
 - With reference to the *ex-ante* assessment, please provide a summary of its economic evidence and appropriate justification:
- (d)□ Other:
 - With reference to the *ex-ante* assessment, please provide a summary of its economic evidence and appropriate justification:
- 2.5. Fiscal instruments: the *ex-ante* assessment demonstrates that the following design parameters which do not comply with the GBER are required:
- (a) Fiscal incentives to corporate investors (including financial intermediaries or their managers acting as co-investors).
 - Please provide a summary of its economic evidence and appropriate justification:
- (b) Fiscal incentives to corporate investors for investment in SMEs via an alternative trading platform.
 - Please provide a summary of its economic evidence and appropriate justification:
- (c)□ Other:
 - Please provide a summary of its economic evidence and appropriate justification:
- 2.6. Private investors participating in the measure with equity, loans or guarantees:
- (a) Please provide the characteristics of the private investors participating in the measure (e.g. corporate investors, natural persons, etc.):
- (b) Do the private investors provide equity, loans or guarantees at the level of the financial intermediary (e.g. fund of funds) or at the level of the final beneficiaries? Please specify:
- (c) Do the financial intermediaries implementing the scheme co-invest (and are hence to be considered as private investors)?
 - ☐ Yes. If so, please specify:
 - □ No

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2	.7.	Financial	l intermed	liaries i	mplementi	ng the schem	e:

(See broad definition in paragraph 52 RFG; it also includes funds with and without legal personality)

- (a) Please specify the nature of the financial intermediaries implementing the scheme:
- (b) Does the implementation of the measure involve an 'entrusted entity' (as defined in paragraph 52(v) RFG)?
 - ☐ Yes. If so, please give details:
 - \square No
- (c) Does the entrusted entity co-invest with the Member State out of its own resources?
 - Yes. If so, please provide the reference to the legal basis authorising the entrusted entity to make such co-investment:
 - No. If so, please explain the method for the calculation of its compensation for implementing the measure, to ensure it is not overcompensated:
- (d) Is the entrusted entity selected through an open, transparent, non-discriminatory and objective selection procedure or is it directly appointed? Please specify:
- (e) Does the entrusted entity manage the fund(s) through which the funding is provided under the risk finance scheme?

□yes	□no
<i>y</i>	

- (f) Characteristics of the management company in charge of implementing the measure at the level of the financial intermediary:
- (g) In the case of several levels of financial intermediaries involved in the scheme (including funds of funds), please provide all relevant information for each level of financial intermediary:
- 2.8. Is any party involved in the scheme other than the public authority granting the aid, the target undertakings, the financial intermediaries implementing the scheme mentioned above, and the private investors involved therein?
- ☐ Yes. If so, please specify:
- □ No
- 2.9. Detailed description of the instrument(s):

Note: In order to understand better, please attach a drawing to visualise the structure of the scheme and its instrument(s), indicating all parties involved, the size of their involvement, as well as, if appropriate, an annex summarising the overall design of the notified scheme.

Please outline the design parameters that you have retained for the purposes of soliciting potential financial intermediaries to manifest their interest in participating in the risk finance scheme, by replying to the detailed questions in this section.

2.9.1. Financial instruments

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Risk finance aid measures in the form of financial instruments have to be deployed through financial intermediaries (paragraph 20 RFG). Hence, those measures are composed of, at least, a State intervention for financial intermediaries, and risk finance investments by financial intermediaries into final beneficiary undertakings.

2.9.1.1. Intervention at the level of financial intermediaries

A) State intervention at the level of financial intermediaries

The State provides the following to financial intermediaries (Please tick and complete as applicable):

- □ EQUITY (INCLUDING QUASI-EQUITY) INJECTION BY THE STATE AT THE LEVEL OF THE FINANCIAL INTERMEDIARIES
- 1. Please provide the following information:
- Terms of the equity injection (please include also a comparison with the market terms for such equity injection):
- Type of financial intermediary:
- Type of funding structure of the financial intermediary (e.g. investment fund with a percentage of private and public participation; fund of funds multi-stage structure with specialised sub-funds, public fund co-investing with private investors on a deal-by-deal basis) Please explain in detail:
- 2. In case of quasi-equity, please describe in detail the nature of the envisaged instrument:
- 3. If there is private participation (e.g. private investors provide equity to the financial intermediary alongside the State):
- Please indicate the participation ratios of the public and private investors:
- Please indicate the type of preferential treatment envisaged for the benefit of participating private investors, as described in the call for expression of interest (please give details):
 - □ Upside incentives:
 - □ Downside protection:
- If non-pari passu loss-sharing features go beyond the limits set out in the GBER, please provide economic evidence and justification, with reference to the ex-ante assessment (paragraph 110 RFG):
- If relevant, please indicate whether the first loss piece borne by the public investor is capped (RFG paragraph 110):
 - ☐ Yes; Please specify how that cap has been fixed:
 - □ No; Please explain:
- 4. What is the strategy of the public investor?

Please explain how the chosen instrument supports the public policy objectives pursued by the public investor:

5. Please describe how the instrument is designed to ensure alignment of interests between the financial intermediary's investment strategy and the public policy objectives:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 6. Please provide a detailed explanation of the duration of the instrument or of the exit strategy underpinning the investment in equity, and how the exit is strategically planned by the public investor:
- 7. Other relevant information:
- □ FUNDED DEBT INSTRUMENTS: LOAN INSTRUMENTS (HEREAFTER 'LOANS') AT THE LEVEL OF FINANCIAL INTERMEDIARIES
- 1. Please provide the following information:
- Type of loans (e.g. subordinated, portfolio risk-sharing) please provide details:
- Terms of the loans under the measure (please include also a comparison with the market terms for such loans):
- Maximum size of the loan:
- Maximum duration of the loan:
- Collateral or other requirements:
- Other relevant information:
- 2. Please provide reference to the relevant provisions of the legal basis that prohibit the use of the aid to refinance existing loans (paragraph 115 RFG):
- 3. If private participation takes place at this level (e.g. private investors provide loans to the financial intermediary alongside the State):
- Please indicate the participation ratios of the public and private investors/lenders:
 - In particular, in case of portfolio risk-sharing loans, what is the co-investment rate by the selected financial intermediary? Please note that it should not be lower than 30 % of the value of the underlying loan portfolio (paragraph 114 RFG) %
- Please describe the risk and reward sharing between the public and private investors or lenders:

In particular, if the public investor assumes the first loss, at what level is it capped? Please note that it is recommendable that such cap does not exceed 35 % (RFG paragraph 113): Capped at%

Where the public investor/lender assumes a first loss position exceeding the cap set out in the GBER (25 %), it needs to be justified by reference to a severe market failure identified in the *ex-ante* assessment (RFG paragraph 113). Please provide a summary of such justification:

- If there are other risk-mitigation mechanisms for the benefit of the private investors/lenders, please explain:
- 4. What is the pass-on mechanism (as required by paragraph 104 RFG) ensuring that the financial intermediary passes on the advantage it receives from the State to the final beneficiary undertakings? What requirements does the financial intermediary have to apply (e.g. in terms of interest rate, collateral, risk class) to the final beneficiaries (please provide very precise details)? Please also provide details as to what extent the portfolio to be built under the measure goes beyond the financial intermediary's standard credit risk policy.
- 5. What is the strategy of the public investor?

Please explain how the chosen instrument supports the public policy objectives pursued by the public investor:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 6. Please describe how the instrument is designed to ensure alignment of interests between the financial intermediary's investment strategy and the public policy objectives:
- 7. Please explain in detail the duration of the instrument or of the exit strategy underpinning the investment in debt instruments, and how the exit is strategically planned by the public investor:
- 8. Other relevant information:
- UNFUNDED DEBT INSTRUMENTS: GUARANTEES BY THE STATE AT THE LEVEL OF FINANCIAL INTERMEDIARIES ON UNDERLYING TRANSACTIONS WITH FINAL BENEFICIARIES
- 1. Please provide reference to the relevant provision of the legal basis that requires that eligible transactions covered by the guarantee must be newly originated eligible risk finance loan transactions, including lease instruments, as well as quasi-equity investment instruments, to the exclusion of equity instruments (paragraph 116 RFG):
- 2. Do the guarantees to financial intermediaries cover a portfolio of underlying transactions and not a single underlying transaction?

□yes	□no

- 3. Type of guarantee:
- □ Capped: guarantee cap is %

(Please note that this cap applies to portfolios held by financial intermediaries and it is recommended that the cap rate does not exceed 35 % (paragraph 118 RFG); Please provide reasons for that rate:

Moreover, tick to specify whether:

- a)□ The cap rate only covers expected losses; or
- b) The cap rate covers also unexpected losses; In this case, please show how the pricing of the guarantee reflects this additional risk coverage:
- Uncapped; in that case, please justify the need and how the pricing of the guarantee reflects this additional risk coverage by the guarantee:
- ☐ Counterguarantee (guarantee to guarantee institutions)
- □ Other: please specify:
- 4. Guarantee rate (percentage of loss coverage by the public investor of every underlying transaction (See definition in paragraph 52(xvi) RFG and please note that the guarantee rate must not exceed 90 % (paragraph 117 RFG)): %;

Please give the reason for this level of coverage:

- 5. Underlying transactions covered by the guarantee:
- Nature of the underlying transactions:
- Total nominal size of the underlying transactions (in EUR):
- Maximum nominal amount of the underlying transaction per final beneficiary:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- Duration of the underlying transactions:
- Other relevant characteristics of the underlying transactions (risk rating, other):...
- 6. Please describe the other features of the guarantee (please include also a comparison with the market terms for such guarantee):
- Maximum duration of the guarantee: (Please note that this should normally not exceed 10 years (paragraph 119 RFG))
- Please provide reference to the relevant provision in the legal basis that stipulates that the guarantee must be reduced if the financial intermediary does not include a minimum amount of investment in the portfolio during a specific period, and that commitment fees are required for unused amounts:
- Is a guarantee fee envisaged?

□yes	□по

Please specify which party will have to pay the guarantee fee:

Please describe in detail the pricing:

Other:

- 7. What is the pass-on mechanism (as required by paragraph 104 RFG) ensuring that the financial intermediary passes on the advantage it receives from the State to the final beneficiary undertakings? What requirements does the financial intermediary have to apply (e.g. in terms of interest rate, collateral, risk class) to the final beneficiaries? Please provide very precise details. Please also detail to what extent the portfolio to be built under the measure goes beyond the financial intermediary's standard credit risk policy.
- 8. What is the strategy of the public investor?

Please explain how the chosen instrument supports the public policy objectives pursued by the public investor:

- 9. Please describe how the instrument is designed to ensure alignment of interests between the financial intermediary's investment strategy and the public policy objectives:
- 10. Please provide a detailed explanation of the duration of the instrument or of the exit strategy underpinning the investment in debt instruments, and how the exit is strategically planned by the public investor:
- 11. Other relevant information:
- □ OTHER FINANCIAL INSTRUMENTS

Please describe the financial instrument to be implemented with the measure and provide a detailed description of all elements contained in section 2.9.1.1 above in so far as they are applicable to the chosen financial instrument:

B) Intervention by financial intermediaries in further levels of financial intermediaries

There may be situations (including fund of funds structures), whereby, for instance, the State provides equity, loans or guarantees to a financial intermediary, which, in turn, provides equity, loans or guarantees to a further financial intermediary, which eventually provides risk finance

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

investments to final beneficiaries. In such cases where there is a second, or further levels of financial intermediaries involved in the scheme, please provide all relevant information required in section 2.9.1.1.A on Equity/Loans/Guarantees/Other financial instruments, as applicable, for each additional level of financial intermediary:

2.9.1.2. Risk finance investment by financial intermediaries in final beneficiaries

The risk finance investment in the final beneficiaries takes the following form (Please tick and complete as applicable):

- EQUITY (INCL. QUASI-EQUITY) INVESTMENT BY THE FINANCIAL INTERMEDIARIES IN FINAL BENEFICIARIES
- (a) In case of quasi-equity, please describe in detail the nature of the envisaged instrument:
- (b) Please provide the terms of the equity investment in detail (please include also a comparison with the market terms for that equity investment):
- (c) Please describe all features of the investments to be made by the financial intermediary in detail, including the requirements that the investment strategy of the eligible financial intermediaries should comply with:
- (d) Please provide a detailed explanation of the duration of the instrument or of the exit strategy underpinning the investment in equity:
- (e) If private participation takes place (e.g. private investors also provide equity to the final beneficiaries):
 - Please indicate the private participation ratio:
 - Please indicate the type of preferential treatment envisaged for the benefit of participating private investors, as described in the call for manifestation of interest (please give details):
 - □ Upside incentives:
 - □ Downside protection:
 - If non-pari passu loss-sharing features go beyond the limits set out in the GBER, please provide economic evidence and justification, with reference to the *ex-ante* assessment (paragraph 110 RFG):
 - If relevant, please indicate whether the first loss piece borne by the public investor is capped (paragraph 110 RFG):
 - ☐ Yes; Please specify how the cap has been fixed:
 - □ No; Please explain:
- □ FUNDED DEBT INSTRUMENTS: LOANS BY FINANCIAL INTERMEDIARIES TO FINAL BENEFICIARIES
- Type of loans: please provide details:
- Terms of the loans under the measure (please include also a comparison with the market terms for such loans):
- Maximum size of the loan per beneficiary:
- Maximum duration of the loans:
- Please provide a detailed explanation of the duration of the instrument or of the exit strategy underpinning the investment in debt instruments:
- Risk rating of the final beneficiaries:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- Collateral or other requirements:
- Other relevant information:
- If private participation takes place at this level (e.g. private investors also provide loans to the final beneficiaries):

Please indicate the private participation ratio:

Please describe the risk and reward sharing between the public and the private investors:

In particular, if the public investor assumes the first loss, at what level is it capped? Capped at %. (Please note that it is recommendable that such cap does not exceed 35 % (paragraph 113 RFG))

Where the public investor or lender assumes a first loss position exceeding the cap set out in the GBER (25 %), please justify this by reference to a severe market failure identified in the *exante* assessment (paragraph 113 RFG) and please provide a summary for such justification:

If there are other risk-mitigation mechanisms for the benefit of the private investors/lenders, please explain:

- UNFUNDED DEBT INSTRUMENTS: GUARANTEES BY FINANCIAL INTERMEDIARIES TO FINAL BENEFICIARIES:
- 1. Please provide the nature and terms of the guarantees in detail (please include also a comparison with the market terms for such guarantees):
- 2. Please provide reference to the relevant provision of the legal basis that requires that eligible transactions covered by the guarantee must be newly originated eligible risk finance loan transactions, including lease instruments, as well as quasi-equity investment instruments, to the exclusion of equity instruments (paragraph 116 RFG):
- 3. Please provide the nature and terms of the underlying transactions:
- □ OTHER FINANCIAL INSTRUMENTS

Please describe the financial instrument which the measure seeks to implement and provide a detailed description of all elements contained in section 2.9.1.2 above in so far as they are applicable to the chosen financial instrument:

2.9.2. Fiscal instruments:

Please complete this entire section for each tax incentive:

- Tax incentive granted for:
 - (a)□ Direct investments into undertakings
 - (b) Indirect investments into undertakings (that is to say via financial intermediaries)
 - (c) Indirect investments into undertakings made through an alternative trading platform
- Tax incentive granted to:
 - (a)□ corporate investors
 - (b)□ investors who are natural persons, for investments falling outside the scope of the GBER:
- Form of tax incentive:
 - (a) \Box income tax relief applicable on the taxable base

(b)□

(c)□

Document Generated: 2024-06-28

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

income tax break applicable on the tax liability payable

capital gains tax relief

(e)□		
` '	other:	
	escribe in detail the conditions the cenefit from the fiscal incentive:	at the investment must fulfil in order to be
percenta	ge of the invested amount that th f, maximum tax break amount wh	of the tax incentive (including maximum e investor can claim for the purposes of the ich can be deducted from the investor's tax
		nt, please provide economic evidence and indertakings (paragraph 121 RFG):
a well-s publicity	tructured set of investment requi	n of the eligible undertakings is based on rements, made public through appropriate ics of the eligible undertakings which are paragraph 123 RFG):
	m duration of the tax incentive ave a maximum duration of 10 years.	foreseen:(Please note that fiscal schemes
Please e	xplain the specific characteristics	of the national fiscal system that are relevant
	l understanding of the tax incentives	ve: nt fiscal incentives that already exist in the
		ween them and the notified tax incentive:
Is the ta	ax incentive open to all investo	rs fulfilling the required criteria, withou
discrimi	nation as to their place of establis	hment (paragraph 126 RFG)?
□yes		□по
paramet (paragra Does th	ers (incl. ceilings and caps, maxim ph 126 RFG): e total investment for each bene	num investment amount) of the tax incentive
paramet paragra Does th amount	ers (incl. ceilings and caps, maxim ph 126 RFG): e total investment for each bene	licity regarding the scope and the technical num investment amount) of the tax incentive efficiary undertaking exceed the maximum in the GBER (paragraph 149 RFG)?
paramet (paragra Does th amount yes Are eligundertak	ers (incl. ceilings and caps, maximph 126 RFG): e total investment for each benefixed by the risk finance provision ible shares full-risk ordinary sha	rum investment amount) of the tax incentive efficiary undertaking exceed the maximum in the GBER (paragraph 149 RFG)?
paramet (paragra Does th amount yes Are eligundertal three ye	ers (incl. ceilings and caps, maximph 126 RFG): e total investment for each benefixed by the risk finance provision ible shares full-risk ordinary sharing as defined in the <i>ex-ante</i> ass	rum investment amount) of the tax incentive efficiary undertaking exceed the maximum in the GBER (paragraph 149 RFG)?
paramet (paragra Does th amount yes Are eligundertakthree ye	ers (incl. ceilings and caps, maximph 126 RFG): e total investment for each benefixed by the risk finance provision ible shares full-risk ordinary shating as defined in the <i>ex-ante</i> assars (paragraph 150 RFG)? Yes No. If so, please provide details	eficiary undertaking exceed the maximum in the GBER (paragraph 149 RFG)? □no res which are newly-issued by an eligible essment, and must they be held for at least
paramet (paragra (paragra Does thamount	ers (incl. ceilings and caps, maximph 126 RFG): e total investment for each benefixed by the risk finance provision ible shares full-risk ordinary sharing as defined in the <i>ex-ante</i> assars (paragraph 150 RFG)? Yes No. If so, please provide details lief available to investors who are	eficiary undertaking exceed the maximum in the GBER (paragraph 149 RFG)? □no res which are newly-issued by an eligible essment, and must they be held for at least
paramet (paragra Does th amount yes Are eligundertakthree ye	ers (incl. ceilings and caps, maximph 126 RFG): e total investment for each benefixed by the risk finance provision ible shares full-risk ordinary sharing as defined in the <i>ex-ante</i> assurs (paragraph 150 RFG)? Yes No. If so, please provide details lief available to investors who are graph 150 RFG)?	eficiary undertaking exceed the maximum in the GBER (paragraph 149 RFG)? □no res which are newly-issued by an eligible essment, and must they be held for at least not independent from the company invested
paramet (paragra (paragra Does thamount	ers (incl. ceilings and caps, maximph 126 RFG): e total investment for each benefixed by the risk finance provision ible shares full-risk ordinary shating as defined in the <i>ex-ante</i> assers (paragraph 150 RFG)? Yes No. If so, please provide details itef available to investors who are graph 150 RFG)? No Yes. If so, please provide details ase of income tax relief, what it	eficiary undertaking exceed the maximum in the GBER (paragraph 149 RFG)? □no res which are newly-issued by an eligible essment, and must they be held for at least not independent from the company invested.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Please note that capping the tax relief at 30 % of the invested amount is considered reasonable: %

Can the relief exceed the maximum income tax liability of the investor, as established prior to the fiscal measure?

	No				
	Yes. If so	o, please provide details:			
		vides for multiple forms of tax incentive, please fill in the set of questions in e for each form of aid.			
2.9.3.	Measures supporting alternative trading platforms: Existing platform:				
		Yes			
	□ Is or will	No, to be newly established the platform be a sub-platform or subsidiary of an existing stock exchange?			
		Yes. If so, please identify:			
_		No re already existing alternative trading platforms in the Member State ph 129 RFG)?			
		Yes. If so, please identify:			
_	☐ Is the plant RFG)?	No atform set up by and operating across several Member States (paragraph 128			
		Yes. If so, please specify			
		No			
_	Type of	undertakings traded on the platform:			

Please provide, together with this notification:

- Evidence that the majority of the financial instruments admitted to trading on the alternative trading platforms are or will be issued by SMEs.
- A copy of the business plan of the platform operator demonstrating that the platform can become self-sustainable in less than 10 years (paragraph 127 RFG).
- Plausible counterfactual scenarios comparing the situations with which the tradable undertakings would be confronted in the absence of the platform in terms of access to the necessary finance (paragraph 127 RFG).
- For existing platforms, a copy of the business strategy of the platform that shows that, due to a persistent shortage of listings, and therefore a shortage of liquidity, the platform needs to be supported in the short-term, despite its long-term viability (paragraph 129 RFG).

Form of the measure:

Fiscal incentives to corporate investors in respect of their risk finance investments made through an alternative trading platform in eligible undertakings: Please complete section 2.9.2 on Fiscal instruments above.

П

Document Generated: 2024-06-28

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	Support to platform operators: — Platform operator is: a small enterprise or larger than a small enterprise				
	_	Maximum amount of the measure: EUR.			
		Is the maximum amount more than the start-up aid allowed under the GBER?			
		□yes	□no		
		Investment costs incurred for the establishment of the platform: EUR			
	_	Does the aid to the operator exce (paragraph 153 RFG)?	ed 50 % of those investment costs		
		□yes	□no		
		Aid is allowed up to how many years	of start-up of the platform?		
	_	*	o-platform or subsidiary of an existing nee for the lack of finance that such a		
	_	Other relevant information:			

3. Further information for the compatibility assessment of the aid scheme

3.1. Contribution to a common objective and need for State intervention (3.2 & 3.3 RFG)

A risk finance aid scheme can only be justified if it is targeted at addressing a specific market failure, in the form of the existence of a funding gap affecting specific undertakings in a specific development stage, geographic area and, if applicable, economic sector.

Please submit the in-depth *ex-ante* assessment that proves the specific market failure, together with this notification.

3.1.1. Information on the *ex-ante* assessment (paragraphs 65-66 RFG):

Date of the *ex-ante* assessment:

The assessment has been carried out by:

- an independent entity
- an entity linked to the following public authority:

Data on which the assessment is based:

Please tick to confirm that the *ex-ante* assessment is based on data covering 5 years preceding the notification:

The risk finance scheme is financed partially from the European Stuctural and Investment Funds and the assessment was prepared in accordance with Article 37(2) of Regulation (EU) No 1303/2013 (the Common Provisions Regulation)⁽¹⁶⁸⁾:□

3.1.2. Identification in the *ex-ante* assessment of the specific policy objectives and performance indicators for the risk finance scheme (paragraphs 58–59 RFG):

Please list the identified specific policy objectives and make reference to the relevant section in the *ex-ante* assessment:

Please list the defined performance indicators (see examples in paragraph 58 RFG) and make reference to the relevant section in the *ex-ante* assessment:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 3.1.3. Economic evidence and justification in the *ex-ante* assessment for the need for State intervention (section 3.3 RFG): see sections 2.3, 2.4, and 2.5 of this form.
- 3.2. Appropriateness and incentive effect of the risk finance scheme (sections 3.4 & 3.5 RFG)
- 3.2.1. General:
- A) By reference to the *ex-ante* assessment, please explain why the existing and envisaged national and Union policy actions targeting the same identified market failures cannot adequately address the identified market failures (paragraphs 90–91 RFG):
- B) Please explain why the proposed State aid instrument has the most appropriate design to ensure an efficient funding structure (paragraphs 92–93 RFG):
- 3.2.2. Appropriateness conditions for financial instruments (section 3.4.2 RFG):
- Minimum private investment ratios (paragraphs 95–97 RFG):
 What is the minimum aggregate (that is to say total, including all levels) independent private participation rate in the risk finance investment in the final beneficiary?:% of the risk finance (public and private) provided to the final beneficiary.
 - In case of independent private investors' participation below the ratios required in GBER, please summarize the economic evidence and provide detailed justification for this ratio (as per paragraph 95 RFG), with reference to the *ex-ante* assessment:
 - Does the *ex-ante* assessment demonstrate that the scheme leverages additional private funding that would not have been provided otherwise, or in different forms or amounts or on different terms? Please explain:
 - Is private participation in the risk finance scheme of a non-independent nature acceptable (paragraph 96 RFG)?
 - Yes. If so, please provide economic evidence and justification:
 - In the case of undertakings receiving the initial risk finance investment more than seven years after their first commercial sale, what adequate restrictions does the scheme contain? Does the private participation ratio amount to minimum 60 %?
 - □ Yes
 - □ No (paragraph 97 RFG)
- 2. Balance of risks and rewards between public and private investors (paragraphs 98 to 100 RFG):

Please explain why the allocation of risks and rewards between public and private investors as described above in the sections on the relevant financial instruments can be considered as balanced (paragraph 98 RFG):

3. Nature of incentives to be determined via selection of financial intermediaries, as well as fund managers or investors (paragraphs 101-102 RFG)

Please confirm by ticking as appropriate:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- A) Selection of financial intermediaries implementing the scheme:
 - (a) The financial intermediaries are selected via an open and non-discriminatory process, through which the exact nature of incentives is determined.
 - If not, please state the reason (explaining the selection of investors):
 - Please describe the competitive process and describe how the process of selection complies with the requirements:
 - Please provide the reference to the relevant provision of the legal basis containing the requirements in the RFG that the selection process must be open and nondiscriminatory:
 - Please list the selection criteria for financial intermediaries, as listed in the call for interest:
 - Please provide, together with this notification, the evaluation grid used for the screening of the financial intermediaries during the selection process.

Please describe the due diligence process on the selected financial intermediaries:

- Please describe how compliance with the conditions of commercial management and profit-oriented decisionmaking set out in the GBER (Article 21(14) and (15)) is ensured (paragraph 160 RFG):
- Please provide evidence, and reference to the legal basis:
- (b) As part of this selection process, financial intermediaries must demonstrate how their proposed investment strategy contributes to the achievement of the policy objectives and targets (based on the performance indicators identified in the *ex-ante* assessment).
 - For each of the selected financial intermediaries, please submit, together with this notification, the documents from the financial intermediary detailing its investment strategy, including pricing policy, and how it contributes to each of the policy objectives and targets.
 - Please provide a detailed description of the mechanism foreseen in the risk finance scheme, by which the Member State will ensure that the investment strategy of the intermediaries remains at all times aligned with the agreed policy targets (e.g. via monitoring, reporting, participation in the representation bodies), and that material changes to the investment strategy require the prior consent of the Member State.
 - Please also provide the reference to the relevant provision of the legal basis:
- (c) Each of the selected financial intermediaries has been selected in a competitive process taking into account its pricing policy on the instruments deployed in the risk finance scheme (including cost of funding, credit risk premiums, administrative and all other

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- fees). Please provide evidence to that effect for each of the selected financial intermediaries.
- (d) The manager of the financial intermediary or the management company ('the manager') is chosen through an open, transparent, non-discriminatory and objective selection procedure or the manager's remuneration fully reflects market levels.
 - If not, please state the reason (including an explanation on the selection of investors):

Please describe the competitive process and describe how the selection process complies with the requirements of this point:

- Please provide the reference to the relevant provision of the legal basis containing those requirements:
- (e) The managers of the fund of funds are required to legally commit as part of their investment mandate to determine via a competitive process the preferential conditions which could apply at the level of the sub-funds (paragraph 101 RFG).
- B) Selection of private investors
 - The private investors are selected via an open and non-discriminatory process through which the exact nature of incentives is determined (paragraph 101 RFG). Please describe the modalities for identification and selection of private investors:
- 4. Co-investing financial intermediary or fund manager taking at least 10 % of first loss piece (paragraph 103 RFG)
 - Where the financial intermediary or fund manager co-invest alongside the Member State, any potential conflict of interest should be avoided and they must take at least 10 % of the first loss piece (paragraph 103 RFG). Please confirm that this is the case (if applicable):
- 5. Pass-on mechanism in the case of debt instruments (loans or guarantees) (paragraph 104 RFG):
 - (a) The risk finance scheme provides for a pass-on mechanism (as described in section 2.9.1.1.A.) ensuring that the financial intermediary passes on the advantage it receives from the State to the final beneficiary undertakings. Please indicate the relevant provisions in the legal basis:
 - (b)□ The pass-on mechanism includes monitoring arrangements and a claw-back mechanism. Please describe and indicate the relevant provisions in the legal basis:
- 3.2.3. Appropriateness conditions for fiscal instruments (section 3.4.3 RFG):

For the purposes of these requirements, the information you have provided under section 2.9.2 will be considered.

Please indicate any further information you consider relevant with regard to the appropriateness conditions:

3.2.4. Appropriateness conditions for measures supporting alternative trading platforms (section 3.4.4 RFG):

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

For the purposes of these requirements, the information you have provided under section 2.9.3 will be considered.

Please indicate any further information you consider relevant with regard to the appropriateness conditions:

- 3.3. Proportionality of the aid (section 3.6 RFG)
- 3.3.1. Proportionality in relation to the identified market failure:
- Please describe and quantify the sources of financing available to the targeted undertakings, as analysed in the *ex-ante* assessment (cf. paragraph 65 RFG):
- With reference to the *ex-ante* assessment, please provide a summary description of the nature and size of the funding gap faced by each category of targeted undertaking as demonstrated by the *ex-ante* assessment (that is to say the level of demand for finance from the targeted undertakings that is not met by the sources of financing described in point 3.3.1; please specify how the funding gap is calculated):
- Please describe how the total amount of syndicated funding (public and private) provided under the risk finance measure is limited to the size of the funding gap (paragraph 134 RFG):
- Please explain, by reference to the *ex-ante* assessment, how the preferential treatment of private investors is limited to the minimum necessary to achieve the minimum ratios of private capital participation required by the scheme (paragraph 134 RFG):
- Duration of the funding gap faced by each category of targeted undertaking as estimated by the *ex-ante* assessment:
 - Please provide a summary of the economic evidence:...
- The *ex-ante* assessment provides evidence of the above market failure referred to in point 3.3.1 in the following sector(s):and in the following geographic area:

Please provide a summary of the economic evidence:...

- 3.3.2. Proportionality conditions for financial instruments (section 3.6.1 RFG):
- 1. In relation to the financial intermediaries/fund managers:

Is the exact value of incentives determined in the selection process of the financial intermediaries or fund managers (paragraph 136 RFG)?

interme	diaries or fund manage	rs (paragraph 136 KFG)?	
□Yes		□No	
	diaries or fund manage	g information on the remuneration of the rs (paragraph 143 RFG): unnual management fee in accordance with the property of the remuneration of the remuneration of the rs (paragraph 143 RFG):	
	□Yes	□No; please provide details:	
_		performance-based incentives, including es and policy-related incentives, in accordance RFG)?	

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□Yes	□No; please provide details:
Please specify what penalties not met:	are provided for in case the policy targets are
Please specify the perform comparison with market pract	nance-based remuneration and provide a cice (paragraph 145 RFG):
Please specify the total mana market practice (paragraph 14	agement fees and provide a comparison with 6 RFG):
	aluated as part of the scoring of the selection remuneration established as a result of that it?
□Yes	□No; please explain why not:

If the financial intermediary and its manager are public entities and were not selected through an open, transparent, non-discriminatory and objective selection procedure, please tick to confirm and provide evidence of the following (paragraph 41 RFG):

- (a) Their management fee is capped, their overall remuneration reflects normal market conditions and is linked to performance:
- (b)

 The public financial intermediaries are managed commercially and their managers take investment decisions in a profit-oriented manner at arm's length from the State. Please explain in particular the mechanisms established to exclude any possible interference by the State in the day-to-day management of the public fund:
- (c)

 The private investors are selected through an open, transparent, non-discriminatory and objective selection process, on a deal-by-deal basis.

In the case of direct appointment of an entrusted entity, what is its annual management fee, excluding performance-based incentives?: % of the capital to be contributed to the entity. Please note that it should not exceed 3 % (paragraph 148 RFG).

2. In relation to the private investors:

In the case of co-investment by a public fund with private investors participating on a deal by deal basis, are the private investors selected through a separate competitive process in respect of each transaction, so as to establish the fair rate of return (paragraph 137 RFG)?

Yes.	If so,	please	provide	support	ing evid	ence.

□ No

Where private investors are not selected through such a process, is the fair rate of return established by an independent expert on the basis of an analysis of market benchmarks and market risk using the discounted cash flow valuation methodology, and detailing the calculation of a minimum level of fair rate of return and an appropriate margin to reflect the risks (paragraph 138 RFG), and are all conditions of paragraph 139 RFG fulfilled?

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

identify the expert, describe the provide the relevant evidence: k to confirm that the same indep 3-year period plain how the risk adjusted return of return (paragraph 140 RFG): plain, on the basis of the ex-anteric financial parameters underpinently conditions for fiscal instruction of the requirements, the information of the further information of the ex-anteric further information of the executive formation of the executive further information of the executive formation	iments (section 3.6.2 RFG): nation provided under section 2.9.2 will be r relevant with regard to the proportionality
3-year period plain how the risk adjusted return of return (paragraph 140 RFG): plain, on the basis of the <i>ex-ante</i> ic financial parameters underpin nality conditions for fiscal instru f these requirements, the inform further information you consider nality conditions for alternative t	e assessment, the economic justification for ining the measure: ments (section 3.6.2 RFG): nation provided under section 2.9.2 will be relevant with regard to the proportionality
of return (paragraph 140 RFG): plain, on the basis of the <i>ex-ante</i> ic financial parameters underpinality conditions for fiscal instruction of these requirements, the information you consider that the conditions for alternative to the second of the secon	e assessment, the economic justification for ning the measure: uments (section 3.6.2 RFG): nation provided under section 2.9.2 will be r relevant with regard to the proportionality
fic financial parameters underpin nality conditions for fiscal instru f these requirements, the inform further information you consider nality conditions for alternative to	ning the measure: ments (section 3.6.2 RFG): nation provided under section 2.9.2 will be r relevant with regard to the proportionality
f these requirements, the inform further information you consider nality conditions for alternative t	nation provided under section 2.9.2 will be relevant with regard to the proportionality
further information you considernality conditions for alternative t	r relevant with regard to the proportionality
nality conditions for alternative t	
·	1: 1.C (.: 2.C2.DEC)
f these requirements the inform	trading platforms (section 3.6.3 RFG):
requirements, the milling	nation provided under section 2.9.3 will be
further information you consider	r relevant with regard to the proportionality
by ovide, as part of the <i>ex-ante</i> assess the risk finance scheme. It should els, that is to say in the market for ng out of private investors), at the and at the level of final beneficities are active). risk finance scheme ensure that	ompetition and trade (section 3.7 RFG) sment, information on the potential negative d include the potential negative effects at all or the provision of risk finance (e.g. the risk te level of financial intermediaries and their taries (including in the markets in which the latt the only undertakings targeted with risk
tate aid are those that are potential	ally viable?
	□по
wer to the above is yes, please de provisions in the legal basis: a finance scheme geographically	escribe how this is ensured and indicate the or regionally limited?
	по
~ -	egal basis (de iure) to specific sectors?
i imance scheme illinted ill the le	□по
	ase specify: x finance scheme limited in the le

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

TC	1			
If so,	n	Pace	cnec	11177
11 50,	\mathbf{p}	casc	Spec	II y .

— Is the risk finance scheme in practice targeted at certain sectors?

DVAC	
□yes	

If so, please specify:

— How are the negative effects minimised as much as possible?

4. Cumulation of the aid (section 3.9 RFG)

Risk finance aid may be cumulated with other State aid measures without identifiable eligible costs, or with *de minimis* aid, up to the highest relevant total financing ceiling fixed in the specific circumstances of each case by a block exemption regulation or a decision adopted by the Commission (paragraph 168 RFG).

- Please tick to confirm compliance with this rule:
- Please provide reference to the legal basis:
- Please explain in what way is conformity with the cumulation rules achieved:

5. **Other information**

Please indicate here any other information you consider relevant to the assessment of the measure(s) concerned under the RFG:

PART III.8

Supplementary Information Sheet for the notification of an evaluation plan

Member States must use this sheet for the notification of an evaluation plan pursuant to Article 1(2)(a) of Regulation (EU) No $651/2014^{(169)}$ and in the case of a notified aid scheme subject to an evaluation as provided in the relevant Commission guidelines.

Please refer to the Commission Staff Working Document 'Common methodology for State aid evaluation' (170) for guidance on the drafting of an evaluation plan.

1. Identification of the aid scheme to be evaluated

- (1) Title of the aid scheme:
- (2) Does the evaluation plan concern:
 - (a) a scheme subject to evaluation pursuant to Article 1(2)(a) of Regulation (EU) No 651/2014?
 - (b)□ a scheme notified to the Commission pursuant to Article 108(3) TFEU?
- (3) Reference of the scheme (to be completed by the Commission):
- (4) Please list any existing *ex-ante* evaluations or impact assessments for the aid scheme and *ex-post* evaluations or studies conducted in the past on predecessors of the aid scheme or on similar schemes. For each of those studies, please provide the following information: (a) a brief description of the study's objectives, methodologies used, results and conclusions, and (b) specific challenges that the evaluations and studies might have faced from a methodological point of view, for example data availability

Status: Point in time view as at 31/01/2020. Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

that are relevant for the assessment of the current evaluation plan. If appropriate, please identify relevant areas or topics not covered by previous evaluation plans that should be the subject of the current evaluation. Please provide the summaries of such evaluations and studies in annex and, when available, the internet links to the documents concerned:

2. Objectives of the aid scheme to be evaluated (171)

- 2.1. Please provide a description of the aid scheme specifying the needs and problems the scheme intends to address and the intended categories of beneficiaries, for example size, sectors, location, indicative number:
- 2.2. Please indicate the objectives of the scheme and the expected impact, both at the level of the intended beneficiaries and as far as the objective of common interest is concerned:
- 2.3. Please indicate possible negative effects, on the aid beneficiaries or on the wider economy, that might be directly or indirectly associated with the aid scheme⁽¹⁷²⁾:
- 2.4. Please indicate (a) the annual budget planned under the scheme, (b) the intended duration of the scheme⁽¹⁷³⁾, (c) the aid instrument or instruments and (d) the eligible
- 2.5. Please provide a summary of the eligibility criteria and the methods for selecting the aid beneficiaries. In particular, please describe the following: (a) the methods used for selecting beneficiaries (e.g. such as scoring), (b) the indicative budget available for each group of beneficiaries, (c) the likelihood of the budget being exhausted for certain groups of beneficiaries, (d) the scoring rules, if they are used in the scheme, (e) the aid intensity thresholds and (f) the criteria the authority granting the aid will take into account when assessing applications:
- Please mention specific constraints or risks that might affect the implementation of 2.6. the scheme, its expected impacts and the achievement of its objectives:

3. **Evaluation questions**

3.1. Please indicate the specific questions that the evaluation should address by providing quantitative evidence of the impact of aid. Please distinguish between (a) questions related to the direct impact of the aid on the beneficiaries, (b) questions related to the indirect impacts and (c) questions related to the proportionality and appropriateness of the aid. Please explain how the evaluation questions relate to the objectives of the scheme:

4. **Result indicators**

4.1. Please use the following table to describe which indicators will be built to measure outcomes of the scheme, as well as the relevant control variables, including the sources of data, and how each result indicator corresponds to the evaluation questions. In particular, please mention (a) the relevant evaluation question, (b) the indicator, (c) the source of data. (d) the frequency of collection of data (for example, annual, monthly, etc.), (e) the level at which the data is collected (for example, firm level, establishment level, regional level, etc.), (f) the population covered in the data source (for example, aid beneficiaries, non-beneficiaries, all firms, etc.):

Evaluation	Indicator	Source	Frequency	Level	Population
question					_

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Please explain why the chosen indicators are the most relevant for measuring the expected impact of the scheme:

5. Envisaged methods to conduct the evaluation

- 5.1. In light of the evaluation questions, please describe the envisaged methods to be used in the evaluation to identify the causal impact of the aid on the beneficiaries and to assess other indirect impacts. In particular, please explain the reasons for choosing those methods and for rejecting other methods (for example, reasons related to the design of the scheme)⁽¹⁷⁴⁾:
- 5.2. Please describe precisely the identification strategy for the evaluation of the causal impact of the aid and the assumptions on which the strategy relies. Please describe in detail the composition and the significance of the control group:
- 5.3. Please explain how the envisaged methods address potential selection bias. Can it be claimed with sufficient certainty that observed differences in the outcomes for the aid beneficiaries are due to the aid?
- 5.4. If relevant, please explain how the envisaged methods intend to address specific challenges related to complex schemes, for example schemes that are implemented in a differentiated manner at regional level and schemes that use several aid instruments:

6. **Data collection**

- 6.1. Please provide information on the mechanisms and sources for collecting and processing data about the aid beneficiaries and about the envisaged counterfactual. (175) Please provide a description of all the relevant information that relates to the selection phase: data collected on aid applicants, data submitted by applicants and selection outcomes. Please also explain any potential issue as regards data availability:
- 6.2. Please provide information on the frequency of the data collection relevant for the evaluation. Are observations available on a sufficiently disaggregated level, that is to say at the level of individual undertakings?
- 6.3. Please indicate whether the access to the necessary data for conducting the evaluation might be hindered by laws and regulations governing confidentiality of data and how those issues would be addressed. Please mention other possible challenges related to data collection and how they would be overcome:
- 6.4. Please indicate whether surveys of aid beneficiaries or of other undertakings are foreseen and whether complementary sources of information are intended to be used:

7. **Proposed timeline of the evaluation**

- 7.1. Please indicate the proposed timeline of the evaluation, including milestones for data collection, interim reports and involvement of stakeholders. If relevant, please provide an annex detailing the proposed timeline:
- 7.2. Please indicate the date by which the final evaluation report will be submitted to the Commission:
- 7.3. Please mention factors that might affect the envisaged timeline:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

8. The body conducting the evaluation

- 8.1. Please provide specific information on the body conducting the evaluation or, if not yet selected, on the timeline, procedure and criteria for its selection:
- 8.2. Please provide information on the independence of the body conducting the evaluation and on how possible conflict of interest will be excluded during the selection process:
- 8.3. Please indicate the relevant experience and skills of the body conducting the evaluation or how those skills will be ensured during the selection process:
- 8.4. Please indicate which arrangements the granting authority will make to manage and monitor the conduct of the evaluation:
- 8.5. Please provide information, even if only of an indicative nature, on the necessary human and financial resources that will be made available for carrying out the evaluation:

9. **Publicity of the evaluation**

- 9.1. Please provide information on the way the evaluation will be made public, that is to say, through the publication of the evaluation plan and the final evaluation report on a website:
- 9.2. Please indicate how the involvement of stakeholders will be ensured. Please indicate whether the organisation of public consultations or events related to the evaluation is envisaged:
- 9.3. Please specify how the evaluation results are intended to be used by the granting authority and other bodies, for example for the design of successors of the scheme or for similar schemes:
- 9.4. Please indicate whether and under which conditions data collected for the purpose or used for the evaluation will be made accessible for further studies and analysis:
- 9.5. Please indicate whether the evaluation plan contains confidential information that should not be disclosed by the Commission:

10. **Other information**

- 10.1. Please indicate here any other information you consider relevant for the assessment of the evaluation plan:
- 10.2. Please list all documents attached to the notification and provide paper copies or direct internet links to the documents concerned:]

IF6PART III.12

GENERAL INFORMATION SHEET FOR THE EU GUIDELINES FOR STATE AID IN THE AGRICULTURE AND FORESTRY SECTORS AND IN RURAL AREAS

Please note that this general information sheet for the notification of State aid applies to all sectors covered by the European Union Guidelines for State aid in the agriculture and forestry sectors and in rural areas 2014–2020⁽¹⁷⁶⁾ ('the Guidelines'). In addition for all measures covered by the Guidelines the relevant Supplementary Information Sheet must be completed.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

0.	COMMON	A	ASSESSMENT PRINCIPLE	S
v.	COMMON	Δ		N

U.	COMMON F	ASSESSIVIENT I KINC	II LES	
1.	Does the State	e aid measure satisfy the	e following common ass	sessment principles?
		the need for State intertowards a situation who	evention: a State aid me ere aid can bring about	tive of common interest; easure must be targeted a material improvement nedying a well-defined
		the appropriateness of t		posed aid measure must he objective of common
		the incentive effect: undertaking(s) concerne	ed in such a way that it on not carry out without t	the behaviour of the engages in an additional he aid or it would carry
			e limited to the minimu	m needed to induce the
		the avoidance of major	undue negative effects of s: the negative effects of	on competition and trade aid must be sufficiently are is positive;
		transparency of aid:	Member States, the Comust have easy access	Commission, economic s to all relevant acts and
2.	Does the Stat European Uni		ne of the following non	-severable violations of
				dquarters in the relevant lished in that Member
		the obligation for the b national services;	eneficiary to use nation	ally produced goods or
		a restriction on the presearch, development	and innovation results in	
		other non-severable vio	•	
		s answered positively, p cannot be declared con		lance with point (41) of l market.
1.	CONTRIBU	TION TO A COMMO	N OBJECTIVE	
1.1.		nsure viable food produces in order to achieve in		efficient and sustainable e growth?
		yes		no
1.2.		ated closely to the Compared development objectives		y (CAP), and consistent) of the Guidelines?
		yes		no

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

If the answer is no, please note that in accordance with point (44) of the Guidelines, such aid cannot be declared compatible with the internal market.

1.3.		ral products, is the aid of the markets in agricul		rules on the common
		yes		no
cannot b		patible with the internal		the Guidelines, such aid
1.4.			asures, can the Member the relevant rural devel	State demonstrate how opment programmes?
		yes		no
			ecompanied by such doc investment aid on the ba	
1.5.	scheme, can t towards the o agricultural an may rely on th	he granting authority considerations of the scheme and forestry sectors and in	onfirm that the selected e and thus towards the rural areas? For that pu	ojects on the basis of a I project will contribute objectives of aid in the rpose, the Member State where the positive effects
		yes		no
Environ	mental objectiv	es		
1.6.		e aid notification conta ected to have any environ		hether or not the aided
		yes		no
1.7.	Will the aid ha	ave an environmental in	npact?	
		yes		no
demons				notification information le Union environmental
1.8.	is the enviro	nmental requirement f		evelopment programme, sure identical with the exercise.
		yes		no

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

If the answer is no, please note that in accordance with point (52) of the Guidelines the aid cannot be declared compatible with the internal market.

2. APPROPRIATENESS OF AID

If the answer is yes, could the Member State demonstrate the advantages of such a national aid instrument compared to the rural development programme measure at stake?

2.2. As regards investment aid, which is not covered by Regulation (EU) No 1305/2013 of the European Parliament and of the Council⁽¹⁷⁸⁾ as part of the rural development programme or as additional financing for such a rural development measure, is the aid granted in forms that provide a direct pecuniary advantage (for example direct grants, exemptions or reductions in taxes, social security or other compulsory charges, etc.)?

yes	no

If the answer is yes, the Member State must demonstrate why other potentially less distortive forms of aid, such as repayable advances or forms of aid that are based on debt or equity instruments (for example, low-interest loans or interest rebates, State guarantees or an alternative provision of capital on favourable terms) are less appropriate.

2.3. Does the aid fall within the framework of aid to the forestry sector with ecological, protective and recreational objectives referred to in Section 2.8 of Chapter 2 of Part II of the Guidelines?

	VAC		no
Ш	yes	│ [□]	110

If the answer is yes, the Member State must demonstrate that the ecological, protective and recreational objectives aimed at cannot be achieved with the rural development like forestry measures referred to in sections 2.1 to 2.7 of Chapter 2 of Part II of the Guidelines.

2.4. Does the measure cover one of the following categories of aid?

	and towards the costs of market research activities, product conception
	and design and for the preparation of applications for the recognition of
	quality schemes
	aid for knowledge transfer and information actions
	aid for advisory services
	aid for farm replacement services
	aid for promotion measures
	aid to compensate for the costs of the prevention and eradication of
	animal diseases and plant pest
П	aid to the livestock sector

Please note that the aid must be granted to the final aid beneficiaries indirectly, in kind, by means of subsidised services. In these cases the aid must be paid to the provider of the service or activity in question.

3. **INCENTIVE EFFECT**

Status: Point in time view as at 31/01/2020. Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

3.1.	Will the beneficiary submit an application for the aid to the Member State before wherever work on the project or activity has started, which include at least the applicant's name and the size of the undertaking, a description of the project or activity including its location and start and end dates, the amount of aid needed to carry it our and the eligible costs?							
		yes		no				
cannot	t be declared con	patible with the		int (70) of the Guidelines the aid ss the aid is included in one of the Sheet.				
3.2.	Will the aid b	Will the aid be granted to large enterprises?						
		yes		no				
aid (na		erfactual scenar	io) and submit docum	ration, what would happen without tentary evidence in support of the				
3.3.			ent aid for meeting sta 3)(c) of the Guidelines	ndards granted to large enterprises?				
		yes		no				
	answer is yes, wald face the risk o		ng concerned be requ	ired to prove that without the aid,				
		yes		no				
3.4.		check of the co		I the granting authority carry out and confirm that the aid has the				
		yes		no				
counte	erfactual scenario	is credible if i	t is genuine and relat	a point (73) of the Guidelines a es to the decision-making factors				

prevalent at the time of the decision by the beneficiary regarding the project or activity concerned.

- 3.5. Is the aid in the form of tax advantages, is it granted to SMEs and are the following conditions complied with:
- the aid scheme establishes a right to aid in accordance with objective criteria and (a) without further exercise of discretion by the Member State; and

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

(b)		the aid scheme has been adopted and in force before work on the aided project or activity has started ⁽¹⁷⁹⁾ ?						
		yes			1	10		
If the	answer is	yes, please no	ote that points	(70) to (73) of the	e Guideline	s do not apply.		
3.6.	Does t	Does the aid fall under one of the following aid categories of the Guidelines:						
(a)	Sectio sector	aid schemes for agricultural and forestry land consolidation in accordance with Sections 1.3.4. and 2.9.2. of Part II of the Guidelines and aid schemes for the forestry sector with ecological, protective and recreational objectives in accordance with Section 2.8 of Part II of the Guidelines and where:						
	(i)			es a right to aid in cise of discretion		with objective criteria mber State;		
	(ii)	the aid scheme has been adopted and in force before eligible costs in accordance with Sections 1.3.4 and 2.9.2 and Section 2.8. of Part II of the Guidelines are incurred by the beneficiary; and						
	(iii)	•						
			yes			no		
(b)	of the	aid for disadvantages related to Natura 2000 areas and to the Directive 2000/60/EC of the European Parliament and of the Council (the Water framework directive) ⁽¹⁸⁰⁾ granted to SMEs in accordance with Section 1.1.6 of Part II of the Guidelines?						
			yes			no		
(c)	aid to	If the answer is yes, please note that points (70) to (74) of the Guidelines do not apply aid to areas facing natural or other specific constraints in accordance with Section 1.1.7 of Part II of the Guidelines?						
			yes			no		
	If the a	If the answer is yes, please note that points (70) to (74) of the Guidelines do not apply						
(d)		aid to make good the damage caused by natural disasters or exceptional occurrences in accordance with Section 1.2.1.1 of Part II of the Guidelines?						
			yes			no		
(e)	aid to	compensate	for the damage	e caused by adve	rse climation	uidelines do not apply.		
		assimilated to natural disaster in accordance with Section 1.2.1.2 of Part II of the Guidelines?						

(1)

Document Generated: 2024-06-28

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	yes		no
If the answe	er is yes, please note tha	t points (70) to (74	4) of the Guidelines do r
diseases and		ses caused by thos	ontrol and eradication of the animal diseases and plantage of the plantage of
	yes		no
If the answe	er is yes, please note tha	t points (70) to (74	4) of the Guidelines do r
	the costs of the remova 1.4 of Part II of the Gu		of fallen stock in accorda
	yes		no
Section 1.2.	pensate for the damage 1.5 of Part II of the Gu yes	idelines?	no
If the answe aid to make	yes yes r is yes, please note tha e good the damage in	t points (70) to (74)	no 4) of the Guidelines do r
If the answe aid to make accordance	yes yes er is yes, please note that e good the damage in with Section 2.8.5 of P	t points (70) to (74) forests caused by art II of the Guide	no 4) of the Guidelines do roy animals regulated belines?
If the answe aid to make accordance	yes er is yes, please note that e good the damage in with Section 2.8.5 of P	t points (70) to (74) forests caused bart II of the Guide	no 4) of the Guidelines do roy animals regulated belines?
If the answer aid to make accordance	yes er is yes, please note that e good the damage in with Section 2.8.5 of P yes yes yes	t points (70) to (74) forests caused to art II of the Guident points (70) to (74)	no 4) of the Guidelines do roy animals regulated belines?
If the answer aid to make accordance	yes er is yes, please note that e good the damage in with Section 2.8.5 of P yes yes yes	t points (70) to (74) forests caused to art II of the Guident points (70) to (74)	no 4) of the Guidelines do reputated by animals regulated belines? no 4) of the Guidelines do reputated by the Guidelines
If the answer aid to make accordance If the answer accordance Guidelines?	yes er is yes, please note that e good the damage in with Section 2.8.5 of P yes er is yes, please note that yes yes yes yes	t points (70) to (74) forests caused by art II of the Guide t points (70) to (74) t points (70) to (74) ds in accordance of	no 4) of the Guidelines do representation of the Guidelines? no 4) of the Guidelines do rewith point (148)(a) and (148)(b)
If the answer aid to make accordance If the answer investment a Guidelines?	yes er is yes, please note that e good the damage in with Section 2.8.5 of P yes er is yes, please note that yes er is yes, please note that aid for meeting standar yes er is yes, please note that aid for meeting standar	t points (70) to (74) forests caused by art II of the Guide t points (70) to (74) ds in accordance with the points (70) to (74) t points (70) to (74)	no 4) of the Guidelines do replaced by animals regulated belines? no 4) of the Guidelines do rewith point (148)(a) and (148)(b)

aid for investments in favour of conservation of cultural and natural heritage on the

agricultural holding in accordance with Section 1.1.1.2 of Part II of the Guidelines,

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	yes		no
If the answ	er is yes, please note tha	t points (70) to (74)	of the Guidelines do n
aid for pro Guidelines	omotion measures in ac	cordance with poin	nt (464)(b), (c) and (d
	yes		no
If the answ	er is yes, please note tha	t points (70) to (74)	of the Guidelines do n
	earch and development in ons 1.3.6 and 2.9.1. of Po		
	yes		no
If the answ	er is yes, please note tha	t points (70) to (74)	of the Guidelines do r
	yes		no
	e notification thresholds	Set out in point (37)(e) of the Guidennes
		·	•
If the answ	er is yes, please note tha	t points (70) to (74)	of the Guidelines do n
aid for the and village plans relati	drawing up and updates in rural areas and their ing to Natura 2000 sites (644)(a) of the Guidelin	ng of plans for the r basic services and and other areas of h	of protection and man
aid for the and village plans relati	drawing up and updati es in rural areas and their ng to Natura 2000 sites	ng of plans for the r basic services and and other areas of h	development of muni- of protection and man
aid for the and village plans relati with point	drawing up and updaties in rural areas and their ing to Natura 2000 sites (644)(a) of the Guidelin	ng of plans for the r basic services and and other areas of has?	development of muni of protection and mar igh nature value, in ac
aid for the and village plans relati with point If the answ aid for the climatic ev	drawing up and updaties in rural areas and their ng to Natura 2000 sites (644)(a) of the Guidelin yes	ng of plans for the r basic services and and other areas of has? It points (70) to (74) to forests from fill diseases, catastrop	no of the Guidelines do nother, natural disasters, whice events and climat
aid for the and village plans relati with point If the answ aid for the climatic ev	drawing up and updatices in rural areas and their ing to Natura 2000 sites (644)(a) of the Guideling yes The restoration of damage rents, plant pests, anima	ng of plans for the r basic services and and other areas of has? It points (70) to (74) to forests from fill diseases, catastrop	no of the Guidelines do nother, natural disasters, whice events and climat

Status: Point in time view as at 31/01/2020.
Changes to legislation: There are currently no known outstanding effects for the
Commission Regulation (EC) No 794/2004. (See end of Document for details)

		yes		no				
				no no				
Individ		If the answer is yes, please note that points (70) to (74) of the Guidelines do not apply. ally notifiable investment aid						
3.7.	For individually notified investment aid, does the Member State provide in notification clear evidence that the aid effectively has an impact on the investment choice?							
		yes		no				
If the a	nswer is yes, p	please specify how	the aid has such an imp	pact:				
a compathe aide	rehensive asseed project but	essment, the Memb also a comprehensi	er State must provide r	(76) of the Guidelines, to allow not only information concerning ounterfactual scenario, in which				
		please note that in impatible with the i		t (76) of the Guidelines the aid				
assume present	d when there	is a funding gap, tl	nat is to say when the	own, the incentive effect can be investment costs exceed the net need need the net need to be a six of an expectation of the basis of an expectation.				
4.	PROPORT	TIONALITY AND	CUMULATION OF	THE AID				
4.1.	Will the aid	amount exceed the	e eligible costs?					
		yes		no				
			accordance with point and therefore cannot be	(82) of the Guidelines, such aid granted.				
4.2.	Does the aid	d fall under Section	s 1.1.3 and 1.2.2 of Pa	rt II of the Guidelines?				
		yes		no				
If the a	nswer is yes, p	please note that poin	nt (82) of the Guideline	es does not apply.				
4.3.	Will the maximum aid intensity and aid amount be calculated by the granting authority when granting the aid?							
		yes		no				
If the ar	nswer is no, pl	ease note that this i	s a condition laid dowr	n in point (85) of the Guidelines.				
4.4.	Will the eligible costs be supported by documentary evidence which is clear, specific and contemporary?							

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Please note that for the purposes of calculating the aid intensity and the eligible costs, all figures used must be taken before any deduction of tax or other charge. Please also note that value added tax (VAT) is not eligible for aid, except where it is not-recoverable under national VAT legislation.

4.5. Is the aid gr	ranted in a form other tha	n a grant?	
	yes		no
If the answer is yes, i	is the aid amount the gros	s grant equivalent of the	aid?
	yes		no
4.6. Is the aid pa	ayable in several instalme	ents?	
	yes		no
If the answer is yes,	will the aid be discounted	to its value at the mome	ent of granting the aid?
	yes		no
the aid. Furthermore applicable on the date	eligible costs must be disc , the interest rate to be us e of granting the aid.	sed for discounting purp	
	yes		no
	does the discounting of ai		the basis of the discoun
	yes		no
4.8. Does the aid	d regard investment aid in	n rural areas?	
	yes		no

If the answer is yes, please note that the maximum aid intensity for large investment projects must be scaled down to the adjusted aid amount as defined in point (35).31 of the Guidelines. In addition, large investment projects cannot benefit from the increased aid intensities for SMEs.

4.9. For commitments under Sections 1.1.5.1, 1.1.8, 2.3 and 3.4 of Part II of the Guidelines, if they are expressed in units other than those set out in Annex II to Regulation (EU) No 1305/2013, Member States may calculate payments on the basis of those other units. In such cases, does the Member State ensure that the maximum amounts per year are complied with?

		es to legislation: There are curr	e view as at 31/01/2020. rently no known outstanding effec				
	Comm	ission Regulation (EC) No 794	/2004. (See end of Document for	details)			
		yes		no			
4.10.	For the measures or types of operations referred to in Sections 1.1.5, 1.1.6, 1.1.7, 1.1.8 2.2, 2.3, 3.4 and 3.5 of Part II of the Guidelines, Member States may fix the aid amour on the basis of standard assumptions of additional costs and income foregone. In thes cases, the Member State must ensure that the calculations and the corresponding aid						
	onal conditions rises under notifi	indicate clearly the sou are differentiated to tal actual land use, where do not contain elemen- for individually notific	stablished by appropriate arce of the figures used we account of regional or applicable and ts linked to investment of	local site conditions and			
4.11.	the net extra	In the case of individually notified investment aid, does the aid amount correspond to the net extra costs of implementing the investment in the area concerned, compared to the counterfactual scenario in the absence of aid?					
		yes		no			
be cons	sidered to be lim	ited to the minimum.	·	ified investment aid will			
4.12.	Is the investm		e enterprises under notif	led schemes?			
		yes		no			
		s the Member State ens stra cost approach'?	sure that the aid amount i	s limited to the minimum			
		yes		no			
sufficie return (investn beyond	ently profitable, (IRR) beyond the nent projects of a lithe cost of capited in the sector of the answer	for example, it should e normal rates of return a similar kind or, if thes al of the undertaking as concerned.	d not lead to an increase in applied by the underta- te rates are not available, as a whole or beyond the rates, does the Member S	sary to render the project se of its internal rate of aking concerned in other to an increase of its IRR rates of return commonly			
			costs of implementing the actual scenario in the ab	ne investment in the area psence of aid?			
		yes		no			
<u></u>							

The method explained in point (96) of the Guidelines must be used together with maximum aid intensities as a cap.

4.14. Does the aid concern individually notified investment aid?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the

Commission Regulation (EC) No 794/2004. (See end of Document for details)

		I	I	T
		yes		no
the mini in point	mum necessary (96) of the Gu	to render the project su	fficiently profitable, by used for the analysis of	r the aid amount exceeds using the method set out the incentive effect can
4.15. Cumula	referred to in		idelines. This requiren	umentation such as that nent does not apply to
4.16.	Will the notifi ad hoc aid?	ed aid be granted concur	rrently under several sch	nemes or cumulated with
		yes		no
		es the total amount of Sta aid down in the Guidelin		project exceed the limits
		yes		no
4.17.	Will the notifi	ed aid have identifiable	eligible costs?	
		yes		no
If the an	swer is yes, wil	Il this aid be cumulated	with any other State aid	?
		yes		no
If yes, w	vill those measu	res concern different id	entifiable eligible costs	?
		yes		no
identifia costs, ca	able eligible cos an be partly or f	sts cumulated with any ully overlapping. Howe	other State aid, in relat	Guidelines, the aid with ion to the same eligible n result in exceeding the delines?
		yes		no
4.18.	Will the aid a	uthorised under the Gui	delines be cumulated wi	ith de minimis aid?
		yes		no

If yes, is the aid in this case cumulated in respect of the same eligible costs and will such cumulation result in an aid intensity or aid amount exceeding that fixed in the Guidelines?

				Oocument Generated: 2024-06-28
	Chang		view as at 31/01/2020. ently no known outstanding effec	ts for the
			/2004. (See end of Document for	
		yes		no
		yes		110
4.19.	Articles 81(2) costs and will	and 82 of Regulation (I	sector cumulated with EU) No 1305/2013 in res in an aid intensity or aid	spect of the same eligible
		yes		no
4.20.		bining State aid with Use t undertakings or other	nion funds centrally mar bodies of the Union?	naged by the institutions
		yes		no
	able funding rate Does the aid	e(s) laid down in the appropriate concern aid for investigation.	the same eligible costs of colicable rules of Union I tments aimed at the respoint (143)(e) of the Gu	aw. storation of agricultura
		yes		no
			ot be cumulated with aid , 1.2.1.2 and 1.2.1.3 of I	
environ equival Parlian (724) o	nment under Se lent practices ref nent and of the C of the Guidelines	ections 1.1.5.1, 1.1.6, 1 Ferred to in Article 43 of Council ⁽¹⁸¹⁾ should be exc is also to ensure avoid		of the Guidelines and 07/2013 of the Europear use provided for in poin
4.22.			for producer groups as ection 1.1.4 of Part II of	
		yes		no
of prod		d organisations in the	ould not be cumulated agricultural sector laid	
4.23.			for young farmers ared to in Section 1.1.2 of I	
П		ves	П	no

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

If the answer is yes, please note that it should not be cumulated with business start-up aid for young farmers or the development of small farms as referred to in Article 19(1)(a)(i) and (iii) of Regulation (EU) No 1305/2013 if such cumulation would result in an aid amount exceeding those laid down in the Guidelines.

5. EFFECTS ON COMPETITION AND TRADE

yes

yes

- 5.1. As regards investment aid schemes for the processing of agricultural products and the marketing of agricultural products, in the forestry sector and in rural areas, could the Member State demonstrate that the negative effects will be limited to the minimum taking into account, for example, the size of the projects concerned, the individual and cumulative aid amounts, the expected beneficiaries as well as the characteristics of the targeted sectors?
- 5.2. As regards investment aid schemes for the processing of agricultural products and the marketing of agricultural products, in the forestry sector and in rural areas, has the Member State submitted any impact assessment at its disposal as well as *expost* evaluations carried out for similar predecessor schemes in order to enable the Commission to assess the likely negative effects of the aid scheme?

5.2 Concerning the negative effects of individual investment aid for the processing	5.3. Concerning the negative effects of individual investment aid for the processing agricultural products and the marketing of agricultural products and in rural areas, order to identify and assess the potential distortions of competition and trade, has the Member State provided in the notification evidence, permitting the Commission identify the product markets concerned (that is to say products affected by the charge).
agricultural products and the marketing of agricultural products and in rural areas order to identify and assess the potential distortions of competition and trade, has Member State provided in the notification evidence, permitting the Commission identify the product markets concerned (that is to say, products affected by the char	in the behaviour of the aid beneficiary) and to identify the competitors and customer

П

If the answer is yes, please specify:

6. TRANSPARENCY

6.1. Will the Member State ensure the publication of the following information on a comprehensive State aid website at national or regional level?

the full text of the aid scheme and its implementing provisions or legal basis for individual aid, or a link to it;
the identity of the granting authority/(ies);
the identity of the individual beneficiaries, the form and amount of aid granted to each beneficiary, the date of granting, the type of undertaking (SME/large enterprise), the region in which the beneficiary is located

granted to each beneficiary, the date of granting, the type of undertaking (SME/large enterprise), the region in which the beneficiary is located (at NUTS level II) and the principal economic sector in which the beneficiary has its activities (at NACE group level). Such a requirement can be waived with respect of individual aid awards not exceeding the following thresholds:

no

no

(i) EUR 60 000 for beneficiaries in the primary agricultural production;

Document Generated: 2024-06-28

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

		of agricu products,	iltural products, the	the sectors of the processing marketing of agricultural ctivities falling outside the	
6.2.			es in the form of tax add in the following ranges	vantages the information on (in EUR million):	
		0,06 to 0,5 only for 0,5 to 1 1 to 2 2 to 5 5 to 10 10 to 30 30 and more	primary agricultural pro	oduction	
6.3.	Please confirm that such information:				
	will be published after the decision to grant the aid has been taken will be kept for at least 10 years will be available for the general public without restrictions ⁽¹⁸²⁾				
Please n 2016 ⁽¹⁸³⁾		er States will not be	required to publish suc	h information before 1 July	
6.4.	In the case of an individual aid award, will the Member State publish the individual aid award on the State aid website referred to in point (128) of the Guidelines?				
		yes		no	
6.5.	If the answer i	s no, the individual	aid award is not publish	ned because:	
	it falls within the scope of Regulation (EU) No 1305/2013 and it is either co-financed by the EAFRD or granted as additional national financing for such co-financed measures and				
	the individual aid award has already been published in accordance with Articles 111, 112 and 113 of Regulation (EU) No 1306/2013 ⁽¹⁸⁴⁾ .				
	lation (EU) No			ite referred to in Article 111 red to in point (128) of the	
7.	OTHER QUI	ESTIONS			
7.1.	OTHER QUESTIONS Does the aid measure concern aid for export-related activities to third countries or to Member States which would be directly linked to the quantities exported, aid contingent upon the use of domestic over imported goods, or aid to establish and operate a distribution network or to cover any other expenditure linked to export activities?				

If the answer is yes, please note that such aid will not be authorised.

yes

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Please note that aid towards the cost of participating in trade fairs, or of studies or consultancy services needed for the launch of a new or existing product on a new market in principle does not constitute export aid.

7.2. Does the system of financing, for example, by parafiscal levies, form an integral part of the aid measure?

yes	no

If the answer is yes, the system of financing must be notified.

8. **TYPE OF AID**

List of types of aids included in the Guidelines:

- 1. Aid in favour of undertakings active in the primary production, processing and marketing of agricultural products
- 1.1. Rural development measures
- 1.1.1. Aid for investment
- 1.1.1.1. Aid for investment in tangible assets and intangible assets on agricultural holdings linked to primary agricultural production
- 1.1.1.2. Aid for investments in favour of the conservation of cultural and natural heritage located on agricultural holdings
- 1.1.1.3. Aid for investment concerning the relocation of farm buildings
- 1.1.1.4. Aid for investments in connection with the processing of agricultural products and the marketing of agricultural products
- 1.1.2. Start-up aid for young farmers and for the development of small farms
- 1.1.3. Aid for the transfer of agricultural holdings
- 1.1.4. Start-up aid for producer groups and organisations in the agricultural sector
- 1.1.5. Aid for agri-environment-climate and animal welfare commitments
- 1.1.5.1. Aid for agri-environment-climate commitments
- 1.1.5.2. Aid for animal welfare commitments
- 1.1.6. Aid for disadvantages related to Natura 2000 areas and to the Water Framework Directive
- 1.1.7. Aid to areas facing natural or other specific constraints
- 1.1.8. Aid for organic farming
- 1.1.9. Aid for the participation of producers of agricultural products in quality schemes
- 1.1.10. Aid for the provision of technical support in the agricultural sector
- 1.1.10.1. Aid for knowledge transfer and information actions
- 1.1.10.2. Aid for advisory services

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 1.1.10.3. Aid for farm replacement services
- 1.1.11. Aid for cooperation in the agricultural sector
- 1.2. Risk and crisis management
- 1.2.1. Aids to compensate for the damage to agricultural production or the means of agricultural production and to prevent damage
- 1.2.1.1. Aid to make good the damage caused by natural disasters or exceptional occurrences
- 1.2.1.2. Aid to compensate for damage caused by adverse climatic event which can be assimilated to a natural disaster
- 1.2.1.3. Aid for the costs of the prevention, control and eradication of animal diseases and plant pests and aid to make good the damage caused by animal diseases and plant pests
- 1.2.1.4. Aid for fallen stock
- 1.2.1.5. Aid to compensate for the damage caused by protected animals
- 1.2.1.6. Aid for the payment of insurance premiums
- 1.2.1.7. Aid for financial contributions to mutual funds
- 1.2.2. Aid for Closing Production Capacity
- 1.2.2.1. Closing of capacity for animal, plant or human health, sanitary, ethical or environmental reasons
- 1.2.2.2. Closing of capacity for other reasons
- 1.3. Other types of aid in the agricultural sector
- 1.3.1. Aid to the livestock sector
- 1.3.2. Aid for promotion measures in favour of agricultural products
- 1.3.3. Aid for the outermost regions and the smaller Aegean islands
- 1.3.4. Aid for agricultural land consolidation
- 1.3.5. Aid for rescuing and restructuring undertakings in difficulty
- 1.3.6. Aid for research and development in the agricultural sector
- 2. Aid for the forestry sector which is co-financed by the European Agricultural Fund for Rural Development (EAFRD), granted as additional national financing to such co-financed measures or granted as a pure State aid
- 2.1. Investments in forest area development and improvement of the viability of forests
- 2.1.1. Aid for afforestation and creation of woodland
- 2.1.2. Aid for the establishment of agro-forestry systems
- 2.1.3. Aid for the prevention and restoration of damage to forests from forest fires, natural disasters, adverse climatic events which can be assimilated to a natural disaster, other adverse climatic events, plant pests and catastrophic events

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 2.1.4. Aid for investments improving the resilience and environmental value of forest ecosystems
- 2.1.5. Aid for investments in forestry technologies and in processing, in mobilising and in the marketing of forest products
- 2.1.6. Aid for investments in infrastructure related to the development, modernisation or adaptation of forestry
- 2.2. Aid for disadvantages related to Natura 2000 forest areas
- 2.3. Aid for forest-environment and climate services and forest conservation
- 2.4. Aid for knowledge transfer and information actions in the forestry sector
- 2.5. Aid for advisory services in the forestry sector
- 2.6. Aid for cooperation in forestry sector
- 2.7. Start-up aid for producer groups and organisations in the forestry sector
- 2.8. Other aid to the forestry sector with ecological, protective and recreational objectives
- 2.8.1. Aid for specific forest actions and interventions with the primary objective to contribute to maintaining or to restoring forest ecosystem and biodiversity or the traditional landscape
- 2.8.2. Aid for maintaining and improving the soil quality and ensuring a balanced and healthy tree growth in the forestry sector
- 2.8.3. Restoration and maintenance of natural pathways, landscape elements and features and natural habitat for animals in the forestry sector
- 2.8.4. Aid for maintaining roads to prevent forest fires
- 2.8.5. Aid to make good the damage in forests caused by animals regulated by law
- 2.8.6. Aid for establishing forest management plans
- 2.9. Aid in the forestry sector aligned with the agricultural aid measures
- 2.9.1. Aid for research and development in the forestry sector
- 2.9.2. Aid for forestry land consolidation
- 3. Aids in rural areas which are co-financed by the EAFRD or granted as additional national financing to such co-financed measures
- 3.1. Aid for investments concerning the processing of agricultural products into non-agricultural products, the production of cotton or investments in the creation and development of non-agricultural activities
- 3.2. Aid for basic services and village renewal in rural areas
- 3.3. Business start-up aid for non-agricultural activities in rural areas
- 3.4. Aid for agri-environment-climate commitments to other land managers and undertakings in rural areas not active in the agricultural sector
- 3.5. Aid for disadvantages related to Natura 2000 areas to other land managers

Document Generated: 2024-06-28

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 3.6. Aid for knowledge transfer and information actions in rural areas
- 3.7. Aid for advisory services in rural areas
- 3.8. Aid for new participation of active farmers in quality schemes for cotton and foodstuffs
- 3.9. Aid for information and promotion activities concerning cotton and foodstuffs covered by a quality scheme
- 3.10. Aid for cooperation in rural areas
- 3.11. Aid for the setting-up of mutual funds
- 1.1.1.1. SUPPLEMENTARY INFORMATION SHEET ON AID FOR INVESTMENT IN UNDERTAKINGS ACTIVE IN PRIMARY AGRICULTURAL PRODUCTION

This information sheet relates to State aid for investments in tangible and intangible assets on agricultural holdings linked to primary agricultural production as described in Section 1.1.1.1 Chapter 1 of Part II of the European Union Guidelines for State aid in the agricultural and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

1. GENERAL ELIGIBILITY CRITERIA

yes

1.1.	Will the investments at which the aid is aimed, increase production beyond restrictions
	or exceed limitations on Union support at the level of individual undertakings,
	holdings or processing plants set by a common organisation of the market, including
	direct support schemes, financed by the European Agricultural Guarantee Fund (EAGF)?

		yes			no	
If the armarket.	, , I	ease note that t	he aid cannot be de	eclared comp	eatible with the inter	na
1.2.	Are undertaki of this aid?	ngs active in th	e primary agricultur	ral productio	n the sole beneficia	ries

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

no

- 2. AID FOR INVESTMENT IN TANGIBLE ASSETS AND INTANGIBLE ASSETS ON AGRICULTURAL HOLDINGS LINKED TO PRIMARY AGRICULTURAL PRODUCTION
- 2.1. Does one or more beneficiaries carry out the investment in tangible assets and intangible assets on agricultural holdings linked to primary agricultural production?

u yes u no	
------------	--

2.2. If the answer is no, does the investment concern tangible assets or intangible assets used by one or more beneficiaries?

household?

	Status: Point in time view as at 31/01/2020. Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)					
		yes			no	
2.3.	Is the aid targeted at investment in tangible assets and intangible assets linked the production of energy from renewable sources or the production of biofuels choldings?					
		yes			no	
If the a	answer is no, ple	ase do not reply t	o questions 2.4	to 2.17.		
2.4.		nent made for the of the European P	•		e meaning of Directive	
		yes			no	
	transport fuel	yes yes	the agricultural	l holding?	no atible with the internal	
marke 2.6.		ed biofuel sold or	the market?			
		yes			no	
If the marke		lease note that th	e aid cannot b	e declared comp	atible with the internal	
2.7.	Is the investment made for the production of thermal energy and/or electricity from renewable sources on agricultural holdings?					
		yes			no	
2.8.	If the answer to question 2.7 is yes:					
(a)		the renewable end only to serve the			he agricultural holding	
		yes			no	
	and					
(b)					on facilities eligible for al energy consumption	

of thermal energy and electricity on the agricultural holding, including the farm

Status: Point in time view as at 31/01/2020.
Changes to legislation: There are currently no known outstanding effects for the
Commission Regulation (EC) No 794/2004. (See end of Document for details)

		yes			no		
	eply to either poi	nts (a) or (b) is no, pleas t.	se note that the	aid cannot b	e declared compatible		
2.9.	Regarding ele	Regarding electricity, is the annual self-consumption limit respected?					
		yes		n	o		
If the a		ease note that the aid of	cannot be decl	lared compat	tible with the internal		
2.10.		How is the annual average consumption accumulated where more than one agricutura holding carry out the investment in the production of energy or biofuel?					
2.11.		Are there any minimum standards for energy efficiency for investments that consume or produce energy at the national level?					
		yes		n	0		
2.12.	If the answer to question 2.11 is yes, is there a requirement at national level that the minimum standards referred in question 2.11 are complied with? yes no						
If the a		ease note that the aid of	cannot be dec	lared compat	tible with the internal		
2.13.		Is the aid targeted specifically at investments in installations, the primary purpose o which is electricity production from biomass?					
		yes		n	0		
2.14.	If the answer to question 2.13 is yes, do the installations use a minimum percentage of the heat energy produced as determined by the Member State?						
		yes		n	o		
If the a		ease note that the aid of	cannot be decl	lared compat	tible with the internal		
2.15.	proportions of	Are there thresholds established at the level of the Member State, for the maximum proportions of cereals and other starch rich crops, sugar and oil crops used for bioenergy production, including biofuels, for different types of installations?					
		yes		n	0		
	<u></u>				-		

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2.16.	Is aid to bioenergy projects limited to bioenergy meeting the applicable sustain criteria laid down in Union legislation including Article 17(2) to (6) of Dir 2009/28/EC?						
		yes		no			
If the a	_	ease note that the aid ca	annot be declared comp	patible with the internal			
2.17.		production capacity of the installation exceed the average annular of the beneficiary(ies)?					
		yes		no			
in the (Guidelines on S		tal protection and energ	the conditions laid down gy 2014-2020 ⁽¹⁸⁶⁾ unless R ⁽¹⁸⁷⁾).			
2.18.	Which of the	following objectives do	es the investment pursu	e?			
□(a)		the improvement of the overall performance and sustainability of the agricultural holding, in particular through a reduction in production costs or the improvement and re-deployment of production;					
□(b) □(c)		the improvement of the natural environment, hygiene or animal welfare standards, provided that the investment concerned aims at going beyond the Union standard in force; the creation and improvement of infrastructure related to the development, adaptation and modernisation of agriculture, including access to farm land, land consolidation and land improvement, the supply and saving of energy and water.					
□(d)		biodiversity conservati	ri-environmental-clima on status of species a nenity value of a Natura	te objectives, including and habitats as well as a 2000 area or other high			
□(e)		disasters, exceptional of be assimilated to a nat	altural production poter courrences or adverse of cural disaster, animal d the prevention and ris	ntial damaged by natural limatic events which can iseases and plant pests, k mitigation of damage			
□(f)		holding as head of the	g farmers for the first holding in respect of	objective is covered: time in an agricultural investments to comply al production, including			

Please note that this objective justifies investment aid for a maximum period of 24 months from the date of the setting up. Is that deadline respected?

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

		yes		no
	(the Nitrate the date of	es Directive) with	in a maximum p	Directive 91/676/EEC ⁽¹⁸⁸⁾ period of four years from) and Article 5(1) of that
		ance with new recicultural producti		ndertakings active in the Jnion law.
	period of 1 imposed by	2 months from t	he date on which become manda	tment aid for a maximum the the new requirements tory for the undertaking
		yes		no
	other (pleas	se specify):		
1م	investments	s pursuing one or igible for support	more of the obje	s, please note that only ectives listed in points (a) in agricultural holdings.
ic	_	ction, acquisition	, including leas	sing, or improvement of
		st of the amount o tal eligible costs o		ased equal or less than 10 oncerned?
		yes		no
	If the ans conservatio		the operation	concern environmental
		yes		no
		er is yes, in exce may be permitted		justified cases, a higher
	percentage Please pro justified cir	may be permitted vide information	concerning th	e exceptional and duly on may assess the case in
	percentage Please pro justified cir question. the purchas	may be permitted vide information cumstances so that e or lease purcha	concerning that the Commission	e exceptional and duly
	Please pro justified cir question. the purchas market valuate general and (b), such to advice of feasibility shased on the	may be permitted vide information cumstances so that the or lease purchase of the assets; costs linked to the as architect, encorn environmental studies; feasibility	concerning that the Commissions of machinery the expenditure gineer and constant and economic studies remain	e exceptional and duly on may assess the case in
	Please pro justified cir question. the purchas market valuate general and (b), such to advice of feasibility such based on the is incurred; the acquisit	may be permitted vide information cumstances so that the or lease purchase of the assets; costs linked to the as architect, enon environmental studies; feasibility their results, no expense of the property of the control of the con	concerning that the Commissions of machinery the expenditure gineer and consument and economic studies remain penditure referrent of computer so	e exceptional and duly on may assess the case in and equipment up to the referred to in points (a) altation fees, fees relating sustainability, including eligible costs even where d to in points (a) and (b) ftware and the acquisition

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□(f)		production po	otential	damaged by natur	restoration of agricultural al disasters, exceptional
□(g)		natural disaster eligible costs n potential up to events; in the case of	r, animal nay inclu the lev investme	diseases or plant pest de the costs incurred fel as it was at beforents aimed at the pre	ch can be assimilated to a s and protected animals the for restoring the production e the occurrence of those vention of damage caused
		which can be a pests and by pr	ssimilate otected a vention	ed to a natural disaster animals, the eligible o	es, adverse climatic events e, animal diseases and plant costs may include the costs acing the consequences of
□(h)		other (please s			
2.20.	Do the eligibl	e costs include	?		
□(a)		-	of produ	ction rights, paymen	t entitlements and annual
□(b)		plants; the planting of			
□(c)		the purchase of animals, with the exception of investments carried out for:			
			urchase ouideline		ective of point (143)(e) of
		and			
		of th	e geneti itions se	c quality of the her	mals for the improvement rd; for this exception the 23 of this Supplementary
□(d)		investments to	compl	y with Union stand	dards in force, with the
□(e)		exceptions referred to in point (148) of the Guidelines; costs, other than those referred to point (144) of the Guidelines connected with leasing contracts, such as lessor's margin, interest			
□(f)		refinancing costs, overheads and insurance charges; working capital.			
	of the costs refer d compatible wi			f) is included, please	note that the aid cannot be
2.21.		le costs include of the Guideli		nent carried out for t	he objective referred to in
		yes			no
2.22.		e costs include quality of the l		chase of breeding ani	mals for the improvement
		yes			no

2.23. If the answer to question 2.22 is yes, are the following conditions fulfilled?

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

\Box (a)			be granted only for the purchase of breeding animals ment of the genetic quality of the herd in beef cattle,
□(b)		sheep and goats;	
□(0)		through the pure	chase of high quality breeding animals, both male and e registered in herd books are eligible;
□(c)		in the case of th	he replacement of existing breeding stock aid may be the replacement of animals which were not registered
\Box (d)		only active farm	ers are eligible for aid;
□(e)		period of time ar	ich ensure optimal reproductive potential for a certain e purchased; to that end, only females purchased before first offspring are eligible;
□(f)			als are to be kept in the herd for a period of at least four
			in points (a) to (f) must be cumulatively fulfilled for be compatible with the internal market.
2.24.	With regard conditions fu		new and existing irrigated areas are the following
□(a)		Framework Dire the entire area in	nagement plan, as required under the terms of the Water ective, has been communicated to the Commission for which the investment is to take place, as well as in any se environment may be affected by the investment;
□(b)		the measures tal accordance with	king effect under the river basin management plan in Article 11 of the Water Framework Directive and of agricultural sector have been specified in the relevant
□(c)		water metering	enabling the measurement of water use at the level of vestment is in place or will be put in place as part of
□(d)		an investment ir or element of ir	an improvement to an existing irrigation installation rigation infrastructure is eligible only if it is assessed
			ng potential water savings of a minimum of between according to the technical parameters of the existing infrastructure;
□(e)		status has been i	nt affects bodies of ground-or-surface water whose identified as less than good in the relevant river basin n for reasons related to water quantity:
		□(i)	the investment must ensure an effective reduction in
			water use, at the level of the investment, amounting to at least 50 % of the potential water saving made possible by the investment;
		□(ii)	in the case of an investment on a single agricultural holding, it must also result in a reduction to the agricultural holding's total water use amounting to at least 50 % of the potential water saving made possible
			at the level of the investment; the total water use of the holding must include water sold by the holding;
$\Box(f)$			nditions referred to in point (e) apply because the

investment is made in an existing installation which affects only energy

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

efficiency, or in the creation of a reservoir or in the use of recycled water which does not affect a body of ground or surface water;

for an investment resulting in a net increase of the irrigated area affecting a given body of ground or surface water:

the status of the water body has not been identified as less than good in the relevant river basin management plan for reasons related to water quantity; and

an environmental analysis shows that there will be no significant negative environmental impact from the investment. Such an environmental impact analysis must be either carried out or approved by the competent authority and may also refer to groups of holdings.

Please note that the two criteria referred to in (i) and (ii) must both be fulfilled for investment aid in this particular case to be declared compatible with the internal market;

the condition referred to in point (g)(i) does not apply to investments resulting in a net increase of the irrigated area if:

□(i) the investment is combined with an investment in an existing irrigation installation or element of irrigation infrastructure assessed *ex-ante* as offering potential water savings of a minimum of between 5 % and 25 % according to the technical parameters of the existing installation or infrastructure; and

□(ii) the investment ensures an effective reduction in water use, at the level of the investment as a whole, amounting to at least 50 % of the potential water saving made possible by the investment in the existing irrigation installation or element of infrastructure;

Please note that the two conditions referred to in (i) and (ii) of this point must both be fulfilled in order for the condition referred to in point (g) (i) not to apply.

the condition referred to in point (g)(i) does not apply to investments in the establishment of a new irrigation installation supplied with water from an existing reservoir approved by the competent authorities before 31 October 2013, if the following conditions are met:

the reservoir in question is identified in the relevant river basin management plan and is subject to the control requirements provided for in Article 11(3)(e) of the Water Framework Directive;

on 31 October 2013, there was in force either a maximum limit on total abstractions from the reservoir or a minimum required level of flow in water bodies affected by the reservoir;

referred to in (ii) of this point, complies with the conditions set out in Article 4 of the Water Framework Directive; and

the investment in question does not result in abstractions beyond the maximum limit in force on

□(g)

□(h)

□(i)

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

31 October 2013 or result in a reduction of the level of flow in affected water bodies below the minimum required level in force on 31 October 2013.

Please note that the four conditions referred to in (i) to (iv) of this point must be cumulatively fulfilled in order for the condition referred to in point (g)(i) not to apply.

2.25.	Were areas which are not irrigated but in which an irrigation installation was active
	in the recent past, to be established and justified by the Member State, considered as
	irrigated areas for the purpose of determining the net increase of the irrigated area?

		yes		no
2.26.	respect of the of the differe agricultural se	river basin district in when the water uses to the re- ector consistent with A	ary 2017, is the Member ich the investment will to ecovery of the costs of tricle 9(1) first indent opriate to the social envir	ake place, a contribution f water services by the f the Water Framework

yes	no

effects of the recovery as well as the geographic and climatic conditions of the region

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

- 2.27. Please state the maximum aid intensity, expressed as a percentage of eligible investment:
- (a) of the amount of the eligible costs in the outermost regions;

or regions affected?

- (b) of the amount of the eligible costs in the smaller Aegean Islands;
- of the amount of the eligible costs in Croatia for the implementation of the Nitrates Directive in accordance with point (148)(b) of the Guidelines;
- (d) of the amount of the eligible costs in less developed regions and in all regions whose the gross domestic product (GDP) per capita for the period from 1 January 2007 to 31 December 2013 period was less than 75 % of the average of the EU-25 for the reference period but whose GDP per capita is above 75 % of the GDP average of the EU-27:
- (e) of the amount of the eligible costs in other regions;
- (f) of the amount of the eligible costs for the purchase of breeding animals referred to in point (147) of the Guidelines.
- 2.28. If the aid intensity rates referred to in question 2.27 of this Supplementary Information Sheet are higher than the those set out in point (152) of the Guidelines, please clarify whether any of the following exceptions apply allowing for an increase by 20 percentage points:
- \Box (a) young farmers or farmers who have set up during the five years preceding the application for aid;

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□(b)	collective investments, such as storage facilities which are used by a group of farmers or facilities to prepare the agricultural products before marketing; and integrated projects covering several measures provided for in Regulation (EU) No 1305/2013, including those linked to the merger of producer organisations;
□(c)	investments in areas facing natural or other specific constraints pursuant to Article 32 of Regulation (EU) No 1305/2013;
□(d)	operations financed in the framework of the European Innovation Partnership (EIP), such as an investment in a new stable, allowing the testing of a new practice of animal housing, which have been developed in an operational group composed of farmers, scientists and animal welfare non-governmental organisations;
□(e)	investments aimed at the improvement of the natural environment, hygiene conditions or animal welfare standards, as referred to in point (143)(b) of the Guidelines; in that case the increased aid intensity as provided for in that point only applies to the additional costs necessary to obtain a level exceeding the Union standards in force and not leading to an increase in production capacity;
□(f)	investments aimed at the improvement of the sustainability of the agricultural holding, as referred to in point (143)(a) of the Guidelines, which are linked to agri-environmental-climate commitments and organic farming under Sections 1.1.5.1. and 1.1.8 of Part II of the Guidelines.

Please note that the maximum combined aid cannot exceed 90 % of the investment for the aid to be declared compatible with the internal market.

2.29. As a derogation from the ceilings of eligible costs laid down in points (152) and (153) of the Guidelines, please state the maximum aid intensity expressed as a percentage of the eligible costs for non-productive investments referred to in point (143)(d) of the Guidelines and investments for the restoration of production potential referred to in point (143)(e) of the Guidelines:

of the amount of the eligible costs.

Please note that the maximum aid intensity cannot exceed 100 % of eligible costs.

2.30. As a derogation from the ceilings of eligible costs laid down in points (152) and (153) of the Guidelines, please state the maximum aid intensity expressed as a percentage of the eligible costs for investments with preventive objectives referred to in point (143) (e) of the Guidelines:

of the amount of the eligible costs.

Please note that the maximum aid intensity cannot exceed 80 % of eligible costs, except for one exception set out in questions 2.31 and 2.32.

2.31. Is the investment with preventive objectives carried out collectively by more than one beneficiary?

□ yes □ no	
------------	--

2.32. If the answer to question 2.31 is yes, please state the maximum aid intensity expressed as a percentage of the eligible costs

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

of the amount of the eligible costs.

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

1.1.1.2. SUPPLEMENTARY INFORMATION SHEET ON AID FOR INVESTMENTS IN FAVOUR OF THE CONSERVATION OF CULTURAL AND NATURAL

and natural he ter 1 of Part II	ritage located on agric	r investments in favour		
	rural areas 2014 to 2020	Guidelines for State aid		
GENERAL ELIGIBILITY CRITERIA				
Will the investments at which the aid is aimed, increase production beyond restrictions or exceed limitations on Union support at the level of individual undertakings, holdings or processing plants set by a common organisation of the market, including direct support schemes, financed by the European Agricultural Guarantee Fund (EAGF)?				
	yes		no	
		•		
	yes		no	
nswer is no, ple	ease note that the aid ca	annot be declared comp	patible with the internal	
Is the heritage in the form of natural landscapes and buildings benefiting from aid formally recognised as cultural or natural heritage by the competent public authorities of the Member State?				
	yes		no	
	or exceed lin holdings or pridirect support (EAGF)? swer is yes, place and the said? Are undertaking of this aid? swer is no, place and the said? AID FOR I CULTURAL HOLDINGS Is the heritage formally recognized formally recognized to the said?	or exceed limitations on Union supholdings or processing plants set by a direct support schemes, financed by (EAGF)? yes were is yes, please note that the aid control of this aid? yes were is no, please note that the aid control of this aid? AID FOR INVESTMENTS IN INCULTURAL AND NATURAL HELDINGS Is the heritage in the form of natural formally recognised as cultural or natural of the Member State?	or exceed limitations on Union support at the level of in holdings or processing plants set by a common organisation of direct support schemes, financed by the European Agricu (EAGF)? yes yes wer is yes, please note that the aid cannot be declared comparts and the primary agricultural production of this aid? yes wer is no, please note that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid cannot be declared comparts and processes are noted that the aid c	

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

- 2.2. Do the eligible costs include?
- investment costs in tangible assets; \Box (a)
- capital works; □(b)
- other (please specify): □(c)

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

If the eligible costs are costs other than those indicated at points (a) and (b), please note that the aid cannot be declared compatible with the internal market.

- 2.3. Please state the maximum aid intensity, expressed as a percentage of eligible investment and in point (f) below please give an amount in EUR per year:
- (a) for investments aimed at the conservation of productive heritage features located on agricultural holdings and provided that the investment does not entail any increase in the production capacity:
 - (i) of the amount of the real costs incurred in areas facing natural or other specific constraints pursuant to Article 32 of Regulation (EU) No 1305/2013;
 - (ii) of the amount of the real costs incurred in less developed regions;
 - (iii) of the amount of the real costs incurred on other areas;
- (b) where there is an increase in production capacity:
 - (i) of the amount of the eligible costs in the outermost regions;
 - (ii) of the amount of the eligible costs in the smaller Aegean Islands;
 - of the amount of the eligible costs in Croatia for the implementation of the Nitrates Directive in accordance with point (148)(b) of the Guidelines;
 - (iv) of the amount of the eligible costs in less developed regions and in all regions whose the gross domestic product (GDP) per capita for the period from 1 January 2007 to 31 December 2013 was less than 75 % of the average of the EU-25 for the reference period but whose GDP per capita is above 75 % of the GDP average of the EU-27;
 - (v) of the amount of the eligible costs in other regions;
 - (vi) of the amount of the eligible costs for the purchase of breeding animals referred to in point (147) of the Guidelines.

If the aid intensity rates referred to in (i) to (vi) of this question are higher than the ones laid down in point (152) of the Guidelines, please clarify whether any of the following exceptions apply allowing for an increase by 20 percentage points:

□(a)	young farmers or farmers who have set up during the
	five years preceding the application for aid;
□(b)	collective investments, such as storage facilities which
	are used by a group of farmers or facilities to

are used by a group of farmers or facilities to prepare the agricultural products before marketing; and integrated projects covering several measures provided for in Regulation (EU) No 1305/2013, including those linked to merger of producer organisations;

investments in areas facing natural or other specific constraints pursuant to Article 32 of Regulation (EU) No 1305/2013;

operations financed in the framework of the European Innovation Partnership (EIP), such as an investment in a new stable, allowing the testing of a new practice of animal housing, which have been developed in an operational group composed of farmers, scientists and animal welfare non-governmental organisations;

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□(e)

investments aimed at the improvement of the natural environment, hygiene conditions or animal welfare standards, as referred to in point (143)(b) of the Guidelines; in that case the increased aid intensity as provided for in that point only applies to the additional costs necessary to obtain a level exceeding the Union standards in force and not leading to an increase in production capacity;

□(f)

investments aimed at the improvement of the sustainability of the agricultural holding, as referred to in point (143)(a) of the Guidelines, which are linked to agri-environmental-climate commitments and organic farming under Sections 1.1.5.1. and 1.1.8 of Part II of the Guidelines.

Please note that the maximum combined aid cannot exceed 90 % of the investment for the aid to be declared compatible with the internal market.

- (c) of the amount of the eligible costs for additional aid granted to cover the extra costs incurred by using traditional materials necessary to maintain the heritage features of buildings on agricultural holdings;
- (d) of the amount of the eligible costs for the aids referred to in points (a), (b) and (c), where the investment concerns small-scale infrastructures;
- (e) of the amount of the incurred costs for investments aimed at the conservation of nonproductive heritage features located on agricultural holdings, such as archaeological or historical features;
- (f) EUR per year for capital works.

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

1.1.1.3. SUPPLEMENTARY INFORMATION SHEET ON AID FOR INVESTMENTS CONCERNING THE RELOCATION OF FARM BUILDINGS

This information sheet relates to State aid for investments concerning the relocation of farm buildings as described in Section 1.1.1.3 of Chapter 1 of Part II of the European Union Guidelines for State aid in the agricultural and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

1. GENERAL ELIGIBILITY CRITERIA

1.1. Will the investments at which the aid is aimed, increase production beyond restrictions or exceed limitations on Union support at the level of individual undertakings, holdings or processing plants set by a common organisation of the market, including direct support schemes, financed by the European Agricultural Guarantee Fund (EAGF)?

yes	no
·	

If the answer is yes, please note that the aid cannot be declared compatible with the internal market.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.2.	Are undertaki of this aid?	ngs active in the prima	ry agricultural production	on the sole beneficiaries
		yes		no

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

2. AID FOR INVESTMENT CONCERNING THE RELOCATION OF FARM BUILDINGS

2.1. Does the relocation of the farm building pursue an objective of public interest specified in the relevant provisions of the Member State?

yes	no

Please note that the legal basis for the aid in the relevant provisions of the Member State must explain the public interest served by the relocation of the farm building.

- 2.2. Do the eligible costs related to the relocation include?
- real costs incurred for the dismantling, removal and re-building of existing facilities:
- in addition to those referred to in point (a), a modernisation of the facilities;
- \Box (c) in addition to those referred to in (a), an increase in production capacity;
- activities close to rural settlements, with a view to improving the quality of life or increasing the environmental performance of the rural settlement and concerning small-scale infrastructures;
- \Box (e) other (please specify):

If the eligible costs are costs other than those referred to in points (a) to (d), please note that the aid cannot be declared compatible with the internal market.

- 2.3. Please state the maximum aid intensity, expressed as a percentage of eligible investment:
- (a) of the amount of the real costs incurred for the dismantling, removal and re-building of existing buildings or facilities;
- (b) where, in addition to the costs referred to in point (a), the relocation results in a modernisation of facilities (189) or in an increase in production capacity:
 - (i) of the amount of the costs relating to the modernisation of the facilities or the increase of the production capacity ('the relevant costs') in the outermost regions;
 - (ii) of the amount of the relevant costs in the smaller Aegean Islands;
 - (iii) of the amount of the relevant costs in Croatia for the implementation of the Nitrates Directive in accordance with point (148)(b) of the Guidelines;
 - (iv) of the amount of the relevant costs in less developed regions and in all regions whose the gross domestic product (GDP) per capita for the period from 1 January 2007 to 31 December 2013 was less than 75 % of the average

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

of the EU-25 for the reference period but whose GDP per capita is above 75 % of the GDP average of the EU-27;

(v) of the amount of the relevant costs in other regions.

If the rates of aid intensity referred to in (i) to (v) are higher than those set out in point (152) of the Guidelines, please clarify whether any of the following exceptions apply allowing for an increase by 20 percentage points:

young farmers or farmers who have set up during the
five years preceding the application for aid; collective investments, such as storage facilities which are used by a group of farmers or facilities to prepare the agricultural products before marketing; and integrated projects covering several measures provided for in Regulation (EU) No 1305/2013, including those linked to the merger of producer organisations;
investments in areas facing natural or other specific constraints pursuant to Article 32 of Regulation (EU) No 1305/2013;
operations financed in the framework of the European Innovation Partnership (EIP), such as an investment in a new stable, allowing the testing of a new practice of animal housing, which have been developed in an operational group composed of farmers, scientists and
animal welfare non-governmental organisations; investments aimed at the improvement of the natural environment, hygiene conditions or animal welfare standards, as referred to in point (143)(b) of the Guidelines; in that case the increased aid intensity as provided for in that point only applies to the additional costs necessary to obtain a level exceeding the Union standards in force and not leading to an increase in
production capacity; investments aimed at the improvement of the sustainability of the agricultural holding, as referred to in point (143)(a) of the Guidelines, which are linked to agri-environmental-climate commitments and organic farming under Sections 1.1.5.1. and 1.1.8 of Part II of the Guidelines.

Please note that the result of the application of these exceptions to the aid intensities referred to in (i) to (v) cannot exceed 90 % of the investment for the aid to be declared compatible with the internal market.

(c) of the amount of the eligible costs for relocation activities close to rural settlements, with a view to improving quality of life or increasing environmental performance of the settlement, and concerning small-scale infrastructures.

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.1.1.4. SUPPLEMENTARY INFORMATION SHEET ON AID FOR INVESTMENT IN CONNECTION WITH THE PROCESSING OF AGRICULTURAL PRODUCTS AND THE MARKETING OF AGRICULTURAL PRODUCTS

of agric	ultural product 1.1.1.4 of Chap	ts ⁽¹⁹⁰⁾ and the marketin ter I of Part II of the I	g of agricultural produ	tion with the processing acts ⁽¹⁹¹⁾ , as described in lines for State aid in the Guidelines').			
1.	or exceed lin	Will the investments at which the aid is aimed, increase production beyond restrictions or exceed limitations on Union support at the level of individual undertakings, holdings or processing plants set by a common organisation of the market, including direct support schemes, financed by the European Agricultural Guarantee Fund (EAGF)?					
		yes		no			
If the armarket.	nswer is yes, pl	ease note that the aid c	annot be declared comp	patible with the internal			
2.	Is the aid gran	ated for food based biof	uels?				
		yes		no			
3.	Is the aid for the processing	investments in tangible	ts and the marketing of	ssets in connection with agricultural products as			
		yes		no			
If the an market.	nswer is no, ple	ease note that the aid c	annot be declared comp	patible with the internal			
4.	Member States may grant aid for investments in connection with the processing of agricultural products and the marketing of agricultural products if the aid fulfils all the conditions of one of the following aid instruments. Please specify under the provisions of which of the following aid instruments that aid is meant to be granted:						
		Commission Regulation	n (EU) No 651/2014 (G	BER) ⁽¹⁹²⁾ ;			
	Guidelines on regional State aid for 2014-2020 ⁽¹⁹³⁾ ; the conditions of Section 1.1.1.4. of Chapter 1 of Part II of the Guidelines.						
5.	If the aid is gra	anted under the provision	ns of the Regulation (EU) No 651/2014 (GBER):			
the basis	of the Guidelin	nes. In this case, please t	fill in the relevant part of	submit a notification on f the general notification of Pagulation (EC) No			

form set out in Part I and the specific form of Part III of Annex I of Regulation (EC) No 794/2004⁽¹⁹⁴⁾ or any provision replacing it.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

6. If the aid is granted under the provisions of the Guidelines on regional State aid for 2014-2020:

Does the aid fulfil the conditions set out in the Guidelines on regional State aid for 2014-2020?

	yes			no	
If the answer is no, the aid cannot be declared compatible with the internal market in accordance with point (168)(b) of the Guidelines.					
	nal State aid for 20	014-2020. Please t	fill in the rele	arried out on the basis of evant part of the general No 1627/2006 ⁽¹⁹⁵⁾).	
	oing to be granted elines, do the eligi			.4. of Chapter 1 of Part	
□(a)	the construction, immoveable prop		uding leasin	g, or improvement of	
	Was the cost of the % of the total elig			ed equal or less than 10 accrned?	
		yes		no	
□(b) □(c)	If the answer is no, please note that the purchase of the land is not eligible for aid. the purchase or lease purchase of machinery and equipment up to the market value of the asset; the general costs linked to the expenditure referred to in points (a) and (b), such as architect, engineer and consultation fees, fees relating to advice on environmental and economic sustainability, including feasibility studies;				
□(d)	Please note that feasibility studies remain eligible expenditure even where, based on their results, no expenditure referred to in points (a) and (b) is incurred; the acquisition or development of computer software and the acquisition of patents, licences, copyrights and trademarks.				
	Other (please specify):				

If the investment pursues other objectives other than those referred to in points (a) to (d), please note that only investments in connection with the processing of agricultural products and the marketing of agricultural products pursuing one or more of the objectives listed in those points are eligible for support.

8. Do the eligible costs include the following expenses?

\Box (a)	the costs, other than those referred to in the question 6 (see point (169)
	of the Guidelines) connected with leasing contracts, such as lessor's
	margin, interest refinancing costs, overheads and insurance charges;
□(b)	working capital;
$\Box(c)$	costs related to investments to comply with Union standards in force.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

If any of the expenses referred to in (a), (b) or (c) is included, please note that the aid cannot be declared compatible with the internal market.

- 9. Please state the maximum aid intensity, expressed as a percentage of eligible investment:
- (a) of the amount of the eligible costs in the outermost regions;
- (b) of the amount of the eligible costs in the smaller Aegean Islands;
- (c) of the amount of the eligible costs in less developed regions and in all regions whose the gross domestic product (GDP) per capita for the period from 1 January 2007 to 31 December 2013 period was less than 75 % of the average of the EU-25 for the reference period but whose GDP per capita is above 75 % of the GDP average of the EU-27;
- (d) of the amount of the eligible costs in other regions.
- 10. If the aid rates referred to in question 8 are higher than those referred to in point (171) of the Guidelines, please clarify whether any of the following exceptions apply allowing for an increase by 20 percentage points for operations:
- \Box (a) linked to a merger of producer organisations;
- \Box (b) supported in the framework of the EIP.

If one of these exceptions is applied, please provide the documentation proving it here or in an annex attached to this supplementary information sheet:

Please note that the maximum combined aid cannot exceed 90% of the investment for the aid to be declared compatible with the internal market.

What is the amount in EUR of the eligible costs for individual investment aid for the processing of agricultural products and the marketing of agricultural products?

EUR

If that amount exceeds EUR 25 million (see point (37)(a) of the Guidelines), please note that the individual aid must be specifically notified to the Commission in accordance with Article 108(3) of the Treaty.

12. What is the amount in EUR of the gross grant equivalent for individual investment aid for the processing of agricultural products and the marketing of agricultural products?

EUR

If that amount exceeds EUR 12 million (see point (37)(a) of the Guidelines), please note that the individual aid must be specifically notified to the Commission in accordance with Article 108(3) of the Treaty.

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

1.1.2. SUPPLEMENTARY INFORMATION SHEET ON START-UP AID FOR YOUNG FARMERS AND FOR THE DEVELOPMENT OF SMALL FARMS

This form must be used by Member States for the notification of State aid measures which are designed to grant start-up aid for young farmers and for the development of small farms as

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

described in section 1.1.2 of Chapter 1 of Part II of the European Union Guidelines for State aid in the agriculture and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

small undertakings the beneficiaries of this aid? yes	1.	Are undertakings action of this aid?	ve in the prim	nary agricultural p	production	on the sole beneficiaries		
Are young farmers as defined in point (35).29. of the Guidelines who are micro- and small undertakings the beneficiaries of this aid? yes		yes				no		
small undertakings the beneficiaries of this aid? yes			e that the aid	cannot be declar	red comp	patible with the internal		
3. Are small farms who are micro- and small undertakings the beneficiaries of this aid? yes	2.		Are young farmers as defined in point (35).29. of the Guidelines who are micro- and small undertakings the beneficiaries of this aid?					
4. If the answer to question 3 is yes, please indicate the criteria on the basis of which the beneficiaries qualify as small farms. Please note that such criteria must be objective. 5. Are the upper and lower thresholds for access to start-up aid for young farmers and the development of small farms defined in terms of the production potential of the agricultural holding, measured in standard output, as defined in Article 5b of Council Regulation (EC) No 1217/2009 ⁽¹⁹⁶⁾ and Article 6 of Commission Implementing Regulation (EU) 2015/220 ⁽¹⁹⁷⁾ , or an equivalent? 1.		yes				no		
4. If the answer to question 3 is yes, please indicate the criteria on the basis of which the beneficiaries qualify as small farms. Please note that such criteria must be objective. 5. Are the upper and lower thresholds for access to start-up aid for young farmers and the development of small farms defined in terms of the production potential of the agricultural holding, measured in standard output, as defined in Article 5b of Council Regulation (EC) No 1217/2009 ⁽¹⁹⁵⁾ and Article 6 of Commission Implementing Regulation (EU) 2015/220 ⁽¹⁹⁷⁾ , or an equivalent? yes	3.	Are small farms who	Are small farms who are micro- and small undertakings the beneficiaries of this aid?					
beneficiaries qualify as small farms. Please note that such criteria must be objective. Are the upper and lower thresholds for access to start-up aid for young farmers and the development of small farms defined in terms of the production potential of the agricultural holding, measured in standard output, as defined in Article 5b of Council Regulation (EC) No 1217/2009 ⁽¹⁹⁶⁾ and Article 6 of Commission Implementing Regulation (EU) 2015/220 ⁽¹⁹⁷⁾ , or an equivalent? yes		yes				no		
If the answer is no, please note that the aid cannot be declared compatible with the internal market. 6. Please indicate the upper and lower thresholds for access to start-up aid for young farmers and the development of small farms. Voung farmers Development of small farms	5.	the development of s agricultural holding, 1 Regulation (EC) No	mall farms de neasured in st 1217/2009 ⁽¹⁹	efined in terms of andard output, as of and Article 6	f the prodefined	duction potential of the in Article 5b of Council		
Are young farmers who receive start-up aid setting up a holding in the form of a legal person? Please indicate the upper and lower thresholds for access to start-up aid for young farmers. Poevelopment of small farms Development of small farms Development of small farms Outper threshold Lower threshold Are young farmers who receive start-up aid setting up a holding in the form of a legal person?		yes				no		
Young farmers Development of small farms Upper threshold Lower threshold Please note that the lower threshold for access to start-up aid for young farmers must be higher than the upper threshold for access to aid for the development of small farms. 7. Are young farmers who receive start-up aid setting up a holding in the form of a legal person?	marke	t.			Î			
Upper threshold Lower threshold Please note that the lower threshold for access to start-up aid for young farmers must be higher than the upper threshold for access to aid for the development of small farms. 7. Are young farmers who receive start-up aid setting up a holding in the form of a legal person?		farmers and the devel	opment of sm	all farms.				
Please note that the lower threshold for access to start-up aid for young farmers must be higher than the upper threshold for access to aid for the development of small farms. 7. Are young farmers who receive start-up aid setting up a holding in the form of a legal person?			Young farn	iers				
Please note that the lower threshold for access to start-up aid for young farmers must be higher than the upper threshold for access to aid for the development of small farms. 7. Are young farmers who receive start-up aid setting up a holding in the form of a legal person?								
than the upper threshold for access to aid for the development of small farms. 7. Are young farmers who receive start-up aid setting up a holding in the form of a legal person?	Low	er threshold						
person?								
u yes u no	7.		no receive star	t-up aid setting u	p a holdi	ng in the form of a legal		
		yes				no		

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

8.	If the answer to question 7 is yes, do those young farmers exercise effectiong-term control over the legal person in terms of decisions related to many benefits and financial risks?				
		yes		no	
Please	indicate how suc	ch control is ascertair	ned:		
9.	are not young does the youn	g farmers, participate	in the capital or	al persons, including persons who management of the legal person, erm control either solely or jointly	
		yes		no	
Please	indicate how thi	s is ascertained:			
10.	controlled by exercise effect	a legal person other	er than the young control either solo	on in question is solely or jointly g farmer, does the young farmer ely or jointly together with other	
		yes		no	
Please			ssion of a busines	ss plan to the competent authority	
		yes		no	
If the marke		ease note that the aid	d cannot be decla	ared compatible with the internal	
12.		lementation of the bu on of the decision gra		within nine months from the date	
		yes		no	
If the market		ease note that the aid	d cannot be decla	ared compatible with the internal	
13.	beneficiary ha	as to comply with the	definition of act	the business plan provide that the ive farmer provided for in Article of this from the date of setting up?	

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

14.	skills and concommitment t	he case of start-up aid for young farmers who do not possess adequate occupational alls and competences, does the business plan provide that they are to make a miniment to acquire those skills within 36 months from the date of the adoption of decision on granting the aid?				
		yes		no		
If the a market.		ease note that the aid ca	annot be declared comp	patible with the internal		
15.	In the case of following?	f start-up aid for young	g farmers, does the bus	siness plan describe the		
□(a) □(b)		the initial situation of the milestones and targets		of the activities of the		
□(c)		agricultural holding; details of the actions including those related to environmental sustainability and resource efficiency, required for the development of the activities of the agricultural holding, such as investments, training, advice or any other activity.				
Please cumula		rerequisites referred to	o in points (a), (b) and	d (c) must be fulfilled		
16.	In the case of describe the fo		elopment of small farms	, does the business plan		
□(a) □(b)		the initial situation of the agricultural holding; details of actions including those related to environmental sustainability and resource efficiency, that could support the achievement of economic viability, such as investments, training, cooperation or any other action.				
Please r	note that the prer	requisites referred to in p	points (a) and (b) must b	e fulfilled cumulatively.		
17.	When is the air	id provided?				
□(a) □(b)		annually; in at least two instalments over a period of five years.				
Please g	give exact detail	s:				
18.	In the case of start-up aid for young farmers, is the payment of the last aid tranche or the payment of the last instalment of the aid conditional upon the correct implementation of the business plan referred to in point (179) of the Guidelines?					
		yes		no		

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

- 19. What is the maximum aid intensity in EUR?
- (a) per young farmer:EUR
- (b) per small farm:EUR

OTHER INFORMATION

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

1.1.3. SUPPLEMENTARY INFORMATION SHEET ON AID FOR THE TRANSFER OF AGRICULTURAL HOLDINGS

This form must be used for the notification of any State aid measure to support the transfer of agricultural holdings as described in Section 1.1.3. of Chapter 1 of Part II of the European Union Guidelines for State aid in the agriculture and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

	aid granted only		vith all the rules listed in	question 1 to 7:		
1.	The aid is grapermanently	anted to undertakings a	ctive in primary agricu	Itural production which undertaking active in		
		yes		no		
2.		nted to undertakings eli Title V of Regulation		e small farmers scheme,		
		yes		no		
3.		nted to undertakings whi eligible for a period of a		ing their aid application,		
		yes		no		
4.	entire agricul		corresponding payment	rmanently transfer their entitlements to another		
		yes		no		
5.	The aid is paid	The aid is paid as:				
		an annual payment		one-off payment		
	se note that under ent or as a one-of		delines, the aid must be	paid either as an annual		
6.	The aid is paid from the date of the transfer of the agricultural holding until 31 December 2020;					
		yes		no		
			•	•		

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

7.	The aid corresponds to 120 % of the annual payment that the beneficiary is eligi to receive under the small farmer scheme?						
		yes		no			
OTHE	R INFORMAT	ION					
		ner information conside ection of the Guidelines		essment of the measure			
1.1.4.		SUPPLEMENTARY INFORMATION SHEET ON START-UP AID FOR PRODUCER GROUPS AND ORGANISATIONS IN THE AGRICULTURAL SECTOR					
produc Chapte	er groups and o r 1 of Part II of	rganisations in the agri	cultural sector as descr Guidelines for State aid	roviding Start-up aid for ribed by Section 1.1.4.of in the agricultural and			
1.	TYPE OF A	ID					
1.1.	Does the aid organisations		newly established produc	cer groups and producer			
		yes		no			
1.2.				nised by the competent submission of a business			
		yes		no			
1.3.	Is the aid granted only after verification by the Member State that the objectives of the business plan have been reached within five years from the date of the official recognition of the producer group or organisation?						
		yes		no			
1.4.	Is the aid granted to other production organisations, entities or bodies, such as companies or cooperatives, the objective of which is the management of one or more agricultural holdings and which are therefore in effect single producers?						
		yes		no			

If the answer is yes, please note that aid to those organisations, entities or bodies is not covered by Section 1.1.4. of Chapter 1 of Part II of the Guidelines.

 \Box (a)

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.5.	as mutual sur	port and farm relief ar		n undertake tasks, such rvices, in the members' pply to the market?	
		yes		no	
		ease note that aid to the of the Guidelines.	ose associations is not c	overed by section 1.1.4.	
1.6.			s or organisations to cos investments or promot	ver expenses, which are ion activities?	
		yes		no	
			in accordance with the of the notification form.	specific rules governing	
2.	BENEFICIA	RY			
2.1.	•	aid granted exclusively ne definition of SME's ⁽¹		d producer organisations	
		yes		no	
2.2.		es in the regulations gov		adjusted to take account anisation of the markets	
		yes		no	
		ase note that in accorda		of Chapter 1 of Part II of	
3.	AID INTENS	SITY AND ELIGIBLI	E COSTS		
3.1.		n that the total amount of EUR 500 000.	of aid granted to a produc	eer group or organisation	
		yes		no	
3.2.	the fifth year		which the producer gro	ct of costs incurred after oup or organisation was	
		yes		no	
3.3.	Do the eligible costs include only the costs of:				

the rental of suitable premises;

			D	Oocument Generated: 2024-06-28
		Status: Point in time ses to legislation: There are currentsission Regulation (EC) No 794/2		
□(b)		the acquisition of office software;	e equipment, including	computer hardware and
□(c)		administrative staff cos	ts;	
□(d) □(e)		overheads; legal and administrative	e fees?	
				,
		yes		no
	nswer is no, ple I of the Guideli		eligible costs set in Sec	tion 1.1.4. of Chapter 1
3.4.	years following		e producer group or org	ments for the first five ganisation was officially iness plan?
		yes		no
3.5.		nstalment be paid by the nentation of the busines		after having verified the
		yes		no
ОТНЕН	R INFORMAT	ION		
		ner information conside ection of the Guidelines		essment of the measure
1.1.5.1.		NTARY INFORMATI ENT-CLIMATE COMN		AID FOR AGRI-
producti environi Europed	ion methods des nent-climate co	signed to protect the env ommitments) covered by clines for State aid in th	ironment and to mainta Section 1.1.5.1. of Cha	to support agricultural in the countryside (agri- apter 1 of Part II of the try sectors and in rural
1.	Does the aid Guidelines)?	d only concern enviro	onmental investments	(Section 1.1.1. of the
		yes		no
investm				Sheet 1.1.1.4. on aid for ts and the marketing of
2.		-environmental aid purs lp agricultural producer		as training and advisory Guidelines)?
П		ves		no

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

If yes, please refer to the Supplementary Information Sheet 1.1.10. on 'Aid for provision of technical support in the agricultural sector'.

3. Others?

Please provide a complete description of the measure(s)

4.	Is documentation demonstrating that the aid fits into and is coherent with the relevant
	rural revelopment program attached to the notification?

□ yes □ no	
------------	--

If the answer is yes, please provide that documentation hereunder or in an annex to this supplementary information sheet.

If the answer is no, please note that such documentation is required in accordance with point (47) of the Guidelines.

1. **OBJECTIVE OF THE MEASURE**

1.1. Please confirm that the support measure aims at the preservation as well as at the promotion of the necessary changes to agricultural practices that make a positive contribution to the environment and climate:

	yes		no
--	-----	--	----

If the answer is no, please note that this is a requirement in accordance with point (209) of the Guidelines.

- 1.2. Which one of the following specific objectives does the support measure promote?
- □(a) ways of using agricultural land which are compatible with the protection and improvement of the environment, the landscape and its features, natural resources, the soil and genetic diversity, and the reduction of production costs:
- an environmentally-favourable extensification of farming and management of low-intensity pasture systems, improve and re-deploy production:
- the conservation of high nature-value farmed environments, which are under threat, and an increase of quality;
- \Box (d) the upkeep of the landscape and historical features on agricultural land;
- \Box (e) the use of environmental planning in farming practice.

If the measure does not pursue any of the objectives listed in points (a) to (e), please indicate what are the objectives aimed at in terms of environmental protection? Please submit a detailed description.

If the measure in question has already been applied in the past, what have the results been in terms of environmental protection?

2. ELIGIBILITY CRITERIA

2.1. Will the aid be granted to agricultural undertakings or groups of agricultural undertakings who provide agri-environmental-climate commitments for a period between five and seven years?

		Status: Point in time	Diew as at 31/01/2020.	Occument Generated: 2024-06-28			
			ently no known outstanding effect 2004. (See end of Document for d				
		I	T	T			
		yes		no			
2.2.	Will a longer period be necessary for all or particular types of commitments?						
		yes		no			
If the	answer is yes, ple	ease provide the reasons	justifying that period:				
2.3.	commitments Chapter I of T established pu relevant minii	that go beyond the relevittle VI of Regulation (Eursuant to Article 4(1)(c) mum requirements for for	vant mandatory standard U) No 1306/2013 and o) (ii) and (iii) of Regulat	i-environmental-climate s established pursuant to ther relevant obligations ion (EU) No 1307/2013, tion products use as well onal legislation.			
		yes		no			
agri-ei		nate commitments that o		es not permit for aid for the application of those			
2.4.	Please describe what the mandatory standards and requirements referred to in question 2.3 are and explain how the agri-environmental-climate commitments involve more than their application.						
2.5.	Member States must endeavour to ensure that the agricultural undertakings or groups of agricultural undertakings who give agri-environmental-climate commitments are provided with the knowledge and information required to implement them, such as by commitment-related expert advice and/or by making aid under this measure conditional on obtaining the relevant training. Please confirm and describe if/how this obligation is complied with.						
		yes		no			
2.6.	If relevant, please confirm that the rules for area related payments provided for in Article 47 of Regulation (EU) No 1305/2013 and in any delegated act adopted pursuant to that provision are complied with.						
		yes		no			

3. AID AMOUNT

- 3.1. Please specify the maximum amount of aid to be granted based on the area of the holding to which agri-environmental-climate commitments apply:
- for specialised perennial crops (maximum payment of EUR 900 per \Box (a) hectare per year);
- for annual crops (maximum payment of EUR 600 per hectare per year); □(b)
- for other land uses(maximum payment of EUR 450 per hectare per \Box (c) year);

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□(d) □(e)		local breeds in danger of being lost to farming(maximum payment of EUR 200 per livestock unit per year); other:				
If the n	justify the com	nts of the aid referred to		is question are exceeded Section 1.1.5.1. of the		
3.2.	Is the aid gran					
		yes		no		
If the a	nswer is no, ple	ase provide the reasons	justifying the other peri	od:		
3.3.	Is the amount	t of annual support calcu	ulated on the basis of:			
—(a) —(b) —(c)	income foregone, additional costs resulting from the commitment given; where necessary, the need to provide compensation for transaction costs?					
		yes		no		
Evaloi	.1 1 1					
	e foregone, addit Is the referen	cional costs and transact ace level for calculating mitments given, the ma	ion costs: g income foregone and	support and specify the additional cost resulting requirements as referred		
income	Is the referent from the com	cional costs and transact ace level for calculating mitments given, the ma	ion costs: g income foregone and	additional cost resulting		
3.4.	Is the referent from the combon to in question	cional costs and transact ace level for calculating amitments given, the ma a 2.3?	ion costs: g income foregone and and and atory standards and	additional cost resulting requirements as referred		
3.4.	Is the referent from the company to in question the company to the	cional costs and transact ace level for calculating mitments given, the man 2.3?	ion costs: g income foregone and andatory standards and	additional cost resulting requirements as referred		
3.4. If the a	Is the referent from the company to in question the company to the	ional costs and transact ace level for calculating mitments given, the management of the cost of the c	ion costs: g income foregone and andatory standards and	additional cost resulting requirements as referred		
If the a 3.5.	Is the referent from the company to in question to in question to in question to in question to the paymanswer is no, plear the paymanswer is yes, aken to ensure the paymanswer the paymanswer to ensure the paymanswer the paymanswer to ensure the paymanswer	yes yes ase explain the reference at the maximum amount elines and in the Annex per State intend to give	ion costs: g income foregone and andatory standards and andatory standards and elevel taken into considered and elevel taken into co	no leration: no thod and the initiatives nion support as set out in 1305/2013 are complied as for the continuation of		
If the a 3.5. If the underta point (2 with.	Is the referent from the company to in question to in question to in question to in question to the paymanswer is no, plear the paymanswer is yes, aken to ensure the paymanswer the paymanswer to ensure the paymanswer the paymanswer to ensure the paymanswer	yes yes please explain the reference at the maximum amount elines and in the Annex	ion costs: g income foregone and andatory standards and andatory standards and elevel taken into considered and elevel taken into co	no leration: no thod and the initiatives nion support as set out in 1305/2013 are complied as for the continuation of		

- 3.7. If the answer is yes, please demonstrate that such costs still continue to occur or that new transaction costs are being borne.
- 4. **REVISION CLAUSE**

Status: Point in time view as at 31/01/2020. **Changes to legislation:** There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

		yes		no				
is obl	iged to introduce dments of the relon 1.5.1.1. of the	it in order to en evant mandatory	nsure the adjustme standards, require	E the Guidelines, the Mement of the operations in the ements or obligations referentments referred to in that	e case of rred to in			
4.2.	Does this aid	extend beyond th	ne rural developme	nt programming period 20	14-2020?			
		yes		no				
order devel		ljustment of the c ning period must	perations to the leg	the Guidelines, a revision gal framework of the follow				
	e indicate any oth			t to the assessment of the	measure			
1.1.5.		2. SUPPLEMENTARY INFORMATION SHEET ON AID FOR ANIMAL WELFARE COMMITMENTS						
produ 1 of P	ction methods des	rigned to improve ean Union Guide	e animal welfare co lines for State aid ii	aid measure to support ago vered by Section 1.1.5.2. o In the agriculture and forest	f Chapter			
1.	Does the aid Guidelines)?	d only concern	environmental in	evestments (Section 1.1.1	. of the			
		yes		no				
invest				formation Sheet 1.1.1.4. oural products and the mar				
2.				ives such as training and 1.10. of the Guidelines)?	advisory			

provision of technical support in the agricultural sector.

3. Others?

Please provide a complete description of the measure(s)

4. Is documentation demonstrating that the State aid complies with and is coherent with the relevant rural development program attached to the notification?

Status: Point in time view as at 31/01/2020.
Changes to legislation: There are currently no known outstanding effects for th
Commission Regulation (EC) No 794/2004. (See end of Document for details)

yes	no

If the answer is yes, please provide that documentation hereunder or in an annex attached to this supplementary information sheet

If the answer is no, please note that such documentation is required in accordance with point (47) of the Guidelines.

1. **OBJECTIVE OF THE MEASURE**

- 1.1. For which of the following areas do the animal welfare commitments provide upgraded standards?
- □(a) water, feed and animal care in accordance with the natural needs of animal husbandry;
- housing conditions such as increased space allowances, flooring surfaces, enrichment materials, natural light;
- □(c) outdoor access;
- practices which avoid mutilation and/or castration of animals or in specific cases when mutilation and/or castration of animals is deemed necessary, provide for the use of anaesthetics, analgesia and anti-inflammatory medication or immunocastration.

Please submit a detailed description:

If the measure in question has already been applied in the past, what have been the results in terms of animal welfare?

2. ELIGIBILITY CRITERIA

2.1. Will the aid be granted to undertakings active in the primary agricultural production which are active farmers within the meaning of Article 9 of Regulation (EU) No 1307/2013?

u yes u no	
------------	--

If the answer is no, please note that in accordance with point (232) of the Guidelines the aid can only be granted to undertakings active in primary agricultural production which are active farmers.

2.2. Please confirm that the aid will only be granted to compensate for animal welfare commitments that go beyond the relevant mandatory standards established pursuant to Chapter I of Title VI of Regulation (EU) No 1306/2013 and other relevant mandatory requirements:

□ yes		no
-------	--	----

If the answer is no, please note that point (233) of the Guidelines does not permit aid for animal welfare commitments that do not involve more than the application of the mandatory standards and requirements.

2.3. Please describe what the mandatory standards and requirements referred to in question 2.2. are and explain how the animal welfare commitments involve more than their application:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2.4.		be exclusively grante for a renewable period		dertake animal welfare
		yes		no
2.5.	Will the renev	val of the contract be au	itomatic?	
		yes		no
renewal	l must be describ			elines, the details of the l must be communicated
3.	AID AMOUN	NT		
3.1.	Please specify	the maximum amount	of animal welfare aid to	be granted:
(maxim	um payment of	EUR 500 per livestock	unit)	
	ed breakdown,			compatibility, including apter 1 of Part II of the
3.2.	Is the aid gran	ated annually?		
		yes		no
If the ar	nswer is no, plea	ase provide the reasons	justifying the other period	od:
3.3.	Is the amount	of annual support calcu	lated on the basis of:	
—(a) —(b) —(c)			ng from the commitmen need to provide compe	t given, ensation for transaction
		yes		no
			ne annual aid amount of possible transaction cost	the measure and specify s.
3.4.		mitments given, the ma		additional cost resulting requirements as referred
		yes		no
			1	

If the answer is no, please explain the reference level taken into consideration:

3.5. Are the payments made per livestock unit?

ANNEX I	on Regulation (EC) No PART III.12 t Generated: 2024-06	o 794/2004 of 21 April 2004 impl 28	lementing Council Regulation (E	(U) 183
Document	Change	Status: Point in time vers to legislation: There are currentission Regulation (EC) No 794/2	ntly no known outstanding effects	
		yes		no
initiativ	es undertaken to	ease explain the reason o ensure that the maxim the Annex to Regulation	um amount per year as	set out in point (240) of
3.6.		nber State intend to givelfare commitments?	e aid for transaction cos	sts incurred by entering
		yes		no
		ease provide proof of su undertakings not enterin		
3.7.		nber State intend to give e commitments already		s for the continuation of
		yes		no
		ase demonstrate that succosts are being borne.	ch transaction costs still	continue to be incurred
3.8.	Will transaction	on costs be calculated on	the basis of average cost	ts and/or average farms?
		yes		no
	, , <u>, , , , , , , , , , , , , , , , , </u>	ease demonstrate, as req prises are not overcomp		of the Guidelines, that,
4.	REVISION (CLAUSE		
4.1.	Is a revision c	lause provided for the o	perations included in th	is aid?
		yes		no
required case of to in Se	d to introduce a ramendments of	ase note that under poir revision clause in order the relevant mandatory Chapter 1 of Part II of on have to go.	to ensure the adjustment standards, requirements	of the operations in the or obligations referred

4.2. Does this aid extend beyond the rural development programming period 2014-2020?

yes

If the answer is yes, please note that under point (725) of the Guidelines, a revision clause in order to allow for the adjustment of the operations to the legal framework of the following rural development programming period must to be included.

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

1.

1.4.

Document Generated: 2024-06-28

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.1.6. SUPPLEMENTARY INFORMATION SHEET ON AID CONCERNING NATURA 2000 PAYMENTS AND WATER FRAMEWORK DIRECTIVE PAYMENTS

This form must be used by Member States to notify State aids for disadvantages related to Natura 2000 and the Water Framework Directive⁽¹⁹⁹⁾, as dealt with in Section 1.1.6. of Chapter 1 of Part II of the European Union Guidelines for State aid in the agriculture and forestry sectors and in rural areas 2014-2020 ('the Guidelines').

OBJECTIVE OF THE MEASURE

1.1.		asure concern compen roduction or to other lan		active in the primary
		yes		no
		ns also other land man 243) of the Guidelines.	agers, please provide	detailed justification in
1.2.	resulting from Council Direc	disadvantages in the autive 92/43/EEC ⁽²⁰⁰⁾ ('the Parliament and of the C	reas concerned related to Habitats Directive'), D	ed and income foregone to the implementation of irective 2009/147/EC of irective') and the Water
		yes		no
compe	nsate for other co	sts than those related to		does not permit aid to ed to the implementation k Directive.
1.3.	to disadvantagenvironmenta No 1306/2013	ges resulting from requir l condition provided for and the relevant criteria	ements that go beyond to in Article 94 and Anne	granted only in relation he good agricultural and ex II of Regulation (EU) s as established pursuant 013?
		yes		no

If the answer is no, please note that point (246) of the Guidelines does not permit aid to compensate for other costs than those related to the specific requirements established in that point.

requirements referred to in point (246) of the Guidelines?

Is aid linked to the Water Framework Directive granted only in relation to specific

no

2. ELIGIBILITY CRITERIA

yes

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2.1.	concerned rel			advantages in the areas tive, the Birds Directive
		yes		no
2.1.1.		is yes, please provide a ve(s) in question	ll the details concerning	the relevant provisions
2.1.2.	for aid to con	npensate for other costs nentation of the Habitat	than those resulting from	idelines does not allow m disadvantages related Directive and the Water
2.2.	Water Framev of the river ba objectives of	work Directive, that are asin management plans	in accordance with the p for the purpose of achie eyond the measures requ	t were introduced by the programmes of measures eving the environmental tired to implement other
		yes		no
2.2.1.2.3.	of the Guideli Is the aid gra protection of a adopted as lai	nes: Inted in relation to spe Union legislation existin d down in Article 4(9) of f land use, and/or majo	cific requirements that g at the time the Water Front that Directive and that	go beyond the level of ramework Directive was at impose major changes g practice resulting in a
		yes		no
2.3.1.		is no, please provide a Section 1.1.6 of the Gu		s compatibility with the
3.	AID AMOU	NT		
3.1.	Please specify (UAA):	y the maximum amoun	t of aid, based on the u	itilised agricultural area
□(a) □(b) □(c)		years of EUR 500 per h (Normal maximum Nat (Minimum amount of	nectare) Tura 2000 payment of EU	eriod not exceeding five UR 200 per hectare) e linked to the Water
3.2.	Please explain avoids overco		ensure that payments ar	re fixed at a level which

4. **OTHER INFORMATION**

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

4.1.			the State aid complies was attached to the notifi	vith and is coherent with cation?
		yes		no
	nswer is yes, pl plementary info		mentation hereunder or	in an annex attached to
	nswer is no, ple the Guidelines.	ase note that such docu	mentation is required in	n accordance with point
4.2.		e any other information erned under this Section		o the assessment of the
1.1.7.		ITARY INFORMATIC R OTHER SPECIFIC (TO AREAS FACING
other sp Part II areas 20	pecific constrain of the EU Guide	ts in certain areas, whi elines for State aid in th eGuidelines'). This Sect	ich is covered by Section the agriculture and fores	pensating for natural or n 1.1.7. of Chapter 1 of stry sectors and in rural ngs active in the primary
1.	with Article 32	2 of the Regulation (EU)) No 1305/2013. Please	lesignated in accordance specify the paragraph of e constraint in question.
2.	which are not	affected by natural or		e) in comparison to areas nts, taking into account J) No 1307/2013.
3.	differentiate, v	where duly justified, the led permanent natural of	level of payment taking	will the Member State into account the severity ming activities, and the
		yes		no
4.	Will be the aid	l granted annually per h	nectare of agricultural ar	rea?
		yes		no

If the answer is no, please note that in accordance with point (257) of the Guidelines this aid cannot be granted.

5. What will be the minimum and maximum amount of the aid per hectare per year on average of the area of the beneficiary receiving the aid?

Minimum: Maximum:

Take into account that the aid must be fixed between the following minimum and maximum amounts: EUR 25 minimum per hectare per year on average of the area of the beneficiary receiving aid, and EUR 250 maximum per hectare per year; the maximum amount may reach

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

EUR 450 per hectare per year in mountain areas as defined in Article 32(2) of Regulation (EU) No 1305/2013.

- 6. If the maximum amounts are higher than the maximum permitted, could the Member State explain the specific circumstances that justify them?
- 7. Member States must provide for the degressivity of aid above a threshold level of area per holding, to be defined, except where the grant of aid only concerns the minimum amount per hectare per year as laid down in point (258) of the Guidelines. To that effect, could the Member State specify the size of the farms that will benefit from this aid?
- 8. In addition to the aid provided for in this scheme, will the Member State grant aid under this measure to beneficiaries in areas which were eligible for aid in accordance with Article 36(a)(ii) of Regulation (EC) No 1698/2005⁽²⁰²⁾?

yes	no

If the answer is yes, for beneficiaries in areas that are no longer eligible following the new delimitation referred to in Article 32(3) of Regulation (EU) No 1305/2013, will this aid be degressive over a maximum period of four years starting on the date that the delimitation in accordance with Article 32(3) of Regulation (EU) No 1305/2013 is completed and at the latest in 2018 at no more than 80 % of the average payment fixed in the rural development programme for the programming period 2007-2013 or, if the measure was granted exclusively from national funds, in the relevant State aid decision, in accordance with Article 36(a)(ii) of Regulation (EC) No 1698/2005, and ending in 2020 at the latest at no more than 20 %?

Specify the amounts of payments:

Take into account that when the level of the payment reaches 25 EUR due to degressivity, the Member State can continue to grant aid at this level until the phasing out period is completed.

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

1.1.8. SUPPLEMENTARY INFORMATION SHEET ON AID FOR ORGANIC FARMING

This form must be used for the notification of any State aid measures in favour of organic farming as described in Section 1.1.8 of Chapter 1 of Part II of the European Union Guidelines for State aid in the agriculture and forestry sector and in rural areas 2014 to 2020 ('the Guidelines').

1. GENERAL CONDITIONS AND SCOPE

1.1. Is the aid earmarked only for primary agricultural production?

□ yes		no
-------	--	----

If the answer is no, please note that in accordance with Section 1.1.8 of Chapter 1 of Part II of the Guidelines, aid may be granted only for primary agricultural production.

Status: Point in time view as at 31/01/2020. **Changes to legislation:** There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.2.	who undertak	te on a voluntarily base methods as defined in	sis to convert to or ma	gricultural undertakings aintain organic farming C) No 834/2007 ⁽²⁰³⁾ and		
		yes		no		
			nce with Section 1.1.8 of the conditions are fulfilled	of Chapter 1 of Part II of d.		
2.	COMMITM	ENTS				
2.1.	Is the aid gran requirements:	•	nts which go beyond the	following standards and		
(a)		the relevant mandatory standards established under Chapter 1 of Title VI of Regulation (EC) No 1306/2013;				
(b)	the relevant criteria and minimum activities as established pursuant to Article 4(1)(c) (ii) and (iii) of Regulation (EU) No 1307/2013;					
(c)	the relevant minimum requirements applicable to the use of fertilisers and plant protection products;					
(d)	other relevant mandatory requirements laid down in national law?					
		yes		no		
of the				of Chapter 1 of Part II ond those standards and		
2.2.	Will commitn	nents be carried out ove	r an initial period of five	e to seven years?		
		yes		no		
the Gu	iidelines, aid ma		nt initial period is comp	of Chapter 1 of Part II of lied with, subject to the		
2.3.				l a shorter initial period period of conversion be		
		yes		no		
If the s	answer is ves wh	at will be the duration of	of that period?			

If the answer is yes, what will be the duration of that period?

years

If the aid is granted for the maintenance of organic farming, will an annual extension 2.4. after the termination of the initial period referred to in question 2.2 provided for?

ANNEX I PART III.12 Document Generated: 2024-06-28

Documen	U	Status: es to legislation: T nission Regulation		ly no known outsi	anding effects j	
		yes]	1	10
2.5.		the initial pe				wing the commitment ill a shorter period be
		yes		<u> </u>	1	10
If the a	nswer is yes, wh	nat will be the	duration of	that period?		
years						
2.6.	Please explain, if relevant, how the rules for area related payments laid down in Article 47 of Regulation (EU) No 1305/2013 and any delegated act adopted pursuant to that provision will be complied with:					
3.	PERIODICITY AND ELIGIBLE COSTS					
3.1.	Will the aid cover the following costs:					
(a)	part of additional costs and loss of income resulting from the commitments;					
		yes				no
	If the answer is yes, which part of the costs: %					
(b)	all additional costs and loss of income resulting from the commitments;					
		yes				no
(c)	transaction costs with a maximum value of 20 % of the premium paid for the commitment;					
		yes				no
(d)						premium paid for the altural undertakings?
		yes				no
3.2.	Will the aids	mentioned in	questions 3.	1.(c) and 3.1	.(d) be paid	l on an annual basis?
		yes]	1	10

If the answer is no, please note that in accordance with Section 1.1.8 of Chapter 1 of Part II of the Guidelines, those aids may be granted only on an annual basis.

3.3. If compensation for transaction costs caused by entering into organic farming commitments is the subject of the notification, please provide convincing proof of such costs, for example by submitting cost comparisons with agricultural undertakings not entering into such commitments.

Status: Point in time view as at 31/01/2020. Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

3.4.		n to grant State aid for nitments already entered		continuation of organic	
		yes		no	
3.5.		is yes, please provide eon costs are being incur		continue to occur or that	
3.6.			ne basis of average cost large enterprises are not	s and/or average farms, overcompensated:	
3.7.	environment o			nts covered by an agrid intended to encourage	
		yes		no	
of the G an agri-e	f the answer is no, please note that in accordance with Section 1.1.8 of Chapter 1 of Part II of the Guidelines, no aid for organic farming may be granted for commitments covered by a agri-environment or climate measure, or for costs covered by an aid intended to encourage articipation in quality systems. 8. Is the intention to grant aid for investment in primary production and the processing and marketing of organic products?				
		yes		no	
If the an or 1.1.1.		ease fill in the correspo	nding supplementary in	formation sheet 1.1.1.1.	
4.	AID TYPE A	AND AMOUNT			
4.1.	Which type of	f aid is intended to be gr	ranted?		
□(a) □(b) □(c)	aid per hectare per year for annual crops; aid per hectare per year for specialised perennial crops; aid per hectare per year for other land use.				
4.2.	Please specify	the amounts of aid inte	ended to be granted:		
(a)	aid for annual	crops: EUR per hec	tare (maximum EUR 60	00 per hectare);	
(b)	aid for special hectare);	alised perennial crops:	EUR per hectare (n	naximum EUR 900 per	
(c)	aid for other l	and use: EUR per he	ectare (maximum EUR 4	450 per hectare).	
4.3.	Is it intended	to exceed the ceiling pro	ovided for the type(s) of	f aid to be granted?	
		T			

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

4.4. If the answer is yes, please indicate the amount of aid intended to be granted, explain the exceptional circumstances underlying the intention to grant that amount of aid and justify the amount proposed with figures:
5. REVISION CLAUSE

	justify the am	ount proposed with figu	res:	
5.	REVISION O	CLAUSE		
5.1.	Is a revision c	lause provided for the o	perations included in th	is aid?
		yes		no
State is on the careferred	obliged to introcase of amendme	the note that in accordance luce a revision clause in tents of the relevant ma .1.8. of the Guidelines	order to ensure the adjundatory standards, requ	stment of the operations irements or obligations
5.2.	Does this aid	extend beyond the rural	development programn	ning period 2014-2020?
		yes		no
clause in developm OTHER Please in concerned 1.1.9. This for to encoudescribe	n order to allow ment programmer RINFORMAT andicate any other under this See SUPPLEMEN OF PRODUCTUMENT OF PROD	ase note that in accordance we for their adjustment along period must to be in ION The information considered the control of the Guidelines. The control of the notification of the producers of the control of the producers of the control of the producers of the control of the	to the legal framework included. The red relevant to the associated and state and measure of agricultural products till of the European Unit in rural areas 2014 to 2	essment of the measure THE PARTICIPATION DUALITY SCHEMES res which are designed in quality schemes as ion Guidelines for State
1.1.	Is the aid earn	narked only for produce	rs of agricultural produc	ets?
		yes		no
	delines, aid may	se note that in accordant be granted only if that erred to in point 280(a)	condition is fulfilled	-
		yes		no

If the answer is no, please note that in accordance with Section 1.1.9. of Chapter 1 of Part II of the Guidelines, aid may be granted only if that condition is fulfilled

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2.	ELIGIBLE COSTS				
2.1.	Does the aid cover referred to in point (2			costs in relation	n to quality schemes
□(a) □(b)	the cos scheme behalf	sts for comes undertake of the comp	en pursuant to Upetent authorities	measures in renion or nationa;	elation to the quality al legislation by or on
□(c)					onception and design of quality schemes.
2.2.	Please confirm that by the beneficiaries to control is to be met be specifying the actual	hemselves, y producers	or where Union s of agricultural p	legislation pro	vides that the costs of
	Confir	med			
3.	TYPE OF SCHEM	E AND AC	CESSIBILITY		
For wh	nich type of scheme is th	ne aid for a	new participatior	n granted?	
	Quality schemes established under the following Regulations and provisions:				
	(i)		2 of Chapter I of 2/2013 ⁽²⁰⁴⁾ as cond		II of Regulation (EU)
			yes		no
	(ii)	_	on (EU) No 115 Council ⁽²⁰⁵⁾ ;	1/2012 of the	European Parliament
			yes		no
	(iii)	Regulati	on (EC) No 834/	2007;	
			yes		no
	(iv)		on (EC) No 110/2	2008 of the Eur	opean Parliament and
			yes		no
	(v)	Regulati the Cour		2014 of the Eur	opean Parliament and

Quality schemes, including farm certification schemes, for agricultural products, recognised by the Member States as complying with the following criteria:

no

yes

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (a) the specificity of the final product under such quality schemes must be derived from clear obligations to guarantee:
 - (i) specific product characteristics, or
 - (ii) specific farming or production methods, or
 - (iii) a quality of the final product that goes significantly beyond the commercial commodity standards as regards public, animal or plant health, animal welfare or environmental protection;
- (b) the quality scheme must be open to all producers;
- (c) the quality scheme must involve binding final product specifications and compliance with those specifications must be verified by public authorities or by an independent inspection body;
- (d) the quality scheme must be transparent and ensure complete traceability of agricultural products.

Voluntary agricultural product certification schemes recognised by the Member State as meeting the requirements laid down in Commission Communication — EU best practice guidelines for voluntary certification schemes for agricultural products and foodstuffs⁽²⁰⁸⁾.

4. ACCESS TO THE SCHEME

Is the aid accessible to all eligible undertakings in the area concerned, based on objectively defined conditions?

	
yes	110

If the answer is no, please note that in accordance with Section 1.1.9. of Chapter 1 of Part II of the Guidelines, aid may be granted only if that condition is fulfilled.

5. **PERIODICITY**

П

Will the aid referred to in point 280(a) of the Guidelines be granted as an annual incentive payment, the level of which is determined according to the level of the fixed costs arising from the participation in quality schemes, for a maximum duration of five years?

yes	no

If the answer is no, please note that in accordance with Section 1.1.9. of Chapter 1 of Part II of the Guidelines, aid may be granted only if those conditions are fulfilled.

6. AID AMOUNT/INTENSITY AND CONDITIONS FOR PAYMENT

6.1. What is the amount of the aid referred to in point 280(a) of the Guidelines, to be granted per beneficiary per year?

Please note that in accordance with Section 1.1.9. of Chapter 1 of Part II of the Guidelines, the aid is limited to EUR 3 000 per beneficiary per year.

6.2. What is the intensity of the aid referred to in point 280(b) and (c) of the Guidelines?

1 2

Document Generated: 2024-06-28

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

% the real costs incurred

Please note that in accordance with Section 1.1.9. of Chapter 1 of Part II of the Guidelines, the aid may reach 100 % of the real costs incurred.

6.3. Please confirm that the aid referred to in point 280(b) and (c) of the Guidelines will not involve direct payments to the beneficiaries and will be paid to the body responsible for control measures, the research provider or the consultancy provider:

□ Confirmed OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

1.1.10. SUPPLEMENTARY INFORMATION SHEET ON AID FOR THE PROVISION OF TECHNICAL SUPPORT IN THE AGRICULTURE SECTOR

This form must be used for the notification of any State aid measure aimed at the provision of technical support in the agricultural sector as described by Section 1.1.10. of Chapter 1 of Part II of the EU Guidelines for State aid in the agricultural and forestry sectors and in rural areas 2014-2020 ('the Guidelines').

- 1. COMMON PROVISIONS (Please fill in this section and the relevant section below depending on the objective of the State aid)
- 1.1. Is the aid applied to the agricultural sector, including primary agricultural production, the processing of agricultural products and the marketing of agricultural products?

	\Box yes \Box no
--	----------------------------

Please note that the aid for farm replacement services can be granted only to undertakings active in the primary agricultural production.

Who are the beneficiaries of the aid?

□(a) □(b) □(c) 1.3.		ole in the area concerne	ed based on objectively
	yes		no

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

1.4. Where the provision of technical support is undertaken by producer groups or other organisations is membership of such groups or organisations a condition for access to the service?

yes	no

If the answer is yes, please note that the aid cannot be declared compatible with the internal market.

Status: Point in time view as at 31/01/2020.

1.5.		nisation concerned, ref		ive costs of the producer 4, limited to the costs of		
		yes		no		
If the a	_	ease note that the aid c	annot be declared com	patible with the internal		
2.	AID FOR I (Section 1.1.1		NSFER AND INFO	RMATION ACTIONS		
2.1.	Which of the measure?	following types of aid o	can be financed by the	aid scheme or individual		
□(a) □(b) □(c) □(d)		vocational training and workshops and coachin demonstration activities information actions; aid to cover short-term	g; s;	cluding training courses,		
2.2.	Which of the following eligible costs does the measure cover?					
□(a)		the costs of organising t demonstration activities		skills acquisition actions,		
□(b)				diem expenses of the		
□(c)		the costs of the provision of replacement services during the absence of the participants; the costs related to demonstration projects.				
□(d)			1 0			
2.3.	In the case of demonstration projects, do the eligible costs include?					
□(a)		immovable property, v	vith land only being e	ing, or improvement of eligible to an extent not peration concerned;		
□(b)		exceeding 10 % of total eligible costs of the operation concerned; the purchase or lease purchase of machinery and equipment up to the market value of the asset:				
□(c)	market value of the asset; the general costs linked to expenditure referred to in point (293) (d)(i) and (ii) of the Guidelines, such as architect, engineer and consultation fees, fees relating to advice on environmental and economic sustainability, including feasibility studies;					
□(d)			opment of computer sof	tware and the acquisition		
□(e)			and income foregone r	elating to the small-scale		
Please	note that the elig	gible costs are limited to	those listed in point (2	93) of the Guidelines.		
2.4.		studies remain eligible e under point (293)(d)(i		re, based on their results, ines is incurred?		
		yes		no		

Status: Point in time view as at 31/01/2020. Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2.5.		If the intention is to grant aid for additional expenses and income foregone relating to a small-scale demonstration project, please justify?					
2.6.		used for a demonstra		of the Guidelines eligible the duration period of the			
		yes		no			
2.7.		apacities in the form o		rmation services have the nd regular training to carry			
		yes		no			
2.8. □(a) □(b)	The aid will be granted in the form of: subsidized services; direct payments of money to producers only under the form of reimbursement of actually incurred costs?						
		referred to in point (2 volve direct payments		int (293)(d)(i) to (iv) of the			
2.9.	As an exception to the question 2.8, will the recipient of the aid referred to in point (293)(a) and (c) and point (293)(d)(i) to (iv) of the Guidelines be the provider of knowledge transfer and information action?						
		yes		no			
(293)(c service	c) of the Guidelines, and the aid for	nes may, alternatively, l	be paid directly to the ration projects, referre	services referred to in point provider of the replacement ed to in point (293)(d)(i) to			
2.10.	If the answer	to question 2.9 is no, p	please justify?				
2.11.	Will the aid referred to in point (293)(d)(v) of the Guidelines be paid directly to the beneficiaries?						
		yes		no			
If the a		ease note that the aid	cannot be declared co	ompatible with the internal			
2.12.	Please state th	ne maximum aid intens	sity (maximum 100 %))			
2.13.				d) of the Guidelines is the od of three fiscal years?			
		yes		no			

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

3. AID FOR ADVISORY SERVICES (Section 1.1.10)		AID FOR	ADVISORY	SERVICES	(Section	1.1.10.2	.)
---	--	---------	-----------------	-----------------	----------	----------	----

3.1. Is the aid designed to help undertakings active in the agricultural sector and young farmers benefit from the use of advisory services for the improvement of the economic and environmental performance as well as the climate friendliness and resilience of their holding and/or investment?

		yes		no			
2.2							
3.2.	Advice will c	over as a minimum one	of the following elemen	nts:			
□(a)		Obligations deriving from the statutory management requirements and/or standards for good agricultural and environmental conditions					
□(b)		provided in Chapter I of Title VI of Regulation (EU) No 1306/2013; The agricultural practices beneficial for the climate and the environment as laid down in Chapter 3 of Title III of Regulation (EU) No 1307/2013 and the maintenance of the agricultural area as referred to in point (c) of Article 4(1) of that Regulation;					
□(c)		Measures aiming at farm modernization, competitiveness building, sectoral integration, innovation and market orientation as well as the promotion of entrepreneurship;					
□(d)		Requirements defined by Member States for implementing Article 11(3) of the Water Framework Directive;					
□(e) □(f) □(g)		Requirements defined by Member States for implementing Article 55 of Regulation (EC) No 1107/2009 of the European Parliament and of the Council ⁽²⁰⁹⁾ , and in particular compliance with the general principles of integrated pest management as referred to in Article 14 of Directive 2009/128/EC of the European Parliament and of the Council ('the Pesticides Directive') ⁽²¹⁰⁾ ; Occupational safety standards and safety standards linked to farms; Specific advice for farmers setting up for the first time, including advice					
⊔(<i>§)</i>		on economic and enviro		st time, merading advice			
If the a	If the answer is no, please note that the aid cannot be declared compatible with the internal						

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

3.3. Is advice about one or more of the options as listed in question 3.2, linked to at least one Union priority for rural development?

yes	no

Please provide details:

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

3.4. Which of the following types of advice is financed by the aid scheme or individual measure?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□(a)	adaptation, biodiversity and protection of waters as laid down i						
□(b)		I to Regulation (EC) No 1306/2013; advice on issues linked to the economic and environmental performance of the agricultural holding, including competitiveness aspects;					
□(c)		advice for the developm	nent of short supply chair	ns, organic farming and			
□(d)		health aspects of anima advice on other issues.	I husbandry;				
Please	describe the envi	isaged measures:					
3.5.		pe granted to the advisone beneficiaries:	ry services provider and	must not involve direct			
		yes		no			
If the a		ease note that the aid ca	annot be declared comp	patible with the internal			
3.6.	of regularly to			te resources in the form nce and reliability with			
		yes		no			
If the a		ease note that the aid ca	annot be declared comp	patible with the internal			
3.7.	Is advice partl	y provided in a group?					
		yes		no			
			oriate, advice may be par dividual user of the adv	rtly provided in a group, isory services.			
3.8.	If the answer	to question 3.7 is yes, p	lease justify?				
3.9.	Is the amount	of aid limited to EUR 1	500 per advice?				
		yes		no			
If the a		ease note that the aid ca	annot be declared comp	patible with the internal			
3.10.				rs undertake to respect of Regulation (EU) No			
		yes		no			
			l	L			

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

Status: Point in time view as at 31/01/2020.

4.1.	Are undertaki of this aid?	ngs active in the prima	ary agricultural production	on the sole beneficiaries				
		Are undertakings active in the primary agricultural production the sole beneficiaries of this aid?						
		yes		no				
If the ans	swer is no, ple	ease note that the aid o	cannot be declared comp	patible with the internal				
4.2.	The aid must be granted to the provider of the farm replacement service and must involve direct payments to farmers:							
		yes		no				
If the ans	swer is no, ple	ease note that the aid o	cannot be declared comp	patible with the internal				
4.3.	a farmer, a naduring their ab	atural person who is a psence from work due to	f the real costs incurred a member of farm hous o illness, including illness tory military service or in	sehold, or farm worker, s of their child, holidays,				
		yes		no				
Please no 4.4.	note that the eligible costs are limited to those listed under point (310) of the Guidelines. Is the total duration of the replacement with the exception of replacement for maternity and parental leave and during mandatory military service covered by the aid limited to three months per year per beneficiary?							
		yes		no				
	ote that in duly ger time period		l duration of the replace	ment may be authorised				
4.5.	If the answer	to question 4.4 is no, pl	lease justify?					
4.6.		ration of the replacement to six months in each	ent for maternity and pacase?	arental leave covered by				
		yes		no				
	ote that in duly		acement for maternity ar	nd parental leave may be				
4.7.	If the answer	to question 4.6 is no, pl	lease justify?					
4.8.	Is the total duration of the replacement for mandatory military service covered by the aid limited to the duration of the service?							

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

4.9. Please state the maximum aid intensity (maximum 100 %):

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

1.1.11. SUPPLEMENTARY INFORMATION SHEET ON AID FOR THE PROVISION OF COOPERATION IN THE AGRICULTURE SECTOR

This form must be used for the notification of any State aid measure whose aim is the provision of cooperation in the agricultural sector as described by Section 1.1.11. of Chapter 1 of Part II of the European Union Guidelines for State aid in the agricultural and forestry sectors and in rural areas 2014-2020 ('the Guidelines').

1. **TYPE OF AIDS**

1.1. Is this cooperation applicable in the agricultural sector, including primary agricultural production, the processing of agricultural products and the marketing of agricultural products?

u yes u no	
------------	--

- 1.2. The cooperation involves at least two entities, such as:
- cooperation approaches among different actors in the agricultural sector, food chain (only if the result of the processing is an agricultural product) and other actors active in the agricultural sector that contribute to achieving the objectives and priorities of rural development policy, including producer groups, cooperatives and inter-branch organisation;
- \Box (b) the creation of clusters and networks in the agricultural sector;
- □(c) the establishment and operation of operational groups of the EIP for agricultural productivity and sustainability as referred to in Article 56 of Regulation (EU) No 1305/2013.

2. ELIGIBLE COSTS AND AID INTENSITY

- 2.1. The aid will be granted for cooperation relating to the following activities:
- \Box (a) pilot projects;
- the development of new products, practices, processes and technologies in the agriculture and food (as far as agricultural products are concerned) sectors;
- cooperation among small operators in the agricultural sector in organising joint work processes, sharing facilities and resources;
- nd(d) horizontal and vertical cooperation among supply chain actors for the establishment of logistic platforms to promote short supply chains and local markets:
- promotion activities in a local context relating to the development of short supply chain and local markets;
- \Box (f) joint action undertaken with a view to mitigating or adapting to climate change;

Status: Point in time view as at 31/01/2020.
Changes to legislation: There are currently no known outstanding effects for the
Commission Regulation (EC) No 794/2004. (See end of Document for details)
·

□(g)		practices including effi	icient water managemer				
□(h)		energy ⁽²¹¹⁾ and the preservation of agricultural landscapes; horizontal and vertical cooperation among supply chain actors in the sustainable production of biomass for use in food production if the result					
□(i)		is an agricultural product and energy production for own consumption implementation, in particular by groups of public and private partner other than those defined in point (b) of Article 32(2) of Regulation (E) No 1303/2013 of the European Parliament and of the Council (212), local development strategies other than those defined in Article 2(1) of that Regulation addressing one or more of the Union priorities to					
2.2.	In the case of		clusters and networks, v and those commencing	vill it be granted only to an activity that is new			
		yes		no			
	be granted. Aid for pilot particular technologies is are concerned.	projects and the develop n the agriculture and th I may be granted also	oment of new products, e food sector only as far to individual actors. V				
		yes		no			
	answer is no, ple be granted.	ase note that in accorda	ance with point (318) of	the Guidelines this aid			
2.4.	to in point (31		idelines, cover only sup	pply chains, as referred ply chains involving no			
		yes		no			
	answer is no, ple be granted.	ase note that in accorda	ance with point (319) of	The Guidelines this aid			
2.5.		omply with the relevant and 102 of the Treaty?	provisions of competition	on law, in particular with			
		yes		no			
		I.	1	<u> </u>			

If the answer is no, please note that in accordance with point (320) of the Guidelines no aid may be granted.

If the answer is yes, please explain how this compliance will be ensured.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2.6. Aid may be granted to cover the following eligible costs in so far as they concern agricultural activities:

\Box (a)	studies of the area concerned, feasibility studies, and the drawing up of
	a business plan or local development strategy other than the one referred
	to in Article 33 of Regulation (EU) No 1303/2013;

- the costs for the animation of the area concerned in order to make feasible a collective territorial project or a project to be carried out by an operational group of the EIP for agricultural productivity and sustainability as referred to in Article 56 of Regulation (EU) No 1305/2013; in the case of clusters animation may also concern the organisation of training, networking between members and the recruitment of new members;
- □(c) the running costs of cooperation, such as the salary of a 'coordinator'; the direct costs of specific projects linked to the implementation of a business plan, environmental plan, a local development strategy other than the one referred to in Article 33 of Regulation (EU) No 1303/2013 or other actions aiming at innovation including testing; the direct costs must be limited to the eligible costs of investment aid, as specified in the Section 1.1.1.1 on aid for investment of Part II of the Guidelines; the costs of promotion activities.
- 2.7. Will the aid be limited to a maximum period of seven years except for collective environmental action in duly justified cases?

yes	no

Please provide the justification for collective environmental actions exceeding a period of seven years:

2.8. The aid will be granted up to:% of the eligible costs.

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

1.2.1.1. SUPPLEMENTARY INFORMATION SHEET ON AID TO MAKE GOOD THE DAMAGE CAUSED BY NATURAL DISASTERS OR EXCEPTIONAL OCCURRENCES

This form must be used by Member States for the notification of any State aid measures which are designed to compensate for damage to agricultural production or the means of agricultural production caused by a natural disaster or exceptional occurrences as described by Section 1.2.1.1. of Chapter 1 of Part II of the European Union Guidelines for State aid in the agriculture and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

- 1. Which natural disaster or exceptional occurrence caused (or, in the case of an *ex-ante* framework aid scheme⁽²¹³⁾, could cause) the damage for which the compensation is required?
- 2. When did the event specified in question 1 occur?
- 3. Please indicate the last date on which aid may be paid out.

Status: Point in time view as at 31/01/2020.

4.	Have the competent authority of the Member State formally recognised that disaster or an exceptional occurrence has occurred?						
		yes		no			
5.	Please demonstrate a direct link between the natural disaster or exceptional occurrence and the damages incurred by the agricultural undertaking active in primary agricultural production, the processing and the marketing of agricultural products.						
6.	Who is the aid paid to? Is it paid directly to the agricultural undertaking concerned, or to a producer group or organisation of which the agricultural undertaking is a member?						
7.				sured that the amount of lundertaking is eligible?			
8.	Please submit an assessment as precise as possible of the damage incurred by the potential beneficiaries.						
9.	Are only those costs of the damage which is a direct consequence of the natural disaste or exceptional occurrence eligible for aid?						
		yes		no			
10.	Is the direct consequence referred to in question 9 assessed by a public authority, by an independent expert recognised by the granting authority or by an insurance undertaking?						
		yes		no			
If the an	swer is yes, ple	ase specify which one:					
11.	What kind of	compensation is covere	d by the aid (multiple re	esponses are possible)?			
□(a)				, equipment, machinery,			
□(b)	stocks and means of production; compensation for loss of income resulting from the full or partial destruction of the agricultural production and the means of the primary agricultural production.						
12.	Will the dama	ge be calculated at the l	evel of the individual b	eneficiary?			
		yes		no			
13.	If material damage as specified in question 11 is concerned, is such damage calculated on the basis of the repair cost or economic value of the affected asset before the natural disaster or the exceptional occurrence?						
		yes		no			

22.

eligible costs?

Document Generated: 2024-06-28

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

14.	Will indexes be used in order to calculate the annual agricultural production beneficiary?						
		yes		no			
15.	If the answer to question 14 is yes, will the calculation method used permit th determination of the real loss of the beneficiary in the given year?						
		yes		no			
16.		surement of the extent		tailored to the specific			
□(a)			antity of biomass loss) farm, local, regional or i	or equivalent yield loss			
□(b)		weather indexes (inc		infall and temperature)			
		yes		no			
17. 18.	If the answer to point (a) or (b) of question 16 is yes, how does the Member State intend to ensure that the respective calculations are representative, and not based on abnormally high yields and do not result in the overcompensation on any beneficiary? Has the natural disaster or the exceptional occurrence affected a wide area in the same way?						
		yes		no			
19.	If the answer	to question 18 is yes, w	vill aid payments be base	ed on average losses?			
		yes		no			
20.	If the answer to question 19 is yes, how does the Member State intend to ensure that the average losses referred to in question 19 are representative and do not result in significant overcompensation of any beneficiary?						
21.	Are other payments received by the beneficiary of the aid, such as under insurance policies, deducted from the amount of eligible costs?						
		yes		no			
		yes		no			

How does the Member State intend to ensure that any other costs not incurred because of the natural disaster or exceptional occurrence will be deducted from the amount of

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

23. Please indicate the maximum gross aid intensity as a percentage of the eligible costs.

The following question applies in the case of notification of an ex ante framework aid scheme to compensate for the damages of natural disasters:

24. Please clearly stipulate the conditions under which aid can be granted in the event of earthquakes, avalanches, landslides and floods as well as for tornadoes, hurricanes, volcanic eruptions and wild fires of natural origin.

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

1.2.1.2. SUPPLEMENTARY INFORMATION SHEET ON AID TO COMPENSATE FOR DAMAGE CAUSED BY ADVERSE CLIMATIC EVENT WHICH CAN BE ASSIMILATED TO A NATURAL DISASTER

This form must be used by Member States for the notification of any State aid measures which are designed to compensate for damage to agricultural production or the means of agricultural production caused by an adverse climatic event which can be assimilated to a natural disaster as described by section 1.2.1.2. of Chapter 1 of Part II of the EU Guidelines for State aid in the agriculture and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

- 1. What adverse climatic event which can be assimilated to a natural disaster has justified the aid?
- 2. When did the climatic event which can be assimilated to a natural disaster referred to in question 1 occur?
- 3. Please indicate the last date on which aid may be paid out.
- 4. Please demonstrate as to why the adverse climatic event can be assimilated to a natural disaster.

5.	Will aid only be granted to primary agricultural production?				
		yes		no	

6. Has the occurrence of the adverse climatic event which can be assimilated to a natural disaster been formally recognised by the competent authority of the Member State?

|--|

- 7. Please demonstrate a direct link between the adverse climatic event which can be assimilated to a natural disaster and the damages incurred by the agricultural undertaking active in primary agricultural production.
- 8. Have criteria been established in advance on the basis of which an adverse climatic event which can be assimilated to a natural disaster is to be formally recognised?

		yes		no
--	--	-----	--	----

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	* * * * * * * * * * * * * * * * * * * *	ting meteorological into	ormation in relation to the		
Is the aid paid	d directly to?				
		ultural undertaking acti	ve in primary agricultural		
	a producer group or o		e agricultural undertakings		
amount of aid	does not exceed the a				
Please indicat	te the costs eligible fo	or aid:			
	or partial destruction production resulting which can be assimit (354)(b) of the Guide	n of the agricultural p from the occurrence of ilated to a natural disas elines;	oroduct and the means of an adverse climatic event ster as referred to in point		
compensation for material damage to farm buildings and farm equipment and machinery, stocks and means of production caused by an adverse climatic event which can be assimilated to a natural disaster as referred to in point (354)(a) of the Guidelines.					
subtracting th in the year of or each follo production, b of multiplyin preceding thr	e result of multiplying the adverse climatic e wing year affected b y the average selling g the average annual ee-year period or a th	g the quantity of the agri- event which can be assing y the full or partial de- g price obtained during quantity of agricultural ree year average based	cultural products produced nilated to a natural disaster, estruction of the means of that year; from the result I products produced in the on the preceding five-year		
	yes		no		
to the adverse	e climatic event that c	an be assimilated to a n	atural disaster be added to		
	yes		no		
and costs not	incurred because of t	he adverse climatic ever	nt (for example because of		
	yes		no		
	If the answer amount of aid undertaking is Please indicated. If the answer subtracting the in the year of or each follo production, bo f multiplying preceding the period (excluded). If the answer to the adverse the amount results and costs not non-harvestiments.	production, or a producer group or or referred to in point (a) If the answer to the question 10 is amount of aid does not exceed the a undertaking is eligible? Please indicate the costs eligible for compensation for the or partial destruction production resulting which can be assimit (354)(b) of the Guida compensation for an equipment and mach an adverse climatic error as referred to in point of the answer to question 12 is possibility in the year of the adverse climatic error each following year affected be production, by the average selling of multiplying the average annual preceding three-year period or a the period (excluding the highest and less yes) If the answer to question 13 is yes, to the adverse climatic event that of the amount resulting of the calcular yes If the answer to question 13 is yes, and costs not incurred because of the non-harvesting of the deducted from of income?	the concerned agricultural undertaking acti production, or a producer group or organisation of which the referred to in point (a) are members. If the answer to the question 10 is point (b), please explain amount of aid does not exceed the amount of aid to which explained to a mount of aid to seligible? Please indicate the costs eligible for aid: compensation for the loss of the beneficial or partial destruction of the agricultural production resulting from the occurrence of which can be assimilated to a natural disase (354)(b) of the Guidelines; compensation for material damage to fequipment and machinery, stocks and mean an adverse climatic event which can be assim as referred to in point (354)(a) of the Guidel If the answer to question 12 is point (a), will the loss of subtracting the result of multiplying the quantity of the agring in the year of the adverse climatic event which can be assim or each following year affected by the full or partial deproduction, by the average selling price obtained during of multiplying the average annual quantity of agricultural preceding three-year period or a three year average based period (excluding the highest and lowest entry) by the average lift the answer to question 13 is yes, will the other costs incure to the adverse climatic event that can be assimilated to a machine the amount resulting of the calculation of the loss of incomes.		

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

16.	Will indexes beneficiary?	be used in order to calc	culate the annual agricu	ltural production of the
		yes		no
17.		to question 16 is yes of the real loss of the b		nethod used permit the year?
		yes		no
18.		od of calculation of the es		pe tailored to the specific
(a)		lexes (that is to say, qualished at farm, local, reg		or equivalent yield loss or
(b)		kes (including the quand or national level?	tity of rainfall and tem	perature) established at
		yes		no
19.		se climatic event which the same way?	can be assimilated to a	natural disaster affected
		yes		no
20.	If the answer	to question 19 is yes, w	ill aid payments be base	d on average losses?
		yes		no
21.	referred to qu		ative, and not based on	hat these average losses abnormally high yields?
22.				economic value of the assimilated to a natural
		yes		no
23.	Will the calcu	lation of loss be made a	t the level of an individ	ual beneficiary?
		yes		no

24. Please state the maximum gross aid intensity as a percentage of the eligible cost.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

25.	Are areas faci be assimilated			ffected by the	he adverse c	limatic event w	hich can
		yes				no	
26.	taken out inst	irance co lated inc	overing at least come and the s	50 % of h	is/her averag	rmer concerned ge annual prodent climatic risl	uction or
		yes				no	
Memb covering was no OTHE	answer is no, ple per State can conv ng the statisticall of available at the ER INFORMAT	vincingly y most fi time the ION	show that, des requent climation damage occur	spite all reas c risks in the red.	sonable effor e Member St	rts, affordable i tate or region co	nsurance oncerned
	indicate any oth rned under this Se				t to the asso	essment of the	measure
1.2.1.3	3. SUPPLEMEN ANIMAL DIS		INFORMATION AND PLANT		T ON AID	FOR COM	BATING
costs of to mak	orm must be used of prevention, con the good the dama 3. of Chapter 1 of s in rural areas 2	ntrol and ige cause Part II o	l eradication of ed by animal d f the EU Guide	f animal dis iseases and lines for Sta	eases and p plant pests	lant pests and as described by	State aid y Section
1.	What animal	disease o	or plant pest is i	nvolved?			
2.	Will aid only	be grante	ed to undertakin	ngs active ir	n primary ag	ricultural produ	action?
		yes				no	
3.	Will aid only	be paid:					
□(a)						which Union or lation or admir	
□(b)		as part o	of a:				
		(i)		ontrol or era		or regional lev he animal disea	
		(ii)	emergency mor	easures imp	osed by con	npetent public a	authority,
		(iii)	measures to e accordance w		_	ant pest implen 000/29/EC ⁽²¹⁴⁾ .	nented in

Status: Point in time view as at 31/01/2020.

4.	Please attach to the notification a description of the prevention, control and eradication measures concerned.					
5.	Article 14(1)		etive ⁽²¹⁵⁾ and Article 55	nber State implemented of Regulation (EC) No ?		
		yes		no		
6.				respect of which Union are to be borne by the		
		yes		no		
7.	Was the anim the beneficiar		pest caused deliberately	or by the negligence of		
		yes		no		
8.	the list of ani	mal diseases established diseases and zoonoses	d by the World Organis	disease is referred to in ation for Animal Health to Regulation (EU) No		
		yes		no		
9. 10.	When did the cost or loss caused by the animal disease or plant pest occur? Please indicate the last date on which aid may be paid out.					
11.		which are not incurred ould have been incurred		nal disease or plant pest ciary eligible for aid?		
		yes		no		
12.		measures (i.e. measures curred) are concerned, J		isease or plant pest that are eligible for aid:		
□(a) □(b) □(c) □(d)			administration and d	istribution of vaccines, nals and plant protection		

Status: Point in time view as at 31/01/2020.

□(e)				imals or destruction of s and the cleaning and			
		disinfection of the hold	ing and equipment.	· ·			
13.	a competent a competent aut	uthority has formally re	cognised an outbreak, o	r plant pests for which a e) are concerned, please			
□(a)		tests and other screening measures in case of animal diseases, including transmissible spongiform encephalopathy (TSE) and bovine spongiform encephalopathy (BSE) tests;					
□(b)		the purchase, storage,	administration and di	stribution of vaccines, of animals and plant			
□(c)		the slaughter or culling and destruction of animals and the destruction of products linked to them or the destruction of plants, including those that die or are destroyed as a result of vaccination or other measures ordered by the competent authority and the cleaning and disinfection of the holding and equipment.					
14.	Please specify	by which means aid w	ill be granted?				
□(a) □(b)		in kind; the real costs are reimb	ursed to the beneficiary.				
15.		the question 14 is point points (374)(d) and (375)		e eligible costs are those			
		yes		no			
16.	If the answer	to question 15 is no, are	plants concerned?				
		yes		no			
17.		to question 16 is yes, p)(e) and (375)(c) of the		ble costs are referred to			
		yes		no			
18.		aid to make good the consation calculated in re		nimal disease or a plant			
(a)	linked to then	n, or the plants destroye	d, as a result of the anir	we died or the products nal disease or plant pest in point (366)(b) of the			
		yes		no			
		<u> </u>					

Status: Point in time view as at 31/01/2020.

yes no Will the aid be limited to the costs and damage caused by animal diseases and pests for which the competent authority: has formally recognised an outbreak, in the case of an animal disease has formally acknowledged their presence, in the case of plants of the cost of plants of the cost of the same eligible costs will not exceed 100 % of the eligible costs of the same eligible costs of the eligible costs of the same eligible costs of the same eligible costs of the eligib	ease, or pests? by the curance
pests for which the competent authority: □(a) has formally recognised an outbreak, in the case of an animal dise has formally acknowledged their presence, in the case of plants of the case of an animal disection of the case of plants of the case of an animal disection of the case of an anim	ease, or pests? by the curance
Does the Member State commit that the aid and any other payments received beneficiary, including payments under other national or Union measures or ins policies for the same eligible costs will not exceed 100 % of the eligible costs?	pests? by the surance
beneficiary, including payments under other national or Union measures or ins policies for the same eligible costs will not exceed 100 % of the eligible costs?	urance
□ yes □ no	
OTHER INFORMATION	
Please indicate any other information considered relevant to the assessment of the m concerned under this Section of the Guidelines.	ıeasure
1.2.1.4. SUPPLEMENTARY INFORMATION SHEET ON AID FOR FALLEN STOC	CK
This form must be used by Member States for the notification of any State aid measures for stock as described by Section 1.2.1.4. of Chapter 1 of Part II of the EU Guidelines for Stain the agriculture and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines'	ate aid
1. Will aid only be granted to undertakings active in primary agricultural product	tion?
□ yes □ no	
 Please indicate the costs eligible for aid and the applicable aid intensities: 	
□(a) costs for the removal of fallen stock:% □(b) costs for the destruction of the fallen stock:% □(c) costs for the removal and destruction of the fallen stock where the financed through fees or through compulsory contributions desting the financing of the removal and destruction of the fallen stock, protection of the fallen stock important stock in the fallen stock is a stock in the fallen stock where the fallen stock is a stock in the fallen stock in the fallen stock in the fallen stock is a stock in the fallen s	ned for ovided
the meat sector:%	
costs for the removal and destruction of the fallen stock where t an obligation to carry out TSE tests on the fallen stock concer in the case of an outbreak of an animal disease referred to in to of animal diseases established by the World Organisation for Health or the animal diseases and zoonoses listed in Annexes I to Regulation (EU) No 652/2014 ⁽²¹⁸⁾ :%	the list Animal
an obligation to carry out TSE tests on the fallen stock concer in the case of an outbreak of an animal disease referred to in a of animal diseases established by the World Organisation for A Health or the animal diseases and zoonoses listed in Annexes I	the list Animal and II

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

4.	Does the aid livestock sector		nt of money to the unc	lertakings active in the
		yes		no
5.	undertakings		sector, and that provide	downstream from the e services linked to the
		yes		no
6.	Will aid be gra	anted towards the costs	of disposal of slaughter	house waste?
		yes		no
7.	Will aid be slaughterhous		ts undertaken in relat	ion to the disposal of
		yes		no
OTHER	RINFORMAT	ION		
		ner information conside ection of the Guidelines		essment of the measure
1.2.1.5.		NTARY INFORMATIO GE CAUSED BY PRO		O COMPENSATE FOR
are design 1.2.1.5. (gned to comper of Chapter 1 of I	nsate for damage cause Part II of the European (d by protected animals	tate aid measures which as described in Section te aid in the agricultural
1.	Are undertaki of this aid?	ngs active in the prima	ry agricultural production	on the sole beneficiaries
		yes		no
If the an market.	swer is no, ple	ease note that the aid c	annot be declared comp	patible with the internal
2.				the beneficiaries, which nimals in the concerned
		yes		no

If the answer is no, please note that the aid can be declared compatible with the internal market only if clear evidence is submitted showing the impossibility to take such preventive measures.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

3.	If the answer to	question 2	is yes, p	lease indicate	what type	of preventive	measures
	were requested	(for example	e, safety	fences where j	possible, liv	estock guard	ing dogs).

- 4. Which protected animal caused the damage for which the compensation is envisaged?
- 5. What kind of damage was caused?
- 6. Please demonstrate a direct causal link between the damages incurred by the agricultural undertaking active in primary agricultural production and the behaviour of the protected animals.
- 7. Is the aid paid directly to the agricultural undertaking concerned or to a producer group or organisation of which that undertaking is a member?

yes	no

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

8. If the aid is paid to a producer group or organisation, does the amount of aid exceed the amount of aid to which the agricultural undertaking is eligible?

u yes u no

If the answer is yes, please note that the aid cannot be declared compatible with the internal market.

9. When did the event that caused the damage occur?

Please note that the aid scheme must be established within a period of three years following the date of occurrence of the loss or damage.

10. Please indicate the last date on which aid may be paid out.

Please note that the aid must be paid out within a period of four years following the date of occurrence of the loss or damage.

11. Is the damage calculated at the level of the individual beneficiary?

yes	no

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

- 12. What kind of compensation is covered by the aid (multiple responses are possible)?
- □(a) compensation for animals killed or plants destroyed;
- compensation for indirect costs incurred;
- compensation for material damage to farm equipment, machinery and farm buildings and stocks.

Please, note that aid for investments in measures to prevent damage caused by protected animals, can be granted in accordance with the conditions of Section 1.1.1.1 of Chapter 1 of Part II of the Guidelines and not Section 1.2.1.5. of the Guidelines.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

13.	If the answer to question 12 is point (a), are the eligible costs calculated on the of the market value of the animals killed or the plants destroyed?			
		yes		no
If the armarket.	nswer is no, ple	ease note that the aid ca	annot be declared comp	patible with the internal
14.	If the answer to question 12 is point (b), please indicate exhaustively all the indirect costs that can be reimbursed (such as veterinary costs resulting from the treatment of wounded animals and labour costs related to the searches for missing animals).			
15.	If the answer to question 12 is point (c), is the material damage calculated on the basis of the repair cost or economic value of the affected asset before the event that caused the damage?			
		yes		no
If the armarket.	nswer is no, ple	ease note that the aid ca	annot be declared comp	patible with the internal
16.	If the answer to question 12 is point (c), does the aid exceed the repair cost or the decrease in fair market value caused by the event that caused the damage, that is to say, the difference between the property's value immediately before and immediately after that event?			
		yes		no
If the armarket.	nswer is yes, pl	ease note that the aid c	annot be declared comp	patible with the internal
17.	Is the aid restricted to damage incurred as a direct consequence of the event that cause the damage?			
		yes		no
If the armarket.	nswer is no, ple	ease note that the aid ca	annot be declared comp	patible with the internal
18.	Which of the	following bodies is resp	onsible for assessing the	e eligible costs?
□(a) □(b) □(c)	a public authority; an independent expert recognised by the granting authority; an insurance undertaking.			
	ote that the cosisted in points (if the assessment was n	nade by one of the three
19.	Are other payments received by the aid beneficiary, for example, under insurance policies, deducted from the amount of eligible costs?			
		yes		no

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

20.	re costs not incurred because of the event that caused the damage, which wo	uld
	herwise have been incurred by the beneficiary, deducted from the amount of the a	id?

u yes u	no
---------	----

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

- 21. How is it intended to ensure that overcompensation as a result of the combination of this aid with other national or Union support instruments or private insurance schemes is avoided?
- 22. Please indicate the gross aid intensity as a percentage of the direct eligible costs.
- 23. Please indicate the gross aid intensity as a percentage of the indirect eligible costs.

Please note that compensation for indirect costs must not exceed 80 % of the total indirect eligible costs.

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

1.2.1.6. SUPPLEMENTARY INFORMATION SHEET ON AID FOR THE PAYMENT OF INSURANCE PREMIUMS

This form must be used for the notification of any State aid schemes designed to compensate for the payment of insurance premiums as described by Section 1.2.1.6. of Chapter 1 of Part II of the EU Guidelines for State aid in the agricultural and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

1. Does the aid measure provide for the payment of insurance premiums in favour of companies active in the processing and the marketing of agricultural products?

yes	no

If the answer is yes, please note that in accordance with point (406) of the Guidelines the Commission can only authorise aid towards the payment of insurance premiums to undertakings active in primary agricultural production.

- 2. Please specify which damages will be covered by the insurance for which the premium will be partly financed under the notified aid measure:
- Damage caused by natural disasters or exceptional occurrences, adverse climatic events which can be assimilated to a natural disaster, animal diseases and plant pests, the removal and destruction of fallen stock and damage caused by protected animals, as referred to in Sections 1.2.1.1. to 1.2.1.5. of the Guidelines, as well as by other adverse climatic events. Damages caused by environmental incidents.
- 3. For insurance premiums for insurance against losses caused by environmental incidents, has the occurrence of the environmental incident been formally recognised as such by the competent authority of the Member State?

ANNEX I PART III.12 Document Generated: 2024-06-28

		es to legislation: There are curr	view as at 31/01/2020. ently no known outstanding effect 2004. (See end of Document for d			
		yes		no		
3.1.	If the answer is yes, has the Member State established in advance criteria on the basis of which the formal recognition referred to is deemed to be granted?					
		yes		no		
3.2.		been used in order to can the extent of the loss	alculate the annual agric	ultural production of the		
		yes		no		
4.	Is the aid limit companies?	ited to insurance provi	ded by a single insuranc	ce company or group of		
		yes		no		
Comn	nission cannot au	thorise aid for the pay	ordance with point (407) rment of insurance prenormy or group of compar	niums if it is limited to		
5.		le subject to the conditi ny established in the M	on that the insurance coember State?	ntract must be taken ou		
		yes		no		
Comn	nission cannot au		ordance with point (407 ment of insurance premark) for insurance services.			
6.	Does the aid o	cover a re-insurance pro	ogramme?			
		yes		no		

If the answer is yes, please provide all necessary information to enable the Commission to check possible aid components at the different levels involved (that is to say, at the level of the insurer and/or re-insurer) and the compatibility of the proposed aid with the internal market. In particular please submit sufficient information to enable the Commission to check that the final benefit of the aid is passed on to the farmer.

- 7. What are the eligible costs?
- \Box (a) the cost of insurance premiums against the losses caused by the events referred to in question 2.
- \Box (b) the costs associated with a reinsurance scheme. Please specify:
- 8. What is the maximum level of aid proposed? (as a percentage)

Please note that the gross aid intensity must not exceed 65 % of the cost of the insurance premium, with the exception of aid for the removal and destruction of fallen stock, where the

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

aid intensity must not exceed 100 % of the cost of the insurance premium as regards insurance

		val of fallen stock and iums for the destruction		e insurance premium as			
9.	Will the amount of the insurance premium eligible for support be limited by tha application of a ceiling?						
		yes		no			
If the ar	nswer is yes, wh	at will the ceiling be?					
10.	Are the insurance payments limited so that they compensate for no more than the cost of making good the damage caused by the events referred to in question 2?						
		yes		no			
11.	Do the insurar	nce payments require or	specify the type or quant	tity of future production?			
		yes		no			
2 and m OTHEI Please i concern 1.2.1.7. This for financia of the E rural ar	any not require of R INFORMAT andicate any other ed under this Season SUPPLEMENT CONTRIBUTE on must be used all contributions furopean Union treas 2014 to 202	or specify the type or quilon Her information consideration of the Guidelines WITARY INFORMATIONS TO MUTUAL From the notification of an anto mutual funds as described Guidelines for State and the Guidelines for State	antity of future product cred relevant to the ass. ON SHEET ON AI UNDS The state aid schemes deserbed by Section 1.2.1.7 d in the agricultural and	essment of the measure D FOR FINANCIAL signed to compensate for 7. of Chapter 1 of Part II d forestry sectors and in			
1.		anies and/or companies		mutual funds in favour ng and the marketing of			
		yes		no			
Commis	ssion can only a			5) of the Guidelines the al funds to undertakings			
2.			be covered by the mui	tual fund for which the ed aid measure:			
		Damage caused by adv to a natural disaster, ar	verse climatic events whimal diseases and plan	which can be assimilated t pests, as referred to in eart II of the Guidelines.			
		Damages caused by en					

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

				For financial contributions to mutual funds to pay compensation for the damages caused by environmental incidents, has the occurrence of the environmental incident been formally recognised as such by the competent authority of the Member State?				
		yes		no				
occurre	nnswer is no, pl nce of the environ ty of the Membe	onmental incident must b	dance with point (419) be formally recognised a	of the Guidelines, the s such by the competent				
3.1.			State established in adva- red to is deemed to be g	ance criteria on the basis ranted?				
		yes		no				
3.2.		been used in order to cand the extent of the loss?		ultural production of the				
		yes		no				
4 . □		The financial contribut		o pay compensation to				
		the amounts paid by t		tion 2, which relate to ancial compensation to sector.				
Please 1	note that there ar	re no other eligible costs	S.					
5.	What is the le	vel of aid proposed? (as	a percentage)					
Please 1	note that the max	ximum aid rate is 65 %	of the eligible costs.					
6.	Will the amou	int of the costs eligible f	for support be limited?					
		yes		no				
6.1.	If the answer	is yes, how will it be lin	nited?					
		Ceiling per fund: Appropriate per membe	er/affiliate to the fund ce	eilings:				
7.	Has the mutual fund been accredited by the competent authority in accordance with national law?							
		yes		no				

8. Does the mutual fund have a transparent policy towards payments into and withdrawals from the fund?

	1 FART 111.12 nt Generated: 2024-06-	28		
		es to legislation: There are cur	e view as at 31/01/2020. rently no known outstanding effe 4/2004. (See end of Document fo	
		yes		no
9.	Does the mu incurred?	tual fund have clear	rules attributing respo	nsibilities for any debts
		yes		no
Supple	ementary Informa n a position to ap Have rules fo particular for	ntion Sheet must be ans prove of the aid. If the constitution and a	swered affirmatively in a management of the mut asation payments, as wel	nestions 7, 8 and 9 of this order for the Commission and fund been defined, in 1 as for the administration
		yes		no
11.	Does the mutu the part of the		provide for penalties in	the case of negligence on
		yes		no
Supple to be in OTHE Please	ementary Informant a position to aper ER INFORMAT indicate any other	ntion Sheet must be ans prove of the aid. ION	swered affirmatively in a	nestions 10 and 11 of this order for the Commission sessment of the measure
1.2.2.		NTARY INFORMAT ON CAPACITY	ION SHEET FOR	AID FOR CLOSING
closing as desc	g of capacity for a cribed by Section	inimal, plant or human 1.2.2. of Chapter 1 of I	health, sanitary ethical Part II of the European U	designed to promote the or environmental reasons Inion Guidelines for State 2020 ('the Guidelines').
Does t	he planned meas	ure provide that,		
(a)	there must be	a counterpart from the	e beneficiary of the aid;	
(b)	enterprises in	difficulty are excluded	I from the measure;	
(c)	there must be	no over-compensation	of loss of capital value	of assets?
		yes		no

If the answer is no, please note that in accordance with Section 1.2.2. of Chapter 1 of Part II of the Guidelines no aid can be granted if those conditions are not fulfilled.

1.5.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.	CLOSING OF CAPACITY FOR ANIMAL, PLANT OR HUMAN HEALTH, SANITARY, ETHICAL OR ENVIRONMENTAL REASONS					
1.1.	What is the re	ason for the closing of	capacity:			
□(a) □(b) □(c) □(d) □(e) □(f)	animal health; plant health; human health; sanitary reasons; ethical reasons; environmental reasons.					
Please o	lescribe the reas	son(s) in full:				
1.2.	Is the measure	e an aid scheme or an ir	ndividual aid?			
□(a) □(b)		aid scheme; individual aid.				
1.2.1.		n aid scheme, is it according on the same condition	essible to all eligible unons?	ndertakings in the same		
		yes		no		
1.3.		be the aid scheme or the of the measure.	e individual measure in	cluding the reasons and		
1.4.	Counterpart fi	rom the beneficiary/ies	of the aid.			
1.4.1.	To what exten	at will the capacity of the	e undertaking(s) concer	ned be closed:		
□(a) □(b)		complete closure of capacital closure of capaci				
If partia	al closure of cap	acity, please justify:				
1.4.2.	closure of the beneficiary w	production capacity co ill not start the same	been obtained from the oncerned is definite and activity elsewhere and the land/facility concerned	irreversible and that the are those commitments		
		yes		no		
1.4.3.	Only undertakings that have actually been producing, and only production capacities that have actually been in constant use over the past five years before the closing of the capacity, are eligible for aid. Is this the case for the beneficiary/ies of this measure?					
		yes		no		

Are only undertakings fulfilling Union standards eligible for aid?

no

yes

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Please note that undertakings which do not fulfil Union standards and which would be obliged to stop production anyway, must be excluded.

- 1.6. Negative environmental effects
- 1.6.1. In order to avoid erosion and other negative effects on the environment, the owner(s) of open farmland taken out of production must undertake to fulfil one of the commitments listed in point in (a), (b) or (c). Which one does the beneficiary/ies of this measure undertake to fulfil?
- □(a) To afforest or turn into a nature area open farmland within a period of two years of closure in such a way as to ensure that negative effects on the environment are avoided.
- □(b) To maintain the land in good agricultural and environmental condition, in accordance with Chapter I of Title VI of Regulation (EU) No 1306/2013⁽²¹⁹⁾, and with the relevant implementing rules, with the view to re-use the farmland after 20 years of effective closure.
- □(c) To ensure that any closure of installations covered by Directive 2010/75/ EU of the European Parliament and of the Council of 24 November 2010⁽²²⁰⁾ are carried out in accordance with Articles 11 and 22 of that Directive which require that the necessary measures are taken to avoid any pollution risk and that the site of operation is returned to a satisfactory state.

Please describe how the beneficiary will fulfil the commitment:

- 1.7. Eligible costs.
- 1.7.1. What are the eligible costs?
- □(a) Loss of value of assets measured as the current selling value of the
- □(b) In case of the closing of capacity for environmental reason, an additional incentive payment, not exceeding 20 % of the value of the assets.
- \Box (c) Costs of destruction of the production capacity.
- \Box (d) Obligatory social costs resulting from the implementation of the closure decision.

Please note that no costs other than those referred to in points (a) to (d) are eligible under this measure.

Aid for afforestation and the conversion of land into nature areas must be granted in accordance with the rules set out in Sections 2.1.1 and 2.1.2. of Chapter 1 of Part II of the Guidelines and the rules on aid for non-productive investment set out in Section 1.1.1.1. of Chapter 1 of Part II of the Guidelines.

- 1.8. Aid intensity.
- 1.8.1. What aid intensities have been chosen?
- □(a) For loss of value of assets (maximum 120 % where the closure is carried out for environmental reasons, maximum 100 % for the other reasons listed in question 1.1 above).
- □(b) For compensation for the cost of the destruction of the production capacity (maximum 100 %).
- \Box (c) To offset the obligatory social costs resulting from the implementation of the closure decision (maximum 100 %).

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

۷.	CLOSING OF CATACITY FOR OTHER REASONS					
2.1.	What is the re	ason for the closing of o	capacity:			
□(a) □(b) □(c)		restructuring of a sector; diversification; early retirement.				
2.2.	Is the measure	e an aid scheme?				
		yes		no		
	note that measure part of an aid s		r the reasons referred t	o in question 2.1 above		
2.3.			be granted which we tion of the markets in ag	ould interfere with the gricultural products?		
		yes		no		
2.4. 2.5.		e sector(s) covered by t		to production limits or		
		yes		no		
If yes,	please describe i	t:				
		emes applying to sectors ase-by-case basis.	which are subject to pro	oduction limits or quotas		
2.6.		or hose sectors referred ther at regional or nation		nsidered to be in excess		
		yes		no		
If the a	answer is yes, ple	ase describe it:				
2.7.			has defined objectives t diversification or at ea	and a specific timetable rly retirement?		
		yes		no		
		1	1	1		

If the answer is yes, please describe the programme:

2.8. What is the duration of the planned aid scheme?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Please note that in accordance with point (442) of the Guidelines, the Commission can only authorise this type of aid when they provide for a limited duration. The duration of aid schemes aimed at reducing capacity for any of the reason referred to in question 2.1 of this Supplementary Information Sheet must normally be limited to a period of not more than six months for collecting applications for participation and a further 12 months for actually closing down.

				not more than six months for his for actually closing down.
If the du	ration is longer	r than prescribed abov	ve, please justify i	t.
years, si				with duration of more than three result in the postponement of the
2.9.	the same cond		arent system of c	ors in the sector(s) concerned on alls for interest which publically icipate used?
		yes		no
				(443) of the Guidelines, the aid tof this condition is not ensured.
2.10.		acilitates anticompetit		in such a way that it neither concerted practices between the
		yes		no
Please d	escribe how th	is is ensured:		
2.11.	Counterpart f	from the beneficiary of	f the aid.	
2.11.1.	To what exter	nt will the capacity of	the undertaking c	concerned be closed:
□(a) □(b)		complete closure of cap		
If partia	l closure of cap	acity, please justify it	:	
2.11.2.	that the closu that the benef	re of the production of the aid will	capacity concerned not start the sam	om the beneficiary/ies of the aid d is definite and irreversible and a activity elsewhere and are the land/facility concerned?
		yes		no
2.11.3.	that have actu	ually been in constant	use over the past	g, and only production capacities t five years before the closing of the beneficiary/ies of this measure?
	the capacity, a			

2.12. Are only undertakings fulfilling Union standards eligible for aid?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

yes	no

Please note that undertakings which do not fulfil Union standards and which would be obliged to stop production anyway, must be excluded.

- 2.13. Negative environmental effects.
- 2.13.1. In order to avoid erosion and other negative effects on the environment, the owner(s) of open farmland taken out of production must undertake to fulfil one of the commitments listed in point (a), (b) and (c). Which one does the beneficiary/ies of this aid undertake to fulfil?
- □(a) To afforest or turn into a nature area open farmland within a period of two years of closure in such a way as to ensure that negative effects on the environment are avoided.
- □(b) To maintain the land in good agricultural and environmental condition, in accordance with Chapter I of Title VI of Regulation (EU) No 1306/2013, and with the relevant implementing rules, with the view to re-use the farmland after 20 years of effective closure.
- □(c) To ensure that any closure of installations covered by Directive 2010/75/ EU of the European Parliament and of the Council is made in accordance with Articles 11 and 22 of that Directive which require that the necessary measures are taken to avoid any pollution risk and that the site of operation is returned to a satisfactory state.

Please describe how the beneficiary of the aid will fulfil the commitment:

- 2.14. Which are the eligible costs?
- □(a) Loss of value of assets, measured as the current selling value of the
- \Box (b) Costs of the destruction of the production capacity.
- \Box (c) Obligatory social costs resulting from the implementation of the closure decision.

Please note that costs other than those referred to in points (a), (b) and (c) are not eligible under this measure.

Aid for afforestation and the conversion of land into nature areas must be granted in accordance with the rules set out in Sections 2.1.1. and 2.1.2. of Chapter 1 of Part II of the Guidelines and the rules on non-productive investments in Section 1.1.1.1. of Chapter 1 of Part II of the Guidelines.

- 2.15. Aid intensity.
- 2.15.1. Which of the following aid intensities have been chosen?
- \Box (a) For the loss of value of assets (maximum 100 %).
- For compensation for the cost of the destruction of the production capacity (maximum 100 %).
- \Box (c) To offset the obligatory social costs resulting from the implementation of the closure decision (maximum 100 %).

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.3.1. SUPPLEMENTARY INFORMATION SHEET ON AID TO THE LIVESTOCK SECTOR

This form must be used for the notification of any State aid measures designed to support the livestock sector as described in Section 1.3.1. of Chapter 1 of Part II of the European Union Guidelines for State aid in the agriculture and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

1. **ELIGIBLE COSTS**

1	1	. Whic	h of the	follo	wing (eligible	costs	does 1	the aid	cover:

- □(a) the administrative costs of the establishment and maintenance of herd books?
- □(b) tests to determine the genetic quality or yield of livestock (namely, tests performed by or on behalf of third parties)?

Controls undertaken by the owner of the livestock and routine controls of milk quality are excluded from the aid.

2. **AMOUNT OF AID**

- 2.1. Please specify the maximum rate of public support expressed as a volume of eligible costs:
- □(a) to cover the administrative costs of the establishment and maintenance of herd books (maximum 100 %);
- \Box (b) for the costs of tests to determine the genetic quality or yield of livestock (maximum 70 %).
- 2.2. What measures have been taken to avoid overcompensation of the beneficiary and to verify compliance with the above aid intensities referred to in question 2.1?
- 2.3. Please describe the eligible costs to be covered by the aid:

Please note that the eligible costs are limited to those listed in point (449) of the Guidelines.

Please note that the aid should be provided in kind and should not involve direct payments to the beneficiaries in accordance with point (447) of the Guidelines.

3. **BENEFICIARIES**

3.1. Is the aid limited to firms which meet the Union definition of SMEs?

□ yes		no
-------	--	----

If the answer is no, please note that, in accordance with point (446) of the Guidelines, large companies must be excluded from receiving aid.

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

1.3.2. SUPPLEMENTARY INFORMATION SHEET ON AID FOR THE PROMOTION OF AGRICULTURAL PRODUCTS

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

This notification form must be used for State aid for the promotion of agricultural products as described in Section 1.3.2. of Chapter 1 of Part II of the European Union Guidelines for State aid in the agricultural and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

1.	Where will the measure be carried out?					
□(a) □(b) □(c)	on the market of another Member State; on the home market; on a third country.					
2.	Who will carry out the campaign?					
□(a) □(b)	producer groups or other organisations, regardless of their size; others (please explain):					
3.	Can the Mem Commission?	ber State submit sample	es or mock-ups of the pr	romotion material to the		
		yes		no		
If the a	nnswer is no, plea	ase explain why not.				
4.	Member State		t to provide this materia	lable at present, can the		
		yes		no		
5.	Please provide	e an exhaustive list of th	ne eligible costs.			
6.	Who are the b	eneficiaries of the aid?				
□(a) □(b) □(c) □(d)	farmers; producer groups and/or producer organisations; enterprises active in the processing and marketing of agricultural products; others (please specify):					
7.	Are the benefit limited to SM		sation of competitions, t	rade fairs or exhibitions		
		yes		no		
8.	Where the promotion measure is undertaken by producer groups and organisations will be the participation be subject to membership of such producer groups and organisations?					
		yes		no		

If the answer is yes, please note that in accordance with point (459) of the Guidelines no aid may be granted for such promotion campaigns.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

9.	Will the promotion campaign be earmarked for quality products covered by a qualit scheme as referred to in point (282) of the Guidelines?				
		yes		no	
10.		l be generic in characte		ssurance that the promotion of all producers of the type	
		yes		no	
11.	Will the promotion campaign comply with Regulation (EU) No 1169/2011 ⁽²²¹⁾ of the European Parliament and of the Council, and, where appropriate, with the specific labelling rules laid down for various products?				
		yes		no	
	nted for such pro	se note that in accordant motion campaigns.	• • • •	of the Guidelines no aid may fEUR 5 million?	
		yes		no	
		please note that in accust be notified individu		(458) of the Guidelines the	
13.	The aid must	be granted:			
□(a) □(b)		in kind; or on the basis of reimbu	rsement of real costs	incurred by the beneficiary.	
14.				promotion campaigns must eans of subsidised services?	
		yes		no	
		se note that in accordant motion campaigns.	nce with point (461)	of the Guidelines no aid may	
15.	Where the a beneficiaries?		, will the aid inclu	de direct payments to the	
		yes		no	
If the	ongryon ig vog =1		dance with naint (46	(2) of the Guidelines the aid	

If the answer is yes, please note that in accordance with point (462) of the Guidelines the aid can be only paid to the provider of the promotion measures.

16. Will the promotion campaign include promotion activities for the dissemination of scientific knowledge and factual information on quality systems, or on generic

23.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

		roducts and on the nutrit, or consumer-targeted ps?		
		yes		no
		lease note that in accor ar undertaking, brand na		
17.		consumer-targeted proposed will those campaigns be companies?		
		yes		no
	king is not perm If the answer t	lease note that in accorditted. o question 17 is yes, will nominations with refere	I the promotion campaig	gn be earmarked for EU-
	recognised de			
		yes		no
19.		to question 18 is yes, yactly to the references v		
		yes		no
20.		otion campaign be earma		another quality schemes
		yes		no
21.	Does the labe	I make any reference to	the national origin of th	e products concerned?
		yes		no
		e Member State must de lary in the message.	monstrate that the reference	ence to the origin of the
22.		on campaign of generic oduct concerned?	character and for the be	nefit of all producers of
		yes		no

If the answer to question 22 is yes, will the promotion campaign be carried out without reference to the origin of the products?

	:1	-28		
		es to legislation: Ther	int in time view as at 31/01) we are currently no known ou C) No 794/2004. (See end o _j	utstanding effects for the
		yes		no
	* A		accordance with Sector such promotion c	tion 1.3.2. of Chapter 1 of Part II o ampaigns.
24.		notion campaig or commercial b		ectly to the products of particula
		yes		no
	· · · ·		accordance with Sector such promotion c	etion 1.3.2. of Chapter 1 of Part II o ampaigns.
25.			of the organization of the the following:	of/and participation in competitions
		up to 100 % (in	dicate the exact rate	: %)
26.	The aid intens	sity in the case o	of promotion campai	gns will be the following:
			ality products becau	rate: %) for promotion campaign se the sector will finance the rest of
		up to 80 % (i		rate: %) for promotion campaign
		up to 100 %	(indicate the exact	rate: %) because the sector will sts, irrespective of the form of the

27. Does the campaign concern promotion measures referred to in Article 45 of Regulation (EU) No 1308/2013?

up to 100 % (indicate the exact rate: %) because the promotion campaign

is generic and in the benefit of all producers of the type of product

contribution;

concerned.

u yes u no

If the answer is yes, please note that in accordance with point (470) of the Guidelines, the Commission will consider national payments granted by Member States compatible with the internal market if they comply with the common assessment principles of the Guidelines and with the rules on aid for promotion measures set out in Section 1.3.2. of Chapter 1 of Part II of the Guidelines.

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

1.3.3. SUPPLEMENTARY INFORMATION SHEET ON AID FOR THE OUTERMOST REGIONS AND THE SMALLER AEGEAN ISLANDS

This form must be used by Member State to notify State aids for the outermost regions and the smaller Aegean islands, as referred to in Section 1.3.3.of Chapter 1 of Part II of the European Union Guidelines for State aid in the agriculture and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

 \Box (a)

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.		ed aid for the outermosions set out in the Gu		aller Aegean Islands related					
		yes		no					
	answer is yes, plof aid notified.	ease complete the Sup	pplementary Information	on Sheet corresponding the					
If the	answer is no, plea	ase complete this Sup	plementary Informatio	n Sheet.					
2.	Does the mea	sure involve the grant	ting of operating aid?						
		yes		no					
3.		In the case of the outermost regions, is the aid intended to mitigate their specific constraints on farming as a result of their isolation, insularity and extreme remoteness?							
		yes		no					
3.1.3.2.4.	If the answer to question 3 is yes, please determine the amount of the additional costs resulting from such specific constraints and the method of calculation: How can the Member State establish the link between the additional costs referred to in question 3.1 and the specific constraints giving rise to them? In the case of the smaller Aegean islands, is the aid intended to mitigate the specific constraints on farming in those islands resulting from their insularity, small size, mountainous terrain and climate, their economic dependency on a small number of products and their distance from the markets?								
		yes		no					
4.1.		If the answer to question 4 is yes, please determine the amount of the additional costs resulting from such specific constraints and the method of calculation:							
4.2.			sh the link between the astraints giving rise to t	additional costs referred to hem?					
5.				osts of agricultural products the smaller Aegean islands?					
		yes		no					
5.1.	If the answer (a) to (d)?	to question 5 is yes,	will this aid fulfil the	conditions set out in points					

the beneficiaries have their production activity in the outermost regions or in the smaller Aegean islands;

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□(b)	the aid is objectively quantifiable in advance on the basis of a fixed sum
□(c)	or per tonne/kilometre ratio or any other relevant unit; the additional transport costs are calculated on the basis of the journey of the products inside the national border of the Member State concerned using the means of transport which results in the lowest costs for the
□(d)	beneficiary, taking into account the external costs to the environment; for outermost regions, the eligible additional transport costs may include the costs of transporting agricultural products from the place of their production to locations in outermost areas where they are further processed.
5.2.	If the aid is intended to offset in part the additional transport costs of agricultural products, please provide proof of the existence of those additional costs and the method of calculation used to determine the amount of the additional transport costs ⁽²²²⁾ :
5.3. OTHER	Indicate also what will be the maximum amount of aid (on the basis of an aid-per-kilometre ratio or on the basis of an aid-per-kilometre and aid-per-unit-weight ratio) and the percentage of the additional costs covered by the aid: **R INFORMATION**
	ndicate any other information considered relevant to the assessment of the measure ed under this Section of the Guidelines.
1.3.4.	SUPPLEMENTARY INFORMATION SHEET ON AID FOR AGRICULTURAL LAND CONSOLIDATION
agriculta Europea	m must be used for the notification of any State aid measure to cover the costs of ural land consolidation as described in Section 1.3.4. of Chapter 1 of Part II of the in Union Guidelines for State aid in the agriculture and forestry sectors and in rural 014 to 2020 ('the Guidelines').

1. Is the aid measure part of a general programme of agricultural land consolidation operations undertaken in accordance with the procedures laid down by the legislation of the Member State?

yes	no

2. Do the eligible costs include exclusively the legal, administrative and survey costs of land consolidation?

□ yes □ no

Please note that the eligible costs are limited to those listed in point (480) of the Guidelines.

3. What is the planned maximum aid intensity (maximum 100 %)?:

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

1.3.6. SUPPLEMENTARY INFORMATION SHEET ON AID FOR RESEARCH AND DEVELOPMENT IN THE AGRICULTURAL SECTOR

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

This form must be used for the notification of any State aid scheme for research and development in the agricultural sector as described in Section 1.3.6. of Chapter 1 of Part II of the European Union Guidelines for State aid in the agricultural and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

1.	Does the aid	concern products	listed in Annex I to	the Treaty?		
		yes		no		
If the	answer is yes, s	pecify the type of	agricultural products	:		
2.		ect of interest to all r concerned?	l undertakings active	in the particular agricultural	sector	
		yes		no		
If the	answer is yes, p	rovide evidence:				
3.		lowing information		the internet prior to the d	ate of	
(a)	that the aide	d project is carried	l out;			
		yes		no		
(b)	the goals of the aided project;					
		yes		no		
(c)	an approximate date of the publication of the results expected from the aided project;					
		yes		no		
(d)	the place of publication on the internet of the results expected from the aided project;					
		yes		no		
(e)	a reference that the results are available to all undertakings active in the particular agricultural sector or sub-sector concerned at no cost?					
		yes		no		
	answer to point ternet address:	(a), (b), (c), (d) or	(e) is yes, please pr	ovide evidence and give det	ails of	
4.	Are the resu	lts of the aided pro	oject to be:			
(a)	on which ar		ncerning those resul	of the aided project or the ts is be given to members		
		yes		no		

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

(b)		made available on the internet, for a period of at least five years from the end date of the aided project?							
			yes				no		
If the	answer to po	int (a) or (b) is yes, p	lease provid	le evidend	ce:			
5.	Is the aid	Is the aid granted directly to the research and knowledge-dissemination organisation?							
		yes				n	0		
If the	answer is yes	s, please pro	vide evid	ence:					
6.		measure in to undertak					ice of the agricultural		
		yes				n	o		
If the	answer is no,	, please pro	vide evide	ence:					
7.	Specify the	he aid inten	sity (%):						
8.	Do the el	igible costs	include th	ne following	j.				
(a)		personnel costs related to researchers, technicians and other supporting staff to the extent employed on the project;							
			yes				no		
	If the ans	wer is yes,	please pro	vide a desc	ription of	such costs:			
(b)	project.(V	Where such ct, only the d on the bas	instrume deprecia	nts and equation costs of	ipment a orrespond	re not used ding to the	ne period used for the for their full life for life of the project, as ples are considered as		
			yes				no		
	If the ans	wer is yes,	please pro	vide a desc	ription of	such costs:			
(c)	project. (\) of the proconsidere	With regard ject, as calc	to buildin ulated on t e. For land	gs, only the he basis of g	depreciat generally	tion costs co	on period used for the rresponding to the life counting principles are er or actually incurred		
			yes				no		
	TC.1					1 .			

If the answer is yes, please provide a description of such costs:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	outside sou	rces at arm's leng		patents bought or licensed from the licensed from the costs of consultancy and the costs of consultancy and the costs of consultance and the costs of costs of consultance and the costs of costs o	
		yes		no	
	If the answ	er is yes, please pro	ovide a description of	f such costs:	
(e)				including the costs of material result of the project.	ıls,
		yes		no	
OTH 9.	ER ELEMEN'	er is yes, please pro TS TO BE CONS be combined with		f such costs:	
	Can the ard			no	
		yes		no	—
will b	e verified: al conditions fo	r repayable advan	се	plicable to the notified aid schen	iic,
		yes		no	
		er to question 10 is		in the form of a repayable advar	nce
		er to question 10 is	yes, is the aid granted	in the form of a repayable advar	nce
In addata o	dition, please pon which that mal conditions fo	yes rovide the comple ethodology has been rax advantage me	yes, is the aid granted expressed as gross granted te methodology applen based:	in the form of a repayable advant equivalent? no ied and the underlining verifial	ble
11.	dition, please pon which that mal conditions for	yes rovide the comple ethodology has been rax advantage me	yes, is the aid granted expressed as gross granted te methodology applen based:	in the form of a repayable advantant equivalent?	ble

13. If the answer to question 12 is yes, please specify how the aid intensities are calculated:

Please provide details of the calculation method applied:

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2. SUPPLEMENTARY INFORMATION SHEET ON AIDS FOR THE FORESTRY **SECTOR**

This form must be used for the notification of any State aid measure (223) for the forestry sector covered by Chapter 2 of Part II of the European Union Guidelines for State aid in the agriculture and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

In addition to this form, please fill in the General Information Sheet for the notification of State aid in the agriculture and forestry sectors and in rural areas (Part III 12) demonstrating the general eligibility conditions for State aid and also the corresponding forestry forms 2.1 to 2.9 depending on the specific type of aid.

Please submit information concerning the legal basis provided for in national law, or the draft act providing the legal basis in national law, together with any additional documents, such as

				ls the State aid measure.			
specific i	to trade and ind		olicable notification fort	mmon to all sectors or n for those sectors when			
1.	GENERAL ELIGIBILITY CRITERIA						
1.1.	Does the aid meet the objectives and fulfil all the conditions, including the conditions concerning the beneficiaries of aid, laid down in Regulation (EU) No 1305/2013 ⁽²²⁴⁾ , and any implementing and delegated acts adopted pursuant to that Regulation?						
		yes		no			
compatil laid dow	ble with Article on in Regulation 2 of Part II of t	e 107(3)(c) of the Treat n (EU) No 1305/2013 (he Guidelines).	ty only if the aid comp	id for the forestry sector lies with the conditions Sections 2.8 and 2.9 of wable energies?			
		yes		no			
1.2.1.	If the answer is yes, does the aid concern investments in energy saving and renewable energies related to the use of wood as a raw material or energy source, which are limited to all working operations prior to industrial processing ⁽²²⁵⁾ ?						
		yes		no			
			l.				

Please note that in accordance with point (495) of the Guidelines any aid other than aid for these investments for energy savings and in renewables is excluded from the scope of Chapter 2 of Part II of the Guidelines as such aid should comply with the Guidelines on State aid for environmental protection and energy 2014-2020⁽²²⁶⁾, unless it is exempt from notification obligation.

1.3. Please confirm that the aid is not intended for forest based industries

	yes		no
--	-----	--	----

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

development and orests.
Natura 2000 forest
climate services and
nformation actions in
orestry sector
ector
and organisations in
tor with ecological, tives
with the agricultural
sure?
)

If the answer is yes, please note the only beneficiaries that can be eligible for aid are those who are listed in Regulation (EU) No 1305/2013 concerning the respective rural development measure. Please describe the eligible beneficiaries:

3.2. For aid measures which are not co-financed from EAFRD, but financed exclusively from national resources, please describe the eligible beneficiaries:

4.1.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

3.3.	In the case of II of the Guid	aid measures covered lelines, please confirm	by Section 2.1.5 or if SMEs can be th	Section 2.7 of Chapter e only beneficiaries of	2 of Part the aid:	
		yes		no		
		by Section 2.1.5 of Chorest holders, municipa		of the Guidelines, aid a ociations.	may also	
2.1.		NTS IN FOREST ARE LITY OF FORESTS	A DEVELOPME	NT AND IMPROVEM	ENT OF	
1.1.	Is aid conditional on the submission of a forest management plan or equivalent instrument as required in accordance with Regulation (EU) No 1305/2013 in the case of EAFRD co-financed aid forming part of a rural development programme?					
		yes		no		
1.2.	If the answer is yes, please provide detailed information concerning that requirement, such as, when it is required, the size of forest holding, a description on rural development programme:					
2.	Please indicat	te whether the eligible		_	_	
□(a)		immovable property v not exceeding 10 % of in exceptional and do	with land purchased the total eligible c aly justified cases,	g leasing, or improved only being eligible to a osts of the operation co the limit may be raised g environmental conse	an extent incerned; ed above	
□(b)			purchase of mach	inery and equipment		
□(c)		the general costs lin and (b), such as archi- to advice on environ- feasibility studies. Th	ked to the expend feet, engineer and mental and econon e latter remain elig	liture referred to in p consultation fees, fees nic sustainability, inclu- tible even where, based (a) and (b) is incurred	relating iding the lon their	
□(d)			velopment of comp	outer software and acq		
□(e)				ment plans and their ed	ղuivalent	
□(f)		other costs related to interventions), that are (EU) No 1305/2013.	e consistent with the Please describe the	stry measure (such as ne rules laid down in Re ose other costs and jus f the given forestry me	egulation stify how	
3.	Please confirm	m that aid will not be a	granted for:			
□(a) □(b)		working capital; other costs connecte interest refinancing co		racts, such as lessor's d insurance charges.	margin,	
4.	Please indicate	te whether the aid con-	cerns:			

☐ Aid for afforestation and creation of woodland.(Section 2.1.1 of the Guidelines)

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

4.1.1.	Please indicat	e whether the eligible c	osts concern:				
□(a)	the establishment of forest and woodland on						
□(b)	agricultural or non-agricultural land an annual premium per hectare to cover the costs of agricultural income foregone and maintenance, including early and late cleanings, for a maximum period of 12 years.						
Please j	provide more in	formation as to amounts	s of aid and the calculati	on methods:			
4.1.2.			part of a rural developn rivate land-holders and	nent programme, are the their associations?			
		yes		no			
If the a	nswer is yes, aid	I for establishment costs	s and the annual premiu	n can be granted.			
4.1.3.	Can the Mem case of:	ber State confirm that the	ne aid only covers the es	stablishment costs in the			
□(a)		afforestation of land ov	vned by public authorities	es,			
□(b)		or fast growing trees.					
4.1.4.			the case of state-owned rivate body or municipa	land, aid can be granted lity?			
		yes		no			
4.1.5.	1305/2013, ca		nfirm that the aid is not	n Regulation (EU) No an EAFRD co-financed			
		yes		no			
4.1.6.			anted for the planting o	f trees for short rotation roduction:			
		yes		no			
4.1.7.			ed are adapted to the envith minimum environme	rironmental and climatic ntal requirements.			
		yes		no			

4.1.8. Please confirm, and demonstrate with additional description and information, that the aid complies with the following minimum environmental requirements:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

 \Box (a)

□(b)

□(c)

 \Box (d)

the selection of species to be planted, of areas and of methods to be used must avoid the inappropriate afforestation of sensitive habitats such as peat lands and wetlands and negative effects on areas of high ecological value including areas under high natural value farming. On sites designated as Natura 2000 pursuant the Habitats Directive and the Birds Directive only afforestation consistent with the management objectives of the sites concerned and agreed with the Member State's authority in charge of implementing Natura 2000 must be allowed;

the selection of species, varieties, ecotypes and provenances of trees must take account of the need for resilience to climate change and to natural disasters and the pedologic and hydrologic condition of the area concerned, as well as of the potential invasive character of the species under local conditions. The beneficiary must be required to protect and care for the forest at least during the period for which the premium for agricultural income foregone and maintenance is paid. This must include tending, thinning or grazing, as appropriate, in the interest of the future development of the forest and regulating competition with herbaceous vegetation and avoiding the building up of fire prone undergrowth material. As regards fast-growing species, the Member State must define the minimum and maximum time before felling. The minimum time must not be less than eight years and the maximum must not exceed 20 years;

in cases where, due to difficult environmental or climatic conditions, including environmental degradation, the planting of perennial woody species cannot be expected to lead to the establishment of forest cover as defined under the applicable national legislation, the Member State may allow the beneficiary to establish other woody vegetation cover. The beneficiary must provide the same level of care and protection as applicable to forests;

in the case of afforestation operations leading to the creation of forests of a size exceeding a certain threshold, to be defined by Member States, the operation must consist of either:

- (i) the exclusive planting of ecologically adapted species and/or species resilient to climate change in the bio-geographical area concerned, which have not been found, through an assessment of impacts, to threaten biodiversity and ecosystem services, or to have a negative impact on human health; or
- (ii) a mix of tree species which includes either at least 10 % of broadleaved trees by area, or a minimum of three tree species or varieties, with the least abundant making up at least 10 % of the area.
- 4.1.9. Please confirm that, in areas where afforestation is made difficult by severe pedoclimatic conditions, aid may be provided for planting other perennial woody species such as shrubs or bushes suitable to the local conditions.
- 4.1.10. Is aid limited to 100 % of the eligible costs?

□ ye	/es		no
------	-----	--	----

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

4.2.	□ Aid for to Guidelines)	the establis	hment of	agro-forest	ry systems.	(Secti	ion 2.1.2. of the
4.2.1.		vn in combir	nation with				systems in which as defined in point
		yes				no	
Please	describe the aid	measure:					
4.2.2.		pment progra	amme, aid				forming part of a vate land-holders,
		yes				no	
4.2.3.	If there are el confirm that	the aid meas		ced exclusiv			estion 4.2.2, please resources:
		yes				110	
4.2.4.	Please indica	te whether th	ne eligible c	osts concerr	n the followi	ng:	
□(a)		the establish	nment of an	agro-forest	ry system.		
					ed to 80 % o forestry sys		ligible investment
			yes				no
□(b)		an annual p	remium per	hectare for	maintenanc	e.	
		If the answe	er is yes, is t	the aid limit	ed to 100 %	of the	annual premium?
			yes				no
4.2.5.	Please define of five years)		of the max	ximum perio	od (with a m	naxim	um period of time
4.2.6.	and demonst	rate that it t	akes into a s, forestry	account the	local pedo-	climat	tare to be planted, ic conditions and nsure sustainable
4.3.	fire, natural	l disasters,	adverse cl	imatic eve	nts which	can b	rests from forest be assimilated to and catastrophic

events.(Section 2.1.3. of the Guidelines)

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

4.3.1.	programme, p	of EAFRD co-fi please confirm that and to other priva	at the a	aid may on	ly be grante	d to p	private and public
		yes				no	
4.3.2.		other beneficiario				on 4.3	.1, please confirm
		yes				no	
4.3.3. □(a)		e whether the elig the establishment it may also cover Please confirm t activities in areas	of promainte	enance costs aid will l	astructure (is); be granted	for ag	gricultural related
			yes				no
□(b)		local, small scale hazards, including				st fire	or other natural
□(c)		establishing and	improv	ving forest	fire, pest a	nd dis	seases monitoring
□(d)		facilities and com- restoring forest p climatic events v adverse climatic change related ev	otentia vhich o events	al damaged can be assi	by fires, namilated to a	natu	ral disaster, other
4.3.4.	Please confirm that there has been a formal recognition by the competent public authorities that an event referred to in point (d) of question 4.3.3 has occurred and that either this event, or measures adopted in accordance with Directive 2000/29/EC to combat, eradicate or contain harmful organisms, has caused the destruction of at least 20 % of the relevant forest potential.						
		yes				no	
4.3.5.	provide the sorganisation of	aid for the prevence that the risk of the state of the risk of the state of the sta	e and occurr	the acknown the acknown the	wledgement plant pest.	by a Wher	scientific public e relevant, please
4.3.6.	established by	n that the eligible of the Member Sta led in the forest pr	te and	especially			
		yes				no	

4.4.2.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

4.3.7.		ncerned classified as a son plan established by t		ire risk according to the		
		yes		no		
If the ar	nswer is yes, it s	hall be eligible for aid f	or prevention of fire.			
4.3.8.	natural disast	ers, an adverse climatic adverse climatic even	c event which can be	ome resulting from fires, assimilated to a natural phic events and climate		
		yes		no		
4.3.9.	Is the aid targe	eted at restoration of da	mage due to plant pests	?		
		yes		no		
	Community a	action to achieve the start (C) No 1107/2009 conc	ustainable use of pestion	lishing a framework for cides and Article 55 of lant protection products		
4.3.11.	Is the aid for costs?	the costs referred to in	question 4.3.3 limited	to 100 % of the eligible		
		yes		no		
4.3.12.	particular to e received by	ensure that aid granted the beneficiary, includ	for the eligible costs a	e overcompensation, in and any other payments ther national or Union limited to 100 %.		
4.4.		estments improving th Section 2.1.4. of the Gu		nmental value of forest		
4.4.1.	confirm that the	For EAFRD co-financed aid forming part of a rural development programme, please confirm that the aid may only be granted to natural persons, private and public forest holders and other private and public bodies and their associations.				
		yes		no		
						

In the case of beneficiaries other than those referred to in question 4.4.1, please confirm

that the measure is financed exclusively from national resources:

Document	t Generated: 2024-06-			
		es to legislation: There are cur	e view as at 31/01/2020. rently no known outstanding effec 4/2004. (See end of Document for	
		yes		no
4.4.3.	Please indicat	e whether the eligible	costs concern the follow	ing:
□(a)		the achievement of co		for environmental aims
□(b)			ne public amenity value	of forests and woodland
□(c)				mitigation potential of
If there	is any economic	c benefit in the long te	rm, please describe it:	
4.4.4.	Is the aid limi	ted to 100 % of the eli	gible costs?	
		yes		no
4.5. 4.5.1.	and in marker Please confirmation such standard	eting of forest produce on compliance with the	ets.(Section 2.1.5. of the minimum standards for l, for supported investments.)	cocessing, in mobilising Guidelines) energy efficiency, where ents in renewable energy
		yes	0	no
Please	describe any suc	h minimum standards	and their application to t	the measure:
4.5.2.		om biomass, please co		se of which is electricity ercentage of heat energy
		yes		no
		ach requirements conc plication to the measur		rcentage of heat energy
4.5.3.				applicable sustainability 17(2) to (6) of Directive
		yes		no
		I.	<u> </u>	1

Please describe any such requirements and their application to the measure:

4.5.4. Please confirm that aid may be granted only to private forest holders, municipalities and their associations or SMEs or also to undertakings that are not SMEs in the territories of Azores, Madeira, the Canary islands, the smaller Aegean islands within the meaning of Regulation (EU) No 229/2013⁽²²⁷⁾ and the French overseas departments.

						Doci	iment Generated: 2024-06-28
		iges to legisle	Status: Point in time ution: There are curr ulation (EC) No 794	ently no	known outstandin		
	Con	nmission Keg	utation (EC) No 794	/2004. (see ena oj Docume	eni jor dell	uis)
		yes				n	0
4.5.5.	Please descr	ribe the el	igible beneficia	ries:			
4.5.6.	Please indic	ate wheth	er the eligible of	costs	concern the fo	llowing	; :
□(a)	investments enhancing forestry potential:						
		□(i)			ts for soil and and and practices		ce friendly harvesting
□(b)		□(ii) process			stments. narketing addi	ing valu	e to forest products.
4.5.7.	Please descr	ribe the m	easure in more	detai	ls:		
4.5.8.	If investments are related to the improvement of the economic value of forests, please provide justification in relation to the expected improvements to forests on one or more holdings and please specify whether investments for soil-friendly and resource-friendly harvesting machinery and practices are included:						
4.5.9.	In case investments are related to the use of wood as a raw material or energy source, are those investments limited to all working operations prior to industrial processing?						
		yes				n	o
4.5.10.	Dlagga india	ata xybath	er aid intensitie	og oro	the following		
(a)		nited to 7	5 % of the amo		C		the outermost regions
			yes				no
(b)	and in all re December 2	egions wh 2013 was	ose GDP per o less than 75 %	apita of tl	for the period ne average of	d from the EU	ess developed regions 1 January 2007 to 31 J-25 for the reference erage of the EU-27?
			yes				no
(c)	Is the aid lin	nited to 4	0 % of the amo	unt o	f the eligible of	costs in	other regions?
			yes				no
4.6.			ments in intage				the development, e Guidelines)
4.6.1.	Please indic	ate wheth	er the eligible of	costs	concern the fo	llowing	; :
(a)	investment i	n:					
	□(□(tangib intang		sets and/or assets.		

(b) infrastructure related to the following:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	□(i) □(ii) □(iii)		the modern	ment of forests isation of forest on of forests.			
the inv	estments inclu	de:					
	□(i) □(ii) □(iii)			orest land; idation and imp and saving of ea			
Please	describe the m	easure in	n more detail	ls:			
Please	indicate if aid	intensitie	es are the fol	lowing:			
Is the	aid limited to 1	00 % of	the eligible	costs in the case	e of:		
	□(i) □(ii) □(iii)		investments environment investments the public	ntal value of for s for forest re	usively ests; oads, v ge and	at improving which are ope d which serve	en to
		yes			1	no	
forests		ited to 75	5% of the ar		gible co	conomic potent osts in the outer	
forests region 31 Dec	s, is the aid liming and in all regions and in all regions when the second regions in the second regions.	ted to 50 ons who as less th	% of the anse GDP per chan 75 % of	nount of the eligoapita for the pe the average of	gible co riod fro the EU-	conomic potent sts in less devel om 1 January 20 -25 for the refer rage of the EU-	loped 07 to rence
		yes			1	no	
						conomic potent osts in other reg	
		yes			1	no	
				forest land are in the area con		rted, please inco	

OTHER INFORMATION

investment (in metre/hectare)

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

2.2. AID FOR DISADVANTAGES RELATED TO NATURA 2000 FOREST AREAS

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.1.			aral development measurest holders and/or associ			
		yes		no		
1.2.			er than those referred to exclusively from nation			
		yes		no		
2.	Is the aid gran	nted annually and per he	ectare of forest?			
		yes		no		
3.	Please indicat	e the areas concerned by	y the following:			
□(a)			eas designated in accord	dance with the Habitats		
□(b)	Directive and the Birds Directive; other delimited nature protection areas with environmental restrictions applicable to forests which contribute to the implementation of Article 10 of the Habitats Directive; provided that, where the measure is co-financed by EAFRD as a rural development measure, per rural development programme, those areas do not exceed 5 % of the designated Natura 2000 areas covered by its territorial scope; for aid measures financed exclusively from national funds, this latter territorial restriction does not apply.					
4.	Please indicat	e the eligible costs:				
□(a)		additional costs incurr Directive and the Birds	ed due to the impleme	entation of the Habitats		
□(b)			ulting from the disad	vantages in the areas		

Please describe the calculation methodology:

5. Please indicate the amount per hectare per year:

Please note that aid under this measure is limited to EUR 500 per hectare per year maximum in the initial period not exceeding five years and EUR 200 per hectare per year maximum thereafter. If the maximum amounts are higher, please justify those higher aid amounts, taking into account and describing the specific circumstances as described in the rural development programme or otherwise (where the measure is financed exclusively from national resources):

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

2.3. AID FOR FOREST-ENVIRONMENT AND CLIMATE SERVICES AND FOREST CONSERVATION

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.	COMM	ON PRO	VISIONS

1.	COMMON I	PROVISIONS					
1.1.	1. In the case of EAFRD co-financed aid forming part of a rural development prograplease confirm that the aid may only be granted to public and private forest hand to other private law and public bodies and their associations and that in the of state-owned forests, aid may only be granted if the body managing such a formula private body or a municipality:						
		yes		no			
1.1.1.			er than those referred to usively from national res				
		yes		no			
1.1.2.		aid for the conservation gible beneficiaries:	n and promotion of forest	genetic resources please			
		(a) public entities		(b) private entities			
Please _j	provide more in	formation on the eligib	le beneficiaries:				
1.2.	they go beyo		nitment(s) to be undertak latory requirements esta il legislation.				
Please i	indicate:						
□(a)			ory requirements as				
□(b)		development programme, in the case of an RDP measure: in the case of aid measures financed exclusively from national funds, the relevant mandatory requirements and describe them in more details or in attached documentation:					
(c)	Please state th	e duration period of the	e commitments undertake	en (between 5 to 7 years):			
(d)	If the duration of the commitment period is longer, please justify why it is considered necessary in the case of the particular type of commitment.						
1.3.	Please indicat	e whether the eligible	costs concern the payme	ents for the following:			
□(a)		compensating benefic from the voluntary con	iaries for additional costs mmitments;	s (all or in part) resulting			
□(b)		Please state the amour compensating benefic voluntary commitmen	ciaries for income foreg	gone resulting from the			
		Please state the amour	nt:				

		Status: Point ges to legislation: There as mission Regulation (EC) N		n outstanding effects				
□(c)			ment voluntary		of the premium paid for Please state the reasons			
□(d)		for operations concerning environmental conservation, in these cases, aid may be granted as a flat-rate or one-off payment per unit for voluntary commitments to renounce the commercial use of trees and forests, calculated on the basis of additional costs incurred and income foregone.						
Please	describe the cal	culation methodolo	ogy:					
1.4.	Is aid granted	d per hectare of fore	est?					
		yes			no			
aid for the sub	forest holdings	above a certain thr	eshold (defined	l by the Membe	nent programme, is the er State) conditional on in line with sustainable			
		yes			no			
in line	with sustainabition of Forests in Is the aid lin	le forest management Europe of 1993 ⁽²²⁾	ent as defined 1 8) . um amount of I	by the Minister	r equivalent instrument rial Conference on the ectare per year (except			
		yes			no			
higher	amount, taking mmes (in the c	into account speci	ific circumstand	ces justified in	de justification for that the rural development e) or otherwise in this			
1.6.	Is the aid gra	nted for operations	for the conserv	ation of geneti	c resources?			
		yes			no			
Please	confirm whether	er the operations inc	clude the follow	ving:				
□(a) □(b)		characterisation, forestry, including conserved in situ, situ collections an concerted actions the conservation,	collection and g web-based invincluding on-f ad databases; actions promo- characterisation	utilisation of rentories of gen- corest holding co- oting the excha- n, collection an	nd ex situ conservation, genetic resources in etic resources currently conservation, and of example of information for dutilisation of genetic organisations in the			

Status: Point in time view as at 31/01/2020.
Changes to legislation: There are currently no known outstanding effects for the
Commission Regulation (FC) No 794/2004 (See end of Document for details)

□(c)	accompanying	actions:	information	, disseminati	ion an	d advis	sory
	actions involving	ng non-go	vernmental o	organisations	and otl	her rele	vant
	stakeholders, tra	aining cou	rses and the	preparation of	f techni	cal repo	rts.

□(c)		accompanying actions: information, dissemination and advisory actions involving non-governmental organisations and other relevan stakeholders, training courses and the preparation of technical reports.				
		e detail the operations referred to in points (a)		ation and promotion of		
1.7.	Is the aid limited to 100 % of the eligible costs?					
		yes		no		
2.	REVISION O	CLAUSE				
2.1.			operations included in th	is aid?		
		yes		no		
2.2.		ed to in that Section have extend beyond the rural yes		ning period 2014-2020?		
clause followi	in order to allow	w for the adjustment og period must be include	f the operations to the	ne Guidelines, a revision legal framework of the		
		ner information consideration of the Guidelines		essment of the measure		
2.4.			ON SHEET ON AID ACTIONS IN THE FO	FOR KNOWLEDGE RESTRY SECTOR		
1.	COMMON PROVISIONS					
1.1.	Please specify	who are the beneficiar	ies of the aid:			
1.2.	Is the aid ava defined condi		ble in the area concerns	ed based on objectively		
		yes		no		

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

Where the provision of knowledge transfer and information actions is undertaken by producer groups or other organisations is membership of such producer groups or 1.3. organisations a condition for access to the service?

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

		yes		no	
If the a		ease note that the aid o	annot be declared com	patible with the internal	
1.4.		ation of non-members t nisation concerned limi		ive costs of the producer iding the service?	
		yes		no	
If the a		ease note that the aid c	annot be declared com	patible with the internal	
2.	AID FOR K	NOWLEDGE TRANS	FER AND INFORMA	ATION ACTIONS	
2.1.	Which of the measure?	following types of aid	can be financed by the	aid scheme or individual	
□(a)		vocational training and workshops and coachir	<u> </u>	cluding training courses,	
□(b)		demonstration activitie			
\Box (c)		information actions;	forest management ave	phonon and forest visits	
□(d)		aid to cover short-term forest management exchange and forest visits.			
2.2.	Which of the	the following eligible costs does the aid measure cover?			
□(a)		demonstration activitie	s or information actions	*	
□(b)		the costs of travel, ac participants;	ecommodation and per	r diem expenses of the	
□(c)			on of replacement servi	ces during the absence of	
\Box (d)		the costs related to dem	nonstration projects.		
2.3.	In the case of following?	f demonstration project	s, do eligible investme	ent expenses include the	
□(a)			with land only being e	ing, or improvement of eligible to an extent not peration concerned:	
□(b)			ourchase of machinery	and equipment up to the	
□(c)		and (b), such as archite	ect, engineer and consu	referred to in points (a) ltation fees, fees relating sustainability, including	
\Box (d)			lopment of computer s	oftware and acquisitions	

Please note that the eligible costs are limited to those listed in points (293) and (565) of the Guidelines.

of patents, licenses, copyrights and trademarks.

2.4. Do the feasibility studies remain eligible expenditure even where, based on their results, no expenditure under point (293)(d)(i) and (ii) of the Guidelines is incurred?

Documen	nt Generated: 2024-06	28				
		Status: Point in time vers to legislation: There are currentission Regulation (EC) No 794/2				
		yes		no		
2.5.	Is the duration forest visits de	n and the content of the efined?	e short-term forest mana	agement exchanges and		
		yes		no		
Please	provide the detail	ils either:				
		from the rural developm or hereby, in this notific				
2.6.	What is the fo	cus of those schemes ar	nd exchanges?			
□(a) □(b) □(c) □(d) □(e)		sustainable forestry methods and/or technologies the development of new business opportunities the development of new technologies the improvement of forest resilience other, please specify:				
2.7.		es providing knowledge spacities in the form of second				
		yes		no		
2.8.	The aid will b	e granted:				
□(a) □(b)		in kind by means of sub in the form of direct pa form of reimbursement diem expenses of the pa	ayments of money to be of the costs of travel, a			
		referred to in point (293) payments to the benefic		to (iv) of the Guidelines		
2.9.		ient of the aid referred the nes be the provider of the				
		1	T	T		

yes	no

Please state the maximum aid intensity (maximum 100 %)

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

SUPPLEMENTARY INFORMATION SHEET ON AID FOR ADVISORY 2.5. SERVICES IN THE FORESTRY SECTOR

1. **COMMON PROVISIONS**

1.1. Who are the beneficiaries of the aid, please specify:

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.2.	defined condi		ose eligible in the area	
		yes		no
If the a		ease note that the	ne aid cannot be decl	ared compatible with the internal
1.3.		, is membership		aken by producer groups or other ganisations a condition for access
		yes		no
If the a market 1.4.	Is the contribution	ution of non-me		ared compatible with the internal ministrative costs of the group or ding the service?
		yes		no
If the a market 2.		ease note that the object of the control of the con		ared compatible with the internal
2.1.	Which of the measure ⁽²²⁹⁾ ?	following type	s of aid can be finance	ced by the aid scheme/individual
□(a) □(b)		use of advisory environmental	y services for the im performance as well eir holding, enterprise	provement of the economic and as the climate friendliness and and/or investment;
` ′	describe the env			
2.2.	Advice to unfollowing elements	•	ne forestry sector will	cover as a minimum one of the
□(a)		the relevant obl	ligations under the	
			Habitats Directive Birds Directive Water Framework I	Directive:
□(b)		forest holding;	the economic and e	nvironmental performance of the
□(c)	Trl	any other issues		:4
2.3.			dvisory services provive in the forestry sect	ider and does not involve direct for (beneficiaries):
		yes		no

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

2.4.	of regularly to		vice have the appropriate ff and advisory experie	
		yes		no
If the a	•	ease note that the aid ca	annot be declared comp	patible with the internal
2.5.	Is the advice p	partly provided in group	?	
		yes		no
	on of the individu	tion where advice is properties of advisory servers of aid limited to EUR 1		taking into account the
		yes		no
If the a		ease note that the aid ca	annot be declared comp	patible with the internal
2.7.			service providers under article 13(2) of Regulation	
		yes		no
OTHE	ER INFORMAT	ION		

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

2.6. SUPPLEMENTARY INFORMATION SHEET ON AID FOR COOPERATION IN FORESTRY SECTOR

This form must be used for the notification of any State aid measure whose aim is the cooperation in the forestry sector as described by section 2.6 of the EU Guidelines for State aid in the agricultural and forestry sectors and in rural areas 2014-2020 ('the Guidelines'). For aid for cooperation in the forestry sector related to horizontal and vertical cooperation among supply chain actors in the sustainable production of biomass for energy production and industrial processes and for rural tourism, the form covering Section 3.10 of Part II of the Guidelines must be filled in.

1. **TYPE OF AIDS**

1	l.]	The	coo	pera	tion	inv	olves	at	least:

two entities in the forestry sector;
one entity in the forestry sector and one in the agricultural sector

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.2.		The aid is granted in order to promote forms of cooperation benefitting the forestry sector, and in particular involves:						
□(a)		cooperation approaches among different actors in the forestry sector and other actors active in the forestry sector that contribute to achieving the objectives and priorities of rural development policy, including producer groups, cooperatives and inter-branch organisation;						
□(b) □(c)		the creation of clusters the establishment and	and networks in the force operation of operationand sustainability as refe					
2.	ELIGIBLE (COSTS AND AID INT	ENSITY					
2.1.	The aid will b	e granted for cooperation	on relating to the follow	ing activities:				
□(a)		pilot projects;						
			v producta practices pro	and tachnalaging				
□(b)		in the forestry sector;		ocesses and technologies				
□(c)		cooperation among sm and sharing facilities an		ng joint work processes				
□(d)	horizontal and vertical cooperation among supply chain actors for the establishment of logistic platforms to promote short supply chains and local markets;							
□(e)				g to the development of				
□(f)				ng or adapting to climate				
□(g)		joint approaches to env practices including effi		ongoing environmental nt, the use of renewable decapes:				
□(h)		horizontal and vertical sustainable production	cooperation among sup	pply chain actors in the d production if the result				
□(i)			anagement plan or equiv					
□(i)				olic and private partners				
⊔(<i>)</i>)		other than those defined No 1303/2013, of local in Article 2(19) of Reg	d in point (b) of Article 3 development strategies	2(2) of Regulation (EU) other than those defined 2013 addressing one or				
2.2.				vill it be granted only to g an activity that is new				
		yes		no				
		I -						

If the answer is no, please note that under the Guidelines this aid cannot be granted.

2.3. Aid for pilot projects and the development of new products, practices, processes and technologies in the forestry sector may be granted also to individual actors. Where aid is granted to individual actors, will the results of the supported project or activity be disseminated?

2.5.

□(e)

activities in the forestry sector:

	PART III.12 nt Generated: 2024-06	28		233		
	Change	Status: Point in time es to legislation: There are curre	view as at 31/01/2020. ntly no known outstanding effect 2004. (See end of Document for a	v .		
		yes		no		
If the a	nnswer is no, plea	ase note that under the C	Guidelines this aid canno	ot be granted.		
2.4.	2.4. Will the aid comply with the relevant provisions of competition law, in particular with Articles 101 and 102 of the Treaty?					
		yes		no		
	answer is no, ple e granted.	ease note that in accord	ance with point (706) o	of the Guidelines no aid		
If the a	answer is yes, ple	ase explain how this co	mpliance will be ensure	d.		

the costs of studies of the area concerned, feasibility studies, and the \Box (a) drawing up of a business plan or a forest management plan or equivalent or local development strategy other than the one referred to in Article 33 of Regulation (EU) No 1303/2013; □(b)

Aid may be granted to cover the following eligible costs in so far as they concern

the costs of animation of the area concerned in order to make feasible a collective territorial project or a project to be carried out by an operational group of the EIP for forestry productivity and sustainability as referred to in Article 56 of Regulation (EU) No 1305/2013; in the case of clusters animation may also concern the organisation of training, networking between members and the recruitment of new members;

the running costs of cooperation, such as the salary of a 'coordinator'; □(c) the direct costs of specific projects linked to the implementation of \Box (d) a business plan, environmental plan or a forest management plan or equivalent, a local development strategy other than the one referred to in Article 33 of Regulation (EU) No 1303/2013 or other actions aiming at innovation including testing; the related direct costs must be limited to the eligible costs and the maximum aid intensities of investment aid in the forestry sector, as specified in Section 2.1 of Part II of the Guidelines on investment aid;

the costs of promotion activities.

environmental action in duly justified cases?

2.6. Is the aid limited to a maximum period of seven years except for collective

yes	no

Please provide the justification for collective environmental actions exceeding seven years:

The aid is granted up to:% of the eligible costs (maximum: 100 %, with the exception 2.7. of direct costs).

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

START-UP AID FOR PRODUCER GROUPS AND ORGANISATIONS IN THE 2.7. FORESTRY SECTOR

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.	and organisations which organisations, entities or thich is the management effect a single producer as mutual support and out being involved in the					
		yes		no		
		ommission does not au large enterprises.	thorize aid under Section	on 2.7 of Part II of the		
2.	Please confirm that the agreements, decisions and concerted practices concluded in the framework of the producer group or organisation comply with the relevant provisions of competition law, and in particular with Article 101 and 102 of the Treaty.					
		yes		no		
3.	Have producer groups or organisations been officially recognised by the competent authority of the Member State concerned on the basis of a submission of a business plan?					
		yes		no		
4.	Is the Member State obliged to verify that the objectives of the business plan have been reached within a period of five years from the date of recognition of the producer group or organisation?					
		yes		no		
5.	If the aid is financed exclusively from national resources, please indicate whether the eligible costs concern:					
□(a) □(b)	the rental of suitable premises, the acquisition of office equipment, including computer hardware and software, administrative staff costs, overheads and legal and administrative fees.					
6.	Where premis rental costs at		on 5 are purchased, will	the costs been limited to		
		yes		no		

7. If the aid is financed exclusively from national resources, have the costs been incurred after the fifth year after recognition of the producer group or organisation by the competent authority on the basis of its business plan?

	1 PART III.12 ant Generated: 2024-06-	28		
	0	es to legislation: There are curre	view as at 31/01/2020. ntly no known outstanding effect 2004. (See end of Document for a	J
		yes		no
If the	answer is yes, ple	ase note that under the	Guidelines this aid cann	ot be granted.
8.	development			nal financing for a rural age marketed production
		yes		no
organi produ	isation, the supportion of the mem	t in the first year shall be	e calculated on the basis organisation over the last	uction of the group or of the average marketed st five years before the
9.	rural develope the first five	nent measure, is the aid years from the date on gnised by the competen	I paid as a flat rate aid in which the producer gro	national financing for a n annual instalments for oup or organisation was of its business plan, and
		yes		no
10.				mber State pay the last on of the business plan?
		yes		no
11.	Is he maximu	m intensity 100 % of eli	igible costs?	
		yes		no
12.	Please confirm	n that the total amount o	of aid is limited to EUR	500 000.
		yes		no
13.	running the gr	oup or organisations du or organisation, can you	ring the first five years	ntributions to the cost of following the formation hay be granted up to the

OTHER INFORMATION

yes

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

no

Status: Point in time view as at 31/01/2020. Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2.8.	OTHER AID TO THE FORESTRY SECTOR WITH ECOLOGICAL, PROTECTIVE
	AND RECREATIONAL OBJECTIVES

2.0.	AND RECREATIONAL OBJECTIVES						
1.	COMMON PROVISIONS						
1.1.	Please describe how the measures directly contribute to maintaining or restoring the ecological, protective and recreational functions of forests, biodiversity and a healthy forest ecosystem. Those objectives must be the primary objectives of the aid measure.						
1.2.	Please confirm that no aid will be granted to forest based industries or for commercially viable extraction of timber or for transportation of timber or the processing of wood or other forestry resources into products or for energy generation.						
		yes		no			
		ase note that in accordantible with the internal		Part II of the Guidelines			
1.3.	the commerci			hary purpose of which is ig where the felled trees			
		yes		no			
		ase note that in accordant atible with the internal		Part II of the Guidelines			
1.4.	Is aid granted	to undertakings active	in the forestry sector?				
		yes		no			
1.5.	Please explain why the measures falling under Section 2.8 of Part II of the Guidelines aiming for ecological, recreational and protective objectives cannot be achieved with the application of the rural development like forestry measures in sections 2.1 to 2.7 of Part II of the Guidelines (point (63) of the Guidelines):						
2.	SPECIFIC P	ROVISIONS					
2.1.	Please confirm that the aid complies with the common assessment principles and the common provisions applicable to Section 2.8 of Part II of the Guidelines.						

3. SECTION 2.8.1.

yes

Aid for specific forest actions and interventions with the primary objective to contribute to maintaining or to restoring forest ecosystem and biodiversity or the traditional landscape.

no

3.1. Please confirm that the primary objective of aid for planting, pruning, thinning and felling of trees and other vegetation in existing forests, the removal of fallen trees, and the planning costs of such measures, aid for the costs of treating and preventing the

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

spreading of pests and tree diseases and aid to make good the damage caused by the pests and tree diseases, is to contribute to maintaining or to restoring forest ecosystem and biodiversity or the traditional landscape.

		yes		no				
3.2.	Please describ	be the measure in more	details:					
3.3.	Does the aid for the costs of treating and preventing the spreading of pests and tree diseases and aid to make good the damage caused by the pests and tree diseases include the following costs?							
□(a)		preventive and treatment measures, including soil preparation for replanting, and the products, appliances and materials necessary for such measures. Biological, physical and other non-chemical mechanical prevention and treatment methods must be preferred to chemical methods, unless it can be shown that such methods are not sufficient to						
□(b)		provide satisfactory control over the disease or pest in question; loss of stock and restocking costs up to the market value of the stock destroyed on the order of the authorities to fight the disease or pest in question. When calculating the increment loss, the potential increment of the stock destroyed until the normal felling age may be taken into consideration.						
3.4.	The maximun	n intensity: (up to 100 %	% of eligible costs).					
4.	SECTION 2.	8.2.						
			nd improving the soil ee growth in the forestr	quality and ensuring a y sector.				
4.1.		for maintaining and imhealthy tree growth?	proving the soil quality	in forests and ensuring				
		yes		no				
4.2.	Please describ	be the measure in more	details:					
4.3.	maintain its i	natural balance, reduci		and other treatments to n density and ensuring r planning costs?				
		yes		no				
4.4.	Please explain how it is ensured that the measures do not reduce biodiversity, cause nutrient leaching or adversely affect natural water ecosystems or water protection zones.							
	zones.		Are planning costs covered?					
4.5.		costs covered?						

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

4.6.	The maximum intensity: (up to 100 % of eligible costs)						
5.	SECTION 2.	SECTION 2.8.3.					
	Aid for restoration and maintenance of natural pathways, landscap elements and features and natural habitat for animals in the forests sector.						
5.1.					ee of natural pathways, nals, including planning		
		yes			no		
5.2.	Please describ	be the measure ar	nd eligible costs in	more details	:		
5.3.	and the Birds	Please confirm that measures aiming at the implementation of the Habitats Directive and the Birds Directive are excluded from this type of aid (those measures must be covered by the form related to Section 2.2).					
		yes			no		
5.4.	The maximum	n intensity: (up to	o 100 % of eligible	e costs).			
6.	SECTION 2.	8.4.					
		Aid for maintain	ing roads to preve	ent forest fire	S		
6.1.	Please describ	be the aid measur	e:				
6.2.	Please describe the road main		en the objective of	f the aid (pre	venting forest fires) and		
6.3.	The maximum	n intensity: (up to	o 100 % of eligible	e costs).			
7.	SECTION 2.	8.5.					
		Aid to make good by law.	od the damage in	forests cause	ed by animals regulated		
7.1.	Are the anima	als causing the da	mage:				
□(a) □(b)	protected animals as defined by Point (35)28 of the Guidelines; species subject to specific national legislation.						
If the ar		b), please demon	strate the interest	for preservir	ng the population of the		
7.2.	Have reasonable preventive measures proportionate to the risk of damage by the animals regulated in the forest area concerned been taken?						
		yes			no		

In case of impossibility to take such reasonable preventive measures, please explain why no preventive measures are possible:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

occurrence of the event causing the damage and that the aid will be parfour years from that date. yes		an the direct causal link between the established?	e damage suffered a	and the behavior of the animals				
occurrence of the event causing the damage and that the aid will be parfour years from that date. yes	<u> </u>	yes		no				
7.5. Is the damage calculated at the level of the individual beneficiary? yes	00	Please confirm that the aid scheme has been established within three years from the occurrence of the event causing the damage and that the aid will be paid out within four years from that date.						
yes □ no 7.6. Have the costs of the damage incurred as a direct consequence of the of the damage been assessed either by a public authority, by an indeper recognized by the granting authority or by an insurance undertaking? □ yes □ no 7.7. Please state the kind of damage: □(a) damage to living trees. The aid may be granted to comper of stock and for restocking costs up to the market value destroyed by the regulated animals. When calculating the of the increment loss, the potential increment of the stock do the normal felling age may be taken into consideration. □(b) other costs incurred by the beneficiary due to the even damage, such as treatment measures, including soil preplanting and the products, appliances and materials necess operations. □(c) the material damage to the following assets: forestry machinery and buildings. The calculation of the material be based on the repair cost or economic value of the abefore the event caused the damage. It does not exceed the	1	yes		no				
7.6. Have the costs of the damage incurred as a direct consequence of the of the damage been assessed either by a public authority, by an indeper recognized by the granting authority or by an insurance undertaking? yes	5. Is	s the damage calculated at the level	of the individual b	peneficiary?				
the damage been assessed either by a public authority, by an indeperence recognized by the granting authority or by an insurance undertaking? yes		yes		no				
7.7. Please state the kind of damage: damage to living trees. The aid may be granted to compe of stock and for restocking costs up to the market value destroyed by the regulated animals. When calculating the of the increment loss, the potential increment of the stock do the normal felling age may be taken into consideration. other costs incurred by the beneficiary due to the even damage, such as treatment measures, including soil properations. c) the material damage to the following assets: forestry machinery and buildings. The calculation of the material be based on the repair cost or economic value of the abefore the event caused the damage. It does not exceed the	th	ne damage been assessed either b	y a public authori	ity, by an independent expert				
damage to living trees. The aid may be granted to compet of stock and for restocking costs up to the market value destroyed by the regulated animals. When calculating the of the increment loss, the potential increment of the stock destroyed the normal felling age may be taken into consideration. □(b) other costs incurred by the beneficiary due to the even damage, such as treatment measures, including soil properly replanting and the products, appliances and materials necessory machinery and buildings. The calculation of the material be based on the repair cost or economic value of the abefore the event caused the damage. It does not exceed the		yes		no				
damage to living trees. The aid may be granted to compet of stock and for restocking costs up to the market value destroyed by the regulated animals. When calculating the of the increment loss, the potential increment of the stock destroyed the normal felling age may be taken into consideration. □(b) other costs incurred by the beneficiary due to the even damage, such as treatment measures, including soil preplanting and the products, appliances and materials necessory machinery and buildings. The calculation of the material be based on the repair cost or economic value of the abefore the event caused the damage. It does not exceed the	7. P	lease state the kind of damage:						
□(b) other costs incurred by the beneficiary due to the even damage, such as treatment measures, including soil preplanting and the products, appliances and materials necessoperations. □(c) the material damage to the following assets: forestry machinery and buildings. The calculation of the material be based on the repair cost or economic value of the abefore the event caused the damage. It does not exceed the		damage to living trees of stock and for resto destroyed by the regul of the increment loss, t	damage to living trees. The aid may be granted to compensate for loss of stock and for restocking costs up to the market value of the stock destroyed by the regulated animals. When calculating the market value of the increment loss, the potential increment of the stock destroyed until					
the material damage to the following assets: forestry machinery and buildings. The calculation of the material be based on the repair cost or economic value of the a before the event caused the damage. It does not exceed the	(b)	other costs incurred to damage, such as trea replanting and the prod	other costs incurred by the beneficiary due to the event causing the damage, such as treatment measures, including soil preparation for replanting and the products, appliances and materials necessary for such					
	(c)	the material damage to the following assets: forestry equipment, machinery and buildings. The calculation of the material damage must be based on the repair cost or economic value of the affected asset before the event caused the damage. It does not exceed the repair cost or the decrease in fair market value caused by the event that is to say the difference between the property's value immediately before and						
·		Has the amount been reduced by any costs not incurred because of the damaging event,						
u yes u no		yes		no				

7.9. The aid intensity is up to (maximum 100 % of the eligible costs).

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

7.10.	7.10. Is aid and any other payments received to compensate the damage, including punder national or Union measures or insurance policies, limited to 100 % of th costs?					
		yes		no		
8.	SECTION 2.	8.6.				
		Aid for establishing for	rest management plans			
8.1.	Does the aid of	comply with the commo	on assessment principles	?		
		yes		no		
8.2.	Is the aid ava		ble in the area concern	ed based on objectively		
		yes		no		
market 8.3.	Where the provision of technical support is undertaken by producer groups or other organisations is membership of such groups or organisations a condition for access to the service?					
		yes		no		
If the a		ease note that the aid o	cannot be declared com	patible with the internal		
8.4.			owards the administration costs of providing the s	ve costs of the group or service?		
		yes		no		
If the a		ease note that the aid c	annot be declared comp	patible with the internal		
8.5.		anted to the service pro active in the forestry see		olve direct payments to		
		yes		no		
If the o	encwer ic no nle	ease note that the aid o	annot be declared com	patible with the internal		

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

8.6. Do the bodies selected to provide advice have the appropriate resources in the form of regularly trained and qualified staff and advisory experience and reliability with respect to the fields they advise in?

	PART III.12 t Generated: 2024-06-	28		203		
	Change					
		yes		no		
If the a		ease note that the aid ca	annot be declared comp	patible with the internal		
8.7.	Is the advice p	partly provided in group	?			
		yes		no		
		ation where advice is prical user of advisory serv	1 2 0 1	taking into account the		
8.8.				take to respect the non- on (EU) No 1306/2013?		
		yes		no		
8.9.	Is the service	provider the body estab	lishing the forest manag	gement plan?		
		yes		no		
Please	R INFORMAT indicate any oth		red relevant to the asso	sts).		
2.9.1.				RESTRY SECTOR		
1.	AID FOR RESEARCH AND DEVELOPMENT IN THE FORESTRY SECTOR Is the aid intensity limited to 100 % of the eligible costs?					
		yes		no		
2.		project of interest to all sectors concerned?	I undertakings active in	n the particular forestry		

3. Will the following information be published on the internet prior to the start of the aided project?

no

(a) that the aided project is to be carried out;

yes

(b) the goals of the aided project;

- (c) an approximate date for the publication of the results expected from the aided project;
- (d) the place of publication of the results expected from the aided project on the internet;

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

(e)	a reference that the results of the aided project are available to all undertakings in the forestry sector or sub-sector concerned at no cost.							
		yes		no				
4.	Please confirm	n that the results of the	aided project will be:					
□(a)		made available on internet, from the end date of the aided project or the date on which any information concerning those results is given to members of any particular organisation whatever comes first; and						
□(b)		remain available on int from the end date of the		least five years starting				
5.	Please confirm that the aid will be granted directly to the research and knowledge dissemination organisation and will not involve the provision of aid based on the price of the forestry products to undertakings active in the forestry sector.							
		yes		no				
6.	Please state th	e costs that are covered	by the aid:					
□(a)		personnel costs related staff to the extent emplo		ans and other supporting				
□(b)		the costs of instruments and equipment to the extent and for the period used for the project. Where such instruments and equipment are not used for their full life for the project, only the depreciation costs corresponding to the life of the project, as calculated on the basis of generally accepted accounting principles are considered as eligible costs;						
□(c)		the costs of buildings and land, to the extent and for the duration period used for the project. With regard to buildings, only the depreciation costs corresponding to the life of the project, as calculated on the basis of generally accepted accounting principles are considered as eligible costs. For land, the costs of commercial transfer or actually incurred						
□(d)		capital costs are eligible costs; the costs of contractual research, knowledge and patents bought or licensed from outside sources at arm's length conditions, as well as the costs of consultancy and equivalent services used exclusively for the						
□(e)				nses, including the costs arred directly as a result				

Please note that the aid must be limited to the costs referred to in points (a) to (e).

7. Please state the aid intensity: (maximum 100 %).

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

2.9.2. AID FOR FORESTRY LAND CONSOLIDATION

Status: Point in time view as at 31/01/2020.
Changes to legislation: There are currently no known outstanding effects for the
Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.	Is the aid inter	nsity limited to 100 % c	of the eligible costs?		
		yes		no	
2.	Please describ		strating that the aid is	aimed at forestry land	
3.	Is the aid limi	ted to the real legal, adr	ministrative and survey	costs incurred?	
		yes		no	
4.	Please describ	be the costs covered by	the aid measure:		
5. OTHE	Please state th	ne aid intensity: (maxim	um 100 %).		
		ner information consideration of the Guidelines		essment of the measure	
3.	SUPPLEMEN	TARY INFORMATIO	N SHEET ON AID IN I	RURAL AREAS	
by Cha	pter 3 of Part II		Guidelines for State ai	the rural areas covered d in the agriculture and	
aid in i genera	the agriculture a l eligibility cond	and forestry sectors and	l in rural areas (Part II	the notification of State I 12) demonstrating the rural areas forms 3.1 to	
act pro	viding the legal	basis in national law, to	ogether with any additio	ational law, or the draft onal documents, such as ls the State aid measure.	
trade a	nd industry, plea		otification form for thos	all sectors or specific to e sectors when notifying	
1.	GENERAL ELIGIBILITY CRITERIA				
1.		be granted in the framith Regulation (EU) No		lopment programme in	
		yes		no	
If the a	· .	ease note that the aid c	annot be declared comp	patible with the internal	

2. Please indicate the relevant rural development programme (RDP) and the measure under which the aid is granted:

RDP : Measure :

6.

□(a)

Do eligible costs include?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

3.	Is it co-finance	ed by the	EAFRD or doe	es it constitute a	dditiona	l nation	al financ	ing?
□(a) □(b)		co-financed by the EAFRD additional national financing						
4.	Is the investm	ent in energy saving and/or renewable energies?						
		yes				no		
II of the protection Question	Guidelines. Suon and energy 2 ns exclusively	ch aid mu 014-2020 <i>for Stat</i>	ast comply with 0, unless it is ex te aid scheme	s excluded from the Guidelines empt from the c s including in of Chapter 3 o	on State bligatio vestmen	e aid for on to not on to not	environi ify it. cural are	mental
5.	Do eligible co	sts includ	le the following	g?				
□(a)			truction, acquis	sition, including	g leasin	g, or in	nprovem	ent of
		(i) Is the amount of the land purchased equal or less than 10 % of the total eligible costs of the operation concerned?						
				yes			no	
		(ii) If the amount of the land purchased is more than 10 % of the total eligible costs of the operation concerned, does the operation concern environmental conservation?						
				yes			no	
□(b)	If the answer is yes, in exceptional and duly justified cases, a higher percentage may be permitted. Please elaborate on that in order to allow the Commission to assess the case at hand. (b) the purchase or lease purchase of machinery and equipment up to the					allow		
		market va	alue of the asse	t;	-		-	
□(c)		general costs linked to the expenditure referred to in points (a) and (b), such as architect, engineer and consultation fees, fees relating to advice on environmental and economic sustainability, including feasibility studies. Please note that feasibility studies remain eligible expenditure even where, based on their results, no expenditure under points (a) and (b) is incurred;						
□(d) □(e)		(b) is incurred; the following investments in intangible assets: acquisition or development of computer software and the acquisition of patents, licenses, copyrights and trademarks. Other (please specify):						
		a	1 3/	to in points (a)	to (d) is	s exhaus	tive	
_ 10000 II		5		pomio (u)	(4) 10			

costs, other than those referred to in question 5 (point (635) of the

Guidelines) connected with leasing contracts, such as lessor's margin,

interest refinancing costs, overheads and insurance charges;

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□(b) working capital.

If any of the expenses referred to in points (a) and (b) is included, please note that the aid cannot be declared compatible with the internal market.

- 7. Please state the maximum aid intensity, expressed as a percentage of eligible investment:
- (a) in less developed regions:
 - (i) % of the amount of the eligible costs for investment in regions whose GDP per capita is below 45 % of the EU-27 average;
 - (ii) % of the amount of the eligible costs for investment in regions whose GDP per capita is between 45 % and 60 % of the EU-27 average;
 - (iii) % of the amount of the eligible costs for investment in regions with a GDP per capita above 60 % of the EU-27 average;
- (b) in outermost regions that have a GDP per capita below or equal to 75 % of the EU-27 average:
 - (i) % of the amount of the eligible costs for investment in regions whose GDP per capita is below 45 % of the EU-27 average;
 - (ii) % of the amount of the eligible costs for investment in regions whose GDP per capita is between 45 % and 60 % of the EU-27 average;
 - (iii) % of the amount of the eligible costs for investment in regions with a GDP per capita above 60 % of the EU-27 average;
- (c) in other outermost regions:
 - (i) % of the amount of the eligible costs for investment in regions whose GDP per capita is below 45 % of the EU-27 average;
 - (ii) % of the amount of the eligible costs for investment in regions whose GDP per capita is between 45 % and 60 % of the EU-27 average;
 - (iii) % of the amount of the eligible costs for investment in regions with a GDP per capita above 60 % of the EU-27 average;
- (d) in 'c' areas:
 - (i) % of the amount of the eligible costs for investment in sparsely populated areas and in NUTS 3 regions or parts of NUTS 3 regions that share a land border with a country outside the European Free Trade Area (EEA) or the European Free Trade Association (EFTA);
 - (ii) % of the amount of the eligible costs for investment in non-predefined 'c' areas;
 - (iii) % of the amount of the eligible costs for investment in the former 'a' areas only for the period from 1 July 2014 to 31 December 2017;
 - (iv) % of the amount of the eligible costs for investment in the NUTS 3 areas or parts of NUTS 3 areas within a 'c' area which are adjacent to an 'a' area.
 - What is the difference in aid intensity between both these areas?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

		yes				no			
	(638)(a) to (Please note that if the answer is yes, the maximum aid intensities set out in point (638)(a) to (c) of the Guidelines cannot be increased by up 10 percentage points for medium-sized undertakings and by up to 20 percentage points for micro and small undertakings							
(f)	in all other a	in all other areas than those referred to in points (a) to (d) of this question:							
% of the amount of the eligible costs									
(g)	for aid for la	arge investment pro	ojects:						
	Please indicate the adjusted aid amount on the basis of the formula described in point (35).31. of the Guidelines (maximum aid amount = $R \times (50 + 0.50 \times B + 0.34 \times C)$, where R is the maximum aid intensity applicable in the area concerned, excluding the increased aid intensity for SMEs. B is the part of eligible costs between EUR 50 million and EUR 100 million. C is the part of eligible costs above EUR 100 million):								
8.	Does the Member State commit to notify pursuant to Article 108(3) of the Treindividual investment aid granted under a notified scheme, if the aid from all sou exceeds the notification threshold, as specified in point (37)(c) of the Guidelines?				e aid from all sources				
		yes			n	0			
	CONCERN NON-AGRI INVESTME AGRICULT	ING THE PROCICULTURAL PROCENTS IN THE URAL ACTIVITI	ESSING OIDUCTS, T CREATION ES relates to a	F AGRICULE THE PRODU N AND DE Sinvestments c	TURAL ICTION EVELOI	OR INVESTMENTS OR PRODUCTS INTO N OF COTTON OR PMENT OF NON- Ting the processing of cotton or investments			
in the	creation and d	evelopment of non	-agricultura	l activities, a	s descri	ibed in Section 3.1 of the agricultural and			
		n rural areas 2014				the agricultural and			
1.	Does the aid	l cover investments	s in tangible	assets and in	tangible	e assets?			
		yes			n	0			
2.	What is the aid for?								
□(a)		the processing	of agricult	ural products	where	e the output of the			
, ,		production proce	ess is a non-	agricultural p	roduct;	·			
□(b) □(c)		the production of investments in n							
		investments in non-agricultural activities which are granted to farmers or members of a farm household who diversify into non-agricultural activities and to micro and small undertakings and natural persons in rural areas.							
OTHE	ER INFORMA								

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

3.2. SUPPLEMENTARY INFORMATION SHEET ON AID FOR BASIC SERVICES AND VILLAGE RENEWAL IN RURAL AREAS

This Supplementary Information Sheet relates to State aid for basic services and village renewal in rural areas, as described in Section 3.2 of Chapter 3 of Part II of the European Union Guidelines for State aid in the agricultural and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

2.	Does the aid c	yes					
2.	Does the aid c			no			
	Does the aid cover investments in the creation, improvement or expansion of all types of small-scale infrastructures as defined in point (35)48 of the Guidelines, excluding investments in renewable energy and energy savings and broadband infrastructure?						
		yes		no			
3.	Does the aid cover investments in the setting-up, improvement or expansion of local basic services for the rural population, including leisure and culture, and the related infrastructure?						
		yes		no			
4.	Does the aid cover investments for public use in recreational infrastructure, tourist information and small-scale tourism infrastructure?						
		yes		no			
5.	restoration an landscapes an	d upgrading of the cu	ultural and natural her es, including related so	with the maintenance, itage of villages, rural cioeconomic aspects, as			
		yes		no			
6.	Does the aid cover investments targeting the relocation of activities and conversion of buildings or other facilities located inside or close to rural settlements, with a view to improving the quality of life or increasing the environmental performance of the settlement?						
		yes		no			

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

7.	Are the relevant operations of the investments implemented in accordance with pl for the development of municipalities and villages in rural areas and their baservices, where such plans exist?						
		yes		no			
8.	Are the relevant operations of the investments consistent with any relevant local development strategy?						
		yes		no			
9.	Is the aid referred to in point (644)(e) of the Guidelines granted for heritage which is formally recognised as cultural or natural heritage by the competent public authorities of the Member State?						
		yes		no			
Eligibi	le costs						
10.	The costs are	eligible if they concern	1:				
□(a)				opment and management vices, and to high nature			
□(b)		investment costs for ta	angible and intangible as	sets; ciated with cultural and			
□(c)		natural heritage, rural	landscapes and high nat	ure value sites;			
□(d) □(e)			rironmental awareness ac rks may be also eligible f	*			
, ,	. •.	the costs of capital works may be also eligible for aid referred to in point (644)(e) of the Guidelines.					
Aid in	·						
11.		m that the aid intensity s do not exceed 100 %		point (644)(a) and (b) of			
		yes		no			
12.	Please confirm that the aid intensity for activities under point (644)(c), (d) and (e) of the Guidelines does not exceed 100 % of the eligible costs.						
		yes		no			
				·			

Please note that the net revenues must be deducted from the eligible costs ex ante or through a clawback mechanism.

Please confirm that aid intensity for activities under point (644)(f) of the Guidelines 13. does not exceed the following amounts:

production capacity.

Document Generated: 2024-06-28

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□(a) where the relocation of the activities or the conversion of buildings or other facilities consists of the dismantling, removal and re-building of existing facilities, 100 % of real costs incurred for such activities; where the relocation of the activities or the conversion of buildings or other facilities results, in addition to the dismantling, removal and re-building of existing facilities as referred to in point (650)(a) of the Guidelines, in a modernisation of these facilities or in an increase in production capacity, the aid intensities for investments laid down in point (638) of the Guidelines must be applied in respect to the costs relating to the modernisation of the facilities or the increase of

Please note that for the purpose of point (650)(b) of the Guidelines, the pure replacement of an existing building or facilities by a new up-to date building or facilities without fundamentally changing the production or the technology involved will not be considered to be related to the modernisation.

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

3.3. SUPPLEMENTARY INFORMATION SHEET ON BUSINESS START-UP AID FOR NON-AGRICULTURAL ACTIVITIES IN RURAL AREAS

This Supplementary Information Sheet relates to State aid for business start-up for non-agricultural activities in rural areas, as described in Section 3.3 of Chapter 3 of Part II of the European Union Guidelines for State aid in the agricultural and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

1.	The aid will be granted to:
□(a) □(b)	farmers; members of the farm household diversifying into non-agricultura activities;
□(c)	micro and small undertakings;
□(d)	natural persons in rural areas;
□(e)	medium-sized and large enterprises in rural areas.

Please note that if the beneficiaries can be medium-sized and large enterprises in rural areas, aid can be granted only for the setting up of farm management, farm relief and farm advisory services, as well as forestry advisory services, including the Farm Advisory System referred to in Articles 12 to 14 of Regulation (EU) No 1306/2013.

		(20)110120	0,2012.				
2.	Please confirm	Please confirm that a business plan will be submitted.					
		yes		no			
3.		n that the implementat the date of the decision	ion of the business pla granting the aid.	n will start within nine			
		yes		no			

4. Please confirm that the business plan describes at least the following:

Status: Point in time view as at 31/01/2020. Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□(a) □(b) □(c)		the initial economic situation of the beneficiary applying for aid; milestones and targets for the development of the new activities of the beneficiary; details of the actions required for the development of the activities of the beneficiary, such as details of investments, training, advice or other activity.				
5.	Will the aid b	e paid in at least two ins	talments over a maximu	um period of five years?		
		yes		no		
If the marke		ease note that the aid v	vill be declared incomp	patible with the internal		
6.	Are the instal	ments degressive?				
		yes		no		
7.	Is the paymer the business p		conditional upon the co	rrect implementation of		
		yes		no		
If the marke 8.	t.		_	patible with the internal nomic situation of the		
		rea in determining the ar				
		yes		no		
9.	Please confirm	n that the aid amount is	limited to EUR 70 000	per undertaking.		
		yes		no		
ОТН	ER INFORMAT	ION				
		ner information conside ection of the Guidelines		essment of the measure		
3.4.	AND UND	ENT-CLIMATE COMN		AID FOR AGRI- R LAND MANAGERS ACTIVE IN THE		
1.	COMMON I	PROVISIONS				
1.1.		inted in accordance wit 1 of Chapter 1 of Part II		nt conditions set out in		
		yes		no		

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

			nate commitments grant al sector and other land	ed to groups formed by managers?
		yes		no
1.3.	undertake, on		arry out operations consis	erred to in question 1.2 sting of one or more agri-
		yes		no
	Is the aid for a	ral area as defined und agri-environment-clim	er point (35)50 of the Gu	d to other land managers
		yes		no
If the a	answer is yes, ple	ease justify such aid in	compliance with point (662) of the Guidelines.
1.5.	• •	-		·
1.3.		re not active in the agr	ate commitments granted icultural sector?	i to undertakings in rurar
				no
If the a	areas which a	yes ease note that in this cause and development	ase the aid may be provi	no ded for the conservation griculture for operations
If the a and for not cor Guidel	areas which a	yes ease note that in this cause and development existing under points (2)	ase the aid may be provi	no ded for the conservation griculture for operations
If the a	areas which a answer is yes, ploof the sustainable vered by the profines.	yes ease note that in this cause and development evisions under points (2) CLAUSE	ase the aid may be provi	no ded for the conservation griculture for operations 1.1.5.1. of Part II of the
If the a and for not cor Guidel 2.	areas which a answer is yes, ploof the sustainable vered by the profines.	yes ease note that in this cause and development evisions under points (2) CLAUSE	ase the aid may be provi of genetic resources in a 208) to (219) of Section	no ded for the conservation griculture for operations 1.1.5.1. of Part II of the
If the a and for not cor Guidel 2. 2.1. If the is oblimated amend Section is section in the and section is section.	areas which a answer is yes, play r the sustainable vered by the pro- lines. REVISION (Is a revision of the rel answer is no, play ge to introduce ments of the rel	yes ease note that in this cause and development evisions under points (2) CLAUSE clause provided for the yes ease note that under points in order to ensure evant mandatory stand 3 of Part II of the Gui	ase the aid may be provious genetic resources in a 208) to (219) of Section operations included in the coint (724) of the Guide the adjustment of the odards, requirements or o	no ded for the conservation griculture for operations 1.1.5.1. of Part II of the his aid? no lines, the Member State perations in the case of bligations referred to in
If the a and for not cor Guidel 2. 2.1. If the is oblimated amend Section is section in the and section is section.	areas which a answer is yes, play the sustainable vered by the prolines. REVISION (Is a revision of answer is no, play to introduce ments of the relation of the relation at Section have the section have the section of the section of the section of the section have the section of the section o	yes ease note that in this cause and development evisions under points (2) CLAUSE clause provided for the yes ease note that under points in order to ensure evant mandatory stand 3 of Part II of the Guitto go.	ase the aid may be provious genetic resources in a 208) to (219) of Section operations included in the coint (724) of the Guide the adjustment of the odards, requirements or odelines beyond which the	no ded for the conservation griculture for operations 1.1.5.1. of Part II of the his aid?

If the answer is yes, please note that under point (725) of the Guidelines, a revision clause in order to allow for the adjustment of the operations to the legal framework of the following programming period needs to be included.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

- 3.5. SUPPLEMENTARY INFORMATION SHEET ON AID FOR DISADVANTAGES RELATED TO NATURA 2000 AREAS TO OTHER LAND MANAGERS
- 1. Please provide justification for granting aid to other land managers:
- 2. Please confirm that the following conditions are complied with:
- aid will be granted to compensate other land managers for the additional costs and income foregone resulting from disadvantages in the areas concerned, related to the implementation of the Habitats Directive and the Birds Directive;
- \Box (b) only measures undertaken in the following areas will be eligible for aid:
 - (i) Natura 2000 agricultural areas designated pursuant to Habitats Directive and the Birds Directive:
 - (ii) other delimited nature protection areas with environmental restrictions applicable to farming which contribute to the implementation of Article 10 of the Habitats Directive.
- □(c) the aid will be limited to the amounts set out in point (668) of the Guidelines:
 - (i) EUR 500 per hectare per year maximum in the initial period not exceeding five years;
 - ii) EUR 200 per hectare per year maximum thereafter.
- if, in exceptional cases, the Member State increases the maximum amounts of EUR 500 and EUR 200, can it State explain the specific circumstances that justify it?

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

3.6. SUPPLEMENTARY INFORMATION SHEET ON AID FOR KNOWLEDGE TRANSFER AND INFORMATION ACTIONS IN RURAL AREAS

This notification form must be used for the notification of State aid for knowledge transfer and information actions in rural areas as described in Section 3.6. of Chapter 3 of Part II of the European Union Guidelines for State aid in the agricultural and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

- 1. Which of the following types of aid will be financed?
- □(a) vocational training and skills acquisitions (including training courses, workshops and coaching);
- □(b) demonstration activities;
- \Box (c) information actions;
- aid for the training for advisors linked to the advisory services referred to in Section 1.1.10.2. and Sections 2.5. and 3.7. of Part II of the Guidelines.

9.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2.				food sector, other land nd SMEs in rural areas?
		yes		no
3.	Please state th	ne maximum aid intensi	ty?	
large en		% in the case of media		ible costs in the case of d to 70 % in the case of
4.	Will the aid for	or the training for advisor	ors be granted in favour	of large enterprises?
		yes		no
5.		aid for the training for per period of three year		m aid amount limited to
		yes		no
If the a		ease note that the aid c	annot be declared comp	patible with the internal
6.	Which of the	following eligible costs	does the support measu	re cover?
□(a) □(b) □(c)		information action; in case of demonstratio	n projects, relevant inve	knowledge transfer or estment costs; per diem expenses of
7.	The aid will b	e granted in the form of	f:	
□(a) □(b)		subsidized services; direct payments of m reimbursement of actua		aly under the form of
	note that the aid payments to the l		2)(a) and (b) of the Guid	delines must not involve
8.			to in point (672)(a) and vledge transfer and info	(b) of the Guidelines be rmation action?
		yes		no
If the a	· .	ease note that the aid c	annot be declared comp	patible with the internal

Is the aid accessible to all those eligible undertakings active in the rural area concerned, based on objectively defined conditions?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

yes	no

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

10. Do the bodies providing knowledge transfer and information services have the appropriate capacities in the form of staff qualifications and regular training to carry out those tasks?

-		
	yes	no

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

OTHER INFORMATION

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

3.7. SUPPLEMENTARY INFORMATION SHEET ON AID FOR ADVISORY SERVICES IN RURAL AREAS

This notification form must be used for the notification of State aid for advisory services in rural areas as described in Section 3.7. of Chapter 3 of Part II of the European Union Guidelines for State aid in the agricultural and forestry sectors and in rural areas 2014 to 2020 ('the Guidelines').

1. Will the aid be granted to help other land managers and SMEs in rural areas benefit from the use of advisory services for the improvement of the economic and environmental performance as well as climate friendliness and resilience of their enterprise and/or investment?

yes	no

2. Advice will cover as a minimum one of the following elements:

□(a)	obligations	deriving	from	the	statutory	management	requirements
	and/or stand	dards for	good	agric	cultural an	d environmen	tal conditions
	provided in	Chapter I	of Tit	le VI	of Regula	tion (EU) No	1306/2013;

the agricultural practices beneficial for the climate and the environment laid down in Chapter 3 of Title III of Regulation (EU) No 1307/2013 and the maintenance of the agricultural area as referred to in point (c) of Article 4(1) of that Regulation;

measures aiming at farm modernization, competitiveness building, sectoral integration, innovation and market orientation as well as the promotion of entrepreneurship;

requirements defined by Member States for implementing Article 11(3) of the Water Framework Directive;

requirements defined by Member States for implementing Article 55 of Regulation (EC) No 1107/2009 of the European Parliament and of the Council⁽²³⁰⁾, and in particular compliance with the general principles of integrated pest management as referred to in Article 14 of Sustainable Use of Pesticides Directive⁽²³¹⁾;

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□(f) □(g)		specific advice for farm	ndards and safety standa ners setting up for the firs onmental sustainability.	ards linked to farms; st time, including advice
3.	Which of the measure?	following types of aid	will be financed by the	e aid scheme/individual
□(a)		advice to SMEs in rural performance of the ben		omic and environmental
□(b)		advice on the informa adaptation, biodiversity	ation related to climate and protection of water	change mitigation and s as laid down in Annex
□(c)				ironmental performance
□(d)			nent of short supply chair	ins, organic farming and
□(e)		advice on other issues.	37	
Please	describe the env	isaged measures:		
4.	The aid must payments to p		ory services provider and	l must not involve direct
		yes		no
If the a		ease note that the aid c	annot be declared comp	patible with the internal
5.	Is the advice p	partly provided in a gro	up?	
		yes		no
			priate, advice may be pa ndividual user of advisor	rtly provided in a group, ry services.
6.	If the answer	to question 5 is yes, ple	ase justify the provision	of advice in a group?
7.	Is the amount	of aid limited to EUR 1	1 500 per advice?	
		yes		no
If the a		ease note that the aid c	annot be declared comp	patible with the internal
8.				rs undertake to respect of Regulation (EU) No
		yes		no
TC :1				

If the answer is no, please note that the aid cannot be declared compatible with the internal market.

OTHER INFORMATION

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

3.8. SUPPLEMENTARY INFORMATION SHEET ON AID **FOR NEW** PARTICIPATION OF ACTIVE FARMERS IN QUALITY SCHEMES FOR

	COTTON AN	ND FOOI	OSTUFFS	_		
of active Chapter	e farmers in qu · 3 of Part II o	uality schoof the Eur	emes for cottor copean Union (n and foodstuffs as	sures for new participation described in Section 3.8. It is aid in the agriculture as	oj
1.	For which typ	e of sche	me is the aid for	or a new participation	on granted?	
□(a) □(b)		quality s	schemes for co		blished by Union legislation recognised by the Memberia:	
		(i)	scheme is der — spec — spec — a qu beyc rega	ived from clear obli- diffic product characterific farming or prod- ality of the final pro- ality of the commercial	luction methods, or oduct that goes significant all commodity standards or plant health, anim	tly
		(ii)	the scheme is	open to all produce	rs;	
		(iii)	compliance v		l product specifications artions is verified by publispection body;	
□(c)		Member practice	of agricultural y agricultural p State as meeting guidelines for	I products. product certification ng the requirements	res the complete traceabilic schemes recognised by the laid down in the Union beautary certification schemuffs.	he est
2.	determined a	ccording	to the level of		ent, the level of which ng from the participation	
		yes			no	
the Guio	delines, aid may	y be grant	ed only if thos	e conditions are fulf		of
3.	What is the a	mount of	the aid to be gr	ranted ner beneficia	rv per vear?	

ne amount of the aid to be granted per beneficiary per year

Please note that in accordance with Section 3.8. of Chapter 3 of Part II of the Guidelines, the aid is limited to EUR 3 000 per beneficiary per year.

OTHER INFORMATION

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Please indicate any other information considered relevant to the assessment of the measure concerned under this Section of the Guidelines.

3.9. SUPPLEMENTARY INFORMATION SHEET ON AID FOR INFORMATION AND PROMOTION ACTIVITIES CONCERNING COTTON AND FOODSTUFFS COVERED BY A QUALITY SCHEME

This form must be used for the notification of State aid for information and promotion activities

1.	and foods	tuffs which are		motion activities concerning cotton neme for which aid is granted unde
		yes		no
	answer is no e granted for		hat in accordance with p	oint (691) of the Guidelines no aid
2.		id be granted on activities?	nly to groups of producer	rs implementing the information and
		yes		no
If the				(602) 0.1 6 .1.1
may b	e granted for	such actions.	^	oint (692) of the Guidelines no aid
	e granted for Will the a	such actions. id cover only t ed to induce of	he costs for actions havir	oint (692) of the Guidelines no aiding the following characteristics: odstuffs or the cotton covered by a
may b	e granted for Will the a be design quality so draw atter to the qua	such actions. id cover only t ed to induce c heme; ntion to specific lity, specific pr	the costs for actions having consumers to buy the footened features or advantages of	ng the following characteristics: odstuffs or the cotton covered by a of the foodstuff or the cotton, notably nimal welfare standards and respec
may b 3. (a)	e granted for Will the a be design quality so draw atter to the qua	such actions. id cover only t ed to induce c heme; ntion to specific lity, specific pr	the costs for actions having consumers to buy the foot consumers or advantages or roduction method, high a red to the quality scheme	ng the following characteristics: odstuffs or the cotton covered by a of the foodstuff or the cotton, notably nimal welfare standards and respec
may b 3. (a)	be design quality so draw atter to the qua for the en	such actions. id cover only t ed to induce c heme; ntion to specific lity, specific pr vironment link ye wer is no, pleas	he costs for actions having consumers to buy the footen features or advantages of roduction method, high a fed to the quality scheme	odstuffs or the cotton covered by a fifthe foodstuff or the cotton, notably nimal welfare standards and respectoncerned?
may b 3. (a)	e granted for Will the a be design quality sc draw atter to the qua for the en	such actions. id cover only t ed to induce of heme; ntion to specific privironment link ye wer is no, pleas y be granted for	he costs for actions having consumers to buy the foot consumers to buy the foot consumers or advantages or roduction method, high a led to the quality scheme s see note that in accordance or such actions. In the aid scheme incidence of the consumer incidence in the consumer in the consumer incidence in the consumer incidence in the consumer incidence in the consumer incidence in the consumer in the consumer incidence in the consumer in the consu	odstuffs or the cotton covered by a of the foodstuff or the cotton, notably nimal welfare standards and respectoncerned?

Will the origin of the foodstuff or cotton be indicated in the actions included in the aid scheme?

				ocument Generated: 2024-06-28
		es to legislation: There are curre	view as at 31/01/2020. ntly no known outstanding effect 2004. (See end of Document for c	
		yes		no
6.		to question 5 is yes, was ordinate to the main mes		rigin of the foodstuff or
		yes		no
If the an		ase note that under poin	t (695) of the Guideline	s no aid may be granted
7.	Will there be or of commerce		tion activities related to	particular undertakings
		yes		no
If the an		ase note that under poin	at (696) of the Guideline	s no aid may be granted
8.	Will the informarket?	rmation and promotion	activities only be impl	emented in the internal
		yes		no
If the an		ase note that under poin	t (697) of the Guideline	s no aid may be granted
9.	The aid rate if following:	in the case of these int	formation and promotic	on activities will be the
OTHER	R INFORMAT	up to 70 % (indicate the ION	e exact rate: %)	
		ner information conside ection of the Guidelines		essment of the measure
3.10.	SUPPLEMEN RURAL ARE		N SHEET ON AID FO	OR COOPERATION IN
describe	ed in Section 3.	10. of Chapter 3 of Par		ation in rural areas as ion Guidelines for State 2020 ('the Guidelines').
1.	active in the a that contribute	gricultural sector, under to achieving the objec	takings active in the foo	ion among undertakings d chain and other actors tral development policy, anizations?
		yes		no

2. If the answer to question 1 is yes, will the cooperation benefit rural areas?

ANNEX I	ion Regulation (EC) No 794/2004 of 21 April 2004 implementing Council Regulation (EU) 28 PART III.12					
Documer	Document Generated: 2024-06-28 Status: Point in time view as at 31/01/2020. Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)					
Ues no						
	yes no					
	If the answer is no, please note that in accordance with point (700) of the Guidelines no aid may be granted.					
3.	Will the cooperation involve at least two entities?					
	yes no					
4.	Will the cooperation involve?					
□(a) □(b) □(c)	cooperation approaches; the creation of clusters and networks; the establishment and operation of operational groups of the EIP for agricultural productivity and sustainability as referred to in Article 5 of Regulation (EU) No 1305/2013.					
5.	The aid will be granted to cooperation relating to the following activities:					
□(a) □(b)	pilot projects; the development of new products, practices, processes and technologie in the food sector;					
□(c)	cooperation among small operators in organizing joint work processe and sharing facilities and resources and for the development and/o marketing of tourism services relating to rural tourism;					
□(d)	horizontal and vertical cooperation among supply chain actors for the establishment and development of short supply chains and local markets;					
□(e)	promotion activities in a local context relating to the development of short supply chain and local markets;					
□(f)	joint actions undertaken with a view to mitigating or adapting to climat change;					
□(g)	joint approaches to environmental projects and ongoing environmental practices, including efficient water management, the use of renewable energy and the preservation of agricultural landscapes;					
□(h)	horizontal and vertical cooperation among supply chain actors in th sustainable provision of biomass for use in food and energy productio and industrial processes;					
□(i)	the implementation, in particular by groups of public and privat partners other than those defined in point (b) of Article 32(2) of Regulation (EU) No 1303/2013, of local development strategies other than those defined in Article 2(19) of Regulation (EU) No 1303/201 addressing one or more of the Union priorities for rural development;					

6. Will the aid for the creation of clusters and networks be granted only to newly formed clusters and networks and those commencing an activity that is new to them?

□(j)

the diversification of farming activities into activities concerning health

care, social integration, community-supported agriculture and education about the environment and food.

|--|

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

If the answer is no,	please note that	in accordance	with point (70)	3) of the	Guidelines 1	no aid
may be granted.						

may b	e granted.		1 , ,	
7.	processes and	technologies in the food		new products, practices, pindividual actors where ramme?
		yes		no
8.		to question 7 is yes, yied out by individual ac		pilot projects and other
		yes		no
	answer is no, ple e granted.	ease note that in accord	ance with point (704) o	of the Guidelines no aid
9.			*	rt supply chains, cover rry between farmer and
		yes		no
	e granted. Will the aid co		• • • •	of the Guidelines no aid on law, in particular with
		yes		no
	answer is no, ple e granted.	ease note that in accord	ance with point (706) o	of the Guidelines no aid
If the a	answer is yes, ple	ease explain how this co	mpliance will be ensure	d.
11.		mited to a maximum l action in duly justified		s except for collective
		yes		no
Please years:	provide the justif	fication for collective en	vironmental actions exc	eeding a period of seven
12.	Will the aid b	e granted to cover the fo	ollowing eligible costs?	
□(a) □(b)		drawing up of a busines the one referred to in A costs for the animatic	ss plan or local developer rticle 33 of Regulation (on of the area concer	sibility studies, and the ment strategy other than (EU) No 1303/2013; ned in order to make roject to be carried out
				ultural Productivity and

Changes to legislation: There are currently no known outstanding effects for the

	Comm	ission Regulation (EC) No 794/2	2004. (See end of Document for a	details)	
□(c) □(d)	Sustainability as referred to in Article 56 of Regulation (EU) No 1305/2013; in the case of clusters, the animation may also concern networking between members and the recruitment of new members; the running costs of cooperation, such as the salary of a 'coordinator'; the direct costs of specific projects linked to the implementation of a business plan, an environmental plan, a local development strategy other than the one referred to in Article 29 of Regulation (EU) No 1303/2013 or other actions targeted towards innovation, including testing; costs for promotion activities.				
	ote that in according those eligible		s) of the Guidelines, the	e aid may be granted to	
13.	in points (635	t costs be limited to the and (636) of the Guide point (634) of the Guide	elines and comply with	stment aid, as specified a the specific conditions	
		yes		no	
If the an may be g		ease note that in accorda	ance with point (709) o	of the Guidelines no aid	
14. OTHER	Please indicat INFORMAT	e the aid intensity as a p	percentage of the eligible	e costs.	
		ner information conside ection of the Guidelines.		essment of the measure	
3.11.	SUPPLEMEN MUTUAL FU		N SHEET ON AID F	FOR SETTING-UP TO	
describe	d by Section 3.	11. of Chapter3 of Part	II of the European Un	ng-up of mutual funds as ion Guidelines for State 2020 ('the Guidelines').	
1.		which losses will be covill be partly financed un		d for which the financial asure:	
□(a) □(b)			diseases and plant pests Chapter 1 of Part II of t	n can be assimilated to a , as specified in Sections he Guidelines;	
2.	What are the	•			
		The administrative cost maximum period of thro		tual fund, spread over a manner.	
Please no	ote that there ar	re no other eligible costs	s. Initial capital stock is	not an eligible cost.	
3.	What is the le	vel of aid proposed? (as	a percentage)		
Please no	ote that the max	ximum aid rate is 65 %	of the eligible costs.		
4.	Will the amou	int of the cost eligible fo	or support be limited?		

yes

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

4.1.	If the answer	If the answer is yes, how will the amount be limited?				
	Ceiling per fund:					
5.	Has the mutual fund been accredited by the competent authority in accordanational law?				ority in accordance with	
		yes			no	
6.		utual fund have a trom the mutual fund?		policy towar	ds payments into and	
		yes			no	
7.	Does the muincurred?	itual fund have clear	rules attr	ributing respon	sibilities for any debts	
		yes			no	
not an					questions 5, 6 and 7 are eme compatible with the	
8.	particular for		nsation pay	yments, as well	al fund been defined, in as for the administration	
		yes			no	
9.	Does the mut part of the un		s provide fo	or penalties in c	ase of negligence on the	
		yes			no	
answe interna		, the Commission can			uestions 8 and 9 are not me compatible with the	

Textual Amendments

concerned under this Section of the Guidelines.]

F6 Substituted by Commission Regulation (EU) 2016/246 of 3 February 2016 amending Annex I to Regulation (EC) No 794/2004 as regards the forms to be used for the notification of State aid in the agricultural and forestry sectors and in rural areas.

Please indicate any other information considered relevant to the assessment of the measure

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

[F7]X1PART III.12.R.

SUPPLEMENTARY INFORMATION SHEET ON AID FOR THE PROMOTION AND ADVERTISING OF AGRICULTURAL PRODUCTS

This notification form must be used for State aid for advertising of products listed in Annex I to the EC-Treaty.

Please note that promotion operations as defined as the dissemination to the general public of scientific knowledge, the organisation of trade fairs or exhibitions, participation in these and similar public relations exercises, including surveys and market research, are not considered as advertising. State aid for such promotion in the broader sense is subject to points IV.j and IV.k of the Community guidelines for State aid in the agriculture and forestry sector 2007 to 2013⁽²³²⁾.

1.	Advertising campaigns within the Community				
1.1.	Where will the measure be carried out?				
	on the market of another Member State; on the home market.				
Who v	vill carry out the	advertising campaign?			
		producer groups or other others (please explain):	er organisations, regardl	ess of their size;	
1.2.	Can your auth Commission?	norities submit samples	or mock-ups of the adv	vertising material to the	
		yes		no	
If not,	please explain w	hy.			
1.3.	Please provide an exhaustive list of the eligible expenses.				
1.4.	Who are the beneficiaries of the aid?				
	farmers; producer groups and/or producer organisations; enterprises active in the processing and marketing of agricultural products; others (please specify)				
1.5.		norities give the assuran nefit from the aid in the	ce that all producers of same manner?	the products concerned	
		yes		no	
1.6.		criteria to be established	marked for quality produ I pursuant to Article 32		
		yes		no	

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.7.	Will the advertising campaign be earmarked for EU-recognized denominations with reference to the origin of the products ?			
		yes		no
1.8.		e said reference correspective Community?	oond exactly to the refe	rences which have been
		yes		no
1.9.	Will the adver		marked for products using	ng a national or regional
		yes		no
1.10.	Does the labe	l make any reference to	the national origin of th	ne products concerned?
		yes		no
1.11. 1.12.	in the messag Is the advertis	e.		ducts will be subsidiary
		yes		no
1.13.	If yes, will the the products?		be carried out without r	eference to the origin of
		yes		no
If no, p		under point VI.D of t	he Guidelines no aid m	nay be granted for such
1.14.	Will the advecompanies?	ertising campaign be d	edicated directly to the	e products of particular
		yes		no
If vec	nlease note that	under point VID of t	he Guidelines no aid n	hav he granted for such

If yes, please note that under point VI.D of the Guidelines no aid may be granted for such campaigns.

1.15. Will the advertising campaign comply with the provisions of Article 2 of Directive 2000/13/EC of the European Parliament and of the Council of 20 March 2000 on the approximation of the laws of the Member States relating to labelling, presentation and

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

advertising of foodstuffs, as well as, where relevant, with the specific labelling rules laid down for various products (wine, dairy products, eggs and poultry)⁽²³⁴⁾?

		1 ()) 1	1 37		
		yes		no		
If no, pleampaign		under point VI.D of th	ne Guidelines no aid m	ay be granted for such		
1.16.	The aid rate w	vill be the following:				
		the rest of the campaign	itself;	e the sector will finance		
				ecause the sector will cal levies or compulsory		
		up to 100 % (indicate the exact rate: %) because the advertising campaign is generic and in the benefit of all producers of the type of product concerned.				
2.	Advertising c	ampaigns in third cou	ntries			
2.1.	Is the advertising campaign in line with the principles of Council Regulation (EC) No 2702/1999 ⁽²³⁵⁾ ?					
		yes		no		
If no, ple		under point VI.D of th	ne Guidelines no aid m	ay be granted for such		
	rovide the ele on (EC) No 270		he compliance with the	e principles of Council		
2.2.	Is the advertis	ing campaign granted to	owards specific enterpris	ses?		
		yes		no		
If yes, p campaig		under point VI.D of the	he Guidelines no aid m	ay be granted for such		
2.3.	2.3. Does the advertising campaign risk endangering sales of or denigrate products from other Member States ?					
		yes		no		
			-			

Editorial Information

campaigns.

X1 Inserted by Corrigendum to Commission Regulation (EC) No 1935/2006 of 20 December 2006 amending Regulation (EC) No 794/2004 implementing Council Regulation (EC) No 659/1999 laying down detailed rules for the application of Article 93 of the EC Treaty (Official Journal of the European Union L 407 of 30 December 2006).

If yes, please note that under point VI.D of the Guidelines no aid may be granted for such

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Textual Amendments

F7 Inserted by Commission Regulation (EC) No 1935/2006 of 20 December 2006 amending Regulation (EC) No 794/2004 implementing Council Regulation (EC) No 659/1999 laying down detailed rules for the application of Article 93 of the EC Treaty.

PART III.12.S

SUPPLEMENTARY INFORMATION SHEET ON AIDS LINKED TO TAX EXEMPTIONS UNDER DIRECTIVE 2003/96/EC

This form must be used for the notification of any State aid measure linked to tax exemptions under directive $2003/96/EC^{(236)}$.

1.	Which measu	Which measure is envisaged?			
		tax reduction for motor fuels used in primary agricultural production; tax reduction for energy products and electricity used in primary agricultural production.			
2.	What is the le	What is the level of the envisaged reduction?			
3.	Under which article of Council Directive 2003/96/EC do you want to apply this exemption?				
4.	Will there be any differentiation in the level of exemption within the sector concerned				
		yes		no	
5.	If the possibility of applying a level of taxation down to zero to energy products and electricity used for agriculture is repealed by the Council, will the exemption envisaged fulfil all the relevant provisions of the directive, without tax differentiation within the sector concerned?				
		yes		no	

Please indicate which article(s) of the directive will be applied:

PART III.12.T

SUPPLEMENTARY INFORMATION SHEET ON AIDS FOR THE FORESTRY SECTOR

This form must be used for the notification of any State aid measure to support forestry covered by Chapter VII of the Community Guidelines on State aid in the agriculture and forestry sector⁽²³⁷⁾.

- 1. Objective of the measure
- 1.1. Does the measure contribute to maintaining, restoring or improving ecological, protective and recreational functions of forests, biodiversity and a healthy forest

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

		does it concern the elif the Guidelines?	gible costs mentioned	in points 175 to 181 in
	- Chapter VII o	yes		no
	please note that o		g at least one of these of	pjectives or eligible costs
2.	Eligibility cri	teria		
2.1.	extraction of		f timber or for the proc	for commercially viable sessing of wood or other
		yes		no
3. 3.1.	Type of aid Does the mea	sure include aid for plan on (point VII.C. a)?		and pruning of trees and
		yes		no
If ves. 1	please indicate v	whether the eligible cost	s concern:	
		planting, felling and proremoval of fallen trees;	uning in general;	mals, storms, floods, fire
of the n	neasure is to ma upe and that no a	aintain and restore fores	st ecosystem and biodiv whose primary purpos	nat the primary objective versity or the traditional e is commercially viable ed by equivalent ones:
		afforestation to increase	e forest cover;	
				to increase forest cover ltivated in the short term:
		afforestation to promote	e biodiversity;	
Please	describe the me	asure and indicate the a	reas concerned:	
		afforestation to create v	vooded areas for recrea	tional purposes;
		areas accessible to the to protect sensitive area		ecreational purposes? If
		afforestation to comba comparable protective		cation or to promote a

Please describe the measures specifying theareas concerned, the protective function envisaged, tree species to be planted and any accompanying and maintenance measures to be undertaken:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

		other (please explain).								
3.2.			nintaining and improving tree growth (point VII.)	g soil quality in forests C. b)?						
		yes		no						
If yes,	, please indicate v	whether the eligible cost	s concern:							
		fertilisation; other soil treatments;								
Please	e specify the type	of fertilisation and/or o	ther soil treatment							
□ □.		reduction of excessive vegetation density; ensuring sufficient water retention and proper drainage.								
or adv		ıral water ecosystems o		cause nutrient leaching is and describe how this						
3.3.	damage and t		ting and treating dama	and treating pests, pest ge done by animals or						
		yes		no						
If yes,	, please indicate w	whether the eligible cost	s concern:							
			nt of pests and tree diseant of damage done by an	ases and pest damage or nimals;						
Please	e indicate the pest	ts and diseases or anime	als in question:							
applia preferi	inces and materio	als. Are biological and g aid? If not, please den	mechanical prevention	any necessary products, and treatment methods not sufficient to fight the						
		targeted measures to pr	event forest fires.							
Please	e describe the med	asures:								
		ensate for the value of st disease or pest in question		als or on the order of the						
		yes		no						
		e value of stock will be lue thus determined:	calculated and confirn	n that the compensation						
3.4.				nce of natural pathways, nimals (point VII.C. d)?						
		yes		no						

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

3.5.	Does the meas	sure include aid for c	constructing imp	roving and maintaining forest roads
5.5.		s' infrastructures (po		to this und manualling for our found
		yes		no
If yes,	please describe t	he measures:		
	ne forests and in tional purposes?	nfrastructures used	for recreation o	ppen to the public at no cost for
		yes		no
				sure the proper and safe use of the usons for imposing them:
3.6.	Does the mea (point VII.C.f		or the costs of in	aformation materials and activities
		yes		no
dissen	ninate general in cts or producers of Does the mea	formation concernit or promote domestic	ng forests and do products: r the costs of pu	e supported actions and materials on not contain references to named archase of forestry land for nature
		yes		no
confir		s entirely and permo		f the forestry land in question and or or nature protection by means of a
3.8.	non-agricultu Natura 2000 p introducing p	ral land, establishm payments, forest-environeration actions a	ent of agro fore vironment payme s well as non pr	of afforestation of agricultural or estry systems on agricultural land, ents, restoring forestry potential and roductive investments, pursuant to 15 ⁽²³⁸⁾ or any replacing legislation?
		yes		no
		te that the measure 98/2005 or any repl		ions laid down in Articles 43-49 oj :
3.9.		sure include aid for nmentally friendly fo		sts and income foregone due to the?

4.

4.1.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

If yes,	please	describe	in detail	the tech	hnology	used a	ınd co	nfirm	that it	goes	beyond	the	relevan
mand	atory re	equireme	nts:										

		yes		no
	please note that to be the commitmen		ized under Chapter VII	of the Guidelines. If yes,
3.10.			e costs of purchase of f ction purposes, see poin	Forestry land (other than t 3.7 above)?
		yes		no
If yes, j	please describe t	he measure and indicat	e the aid intensity:	
3.11.				ltancy services, such as
		of business plans or fo pation in competitions,	restry management plar exhibitions and fairs?	ns, feasibility studies, as
				ns, feasibility studies, as
If yes p	well as partici	yes yes	exhibitions and fairs?	ı
If yes p Exemp	well as partici	yes te that the measure fulf	exhibitions and fairs?	no lown in Article 15 of the
If yes p	well as partici	yes te that the measure fulf	exhibitions and fairs?	no lown in Article 15 of the
If yes p Exemp 3.12.	well as participle well as participle well as participle as a demonstration Regulation: Does the mean	yes te that the measure fulf sure include aid for the	exhibitions and fairs? ils the conditions laid a setting up of forestry as	no Nown in Article 15 of the sociations?
If yes p Exemp 3.12.	well as participal velocition regulation: Does the mean regulation: Does the mean regulation: Does the mean regulation:	yes te that the measure fulf sure include aid for the yes te that the measure fulf sure include aid in favo	exhibitions and fairs?	no lown in Article 15 of the sociations? no down in Article 9 of the new techniques, such as

Is the aid for measures indicated under points 3.1 to 3.7 limited to 100 % of the eligible

no

Please describe how the exclusion of overcompensation will be controlled:

costs and overcompensation excluded?

yes

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	amount laid de			
		yes		no
			being co-financed under th co-financing envisag	er Regulation (EC) No ed or possible?
		yes		no
If yes,	please describe h	now any double funding	leading to overcompen.	sation will excluded:
4.3.	maximum aid	rate for aid under Artic	le 47 fixed in the Annex	.9 be granted above the cof Regulation (EC) No tional costs and income
	1010801101			
	1014801141	yes		no
In boti descri presen	h cases, please in be the specific c nt calculations sho	dicate the aid amount ircumstances and the	and describe how it is ceffect of the measure i	alculated. If yes, please to the environment and
In boti descri presen	h cases, please in be the specific c at calculations sho onal costs and/or Is the aid for	dicate the aid amount ircumstances and the owing that the additional income foregone: the measures indicated aid down in Article 4 o	and describe how it is ceffect of the measure is all amounts of aid are limed under point 3.10 is limited.	calculated. If yes, please to the environment and ited to the demonstrated imited to the maximum
In both descri present additio	h cases, please in be the specific c at calculations sho onal costs and/or Is the aid for aid intensity l	dicate the aid amount ircumstances and the owing that the additional income foregone: the measures indicated aid down in Article 4 o	and describe how it is ceffect of the measure is all amounts of aid are limed under point 3.10 is limited.	no calculated. If yes, please to the environment and ited to the demonstrated imited to the maximum ation for the purchase of
In both descripresen addition 4.4.	h cases, please in be the specific c at calculations sho onal costs and/or Is the aid for aid intensity la	dicate the aid amount dircumstances and the owing that the additional income foregone: the measures indicated aid down in Article 4 ond? yes	and describe how it is confect of the measure of all amounts of aid are limed under point 3.10 is lift the Exemption Regula	calculated. If yes, please to the environment and ited to the demonstrated imited to the maximum ation for the purchase of
In both descripresen addition 4.4.	h cases, please in be the specific cat calculations show and costs and/or Is the aid for aid intensity languicultural languicultural languicultural for languicultur	dicate the aid amount ircumstances and the owing that the additional income foregone: the measures indicate aid down in Article 4 ond? yes e exclusion of overcompressures indicated uncome foregone in a finite foregone.	and describe how it is coeffect of the measure of amounts of aid are limed under point 3.10 is lift the Exemption Regular	calculated. If yes, please to the environment and ited to the demonstrated imited to the maximum ation for the purchase of

f^{F1}PART III.13.A

Supplementary Information Sheet on investment aid to airports

It is recommended that this supplementary information sheet be completed in addition to the 'General information' form, for the notification of any investment aid covered by the Guidelines on State aid to airports and airlines⁽²³⁹⁾.

1. Additional information on the beneficiary, investment project and aid

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.1.	Beneficiary
	Deliciteiai

1.1.1. Is the aid awarded directly to the owner of the airport?

□yes	□no

- 1.1.2. If the answer to point 1.1.1 is no, please describe, if applicable, (i) the legal entity/-ies receiving the aid, and (ii) the legal entity/-ies transferring the aid to an intermediate entity or to the airport carrying out the investment project:
- 1.1.3. If the answer to point 1.1.1 is no, please explain how the national authorities ensure that that there is no advantage granted at the intermediary levels:
- 1.1.4. In the case of individual aid, please describe the legal, organisational and financial relationships between the recipient of the aid and (i) the undertakings with which it forms part of a group of undertakings; (ii) its subsidiaries; (iii) any other associated enterprises, including joint ventures.

In the case of aid schemes, please describe the method by which the aid granting authority will assess the legal, organisational and financial relations enumerated above:

1.1.5. Please specify whether the beneficiary will also operate the infrastructure:

□yes	□no

- 1.1.6. If the answer to the above question is no, please describe (i) the procedure by which the infrastructure operator will be chosen/respectively has been chosen, and (ii) selection criteria:
- 1.1.7. Where the airport or airports are used by the national armed forces, the police force, non-economic air rescue services, or any other air service of a non-economic nature, please specify (a) the nature of the service(s) and (b) the rate of airport capacity utilisation (e.g. use of the runway and other airport facilities, expressed as a percentage of annual aircraft movements):
- 1.1.8. Please provide the following passenger traffic data of the airport(s) receiving the aid:
- (a) Airports with a commercial passenger traffic history of more than two financial years: Average annual passenger traffic during the two financial years preceding the year in which the aid is notified or actually granted.
- (b) Airports with a commercial passenger traffic history of less than two financial years: Forecasted average annual passenger traffic during the two financial years after the beginning of the operation of commercial passenger air traffic.
- (c) For all airports, forecasted average annual passenger traffic during the projected economic lifetime of the subsidised infrastructure.

Please provide the data in the form of a table, as follow:

Year	Total number of passengers

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Passenger numbers must be counted 'one-way' and for each individual route; e.g. a passenger flying to the airport and back would be counted twice. If the airport is part of a group of airports, passenger traffic data must be established on the basis of each individual airport.

- 1.2. The investment project
- 1.2.1. Please describe the investment project and all underlying estimates, and provide the *ex-ante* business plan (in form of an excel table) on which the project is based. The business plan should cover the economic lifetime of the investment. All estimates should be based on sound demand forecasts. Please explain if, and to what extent, those estimates were taken into account in the beneficiary airport's business plan:
- 1.2.2. Please provide the following information on the investment project:

Date of application for the aid:	
Date of start of works on the investment project:	
Planned end of works on the investment project:	
Planned date of the start of operation of the investment:	
Planned date on which the full capacity is reached:	

1.2.3. Please provide, in one single table, a detailed breakdown of all works to be undertaken, their source of financing, their scheduled duration, related cost items and planned date of entry into operation.

Please indicate for each cost item whether and why it should be regarded as an investment cost (i) directly related to infrastructure of a non-economic nature within the public policy remit (e.g. for safety, air traffic control and any other activities for which a Member State is responsible as part of its official powers as a public authority), or (ii) related to aeronautical airport infrastructure of an economic nature (e.g. runway, ground handling infrastructure), or (iii) related to non-aeronautical infrastructure of an economic nature (e.g. parking, hotels):

Type of works	Financing	Cost structure	Timing

1.2.4. Please provide an overview of (i) the total eligible investment costs⁽²⁴⁰⁾ of an economic nature, and (ii) the total eligible costs of a non-economic nature. Costs must be discounted to their current value; the discount rate must be indicated.

Please indicate in that overview which part of the notified aid will support investments belonging to category (i) and which to category (ii):

- 1.2.5. If investment costs related to non-aeronautical activities of an economic nature will also be financed through State aid, please explain on what basis the authorities consider such aid compatible with the internal market:
- 1.2.6. Did you or do you commit to carry out an Environmental Impact Assessment ('EIA') for the investment? (point 20 of the Guidelines on State aid to airports and airlines)?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□yes	□no

If no, please explain why an EIA is not required for this project:

- 1.3. Activities of a non-economic nature falling within the public policy remit
- 1.3.1. Please confirm whether the investment covers activities that fall under the responsibility of the State in the exercise of its official powers as a public authority (e.g. traffic control, police, customs, firefighting, activities necessary to safeguard civil aviation against acts of unlawful interference). Investments relating to the infrastructure and equipment necessary to perform such activities are considered in general to be of a non-economic nature and thus do not fall within the scope of State aid rules. Please include relevant investments in the table provided in section 1.2.3:

□yes	□no

- 1.3.2. Please provide the national, regional or any other legal instrument concerning the notion of activities falling within the public policy remit, and their financing. In the absence of any such legal instrument, please clarify how those activities are usually financed by the relevant authorities:
- 1.3.3. Please provide evidence showing that the public financing of non-economic activities does not lead to undue discrimination between airports. Discrimination arises in situations where, under the relevant legal order, civil airports normally have to bear certain costs inherent to their non-economic activities, whereas certain civil airports do not have to cover such costs. Please specify the material and territorial applicability of the national rules applicable to the financing of non-economic activities of airports and, if applicable, the level of regional competence in that matter:
- 1.3.4. Please confirm, supported by relevant evidence, that the compensation of costs incurred in relation to non-economic activities will be strictly limited to such costs, and that any cross-subsidisation of economic activities through such compensation is effectively excluded:
- 1.3.5. Please confirm that the airport will maintain separate cost accounts for economic and non-economic activities:
- 2. Compatibility assessment of the measure
- 2.1. Contribution to a well-defined objective of common interest
- 2.1.1. Does the investment aid:
- (a) Increase the mobility of Union citizens and the connectivity of the regions by establishing access points for intra-Union flights?
- (b)□ Combat air traffic congestion at major Union hub airports?
- (c)□ Facilitate regional development?

Please specify how the investment aid contributes to each of the objective chosen.

2.1.2. Is the investment aimed at creating new airport capacity?

□yes	□по
------	-----

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 2.1.3. If the answer to the question in point 2.1.2 is 'yes', please demonstrate on the basis of the *ex-ante* business plan referred to in section 1.2 (The investment project) that the new infrastructure will, in the medium-term, meet the forecasted demand of the airlines, passengers and freight forwarders in the catchment area of the airport.
- 2.1.4. In the case of individual investment aid: is the beneficiary airport located in the same catchment area⁽²⁴¹⁾ as another airport that is *not* operating at or near full capacity?

□yes	□no
-	

If yes, please provide the following information: (a) size and shape of the catchment area; (b) the travelling distance and time between the beneficiary airport and airports in the same catchment area; (c) passenger traffic of other airports in the same catchment area over the 5 years preceding the year of notification; (d) the expected total demand and total capacity in the catchment area of the aided airport in at least the next 10 years, according to the business plan in the worst, base and best case scenarios:

- 2.1.5. In the case of aid schemes: Please (a) indicate the location and the catchment areas of eligible airports in the area of territorial applicability of the scheme; (b) the travelling distance and time between eligible airports and airports in the same catchment area; (c) the method and criteria the national authorities will use to define the size and shape of catchment areas and the capacity utilisation of airports in the same catchment area:
- 2.1.6. In the case of individual investment aid: If the answer to point 2.1.4 is 'Yes', please provide information identifying the likely effect of the investment on the use of existing infrastructure in the same catchment area. That information must demonstrate medium-term prospects for use, must be based on sound passenger and freight traffic forecasts, and must be incorporated in the beneficiary airport's *ex-ante* business plan:
- 2.1.7. In the case of individual investment aid: If the answer to point 2.1.4 is 'Yes', please provide traffic projections in terms of passenger numbers in the worst, base and best case scenarios and explain why, in your view, such projections justify investment aid for the creation of additional capacity or for maintaining existing capacity:
- 2.2. Need for State intervention
- 2.2.1. In case of individual investment aid: please explain, on the basis of the business plan of the airport, to what extent the airport's ability to cover its capital costs depends on the airport's size in terms of annual passenger traffic:
- 2.2.2. In case of individual investment aid: please explain why the airport could not obtain sufficient private financing:
- 2.2.3. In the case of aid schemes: Please confirm that the aid granting authority will verify the need for State intervention in each individual case on the basis of (i) the size of the airport⁽²⁴²⁾; (ii) the airport's ability to attract private financing:
- 2.3. Appropriateness of the measure
- 2.3.1. Please demonstrate that the aid in question is appropriate to achieve the intended objective or resolve the problems which it is meant to address. In particular, please explain how the authorities have established that the same objective cannot be reached, and that the same problem cannot be resolved, with less distortive policy or aid instruments and that the aid in question is an appropriate policy instrument. For

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

example, if the aid is granted in forms that provide a direct pecuniary advantage⁽²⁴³⁾, please demonstrate why other potentially less distortive forms of aid such as repayable advances or forms of aid that are based on debt or equity instruments⁽²⁴⁴⁾ are not appropriate:

- 2.4. Incentive effect of the aid
- 2.4.1. In the case of individual investment aid: please confirm that works on the notified individual investment started only after submission of the application for the aid to the granting authority. Please provide a copy of the aid application sent to the granting authority by the beneficiary and documentary evidence of the date of the start of works:
- 2.4.2. In the case of aid schemes: please confirm that works on eligible investment projects will start only after submission of the individual application for the aid to the granting authority:
- 2.4.3. In the case of individual investment aid: please describe the levels of intended activity in a scenario with aid and a counterfactual scenario without aid. Please provide supporting evidence, for example internal documentation relating to alternative activities considered by the beneficiary airport in its internal decision making:
- 2.4.4. In the case of individual investment aid: in the presence of a counterfactual scenario with alternative activities, please compare both scenarios to specify the additional activity that would only be undertaken with the aid (counterfactual analysis).

In the case of individual investment aid: In the absence of a counterfactual scenario with alternative activities, please specify the capital cost funding gap established on the basis of the beneficiary airport's *ex-ante* business plan. The capital cost funding gap is the difference between the positive and negative cash flows, including investment costs into fixed capital assets over the lifetime of the investment in net present value terms:

- 2.4.5. In the case of aid schemes: please confirm (a) that the granting authority will award individual aid under the aid scheme only after it has verified that there is an incentive effect, by comparing the levels of intended activity with aid and without aid (counterfactual analysis) or, in the absence of alternative activities without aid, by establishing the capital cost funding gap on the basis of the beneficiary airport's *ex-ante* business plan⁽²⁴⁵⁾; (b) describe all underlying input data, parameters and assumptions that the aid granting authority will assess for the purposes of the incentive effect analysis:
- 2.5. Proportionality of the aid
- 2.5.1. In the case of individual investment aid: in the presence of a counterfactual scenario with alternative activities, please (a) provide in the form of excel-tables the *ex-ante* business plans for the scenario with aid and the counterfactual scenario without aid; (b) on that basis specify the net extra costs, net of extra revenues, which result from undertaking the aided project or activity rather than the counterfactual project/activity; (c) explain all underlying input data, parameters and assumptions.

The business plans should cover the period of the economic lifetime of the investments:

2.5.2. In the case of individual investment aid: in the absence of a counterfactual scenario with alternative activities, please (a) provide in the form of an excel-table the *ex-ante* business plan of the beneficiary airport; (b) indicate on that basis the capital costs funding gap which is the net present value of the difference between the positive and

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

negative cash flows, including investment costs, over the economic lifetime of the investment; (c) explain all underlying input data, parameters and assumptions:

2.5.3. In the case of aid schemes: please commit (a) to investigate in each case the counterfactual scenario without aid, on the basis of *ex-ante* business plans; (b) to establish the extra costs, net of extra revenues, which result from undertaking the aided project or activity rather than the counterfactual project or activity in cases where a counterfactual scenario with alternative activities exists; (c) to establish the capital costs funding gap, which is the net present value of the difference between the positive and negative cash flows, including investment costs, over the economic lifetime of the investment, in cases where no alternative activities would have taken place.

Please describe all underlying input data, parameters and assumptions that the aid granting authority will assess for the purposes of the investigations and analysis:

2.5.4. Aid intensity:

Please indicate the maximum percentage of the eligible costs to be covered by investment aid ('aid intensity'); including any top-ups on the basic aid intensity:

Where an aid scheme applies to airports of various sizes, please indicate the maximum aid intensity applicable to the following groups of airports:

Size of airport based on average passenger traffic (passengers per annum)	Maximum investment aid intensity
> 3-5 million	
1-3 million	
< 1 million	

- 2.6. Avoidance of negative effects on competition and trade
- 2.6.1. Please confirm that the airport, including all infrastructure and equipment for which investment aid is granted, will be open to all potential users and not be dedicated to one specific user:
- 2.6.2. Please specify the provisions made to ensure that airport capacity will be allocated to users on the basis of pertinent, objective, transparent and non-discriminatory criteria:
- 2.7. Notification of individual aid under an investment aid scheme
- 2.7.1. The following individual aid measures under the investment aid scheme must be notified pursuant to Article 108(3) TFEU:
- (a) investment aid to airports with average annual traffic above 3 million passengers;
- (b) investment aid with an aid intensity exceeding 75 % to an airport with average annual traffic below 1 million passengers, with the exception of airports located in remote regions;
- (c) investment aid granted for the relocation of airports;
- (d) investment aid financing a mixed passenger and freight airport handling more than 200 000 tonnes of freight during the two financial years preceding that in which the aid is notified;

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (e) investment aid aimed at the creation of a new passenger airport (including the conversion of an existing airfield into a passenger airport);
- (f) investment aid aimed at the creation or development of an airport located within 100 kilometres distance or 60 minutes travelling time by car, bus, train or high-speed train from an existing airport.

PART III.13.B

Supplementary Information Sheet on operating aid to airports

It is recommended that this supplementary information sheet is completed, in addition to the 'General information' form, for the notification of any individual operating aid covered by the Guidelines on State aid to airports and airlines⁽²⁴⁶⁾.

- 1. Additional information on the beneficiary and its activities
- 1.1. Beneficiary
- 1.1.1. Is the aid awarded directly to the operator of the airport?

□yes	□no
------	-----

- 1.1.2. If the answer to point 1.1.1 is no, please describe, if applicable, (i) the legal entity/ies receiving the aid, and (ii) the legal entity/ies responsible for transferring the aid as intermediary/ies to the airport providing the services eligible for the aid.
- 1.1.3. If the answer to point 1.1.1 is 'no', please explain how the authorities ensure that that there is no advantage granted at the intermediary levels.
- 1.1.4. Is the beneficiary of the aid also the owner of the airport?

□yes □no	
----------	--

- 1.1.5. If the answer to point 1.1.4 is 'no': please specify who is/will be the owner of the airport and describe the ownership structure.
- 1.1.6. In the case of individual aid, please describe the legal, organisational and financial relationships between the recipient of the aid and (i) the undertakings with which it forms part of a group of undertakings; (ii) its subsidiaries; (iii) any other associated enterprises, including joint ventures.

In the case of aid schemes, please describe the method by which the aid granting authority will assess the legal, organisational and financial relationships referred to in points 1.1.1 to 1.1.5 above.

- 1.2. General information about the airport operator
- 1.2.1. Where the airport or the airports are used by the national armed forces, the police force, non-economic air rescue services, or any other air service of a non-economic nature, please specify (a) the nature of the service(s); and (b) the rate of airport capacity utilisation (e.g. use of the runway and other airport facilities, expressed as a percentage of annual aircraft movements).

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 1.2.2. Please provide the following passenger traffic data of the airport(s) receiving the aid:
- (a) Airports with a commercial passenger traffic history of more than two financial years: average annual passenger traffic during the two financial years preceding the year in which the aid is notified or actually granted.
- (b) Airports with a commercial passenger traffic history of less than two financial years: forecasted average annual passenger traffic during the two financial years after the beginning of the operation of commercial passenger air traffic flows.

Please provide the data in the form of a table, as follows: [copied from page 188, point 2.2.3]

Year	Total number of passengers

Passenger numbers must be counted 'one-way' and for each individual route. For example: a passenger flying to the airport and back would be counted twice. If the airport is part of a group of airports, passenger traffic data must be established on the basis of each individual airport.

1.2.3. In the case of individual operating aid: please provide the business plan that the beneficiary has implemented in the period 2009–2013, and that it intends to implement in the subsequent 10-year period until 4 April 2024. Please describe the assumptions underlying that 10-year period.

The business plan must contain information on traffic and traffic forecasts; costs and cost forecasts; financial data and financial forecasts as regards the level of profitability and cashflows (by reference to methodologies which are demonstrably used by the airport, e.g. using methods for evaluating the net present value of an investment (NPV), the internal rate of return (IRR), and the average return on capital employed (ROCE). The business plan must be provided in an excel format including explanations of all underlying formulas.

In the case of aid schemes, please specify in detail (a) the formal and material criteria to which business plans of eligible airports must conform; (b) the method the national authorities will use to assess the business plans.

1.2.4. In the case of individual operating aid, please provide a summary overview of operating losses⁽²⁴⁷⁾ the beneficiary incurred in the period 2009–2013, as well as of forecasted operating losses in the period until 4 April 2024. Please provide the data in the form of a table, structured as follows:

Revenue		
Operating costs		
Other		

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

• • •		
Operating results		

In the case of aid schemes, please specify the method the authorities will use to establish the operating losses of eligible airports.

1.2.5. In the case of individual operating aid, please provide copies of the eligible airports' financial reports⁽²⁴⁸⁾ for the 5 years preceding the year of application for operating aid.

In the case of aid schemes, please commit to include the above described financial reports in the assessment of individual aid.

- 1.2.6. Please specify the provisions made to avoid overcompensation and recover excess amounts from the beneficiary.
- 1.3. Airport service activities
- 1.3.1. Please specify the eligible airport services⁽²⁴⁹⁾ and the categories of eligible operating costs⁽²⁵⁰⁾ in respect of the provision of those services.
- 1.4. Activities falling within the public policy remit
- 1.4.1. Does the operating aid cover activities that fall under State responsibility in the exercise of its official powers as a public authority (e.g. air traffic control, police, customs, firefighting, activities necessary to safeguard civil aviation against acts of unlawful interference)? Operating costs relating to the infrastructure and equipment necessary to perform such activities are considered in general to be of a non-economic nature and thus do not fall within the scope of State aid rules.

□yes	□no
3	

- 1.4.2. Please provide the relevant national, regional or any other legal instrument concerning the notion of activities falling within the public policy remit and their financing. In absence of any such legal instrument, please clarify how those activities are usually financed by the relevant authorities.
- 1.4.3. Please provide evidence showing that the public financing of non-economic activities does not lead to undue discrimination between airports. Discrimination arises in situations where, under the relevant legal order, civil airports normally have to bear certain costs inherent to their non-economic activities, whereas certain civil airports do not have to cover such costs. Please specify the material and territorial applicability of the national rules applicable to the financing of non-economic activities of airports and, if applicable, the level of regional competences in that matter.
- 1.4.4. Please confirm, supported by relevant evidence, that the compensation of costs incurred in relation to non-economic activities will be strictly limited to such costs, and that any cross-subsidisation of economic activities through such compensation is effectively excluded.
- 1.4.5. Please confirm that the airport will maintain separate cost accounts for economic and non-economic activities.
- 2. Compatibility assessment of the measure

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2.1. Was the aid granted before 4 April 2014?

	T
□yes	□no

- 2.2. Contribution to a well-defined objective of common interest
- 2.2.1. Does the operating aid:
- (a) Increase the mobility of Union citizens and the connectivity of the regions by establishing access points for intra-Union flights?
- (b)□ Combat air traffic congestion at major Union hub airports?
- (c)□ Facilitate regional development?

Please specify how the operating aid contributes to the objective(s) chosen.

2.2.2. Does the notified measure concern the operator of a new airport?

Lyes Lino	□yes [□no
-----------	--------	-----

2.2.3. In the case of individual operating aid: Is the beneficiary airport located in the same catchment area⁽²⁵¹⁾ of another airport with spare capacity?

□yes	□no
2	

2.2.4. In the case of individual operating aid: If the answer to point 2.2.3 is yes, please specify the size and shape of the catchment area. Please provide information, identifying the likely effect on the traffic of the other airport located in that catchment area. That information should be part of the beneficiary airport's business plan and should be based on sound passenger and freight traffic forecasts.

In the case of aid schemes: Please (a) confirm that the authorities are committed to assess the likely effect on the traffic of any other airport(s) located in the same catchment area as an eligible airport on the basis of information that is part of the beneficiary airport's business plan and is based on sound passenger and freight traffic forecasts; (b) explain the method and the criteria the national authorities will use to assess the likely effect on the traffic of that other airport or those other airports.

- 2.3. Need for State intervention
- 2.3.1. Please confirm that the annual traffic of the airport/eligible airports does not exceed 3 million passengers (see also question 1.2.2 above).
- 2.4. Appropriateness of the measure
- 2.4.1. Please demonstrate that the aid in question is appropriate to achieve the intended objective or resolve the problems which it intendeds to address. In particular, please explain how the authorities have established that the same objective cannot be reached, or the same problem cannot be resolved, through less distortive policy or aid instruments. For example, if the aid is granted in forms that provide a direct pecuniary advantage⁽²⁵²⁾, please demonstrate why other potentially less distortive

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

forms of aid such as repayable advances or forms of aid that are based on debt or equity instruments⁽²⁵³⁾ are not appropriate:

2.4.2.	In the case of individual operating aid: Has the aid amount been established ex-ante
	as a fixed sum covering the expected operating funding gap determined on the basis
	of the business plan of the beneficiary, during a 10-year transitional period starting
	on 4 April 2014?

□yes	□no

If yes, please provide relevant information in the business plan.

2.4.3. In the case of aid schemes: will the aid amount in each individual case be established *ex-ante* as a fixed sum covering the expected operating funding gap determined on the basis of the business plan of the beneficiary, during a 10-year transitional period starting on 4 April 2014?

□ves	$ \Box no$
= y es	

If yes, the beneficiary should provide relevant information in the business plan.

- 2.4.4. If the answer to points 2.4.2 and 2.4.3 is 'no', please specify: (a) the degree of uncertainty of cost and revenue forecasts, (b) any information asymmetries that prevent the national authorities from calculating the aid amount *ex ante* on the basis of a business plan.
- 2.4.5. If the answer to points 2.4.2 and 2.4.3 is 'no', please confirm that the maximum amount of compatible operating aid was/will be established according to a model based on the average of the operating funding gaps⁽²⁵⁴⁾ during the five years from 2009 to 2013.
- 2.4.6. Please confirm that the amount of operating aid will not be increased *ex-post*.

□yes	□no

- 2.4.7. If the answer to the above question is 'no', please explain why in your opinion the possibility of an *ex-post* increase would not reduce incentives to manage the airport efficiently.
- 2.5. Incentive effect and proportionality of the aid
- 2.5.1. In the case of individual operating aid, please describe why there is the likelihood that in the absence of the aid the level of economic activity of the airport concerned would be significantly reduced. Please provide the necessary information on the basis of the business plan (see also point 1.2.3), comparing the levels of intended activity with aid and without aid (the counterfactual scenario), and taking into account the possible presence of investment aid and level of traffic.

In the case of aid schemes, please describe the method applied by the aid granting authority in order to assess the business plans as well as the likelihood that in the absence of aid the level of economic activity of the airport concerned would be significantly reduced, taking into account the possible presence of investment aid and level of traffic.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2.5.2. In the case of individual operating aid, please demonstrate that the business plan of the airport will result in full operating cost coverage by 4 April 2024. Please specify the relevant key parameters of the business plan.

In the case of aid schemes, please confirm that the aid granting authority will grant individual operating aid only if it has concluded that the business plan of the beneficiary airport will result in full operating cost coverage by 4 April 2024. Please specify which key parameters of the business plan the aid granting authorities will assess in order to reach that conclusion in each case.

2.5.3. Please indicate:

In the case of individual operating aid: The beneficiary airport's initial funding gap over 10 years, starting with the operating cost coverage on 4 April 2014 at the beginning of the transition period and reaching full operating cost coverage achievement by 4 April 2024, at the end of the transition period.

In the case of aid schemes: please commit (a) that the funding gap of eligible airports will be established using the method referred to in point 2.5.2; (b) that eligible airports must demonstrate that they will achieve full operating cost coverage by 4 April 2024.

The maximum permissible aid amount:

The percentage of the funding gap to be covered by the operating aid:

The period during which operating aid will be granted:

- 2.6. Avoidance of negative effects on competition and trade
- 2.6.1. Please demonstrate that all airports located in the same catchment area as the eligible airport or airports will be able to achieve full operating cost coverage by 4 April 2024.
- 2.6.2. Please confirm that the airport or airports, including any investment for which aid is granted, will be open to all potential users and not be dedicated to one specific user.

□yes	□no
•	

2.6.3. Please specify the provisions made to ensure that airport capacity will be allocated to users on the basis of pertinent, objective, transparent and non-discriminatory criteria.

PART III.13.C

Supplementary Information Sheet on start-up aid to airlines

It is recommended that this supplementary information sheet is completed, in addition to the 'General information' form in Part I, for the notification of any start-up aid covered by the Guidelines on State aid to airports and airlines⁽²⁵⁵⁾.

This form applies both to the notification of aid schemes and individual aid.

- 1. Additional information on the beneficiary, project and aid
- 1.1. Beneficiary
- 1.1.1. Is the aid awarded directly to the airline operating the new route?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□ yes	□ no

- 1.1.2. If the answer to point 1.1.1 is 'no': please describe, if applicable, the legal entity or entities (a) receiving the aid; (b) transferring the aid to an intermediate entity or to the airline that operates the new route.
- 1.1.3. If the answer to point 1.1.1 is 'no', please explain how the authorities ensure that the intermediary levels receive no advantage.
- 1.1.4. In the case of individual aid, please describe the legal, organisational and financial relationships between the recipient of the aid and (a) the undertakings with which it forms part of a group of undertakings; (b) its subsidiaries; (c) any other associated enterprises, including joint ventures.

In the case of aid schemes, please describe the method by which the aid granting authority will assess those legal, organisational and financial relationships:

1.1.5. Selection of the beneficiary: Please describe (a) the procedure by which the beneficiary will be chosen or has been chosen; (b) the medium and level of publicity of the selection procedure; (c) eligibility conditions; (d) operational requirements; (e) selection criteria.

2. Compatibility assessment of the measure

- 2.1. Contribution to a well-defined objective of common interest
- 2.1.1. Does the start-up aid:
- (a) Increase the mobility of Union citizens and the connectivity of the regions by opening new routes?
- (b)□ Facilitate regional development of remote regions?

Please explain how the aid contributes to the objective chosen.

2.1.2. In the case of individual aid, please demonstrate that the route or routes to be covered is or are not already covered by a high-speed rail service⁽²⁵⁶⁾ or from another airport in the same catchment area⁽²⁵⁷⁾ under comparable conditions. If conditions are considered not to be comparable, please explain why.

In the case of aid schemes, please explain how the aid granting authority will ensure that the condition in this point is met in each individual case of start-up aid.

- 2.2. Need for State intervention
- 2.2.1. Please indicate whether the start-up aid concerns:
- (a) Routes linking an airport with less than 3 million passengers *per annum*⁽²⁵⁸⁾ to another airport within the Common European Aviation Area⁽²⁵⁹⁾;
- (b) Routes linking an airport located in a remote region to another airport (within or outside the Common European Aviation Area) irrespective of the size of the airports concerned;
- (c) Routes linking an airport with more than 3 million passengers *per annum* and less than 5 million passengers per annum, not located in remote regions. Please duly substantiate the specific circumstances in this case;

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (d)□ Other (Please specify).
- 2.2.2. In the case of individual aid, please indicate the location of the airports linked by eligible new routes.
- 2.2.3. In the case of individual aid where the start-up aid is for routes linking an airport **not** located in a remote region to another airport, please provide the following passenger traffic data of the airports linked by the new route(s):
- (a) concerning airports with a commercial passenger traffic history of more than 2 financial years: the average annual passenger traffic during the two financial years preceding the year in which the aid is notified or granted;
- (b) concerning airports with a commercial passenger traffic history of less than two financial years: the forecasted average annual passenger traffic during the two financial years after the beginning of the operation of commercial passenger air traffic;

Please provide the data in the form of a table, as follows:

Year	Airport	Airport
Year	Passenger number	Passenger number
Year	Passenger number	Passenger number

Passenger numbers must be counted 'one-way' and for each individual route; e.g. passengers flying to the airport and back have to be counted twice. If an airport is part of a group of airports, passenger traffic data must be established on the basis of the individual airport.

- 2.2.4. In the case of aid schemes, please explain how the aid granting authority will assess on the basis of airport location, passenger traffic data and routes whether there is a need for State intervention in each individual case of start-up aid.
- 2.3. Appropriateness of the measure
- 2.3.1. In the case of individual aid: please demonstrate that the aid in question is appropriate to achieve the intended objective or resolve the problems it intends to address. In particular, please explain how the authorities have established that the same objective cannot be reached and the same problem cannot be resolved, with less distortive policy or aid instruments. For example, if the aid is granted in forms that provide a direct pecuniary advantage⁽²⁶⁰⁾, please demonstrate that other potentially less distortive forms of aid such as repayable advances or forms of aid that are based on debt or equity instruments⁽²⁶⁰⁾ are not appropriate.
- 2.3.2. In the case of individual aid where the beneficiary airline has prepared a business plan *ex-ante* for the route receiving the aid: please provide that business plan. That business plan must indicate whether that route has prospects of becoming profitable for the airline without public funding after 3 years.
- 2.3.3. In the case of individual aid in the absence of an *ex-ante* business plan for the route receiving the aid: please provide a document showing that the airline in question is irrevocably committed to operate the route for a period which is at least equal to the period during which it received start-up aid.
- 2.3.4. In the case of aid schemes, please explain how the aid granting authority will assess whether the measure is appropriate in each individual case of start-up aid.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- 2.4. Incentive effect and proportionality of the measure
- 2.4.1. In the case of individual aid, please explain why, if available on the basis of a business plan, there is the likelihood that in the absence of aid the level of economic activity of the airline at the airport concerned would not be expanded.

In the case of aid schemes, please describe the method applied by the aid granting authority in order to assess the likelihood that in the absence of aid the level of economic activity of the airline at the airport concerned would not be expanded.

2.4.2. In the case of individual aid, please confirm that the new route will start/started operating only after the application for aid was submitted to the granting authority and provide: (a) a copy of the aid application sent to the granting authority by the beneficiary; (b) documentary evidence of the start date of operations on the new route.

In the case of aid schemes, please confirm that eligible new routes will start operating only after the application for aid was submitted to the granting authority.

- 2.4.3. Please confirm that the aid will be granted for each route for a maximum period of three years.
- 2.4.4. Please specify the aid intensity, which is the total aid amount expressed as a percentage of eligible costs⁽²⁶¹⁾. Both figures must be expressed in net present value terms at the moment the aid is granted and before any deduction of tax or other charges.
- 2.4.5. Please specify the provisions made to avoid overcompensation and recover excess amounts of aid from the beneficiary airline.
- 2.4.6. Please specify the measures taken to avoid cross-subsidisation of other routes of the beneficiary airline to and from the airport or airports in question.
- 2.5. Avoidance of undue negative effects on competition and trade
- 2.5.1. In the case of individual aid: please confirm that the connection (for example, citypair) to be operated by the new air route is not already operated by a high-speed rail service or by another airport in the same catchment area under comparable conditions, notably in terms of length of journey. Please see also point 2.1.2.

In the case of aid schemes: Please explain how the aid granting authority will ensure that the condition in this point is satisfied in each individual case of start-up aid.

- 2.5.2. Please provide evidence that in planning to grant start-up aid to an airline for the new route, whether or not via an airport, the public authority's plans were made or will be made public in good time and with adequate publicity, to enable all interested airlines to offer their services.
- 2.5.3. Please confirm that the start-up aid in question cannot be combined with any other type of State aid granted for the operation of the same route.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

PART III.13.D

Supplementary Information Sheet on aid of a social character under Article 107(2)(a) of the Treaty, for air transport services

It is recommended that this supplementary information sheet is completed, in addition to the 'General information' form in Part I, for the notification of any social aid provision covered by the Guidelines on State aid to airports and airlines⁽²⁶²⁾.

- 1. Information on the beneficiary or beneficiaries, the project and the aid
- 1.1. Please explain the social objective(s) of the notified measure and why you considered that the measure achieves that objective or those objectives.
- 1.2. Details on the aid envisaged
- 1.2.1. Please describe (a) the method of awarding and distributing the aid to final consumers; (b) where applicable, please describe the legal entity or entities receiving the aid or transferring the aid to an intermediate entity responsible of distributing it to eligible final consumers.
- 1.2.2. Please describe the categories of eligible final consumers, for instance passengers with particular needs like children, people with disabilities, people on low incomes, students and elderly people, etc. (263).
- 1.2.3. Please confirm that the aid will effectively benefit eligible final consumers.
- 1.2.4. Please describe the routes eligible for aid.
- 1.2.5. Will the aid be granted for passenger transport on a route or on routes linking an airport or airports in a remote region⁽²⁶⁴⁾ with another airport or other airports within the European Economic Area?

- 1.2.6. If the answer to point 1.2.5 is 'yes', please describe the eligible geographic areas and routes.
- 1.2.7. Please specify the costs which are to be compensated by the aid and confirm that eligible costs are limited to the price of a return ticket, including all taxes and charges, invoiced by the carrier to the eligible final consumer.
- 1.2.8. Please confirm that the aid will be granted without discrimination between distribution channels, such as travel agencies, airline ground services and websites.
- 1.2.9. Please describe (a) the procedure by which the air service operator(s) will be selected, has been or have been selected; (b) eligibility conditions and (c) the selection criteria.
- 1.2.10. Please confirm that the aid will be granted without discrimination as to the origin of the services, that is to say irrespective of the airline operating the services.

П

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1.2.11. Please describe the controls and safeguards in force to ensure that the scheme is only used by eligible final consumers and that there is no overcompensation.

PART III.13.E

Supplementary Information Sheet on aid for maritime transport

It is recommended that this supplementary information sheet is completed, in addition to the 'General information' form in Part I, for the notification of any measure covered by the the Community guidelines on State aid to maritime transport (cf.) (the Guidelines').

1. **Information on the type of scheme**

Does the scheme constitute or include:

- (a)□ Tonnage Tax
- (b)□ a reduction in social contributions
- (c) \square a reduction in local taxes
- (d)□ a reduction in registration fees
- (e)□ aids for training
- (f)□ aids for transferring lorries from roads to sea ways
- (g)□ a public service contract or award procedure thereof
- (h)□ aids of social character
- (i)□ other, please describe:

2. **Information on eligibility**

For categories (a) (b) (c) (d) (e) (f) under point 1, please reply to 2.2 to 2.7:

- 2.1. What are the eligibility criteria for companies?
- 2.2. What are the eligibility criteria for boats? In particular, is there an obligation in relation to the flag? What are the flag-related obligations for the fleet of companies entering tonnage taxation scheme after 17 January 2004? Do flag-related obligations apply to the entire fleet of the beneficiary company or only owned fleet and fleet chartered in on bare-boat terms?
- 2.3. What are the eligibility conditions for the fleet chartered out on bare-boat terms?
- 2.4. What are the eligibility conditions for time- or voyage-chartered fleet?
- 2.5. Where appropriate, what are the eligibility criteria for seafarers?
- 2.6. Describe the list of eligible activities. In particular, does the regime concern:

□ tug activities?	□ dredging activities?
-------------------	------------------------

More generally, are shipping activities other than maritime transport of goods and passengers covered by the aid scheme?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

2.7. What are the ring-fence measures to avoid spill-over into other activities of the same company?

For category (g) under point 1:

2.8. What are the public services obligations, the method for calculating the compensations, the different offers submitted in the tender and the reasons for the choice of the designated company?

For category (h) under point 1:

- 2.9. What are the routes concerned, the populations of users concerned and the conditions attached to the award of individual grants?
- 3. Information on aid intensity
- 3.1. What mechanisms are in place to ensure the respect of the aid ceiling specified in Chapter 11 of the Guidelines? How are the relevant records kept?

For category (a) under point 1, please reply to 3.2 to 3.7:

3.2. What are the rates used to calculate the taxable income per 100 NT?

Up to 1 000 NT...

Between 1 001 and 10 000 NT...

Between 10 001 and 20 000 NT...

More than 20 001 NT...

- 3.3. Are companies obliged to set up separate accounts when operating both eligible and non-eligible activities?
- 3.4. What treatment is applied to groups of companies and intra-group transactions?
- 3.5. To what extent are the revenues from ancillary activities covered by tonnage tax scheme?
- 3.6. Are there special taxation rules for ships entering tonnage taxation in a situation when their market value exceeds their tax value?
- 3.7. Is the normal tax level under Member States' general tax rules applied to remuneration of directors and shareholders of the shipping companies?

For categories (b) (c) and (d) under point 1, please reply to 3.8 to 3.10:

- 3.8. What is the aid intensity in terms of percentage of the social/fiscal contributions or of the tax or fees that the seafarer or the shipowner should have normally been subject to?
- 3.9. Or to what level in absolute terms have the contributions, fees or taxes referred to in point 3.8 been limited?
- 3.10. In case of dredgers and tugboats, is the aid strictly limited to maritime transport part of activities?
- 3.11. For category (e) under point 1: What is the aid intensity in terms of the cost of the training or the salary of the trainee?
- 3.12. For category (f) under point 1: What is the amount of aid per tonne kilometre transferred?
- 3.13. For category (h) under point 1: What is the amount of individual grants?

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

[F8PART III.14

SUPPLEMENTARY INFORMATION SHEET FOR STATE AID TO THE FISHERY AND AQUACULTURE SECTOR

This supplementary information sheet must be used in the case of the notification of an aid measure which is covered by the Guidelines for the Examination of State Aid to the Fishery and Aquaculture Sector⁽²⁶⁶⁾ ('the Guidelines').

Aquacu	lture Sector ⁽²⁶⁶⁾ ('the Guidelines').	
1.	Common assessment principles	
1.1.		wing common assessment principles? If yes, or atisfy the principle of incentive effect set out in tick the appropriate box:
	the aid measure contributes to a well-defined objective of common interest;	
	need for State intervention: the State aid measure is targeted towards a situation where aid can bring about a material improvement that the market cannot deliver itself, for example, by remedying a market failure;	
	appropriateness of the aid measure: the to address the objective of common in	e aid measure is an appropriate policy instrument nterest;
	incentive effect: the aid changes the behaviour of the undertaking(s) concerned in such a way that it engages in additional activity which it would not carry out without the aid or it would carry out in a restricted or different manner or location; or the aid not required to have an incentive effect under point (52) of the Guidelines;	
	proportionality of the aid (aid limited to the minimum necessary): the aid amount is limited to the minimum needed to induce the additional investment or activity in the area concerned;	
	avoidance of undue negative effects on competition and trade between Member States the negative effects of the aid are sufficiently limited, so that the overall balance of the measure is positive;	
		s, the Commission, economic operators, and the t acts and to pertinent information about the aid
1.2.		n attached to it, including its financing method everable part of the measure, entail a violation
□Yes		□No
1.3.	namely aid directly linked to the	es towards third countries or Member States, quantities exported, to the establishment and r to other costs linked to the export activity or is mestic over imported goods?
□Yes		□No

 $\Box Yes$

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

Please note that, if the answer to the questions set out in points 1.2 and 1.3 is yes, the aid is incompatible with the internal market, as set out in points (26) and (27) of the Guidelines.

1.4.		which is subject to an outstanding recovery order ecision declaring an aid illegal and incompatible	
	note that this does not apply to aid to mo Article 107(2)(b) of the Treaty.	uke good the damage caused by natural disasters	
□Yes		□No	
If the a	nswer is yes, please indicate the relevan	nt Commission decision:	
2.	Specific principles for the fishery and aquaculture sector		
2.1.	In the case of an aid scheme, are the applications inadmissible if they are from operators that have committed one or more of the infringements or offences set out in Article 10(1) of Regulation (EU) No 508/2014 of the European Parliament and the Council ⁽²⁶⁷⁾ or a fraud as set out in Article 10(3) of that Regulation during the period set out in the delegated acts adopted on the basis of Article 10(4) of that Regulation?		
	note that this principle does not apply in in Sections 4, 5.3, and 5.4 of the Guide	the case of aid that meets the specific conditions lines.	
□Yes		□No	
If the a	nswer is yes, please indicate the specifi	c provisions on inadmissibility:	
2.2.	In the case of individual aid, please confirm that the operator concerned has not committed one or more of the infringements or offences set out in Article 10(1) of Regulation (EU) No 508/2014 or a fraud as set out in Article 10(3) of that Regulation during the period set out in the delegated acts adopted on the basis of Article 10(4) of that Regulation?		
	note that this principle does not apply in Sections 4, 5.3, and 5.4 of the Guide	the case of aid that meets the specific conditions lines.	
□Yes		□№	
2.3.	Does the aid measure explicitly prov	ide that each undertaking must comply with the	

rules of the common fisheries policy (CFP) throughout the period of implementation of the project and for a period of five years after the final payment to the beneficiary?

 $\square No$

Please confirm that a beneficiary that has committed one or more of the infringements 2.4. set out in Article 10(1) of Regulation (EU) No 508/2014 throughout the period of

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	implementation of the project and for a period of five years after the final payment to the beneficiary must reimburse the aid.	
□Yes	□No	
2.5.	Please confirm that no aid is granted for activities that correspond to ineligible operations under Article 11 of Regulation (EU) No 508/2014.	
□Yes	□No	
2.6.	If the answer to the questions set out in Sections 2.3, 2.4 and 2.5 of this Supplementary Information Sheet is yes, please indicate the specific provisions of the relevant national legal act or acts that set out the conditions referred to in those questions:	
2.7.	If the aid measure is of the same kind as an operation that is eligible for funding under Regulation (EU) No 508/2014, does it comply with the relevant provisions in tha Regulation for that kind of operation, in particular, with the provisions on the intensity of public aid?	
□Yes	□No	

3. Contribution to a common objective

3.1. Does the aid measure fall under Sections 4, 5.1, 5.3 or 5.4 of the Guidelines and fulfil the specific conditions set out in the relevant Section?

□Yes	□No

If the answer is yes, please note that the Commission considers that the aid measure contributes to achieving an objective of common interest and disregard Sections 3.2 and 3.3.

- 3.2. Please identify the objective or objectives of common interest set out in Article 107(3) of the Treaty which the aid measure contributes towards achieving:
- 3.3. Please identify the objective or objectives of the CFP which the aid measure contributes towards achieving and clearly demonstrate how the aid measure contributes to achieving this objective or these objectives without negatively affecting other CFP objectives⁽²⁶⁸⁾:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

4.	Need for state intervention		
4.1.	Does the aid measure fall under Sections 4, 5.1, 5.3 or 5.4 of the Guidelines and full the specific conditions set out in the relevant Section?		
□Yes		□No	
	answer is yes, please note that t ention and disregard Sections 4.2	the Commission considers that there is a need for 2, 4.3 and 4.4.	State
4.2.		to be addressed with the aid measure and explain uations where it can bring about a material improve r on its own:	
4.3.	contributes to the efficient fu or whether and how, in the ca	ow the aid measure corrects market failures and the inctioning of markets and to enhancing competitive se of market outcomes which are unsatisfactory from iew, the aid is used to obtain a more desirable, equi	eness m an
4.4.		ow the aid promotes the rationalisation and efficiencector and aims at permanent improvements in orderasis of market factors:	
5.	Appropriateness of aid		
5.1.	Does the aid fall under Sections 4, 5.1, 5.3 or 5.4 of the Guidelines and fulfil the specific conditions set out in the relevant Section?		1 the
□Yes		□No	

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

5.2.		distortive policy instruments exist that would achieving the objectives of the CFP and why d:
5.3.	Has an impact assessment of the notifie	d aid measure been made?
□Yes		iNo
If the an	nswer is yes, please summarise its main co	onclusions:
5.4.	Please indicate the form of aid and expladistortion of competition and trade?	nin why that form is likely to generate the least
5.5.	example, direct grants, exemptions or compulsory charges), please demonstra of aid such as repayable advances or f	at provide a direct pecuniary advantage (for reductions in taxes, social security or other te why other potentially less distortive forms forms of aid that are based on debt or equity oans or interest rebates, state guarantees, or an arable terms) are less appropriate:
6.	Incentive effect	
6.1.	Is the aid compensatory in nature, such as aid that falls under Section 4, 5.3 or 5.4, and does it fulfil the specific conditions set out in the relevant Section, or does the aid fall under Section 5.6 of the Guidelines and meet the conditions set out in that Section?	
□Yes	Г	iNo
IC d		

If the answer is yes, please note that the aid is not required to have an incentive effect and disregard Sections 6.2 to 6.6.

6.2. Please demonstrate how the aid changes the behaviour of an undertaking in such a way that it engages in additional activity which it would not have carried out without the aid or would have carried out in a restrictive or different manner:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

6.3.		activity that the beneficiary would have incurred for the normal business risk of an economic	
□Yes		□No	
	inswer is yes, please note that the aid c (49) of the Guidelines).	annot be considered to have an incentive effect	
6.4.	Is the aid granted in respect of an operation that the beneficiary has already begun to implement before the aid application was submitted by the beneficiary to the national authorities?		
□Yes		□No	
	answer is yes, please note that under pered to have an incentive effect.	point (51) of the Guidelines the aid cannot be	
6.5.	Is the aid operating aid ⁽²⁶⁹⁾ or aid intended to facilitate the achievement of obligatory standards?		
□Yes		□No	
incomp		nt (50) of the Guidelines such aid is in principle exceptions are expressly provided for in Union fied other cases.	
	nswer is yes, please mention the provisio d justification for such aid:	ns which expressly authorise such aid or provide	
6.6.		tages, does the aid measure establish a right to riteria and without further exercise of discretion	
		ot apply in the case of fiscal successor schemes, ious schemes in the form of tax advantages.	
□Yes		□No	

7. **Proportionality of aid**

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

7.1.	does it fulfil the specific conditions se	th as aid that falls under Section 4, 5.3 or 5.4, and et out in the relevant Section, or does the aid fall and meet the conditions set out in that Section?
□Yes		□No
<i>If the ar</i> 7.2 to 7		deemed to be proportionate and disregard points
7.2.		the aid amount corresponds to the net extra costs e area concerned, compared to the counterfactual
7.3.	Regulation (EU) No 508/2014, does the	as an operation that is eligible for funding under ne amount of aid exceed the applicable maximum the 95 of and Annex I to that Regulation?
□Yes		□No
If the ar		stification for and the indispensability of the aid:
□Yes		□No
	nswer is yes, does the total amount of p um aid intensities referred to in the Gui	ublic funding for an activity exceed the relevant delines?
□Yes		□No
8.	Effects on competition and trade	
8.1.	Does the aid fall under Section 4, 5.1, 5.3 or 5.4 of the Guidelines and fulfil the conditions set out in the relevant Section?	
□Yes		□No

If the answer is yes, please note that the Commission considers that the negative effects on competition and trade are limited to the minimum and disregard Sections 8.2 and 8.3.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

8.2. Please demonstrate how the negative effects of the aid measure in terms of distortions of competition and impact on trade between Member States are limited to the minimum and outweighed by the positive effects in terms of its contribution to achieving the objective of common interest. In the case of an aid scheme, please take into account the levels of distortions on a cumulative basis rather than on an individual level only as well as the size of the projects concerned, the individual and cumulative aid amounts, the expected beneficiaries, as well as the characteristics of the targeted sector. In the case of individual aid, please address the negative effects linked to the prevention of exit from the market and substantial market power and provide evidence permitting the identification of the relevant product market, the geographical market, the competitors, as well as the customers and consumers concerned:

Q	3	Hagan	impact assessme	nt of the	notified aid	l maggura haar	made?
ð.		Has an	impact assessme	nt of the	e notified aid	i measure neer	ı made?

□Yes	□No
------	-----

If the answer is yes, please summarise its main conclusions:

9. Transparency

- 9.1. Will the Member State publish at least the following information on a comprehensive State aid website at national or regional level?
- (a) the full text of the aid scheme and its implementing provisions or the legal basis for individual aid, or a link to it;
- (b) the granting authority or authorities;
- the identity of the individual beneficiaries, the form and amount of aid granted to each beneficiary, the date of granting the aid, the type of undertaking (SME/large enterprise), the region in which the beneficiary is located (at NUTS level II), and the principal economic sector in which the beneficiary has its activities (at NACE group level). (Such publication requirement can be waved with respect to individual aid awards not exceeding EUR 30 000. In the case of aid schemes in the form of tax advantages the information can be provided in the following ranges (in EUR million): 0,03 to 0,5; above 0,5 to 1; above 1 to 2; above 2.)

□Yes	□No

 $\Box Yes$

Document Generated: 2024-06-28

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

9.2.	Please confirm that such information will be:		
(a)	published after the decision to grant the aid has been taken;		
(b)	kept for at least 10 years;		
(c)	available for the general public without restrictions ⁽²⁷⁰⁾ .		
□Yes	□No		
Please 2017 ⁽²⁷	note that Member States are not required to publish the information before 1 July		
10.	Categories of aid		
10.1.	Please choose the Section of the Guidelines under which the aid should be assessed and provide detailed information for the chosen option in Sections 11 to 18 of this Supplementary Information Sheet:		
	Section 4.1 of the Guidelines: Aid to make good the damage caused by natural disasters and exceptional occurrences		
	Section 5.1 of the Guidelines: Aid for categories of measures covered by a block exemption Regulation		
	Section 5.2 of the Guidelines: Aid falling within the scope of certain horizontal guidelines		
	Section 5.3 of the Guidelines: Aid to make good the damage caused by adverse climatic events		
	Section 5.4 of the Guidelines: Aid for the costs of prevention, control and eradication of animal diseases in aquaculture		
	Section 5.5 of the Guidelines: Aid financed through parafiscal charges		
	Section 5.6 of the Guidelines: Operating aid in outermost regions		
	Section 5.7 of the Guidelines: Aid for other measures		
11.	Aid to make good the damage caused by natural disasters and exceptional occurrences		
to make	ction must be filled out in the case of the notification of an aid measure which is designed e good the damage caused by natural disasters or exceptional occurrences, as set out in a 4.1 of the Guidelines.		
11.1.	Is the aid measure an <i>ex ante</i> framework scheme to compensate for the damage caused by earthquakes, avalanches, landslides, floods, tornadoes, hurricanes, volcanic eruptions and wild fires of natural origin?		

 $\square No$

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

□Yes		□No
11.11.	Are the costs of the damage assessed by a public authority, by an independent expert recognised by the granting authority, or by an insurance undertaking?	
□Yes		□No
11.10.	Are only the costs of the damage incurred as a direct consequence of the natural disaster or exceptional occurrence eligible for aid?	
11.9.	Please specify which type of damage to assets, loss of income):	is compensated (for example, material damage
11.8.	Please submit an assessment as predundertaking:	cise as possible of the damage incurred by the
11.7.	Please demonstrate the direct causal link between the natural disaster or the exceptional occurrence and the damage suffered by the undertaking:	
□Yes		□No
11.6.	Is the aid paid directly to the undertaking concerned?	
□Yes		□No
11.5.	Has the competent authority of the Member State formally recognised the character of the event as a natural disaster or as an exceptional occurrence?	
11.4.	Please indicate the last date on which aid may be paid out.	
11.3.	When did the event referred to in Section 11.1 occur?	
11.2.	Which type of natural disaster or exceptional occurrence caused (or, in the case of ar <i>ex ante</i> framework aid scheme, could cause) the damage for which compensation is provided for?	

11.12.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

If the answer is yes, please specify which body assesses the costs.

Please describe how the damage is calculated:

11.13.	In the case of material damage to assets, is it calculated on the basis of the repair costs or the economic value of the affected asset before the natural disaster or exceptional occurrence?	
□Yes		□No
11.14.	In the case of material damage to assets, does it exceed the repair cost or the decrease in fair market value caused by the natural disaster or the exceptional occurrence?	
□Yes		□No
11.15.	In the case of a loss of income, is it calculated by subtracting:	
(a)	the result of multiplying the quantity of the fishery and aquaculture products produced in the year of the natural disaster or exceptional occurrence, or in each following year affected by the full or partial destruction of the means of production, by the average selling price obtained during that year, from	
(b)	the result of multiplying the average annual quantity of fishery and aquaculture products produced in the three-year period preceding the natural disaster or exceptional occurrence or a three-year average based on the five-year period preceding the natural disaster or exceptional occurrence, excluding the highest and lowest entry, by the average selling price obtained?	
□Yes		□No
11.16.	Is the damage calculated at the level of the individual beneficiary?	
□Yes		□No
11.17.	Is the aid, taken together with any other payments received to compensate the damage, including payments under insurance policies, limited to 100 % of the eligible costs?	
□Yes		□No

Status: Point in time view as at 31/01/2020.
Changes to legislation: There are currently no known outstanding effects for the
Commission Regulation (EC) No 794/2004. (See end of Document for details)

		delines.
□Yes	es □No	
11.19.	Please provide other information considered relevant for assessing the aid measure under this Section:	
12.	Aid for categories of measures covered by a block exemption Reg	ulation
This Se		which is of the
same k interna the Gui is of the natural	Section must be filled out in the case of the notification of an aid measure kind as aid falling within a category of aid that can be considered compraid market under one of the Regulations on group exemptions referred to in Guidelines, as set out in Section 5.1 of the Guidelines. In the case of an aid the same kind as aid falling within the category of aid to make good the day aral disasters referred to in Article 44 of Commission Regulation (EU) Notes fill out Section 11.	patible with the point (19)(a) of measure which mage caused by
same k interna the Gui is of the natural	e kind as aid falling within a category of aid that can be considered compraint market under one of the Regulations on group exemptions referred to in Guidelines, as set out in Section 5.1 of the Guidelines. In the case of an aid of the same kind as aid falling within the category of aid to make good the dateral disasters referred to in Article 44 of Commission Regulation (EU) Notes fill out Section 11.	patible with the point (19)(a) of measure which mage caused by 1388/2014 ⁽²⁷²⁾ , and be considered
same k interna the Gui is of the natural please	e kind as aid falling within a category of aid that can be considered compared market under one of the Regulations on group exemptions referred to in Guidelines, as set out in Section 5.1 of the Guidelines. In the case of an aid of the same kind as aid falling within the category of aid to make good the dataral disasters referred to in Article 44 of Commission Regulation (EU) Notes fill out Section 11. Is the aid of the same kind as aid falling within a category of aid that category of	patible with the point (19)(a) of measure which mage caused by 1388/2014 ⁽²⁷²⁾ , and be considered
same k interna the Gut is of the natural please 12.1.	e kind as aid falling within a category of aid that can be considered compared market under one of the Regulations on group exemptions referred to in Guidelines, as set out in Section 5.1 of the Guidelines. In the case of an aid of the same kind as aid falling within the category of aid to make good the dataral disasters referred to in Article 44 of Commission Regulation (EU) Notes fill out Section 11. Is the aid of the same kind as aid falling within a category of aid that category of	patible with the point (19)(a) of measure which mage caused by 1388/2014 ⁽²⁷²⁾ , an be considered coup exemptions
same k interna the Gut is of the natural please 12.1.	e kind as aid falling within a category of aid that can be considered comparant market under one of the Regulations on group exemptions referred to in Guidelines, as set out in Section 5.1 of the Guidelines. In the case of an aid of the same kind as aid falling within the category of aid to make good the dataral disasters referred to in Article 44 of Commission Regulation (EU) Notes fill out Section 11. Is the aid of the same kind as aid falling within a category of aid that category of	patible with the point (19)(a) of measure which mage caused by 1388/2014 ⁽²⁷²⁾ , an be considered oup exemptions allation:

12.3. Please provide other information considered relevant for assessing the aid measure under this Section:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

13. Aid falling within the scope of certain horizontal guidelines

This Section must be filled out in the case of the notification of an aid measure which falls within the scope of certain horizontal guidelines or other instruments adopted by the Commission, as set out in Section 5.2 of the Guidelines.

13.1.	Does the aid fall within the scope of adopted by the Commission ⁽²⁷³⁾ ?	certain horizontal guidelines or other instruments
□Yes		□No

If the answer is yes, please indicate the relevant horizontal guidelines or instruments and the relevant provisions of those acts and demonstrate that the aid fulfils all the criteria set out in the relevant provisions of those acts.

13.2. Please provide other information considered relevant for assessing the aid measure under this Section:

14. Aid to make good the damage caused by adverse climatic events

This Section must be filled out in the case of the notification of an aid measure which is designed to make good the damage caused by adverse climatic events, as set out in Section 5.3 of the Guidelines. In the case of aid of the same kind as aid falling within the category of aid to mutual funds for adverse climatic events referred to in Article 20 of Regulation (EU) No 1388/2014, please fill out Section 12.

14.1. Is the aid measure an *ex ante* framework scheme to compensate for the damage caused by adverse climatic events?

□Yes	□No

(If the answer is yes, please disregard Sections 14.3 to 14.6 and 14.9.)

- 14.2. Which type of adverse climatic event caused (or, in the case of an *ex ante* framework aid scheme, could cause) the damage for which compensation is provided for?
- 14.3. When did the event referred to in Section 14.1 occur?
- 14.4. Please indicate the last date on which aid may be paid out.
- 14.5. Does the damage caused by the adverse climatic event amount to more than 30 % of the average annual turnover, calculated on the basis of the preceding three calendar years

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	or a three-year average based on the event, excluding the highest and the	five-year period preceding the adverse climatic lowest entry?
□Yes		□No
	nswer is yes, please provide detailed in 14.5 is met:	nformation showing that the condition set out in
14.6.	Please demonstrate the direct causal damage suffered by the undertaking:	link between the adverse climatic event and the
14.7.	Regulation (EU) No 508/2014, pleas	rse climatic events referred to in Article 35(1) of se justify why the Member State intends to grant on being paid through mutual funds for adverse nat Regulation:
14.8.	Is the aid paid directly to the underta	king concerned?
□Yes		□No
14.9.	Please submit an assessment as pre potential beneficiaries:	cise as possible of the damage incurred by the
14.10.	Please specify which type of damage to assets, loss of income):	e is compensated (for example, material damage
14.11.	Are only the costs of the damage i climatic event eligible for aid?	ncurred as a direct consequence of the adverse
□Yes		□No

14.12. Is the damage assessed by a public authority, by an independent expert recognised by the granting authority, or by an insurance undertaking?

	Document Generated: 2024-06
Status: Point in time	view as at 31/01/2020.
Changes to legislation: There are curr	ently no known outstanding effects for the
Commission Regulation (EC) No 794	/2004. (See end of Document for details)
□Yes	□No

14.13. Please describe how the damage is calculated:

If the answer is yes, please specify which body assesses the costs:

14.14. In the case of material damage to assets, is it calculated on the basis of the repair costs or economic value of the affected asset before the adverse climatic event?

□Yes □No

14.15. In the case of material damage to assets, does it exceed the repair cost or the decrease in fair market value caused by adverse climatic event?

□Yes □No

14.16. In the case of material damage to assets, has the damage resulted in a loss of production that amounts to more than 30 % of the average turnover, calculated on the basis of the preceding three calendar years or a three-year average based on the five-year period preceding the adverse climatic event, excluding the highest and lowest entry?

□Yes □No

If the answer is yes, please provide detailed information showing that the condition set out in point 14.15 is met:

- 14.17. In the case of a loss of income, is it calculated by subtracting:
- (a) the result of multiplying the quantity of the fishery and aquaculture products produced in the year of the adverse climatic event, or in each following year affected by the full or partial destruction of the means of production, by the average selling price obtained during that year, from
- (b) the result of multiplying the average annual quantity of fishery and aquaculture products produced in the three-year period preceding the adverse climatic event or a three year average based on the five-year period preceding the adverse climatic event, excluding the highest and lowest entry, by the average selling price obtained?

□ Yes □NO	□Yes	□No
-------------	------	-----

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

14.18.	Is the damage calculated at the level	of the individual beneficiary?
□Yes		□No
14.19.		ner payments received to compensate the damage policies, limited to 100 % of the eligible costs?
□Yes		□No
14.20.		emes, please confirm that the Member State wil a set out in point (130) of the Guidelines.
□Yes		□No
15.	Aid for the costs of prevention, c	ontrol and eradication of animal diseases in
to supp aquacu falling	ction must be filled out in the case of ort costs relating to the prevention, lture, as set out in Section 5.4 of the Gu	the notification of an aid measure which aims control and eradication of animal diseases in idelines. In the case of aid of the same kind as aid ealth and welfare measures referred to in Article fill out Section 12.
15.1.	Is the aid measure an <i>ex ante</i> fram prevention, control and eradication of	nework scheme to support costs relating to the of animal diseases in aquaculture?
□Yes		□No
(If the a	nswer is yes, please disregard Sections	s 15.5, 15.6 and 15.9.)
15.2.	the World Organization for Animal H	isease(s) included in the list of animal diseases of lealth, Annex II to Regulation (EU) No 652/2014 the Council (274) or Part II of Annex IV to Council (granted:
Please	note that in the case of a disease or di	iseases included in the list of animal diseases o

Please note that in the case of a disease or diseases included in the list of animal diseases of the World Organization for Animal Health, the version of the list in force at the time of the notification of the aid measure applies. If the aid has already been granted or paid, in the case of individual aid, the version of the list published at the time of granting or payment of the aid applies and, in the case of an aid scheme, the list published at the time of the commencement of the scheme applies.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

15.3.	Is the aid granted as a part of a progra prevention, control, or eradication of	amme at Union, national, or regional level for the animal diseases?
□Yes		□No
If the a	nswer is yes, please indicate the relevan	nt programme and specific provisions:
15.4.	Is the aid granted as a part of emerger authority?	ncy measures imposed by the competent national
□Yes		□No
If the a	when did the costs relating to the	
13.3.	diseases in aquaculture occur?	prevention, control and eradication of animal
15.6.	Please indicate the last date on which	n aid may be paid out.
15.7.	Is the aid paid directly to the undertain	king concerned?
□Yes		□No
15.8.	Please confirm that no aid is granted disease deliberately or by negligence	if it is established that the beneficiary caused the
□Yes		□No
If the ai 15.8:	nswer is yes, please indicate the provisio	ons that set out the condition referred to in Section
15.9.	Please submit an assessment as pre- potential beneficiaries:	cise as possible of the damage incurred by the

- 15.10. Please indicate which of the following costs are eligible for compensation. Costs related to:
- (a)□ health checks, analyses, tests, and other screening measures;

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (b)□ the purchase, storage, administration or distribution of vaccines, medicines, and substances for the treatment of animals;
- (c)□ the slaughtering, culling, and destruction of animals;
- (d)□ the destruction of animal products and of products linked to them;
- (e)□ the cleaning and disinfection of the holding and equipment;
- (f) the damage incurred due to the slaughtering, culling, or destruction of animals, animal products, and products linked to them, limited to the market value of such animals and products if they had not been affected by the disease;
- (g)□ loss of income due to difficulties in restocking;
- (h)□ other costs incurred due to animal diseases in aquaculture.

In the case of (h), please specify the costs and justify why such costs should be eligible.

Please note that under point (110)(h) of the Guidelines, costs other than those referred to in (a) to (g), are only eligible in exceptional and duly justified cases.

15.11. Is the aid, taken together with any other payments received to compensate the damage, including payments under insurance policies, limited to 100 % of the eligible costs?

□Yes	□No

15.12. In the case of *ex ante* framework schemes, please confirm that the Member State will comply with the reporting obligation set out in point (130) of the Guidelines.

1 7	No

15.13. Please provide other information considered relevant for assessing the aid measure under this Section:

16. Aid financed through parafiscal charges

This Section must be filled out in the case of the notification of an aid measure which is financed by special charges imposed on certain fishery or aquaculture products irrespective of their origin, in particular, parafiscal charges, as set out in Section 5.5 of the Guidelines.

16.1. Is the aid scheme financed by special charges imposed on certain fishery or aquaculture products irrespective of their origin, in particular parafiscal charges?

□Yes	□No

If the answer is yes, please provide details on how the aid scheme is financed:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

16.2.	Does the aid equally benefit domestic	and imported products?	
□Yes	s □No		
If the aı	nswer is yes, please show how the schem	e benefits both domestic and imported products:	
16.3.	Please indicate how the funds acquire	ed through parafiscal charges are used:	
16.4.	Please provide other information considered relevant for assessing the aid measure under this Section:		
17.	Operating aid in outermost regions		
operati in those	ng aid granted in outermost regions wi	notification of an aid measure which constitutes the a view to alleviating the specific constraints insularity, and extreme remoteness, as set out in	
17.1.	Does the aid constitute operating aid granted in outermost regions with a view to alleviate the specific constraints in those regions as a result of their isolation, insularity, and extreme remoteness?		
□Yes		□No	
	nswer is yes, please describe the kind of egion or regions:	of operating aid that is granted and indicate the	
17.2.	Please indicate the specific constraint alleviate and describe how the aid int	ts in the region or regions which the aid aims to ends to achieve this goal.	

Please note that under point (113) of the Guidelines only constraints resulting from the isolation,

insularity, and extreme remoteness of outermost regions, may be taken into account.

17.3. Please determine the additional costs that result from the specific constraints and the method of their calculation and show that the aid does not go beyond what is necessary to alleviate the specific constraints in the outermost regions.

Status: Point in time view as at 31/01/2020.
Changes to legislation: There are currently no known outstanding effects for the
Commission Regulation (EC) No 794/2004. (See end of Document for details)

17.4.	types of public intervention, including costs in outermost regions for fishery	does the Member State take into account other ag, if applicable, the compensation of additional and aquaculture products under Articles 70, 71 14 and aid for implementing compensation plans
	under Article 73 of that Regulation?	
□Yes		□No
If the an	swer is yes, please specify how overco	ompensation is avoided:
17.5.	Please provide other information counder this Section:	nsidered relevant for assessing the aid measure
18.	Aid for other measures	
correspo	ond to one of the types of aid referred	te notification of an aid measure which does not to in Sections 4 or 5.1 to 5.6 of the Guidelines, nds to provide or provides, as set out in Section
18.1.		ovide or does it provide an aid which does not d referred to in Sections 4 or 5.1 to 5.6 of the
□Yes		□No

18.2. Please describe the aid measure and its objectives in detail:

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

18.3. In addition to the information provided for in Sections 1 to 9, please provide any other information which clearly demonstrates that the aid complies with the principles set out in Section 3 of the Guidelines:

]

Textual Amendments

F8 Substituted by Commission Regulation (EU) 2016/2105 of 1 December 2016 amending Annex I to Regulation (EC) No 794/2004 as regards the form to be used for the notification of State aid to the fishery and aquaculture sector.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

[F2ANNEX II

SIMPLIFIED NOTIFICATION FORM

This form may be used for the simplified notification pursuant to Article 4(2) of Commission Regulation (EC) No 794/2004 implementing Council Regulation (EC) No 659/1999 laying down detailed rules for the application of Article 93 of the EC Treaty (1).

1.	Prior	approved aid scheme (2).	
1.1.	Aid nu	umber allocated by the Commission:	
1.2.	Title:		
1.3.	Date of	of approval [by reference to the letter of the Commission SG()D/]:	
1.4.	Publication in the Official Journal of the European Union:		
1.5.	Primary objective (please specify one):		
1.6.	Legal	basis:	
1.7.	Overall budget:		
1.8.	Durati	ion:	
2.	Instrument subject to notification		
		New budget (please specify the overall as well as the annual budget in the respective national currency):	
		New duration (please specify the starting date from which the aid may be granted and the last date until which the aid may be granted):	
		Tightening of criteria (please indicate if the amendment concerns a reduction of aid intensity or eligible expenses and specify details):	
3.	Validi	ty of commitments	
		Please confirm that the commitments provided by the Member State for the purposes of the prior approved aid scheme are valid in their entirety also for the new notified measure.	
Pleas	e attacl	h a copy (or a web link) of the relevant extracts of the final text(s) of the legal basis.	

⁽¹⁾ Commission Regulation (EC) No 794/2004 of 21 April 2004 implementing Council Regulation (EC) No 659/1999 laying down detailed rules for the application of Article 93 of the EC Treaty (OJ L 140, 20.4.2004, p. 1). Regulation as last amended by Regulation No 1935/2006 (OJ L 407, 30.12.2006, p. 1).

⁽²⁾ If the aid scheme has been notified to the Commission on more than one occasion, please provide details for the latest complete notification that has been approved by the Commission.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

[F1ANNEX III A

Standardised reporting format for existing State aid

(This format covers all sectors except agriculture)

In order to further simplify and improve the annual reporting of State aid, the Commission provides Member States, by 1 March each year, with a pre-formatted online and access based platform, containing detailed information on all existing aid schemes and individual aid. Member States must check and complete that information by 30 June of the same year. That will enable the Commission to publish State aid data in year t⁽²⁷⁶⁾ for the reporting period t-1.

The bulk of the information in the platform is pre-completed by the Commission on the basis of data provided at the time of approval of the aid. Member States will be required to check and, where necessary, modify the details for each scheme or individual aid, and to add the annual expenditure for the latest year (t-1).

Information such as the objective of the aid and the sector to which the aid is directed must refer to the time at which the aid is approved and relate to the legal instrument which forms the basis for the approval of the aid.

The following information must be included.

- (1) Title
- (2) Aid number
- (3) Previous aid numbers (e.g., following the renewal of a scheme)
- (4) Sector

The sectoral classification shall be based largely on NACE⁽²⁷⁷⁾ at the [three-digit level.]

- (5) Objective
- (6) Region(s)

Aid may, at the time of approval, be exclusively earmarked for a specific region or group of regions according to subparagraphs (a) or (c) of Article 107(3) TFEU.

(7) Category of aid instrument(s)

A distinction must be made between several categories (Grant, Subsidised services, Interest rate subsidy, Loan, Guarantee, Tax advantage, Repayable advances, Equity participation, Other)

(8) Type of aid

A distinction must be made between three categories: Scheme, Individual application of a scheme, Individual aid awarded outside of a scheme (ad hoc aid)

(9) Expenditure

As a general rule, figures should be expressed in terms of actual expenditure (or actual revenue foregone in the case of tax advantages). Where payments are not available, commitments or budget appropriations must be provided and flagged accordingly. Separate figures must be provided for each aid instrument within a scheme or individual aid for example grant and loans. Figures must be expressed in the national

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

currency in application at the time of the reporting period. Expenditure must be provided for t-1, t-2, t-3, t-4 and t-5.

ANNEX III B

Standardised reporting format for existing State aid

(This format covers the agricultural sector)

In order to further simplify and improve the annual reporting of State aid, the Commission provides Member States, by 1 March each year, with a pre-formatted online and access based platform containing detailed information on all existing aid schemes and individual aid. Member States must check and complete that information by 30 June of the same year. That will enable the Commission to publish State aid data in year t⁽²⁷⁸⁾ for the reporting period t-1.

The bulk of the information in the platform will be pre-completed by the Commission on the basis of data provided at the time of approval of the aid. Member States will be required to check and, where necessary, modify the details for each scheme or individual aid, and to add the annual expenditure for the most recent year, that is to say t-1.

Information such as the objective of the aid and the sector to which the aid is directed must refer to the time at which the aid is approved and relate to the legal instrument which forms the basis for the approval of the aid.

The following information must be included.

- (1) Title
- (2) Aid number
- (3) Previous aid numbers (e.g., following the renewal of a scheme)
- (4) Sector

The sectoral classification shall be based largely on NACE⁽²⁷⁹⁾ at the [three-digit level.]

- (5) Objective
- (6) Region(s)

Aid may, at the time of approval, be exclusively earmarked for a specific region or group of regions according to subparagraphs (a) or (c) of Article 107(3) TFEU.

(7) Category of aid instrument(s)

A distinction must be made between several categories (Grant, Subsidised services, Interest rate subsidy, Loan, Guarantee, Tax advantage, Repayable advances, Equity participation, Other).

(8) Type of aid

A distinction must be made between three categories: Scheme, Individual application of a scheme, Individual aid awarded outside of a scheme (ad hoc aid).

(9) Expenditure

As a general rule, figures should be expressed in terms of actual expenditure (or actual revenue foregone in the case of tax advantages). Where payments are not available,

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

commitments or budget appropriations must be provided and flagged accordingly. Separate figures must be provided for each aid instrument within a scheme or individual aid for example grant and loans. Figures must be expressed in the national currency in application at the time of the reporting period. Expenditure must be provided for t-1, t-2, t-3, t-4 and t-5.

(10) Aid intensity and beneficiaries

Member States must indicate:

- the effective aid intensity of the support actually granted per type of aid and of region;
- the number of beneficiaries;
- the average amount of aid per beneficiary.]

ANNEX III C

INFORMATION TO BE CONTAINED IN THE ANNUAL REPORT TO BE PROVIDED TO THE COMMISSION

The reports shall be provided in computerised form. They shall contain the following information:

- 1. Title of aid scheme, Commission aid number and reference of the Commission decision
- 2. Expenditure. The figures have to be expressed in euros or, if applicable, national currency. In the case of tax expenditure, annual tax losses have to be reported. If precise figures are not available, such losses may be estimated. For the year under review indicate separately for each aid instrument within the scheme (e.g. grant, soft loan, guarantee, etc.):
- 2.1. amounts committed, (estimated) tax losses or other revenue forgone, data on guarantees, etc. for new assisted projects. In the case of guarantee schemes, the total amount of new guarantees handed out should be provided;
- 2.2. actual payments, (estimated) tax losses or other revenue forgone, data on guarantees, etc. for new and current projects. In the case of guarantee schemes, the following should be provided: total amount of outstanding guarantees, premium income, recoveries, indemnities paid out, operating result of the scheme under the year under review;
- 2.3. number of assisted projects and/or enterprises;
- 2.4. estimated overall amount of:
- aid granted for the permanent withdrawal of fishing vessels through their transfer to third countries:
- aid granted for the temporary cessation of fishing activities:
- aid granted for the renewal of fishing vessels;
- aid granted for modernisation of fishing vessels;
- aid granted for the purchase of used vessels;
- aid granted for socio-economic measures:

Status: Point in time view as at 31/01/2020.

- aid granted to make good damage caused by natural disasters or exceptional occurences;
- aid granted to outermost regions;
- aid granted through parafiscal charges;
- 2.5. regional breakdown of amounts under point 2.1. by regions defined as Objective 1 regions and other areas;
- 3. Other information and remarks.

Status: Point in time view as at 31/01/2020. **Changes to legislation:** There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

[F4ANNEX IV]

FORM FOR THE SUBMISSION OF COMPLAINTS CONCERNING ALLEGED UNLAWFUL STATE AID OR MISUSE OF AID

The mandatory fields are marked with a star "*".			
1.	Information regarding the complainant		
	First name:*		
	Surname:*		
	Address line 1:*		
	Address line 2:		
	Town/City:*		
	County/State/Province:		
	Postcode:*		
	Country:*		
	Telephone:		
	Mobile Telephone:		
	E-mail address:*		
	Fax		
2.	I am submitting the complaint on behalf of somebody (a person or a firm)		
	Yes* No*		
	If yes, please also provide the following information		
	Name of the person/firm you represent*:		
	Registration nr. of the entity:		
	Address line 1:*		
	Address line 2:		
	Town/City:*		
	County/State/Province:		
	Postcode:*		
	Country:*		
	Telephone 1:		
	Telephone 2:		
	E-mail address:*		
	Fax		

Please attach proof that the representative is authorised to act on behalf of this person/firm.*

Status: Point in time view as at 31/01/2020.

3.	Pleas	e select one of the following options, describing your identity*				
	(a)	Competitor of the beneficiary or beneficiaries				
	(b)	Trade association representing the interests of competitors				
	(c)	Non-governmental organisation				
	(d)	Trade union				
	(e)	EU citizen				
	(f)	Other, please specify				
	Pleas comp	e explain why and to what extent the alleged State aid affects your competitive position/the etitive position of the person/firm you represent. Provide as much concrete evidence as possible.				
	1999 Europ forma prese	the be aware that, by virtue of Article 20(2) of Council Regulation (EC) No 659/1999 of 22 March laying down detailed rules for the application of Article 108 of the Treaty on the Functioning of the bean Union, only interested parties within the meaning of Article 1(h) of that Regulation may submit all complaints. Therefore, in the absence of a demonstration that you are an interested party, the form will not be registered as a complaint, and the information provided therein will be kept as a market information.				
4.	Pleas	se select one of the following two options*				
		Yes, you may reveal my identity				
		No, you may not reveal my identity				
	If not,	please specify the reasons:				
	Confidentiality: If you do not wish your identity or certain documents or information to be disclosed, please indicate this clearly, identify the confidential parts of any documents and give your reasons. In the absence of any indication about confidentiality of your identity or certain documents or information, those elements will be treated as non-confidential and may be shared with the Member State allegedly granting the State aid. The information contained in points 5 and 6 cannot be designated as confidential.					
5.	Infor	mation regarding the Member State granting the aid*				
	Please be aware: the information provided under this point is regarded as non-confidential.					
	(a)	Country:				
	(b)	If known, specify which institution or body granted the alleged unlawful State aid:				
	. ,	Central government:				
		Region (please specify):				
		Other (please specify):				

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

6.	Information		46	_	- 1-4	
n	intormation	regarding	Tne :	alleded	aid	measure"

	mati	on regarding the aneged and measure
Plea	se be	aware: the information provided under this point is regarded as non-confidential.
(a)		ase provide a description of the alleged aid, and indicate in what form it was granted (loans, nts, guarantees, tax incentives or exemptions etc.).
(b)	For	what purpose was the alleged aid given (if known)?
(c)		at is the amount of the alleged aid (if known)? If you do not have the exact figure, please vide an estimate and as much justifying evidence as possible.
(d)		o is the beneficiary? Please give as much information as possible, including a description of main activities of the beneficiary/firm(s) concerned.
(e)	То	your knowledge, when was the alleged aid granted?
(f)	Ple	ase select one of the following options:
		According to my knowledge, the State aid was not notified to the Commission.
		According to my knowledge, the State aid was notified, but it was granted before the decision of the Commission. If known, please indicate the notification reference number or indicate when the aid was notified.
		According to my knowledge, the State aid was notified and approved by the Commission, but its implementation did not respect the applicable conditions. If known, please indicate the notification reference number or indicate when the aid was notified and approved.

According to my knowledge, the State aid was granted under a block exemption regulation,

but its implementation did not respect the applicable conditions.

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

7. Grounds of complaint*

Please note that, for a measure to qualify as State aid under Article 107(1) TFEU, the alleged aid has to be granted by a Member State or through State resources, it has to distort or threaten to distort competition by favouring certain undertakings or the production of certain goods, and affect trade between Member States.

a)	Please explain to what extent public resources are involved (if known) and, if the measure was not adopted by a public authority (but for instance by a public undertaking), please explain why, in your view, it is imputable to public authorities of a Member State.
b)	Please explain why, in your opinion, the alleged State aid is selective (i.e. favours certain commercial undertakings or the production of certain goods).
c)	Please explain how, in your opinion, the alleged State aid provides an economic advantage for the beneficiary or beneficiaries.
d)	Please explain why, in your view, the alleged State aid distorts or threatens to distort competition.
:)	Please explain why, in your view, the alleged aid affects trade between Member States.

Status: Point in time view as at 31/01/2020. Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

	se indicate the reasons why in your view the alleged aid is not compatible with the internal market.
	rmation on alleged infringement of other rules of European Union law and on other redures
(a)	If known, please indicate what other rules of European Union law you think have been infringed by the granting of the alleged aid. Please be aware that this does not imply necessarily that those potential infringements will be dealt with within the State aid investigation.
(b)	Have you already approached the Commission's services or any other European institution concerning the same issue? *
	Yes No
	If yes, please attach copies of correspondence.
c)	Have you already approached national authorities or national courts concerning the same issue? *
	Yes No
	If yes, please indicate which authorities or courts; also, if there has already been a decision or judgement, please attach a copy (if available); if, on the contrary, the case is still pending, please indicate its reference (if available).
(d)	Please provide any other information that may be relevant for the assessment of this case.

ANNEX III C

Document Generated: 2024-06-28

Status: Point in time view as at 31/01/2020.

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

10. Supporting documents

Please list any documents and evidence which are submitted in support of the complaint and add annexes if necessary

- Whenever possible, a copy of the national law or other measure which provides the legal basis for the payment of the alleged aid should be provided.
- Whenever possible, please attach any available evidence that the State aid was granted (e.g. press release, published accounts).
- If the complaint is submitted on behalf of someone else (a natural person or a firm) please attach
 proof that you as a representative are authorised to act.
- Where applicable, please attach copies of all previous correspondence with the European Commission or any other European or national institution concerning the same issue.
- If the issue has already been dealt with by a national court/authority, please attach a copy of the judgement/decision, if available.

I hereby declare that all the information in this form and annexes is provided in good faith.

Place, date and signature of complainant'

- (1) OJ L 83, 27.3.1999, p. 1. Regulation as amended by the 2003 Act of Accession.
- (2) OJ L 124, 8.6.1971, p. 1.
- (**3**) OJ L 300, 5.11.2002, p. 42.
- (4) [FICommission Regulation (EC) No 794/2004 of 21 April 2004 implementing Council Regulation (EU) 2015/1589 laying down detailed rules for the application of Article 108 of the Treaty on the Functioning of the European Union (OJ L 140, 30.4.2004, p. 1).]
- (5) [FINACE Rev. 2, or subsequent legislation amending or replacing it; NACE is the statistical classification of economic activities in the European Union as laid down in Regulation (EC) No 1893/2006 of the European Parliament and of the Council of 20 December 2006 establishing the statistical classification of economic activities NACE Revision 2 and amending Council Regulation (EEC) No 3037/90 as well as certain EC Regulations on specific statistical domains (OJ L 393, 30.12.2006, p. 1).]
- (6) [FICommission Recommendation of 6 May 2003 concerning the definition of micro, small and medium-sized enterprises (OJ L 124, 20.5.2003, p. 36).]
- (7) [FIIn the case of partner and linked enterprises, please note that the amounts reported for the aid beneficiary should take into account the number of employees and the financial data of the linked enterprises and/or partner enterprises.]
- (8) [F1As defined in the Guidelines on State aid for rescuing and restructuring non-financial undertakings in difficulty (OJ C 249, 31.7.2014, p. 1).]
- (9) [F1Regulation (EU) 2015/1017 of the European Parliament and of the Council of 25 June 2015 on the European Fund for Strategic Investments, the European Investment Advisory Hub and the European Investment Project Portal and amending Regulations (EU) No 1291/2013 and (EU) No 1316/2013 the European Fund for Strategic Investments (OJ L 169, 1.7.2015, p. 1).]
- (10) [F1Commission's registration number of the authorised or block-exempted scheme.]
- (11) [F1According to Article 1(e) of Council Regulation (EU) 2015/1589 of 13 July 2015 laying down detailed rules for the application of Article 108 of the Treaty on the Functioning of the European Union (OJ L 248, 24.9.2015, p. 9), individual aid means aid that is not awarded on the basis of an aid scheme and notifiable award of aid on the basis of a scheme.]
- (12) [F1The date of the legally binding commitment to award the aid.]
- (13) [FIIn the case of aid to the agriculture sector or fishery and aquaculture sector information on compliance with the common assessment principles is requested in Parts III.12 (Supplementary information sheet for aid to the agriculture and forestry sectors and in rural areas sector) and III.14 (Supplementary information sheet for aid to the fishery and aquaculture sector).]
- (14) [F1Communication from the Commission amending the Communications from the Commission on EU Guidelines for the application of State aid rules in relation to the rapid deployment of broadband networks, on Guidelines on regional State aid for 2014–2020, on State aid for films and other audiovisual works, on Guidelines on State aid to promote risk finance investments and on Guidelines on State aid to airports and airlines (OJ C 198, 27.6.2014, p. 30).]
- (15) [FIGrant/Interest rate subsidy, Loan/Repayable advances/Reimbursable grant, Guarantee, Tax advantage or tax exemption, Risk finance, Other (please specify). If the aid is granted through multiple aid instruments, the aid amount must be provided by instrument.]
- (16) [FISuch a requirement can be waived with respect to individual aid awards below EUR 500 000. For schemes in the form of tax advantages, the information on individual aid can be provided in the following ranges (in EUR million): [0.5-1]; [1-2]; [2-5]; [5-10]; [10-30]; [30 and more].]
- (17) [F1Overall amount of aid planned, expressed in full national currency amounts. For tax measures, estimated overall revenue loss due to tax concessions. If the average annual State aid budget of the scheme exceeds EUR 150 million, please fill in the section on evaluation.]
- (18) [FI In case of an average annual State aid budget of the scheme above EUR 150 million, please fill in the section on Evaluation of this notification form. The Evaluation requirement does not apply to aid schemes covered by the Supplementary information sheet on Agriculture aid.]

- (19) [F1Commission Regulation (EU) No 1407/2013 of 18 December 2013 on the application of Articles 107 and 108 of the Treaty on the Functioning of the European Union to *de minimis* aid (OJ L 352, 24.12.2013, p. 1) and Commission Regulation (EU) No 717/2014 of 27 June 2014 on the application of Articles 107 and 108 of the Treaty on the Functioning of the European Union to *de minimis* aid in the fishery and aquaculture sector (OJ L 190, 28.6.2014, p. 45).]
- (20) [FIUnion funding centrally managed by the Commission that is not directly or indirectly under the control of the Member State does not constitute State aid. Where such Union funding is combined with other public funding, only the latter will be considered for determining whether notification thresholds and maximum aid intensities are respected, provided that the total amount of public funding granted in relation to the same eligible costs does not exceed the maximum funding rates laid down in the applicable Union legislation.]
- (21) [F1The Evaluation requirement does not apply to aid schemes covered by the Supplementary information sheet on Agriculture aid.]
- (22) [F1For guidance, see the Commission Staff Working Document 'Common methodology for State aid evaluation', SWD(2014)179 final of 28.5.2014 under http://ec.europa.eu/competition/state_aid/modernisation/state_aid evaluation methodology en.pdf.]
- (23) [F1Council Regulation (EU) 2015/1589 of 13 July 2015 laying down detailed rules for the application of Article 108 of the Treaty on the Functioning of the European Union (OJ L 248, 24.9.2015, p. 9).]
- (24) [F1For guidance see Article 339 TFEU that refers to 'information about undertakings, their business relations or their cost components'. The Union courts have generally defined 'business secrets' as information 'of which not only disclosure to the public but also mere transmission to a person other than the one that provided the information may seriously harm the latter's interest' in Case T-353/94 *Postbank* v *Commission* ECLI:EU:T:1996:119, paragraph 87'.]
- (25) [F1Communication from the Commission to the Member States on the application of Articles 107 and 108 of the Treaty on the Functioning of the European Union to short-term export-credit insurance (OJ C 392, 19.12.2012, p. 1).]
- (26) [FI Guidelines on certain State aid measures in the context of the greenhouse gas emission allowance trading scheme post 2012 (OJ C 158, 5.6.2012, p. 4).]
- (27) [FICommunication from the Commission on the application, from 1 August 2013, of State aid rules to support measures in favour of banks in the context of the financial crisis (OJ C 216, 30.7.2013, p. 1).]
- (28) [FICommunication from the Commission Criteria for the analysis of the compatibility with the internal market of State aid to promote the execution of important projects of common European interest (OJ C 188, 20.6.2014, p. 4).]
- (29) [FICommunication from the Commission on the application of the European Union State aid rules to compensation granted for the provision of services of general economic interest (OJ C 8, 11.1.2012, p. 4).]
- (30) [F1Guidelines on regional State aid for 2014-2020 (OJ C 209, 23.7.2013, p. 1).]
- (31) [F1:The same or a similar activity' means an activity falling under the same class (four-digit numerical code) of the NACE Rev. 2 statistical classification of economic activities.]
- (32) [FIAs defined in the Guidelines on State aid for rescuing and restructuring non-financial undertakings in difficulty, (OJ C 249, 31.7.2014, p. 1).]
- (33) [F1OJ C 25, 26.1.2013, p. 1.]
- (34) [FIAs defined in Council Regulation (EC) No 723/2009 of 25 June 2009 on the Community legal framework for a European Research Infrastructure Consortium (ERIC) (OJ L 206, 8.8.2009, p. 1).]
- (35) [F1'SMEs' means undertakings that fulfil the conditions laid down in the Commission recommendation of 6 May 2003 concerning the definition of micro, small and medium-sized enterprises (OJ L 124, 20.5.2003, p. 36).]
- (36) I^{F1}The sole acquisition of the shares of an undertaking does not qualify as initial investment.]

- (37) [F1:The same or a similar activity' means an activity falling under the same class (four-digit numerical code) of the NACE Rev. 2 statistical classification of economic activities.]
- (38) [Fi Paragraph 94 RAG does not apply to SMEs or in the case of the acquisition of an establishment.]
- (39) [FIParagraph 101 RAG provides that intangible assets which are eligible for the calculation of the investments costs must remain associated with the assisted area concerned and must not be transferred to other regions. To that end, the intangible assets must fulfil the following conditions:
 - they must be used exclusively in the establishment receiving the aid:
 - they must be amortisable;
 - they must be purchased under market conditions from third parties unrelated to the buyer.

 Paragraph 102 RAG provides that the intangible assets must be included in the assets of the undertaking receiving the aid and must remain associated with the project for which the aid is awarded for at least five years (three years for small and medium enterprises).]
- (40) [F1For the methodology, refer to Commission notice on current State aid recovery interest rates and reference/discount rates for 25 Member States applicable as from 1 January 2007 (OJ C 317, 23.12.2006, p. 2) and the Communication from the Commission on the revision of the method for setting the reference and discount rates (OJ C 14, 19.1.2008, p. 6).]
- (41) [F1The aid amount(s) and eligible costs should be provided both in nominal and discounted values.]
- (42) [FIn order to do so, you could, for example, refer to the criteria enumerated in paragraph 40 RAG and/or to the business plan of the beneficiary.]
- (43) [FIThis question does not apply to subsidised loans, public equity-capital loans or public participations which do not meet the market investor principle, State guarantees containing elements of aid, or public support granted within the scope of *de minimis* rule.]
- (44) [FIFor example, direct grants, exemptions or reductions in taxes, social security or other compulsory charges, or the supply of land, goods or services at favourable prices, etc.]
- (45) [FIFor example, low-interest loans or interest rebates, state guarantees, the purchase of a shareholding or an alternative provision of capital on favourable terms.]
- (46) [FIThe time period considered for the calculation of the IRR should correspond to timeframes usually considered by the industry for similar projects.]
- (47) [FIAII relevant costs and benefits must be taken into account, including for example administrative costs, transport costs, training costs not covered by training aid and also wage differences. However, where the alternative location is in the EEA, subsidies granted in that other location are not to be taken into account.]
- (48) [F1Details of all classifications are available in the Eurostat classifications database, currently under the link http://ec.europa.eu/eurostat/data/classifications.]
- (49) [FIUnderperformance of the market will normally be measured compared to the EEA GDP over the last three years before the start of the project (benchmark rate); it can also be established on the basis of projected growth rates in the coming three to five years. Indicators may include the foreseeable future growth of the market concerned and the resulting expected capacity utilisation rates, as well as the likely impact of the capacity increase on competitors through its effects on prices and profit margins.]
- (50) [F1Guidelines on regional State aid for 2014-2020 (OJ C 209, 23.7.2013, p. 1).]
- (51) [FICommission Regulation (EU) No 651/2014 of 17 June 2014 declaring certain categories of aid compatible with the internal market in application of Articles 107 and 108 of the Treaty (OJ L 187, 26.6.2014, p. 1).]
- (52) [F1Commission Regulation (EU) No 1407/2013 of 18 December 2013 on the application of Articles 107 and 108 of the Treaty on the Functioning of the European Union to *de minimis* aid (OJ L 352, 24.12.2013, p. 1).]
- (53) [F1'The same or a similar activity' means an activity falling under the same class (four-digit numerical code) of the NACE Rev. 2 statistical classification of economic activities.]

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

- (54) [F1Communication from the Commission, EU Guidelines for the application of State aid rules in relation to the rapid deployment of broadband networks (OJ C 25, 26.1.2013, p. 1).]
- (55) [F1'SMEs' means undertakings that fulfil the conditions laid down in Commission recommendation of 6 May 2003 concerning the definition of micro, small and medium-sized enterprises (OJ L 124, 20.5.2003, p. 36).]
- (56) [F1The sole acquisition of the shares of an undertaking does not qualify as initial investment.]
- (57) [F1'The same or a similar activity' means an activity falling under the same class (four-digit numerical code) of the NACE Rev. 2 statistical classification of economic activities.]
- (58) [F1The notification threshold is defined in paragraph 20(n) RAG.]
- (59) [FIn the transport sector, expenditure on the purchase of transport equipment cannot be included in the uniform set of items of expenditure. Such expenditure is not eligible for initial investment.]
- (60) [FIThat provision does not have to apply for SMEs or in case of acquisition of an establishment.]
- (61) [FIParagraph 101 RAG provides that intangible assets which are eligible for the calculation of the investments costs must remain associated with the assisted area concerned and must not be transferred to other regions. To this end, the intangible assets must fulfil the following conditions:
 - they must be used exclusively in the establishment receiving the aid:
 - they must be amortisable;
 - they must be purchased under market conditions from third parties unrelated to the buyer.

Paragraph 102 RAG provides that the intangible assets must be included in the assets of the undertaking receiving the aid and must remain associated with the project for which the aid is awarded for at least five years (three years for SMEs).]

- (62) [F1Operating Programme or development programme defined in the context of the European Regional Development Fund (ERDF), the European Social Fund, the Cohesion Fund, the European Agricultural Fund for Rural Development or the European Maritime and Fisheries Fund (EMFF).]
- (63) [F1This question does not apply to subsidised loans, public equity-capital loans or public participations which do not meet the market investor principle, State guarantees containing elements of aid, or public support granted within the scope of *de minimis* rule.]
- (64) [F1For that purpose reference can be made, among others, to impact assessments of the proposed scheme or *ex post* evaluations of similar schemes.]
- (65) [F1For example, direct grants, exemptions or reductions in taxes, social security or other compulsory charges, or the supply of land, goods or services at favourable prices, etc.]
- (66) [FIFor example, low-interest loans or interest rebates, State guarantees, the purchase of a shareholding or an alternative provision of capital on favourable terms.]
- (67) [F1A counterfactual is credible if it is genuine and relates to the decision-making factors prevalent at the time of the decision by the beneficiary regarding the investment.]
- (68) [FI In order to enable the Commission to assess the likely negative effects the Member State could submit any impact assessment at its disposal as well as *ex-post* evaluations carried out for similar predecessor schemes (as specified in paragraph 125 RAG).]
- (69) [F1Guidelines on regional State aid for 2014-2020 (OJ C 209, 23.7.2013, p. 1).]
- (70) [F1'SMEs' means undertakings that fulfil the conditions laid down in Commission recommendation of 6 May 2003 concerning the definition of micro, small and medium-sized enterprises (OJ L 124, 20.5.2003, p. 36).]
- (71) [FIIn that respect please note that operating aid to compensate for additional transport costs of goods which have been produced in areas eligible for operating aid can only be granted in compliance with the General Block Exemption Regulation ('GBER') in force at the moment of the granting.]
- (72) [F1OJ C 198, 27.6.2014, p. 1.]

- (73) [FICommission Regulation (EU) No 651/2014 of 17 June 2014 declaring certain categories of aid compatible with the internal market in application of Articles 107 and 108 of the Treaty (OJ L 187, 26.6.2014, p. 1).]
- (74) [F1OJ C 249, 31.7.2014, p. 1.]
- (75) [FIDirective 2014/24/EU of the European Parliament and of the Council of 26 February 2014 on public procurement and repealing Directive 2004/18/EC (OJ L 94, 28.3.2014, p. 65), and Directive 2014/25/EU of the European Parliament and of the Council of 26 February 2014 on procurement by entities operating in the water, energy, transport and postal services sectors and repealing Directive 2004/17/EC (OJ L 94, 28.3.2014, p. 243).]
- (76) [F1Without prejudice to procedures that cover both the development and the subsequent purchase of unique or specialised products or services.]
- (77) [FIf the aid application is for an R & D project, it is not excluded that the potential beneficiary would have already carried out feasibility studies which are not covered by the request for aid.]
- (78) [F1OJ C 14, 19.1.2008, p. 6.]
- (**79**) [F1OJ C 249, 31.7.2014, p. 1.]
- (80) [FIThis refers in particular to the types of company mentioned in Annex I to Directive 2013/34/ EU of the European Parliament and of the Council of 26 June 2013 on the annual financial statements, consolidated financial statements and related reports of certain types of undertakings, amending Directive 2006/43/EC of the European Parliament and of the Council and repealing Council Directives 78/660/EEC and 83/349/EEC (OJ L 182, 29.6.2013, p. 19).]
- (81) [FIThis is the case when deduction of accumulated losses from reserves (and all other elements generally considered as part of the own funds of the company) leads to a negative cumulative amount that exceeds half of the subscribed share capital.]
- (82) [F1This refers in particular to the types of company mentioned in Annex II to Directive 2013/34/EU.]
- (83) [F1Communication from the Commission on the revision of the method for setting the reference and discount rates (OJ C 14, 19.1.2008, p. 6) or any future communication which might replace it.]
- (84) [FIncluding any such aid granted before the date on which the Commission began to apply the Guidelines, i.e. before 1.8.2014.]
- [85] [F1Please note that where less than 10 years have elapsed since the rescue aid or temporary restructuring support was granted or the restructuring period came to an end or implementation of the restructuring plan was halted (whichever occurred the latest), further rescue aid, restructuring aid or temporary restructuring support can be only granted: (a) where temporary restructuring support follows the granting of rescue aid as part of a single restructuring operation; (b) where restructuring aid follows the granting of rescue aid or temporary restructuring support as part of a single restructuring operation; (c) where rescue aid or temporary restructuring support has been granted in accordance with these guidelines and that aid was not followed by restructuring aid, if: (i) it could reasonably have been believed that the beneficiary would be viable in the long term when the aid pursuant to these guidelines was granted, and (ii) new rescue or restructuring aid or temporary restructuring support becomes necessary after at least five years due to unforeseeable circumstances for which the beneficiary is not responsible; (d) in exceptional and unforeseeable circumstances for which the beneficiary is not responsible.]
- (86) [F1OJ C 249, 31.7.2014, p. 1.]
- (87) [FIThis refers in particular to the types of company mentioned in Annex I to Directive 2013/34/ EU of the European Parliament and of the Council of 26 June 2013 on the annual financial statements, consolidated financial statements and related reports of certain types of undertakings, amending Directive 2006/43/EC of the European Parliament and of the Council and repealing Council Directives 78/660/EEC and 83/349/EEC (OJ L 182, 29.6.2013, p. 19).]
- (88) [FIThis is the case when deduction of accumulated losses from reserves (and all other elements generally considered as part of the own funds of the company) leads to a negative cumulative amount that exceeds half of the subscribed share capital.]
- (89) I^{FI}This refers in particular to the types of company mentioned in Annex II to Directive 2013/34/EU.]

- (90) [FIRestructuring may involve one or more of the following elements: the reorganisation and rationalisation of the beneficiary's activities to a more efficient basis, typically involving withdrawal from loss-making activities, restructuring of those existing activities that can be made competitive again and, possibly, diversification towards new and viable activities. It typically also involves financial restructuring in the form of capital injections by new or existing shareholders and debt reduction by existing creditors.]
- (91) [FLong-term viability is achieved when an undertaking is able to provide an appropriate projected return on capital after having covered all its costs including depreciation and financial charges. The restructured undertaking should be able to compete in the marketplace on its own merits.]
- (92) [F¹The alternative scenario may concern, for example: debt reorganisation, asset disposal, private capital raising, sale to a competitor or break-up, in each case either through entry into an insolvency or reorganisation procedure or otherwise.]
- (93) [F1Please also see point 56 of the Guidelines.]
- (94) [F1Please also see point 64 of the Guidelines.]
- (95) [FIFor example, where the aid to be granted enhances the beneficiary's equity position, the own contribution should similarly include measures that are equity-enhancing, such as raising fresh equity from incumbent shareholders, the write-down of existing debt and capital notes or the conversion of existing debt to equity, or the raising of new external equity on market terms.]
- (96) [FIF or example where the State provides grants, injects capital or writes off debt.]
- (97) [FIncluding any such aid granted before the date on which the Commission began to apply the Guidelines, i.e. before 1.8.2014.]
- (98) [FIPlease note that where less than 10 years have elapsed since the rescue aid or temporary restructuring support was granted or the restructuring period came to an end or implementation of the restructuring plan was halted (whichever occurred the latest), further rescue aid, restructuring aid or temporary restructuring support can only be granted: (a) where temporary restructuring support follows the granting of rescue aid as part of a single restructuring operation; (b) where restructuring aid follows the granting of rescue aid or temporary restructuring support as part of a single restructuring operation; (c) where rescue aid or temporary restructuring support has been granted in accordance with these guidelines and that aid was not followed by restructuring aid, if: (i) it could reasonably have been believed that the beneficiary would be viable in the long term when the aid pursuant to these guidelines was granted, and (ii) new rescue or restructuring aid or temporary restructuring support becomes necessary after at least five years due to unforeseeable circumstances for which the beneficiary is not responsible; (d) in exceptional and unforeseeable circumstances for which the beneficiary is not responsible.]
- (99) [F1Divestments to limit distortions of competition should take place without undue delay, taking into account the type of asset being divested and any obstacles to its disposal, and in any case within the duration of the restructuring plan.]
- (100) [F¹This could in particular include measures to open up certain markets directly or indirectly linked to the beneficiary's activities to other Union operators, in compliance with Union law. Such initiatives may replace other measures to limit distortions of competition that would normally be required of the beneficiary.]
- (101) [FIPlease see points 32 to 35 of the Guidelines for the types of measures covered by the notion of 'aid to cover the social costs of restructuring'.]
- (**102**) [F1OJ C 249, 31.7.2014, p. 1.]
- (103) [FIAs defined in Commission Recommendation 2003/361/EC of 6 May 2003 concerning the definition of micro, small and medium-sized enterprises (OJ L 124, 20.5.2003, p. 36).]
- (104) [F1'Smaller State-owned undertakings' are economic units with an independent power of decision that would qualify as small or medium-sized enterprises under Recommendation 2003/361/EC but for the fact that 25 % or more of the capital or voting rights are directly or indirectly controlled, jointly or individually, by one or more public bodies.]
- (105) [F1This refers in particular to the types of company mentioned in Annex I to Directive 2013/34/ EU of the European Parliament and of the Council of 26 June 2013 on the annual financial

- statements, consolidated financial statements and related reports of certain types of undertakings, amending Directive 2006/43/EC of the European Parliament and of the Council and repealing Council Directives 78/660/EEC and 83/349/EEC (OJ L 182, 29.6.2013, p. 19).]
- (106) [FIThis is the case when deduction of accumulated losses from reserves (and all other elements generally considered as part of the own funds of the company) leads to a negative cumulative amount that exceeds half of the subscribed share capital.]
- (107) [F1This refers in particular to the types of company mentioned in Annex II to Directive 2013/34/EU.]
- (108) [FIRestructuring may involve one or more of the following elements: the reorganisation and rationalisation of the beneficiary's activities to a more efficient basis, typically involving withdrawal from loss-making activities, restructuring of those existing activities that can be made competitive again and, possibly, diversification towards new and viable activities. It typically also involves financial restructuring in the form of capital injections by new or existing shareholders and debt reduction by existing creditors.]
- (109) [FLong-term viability is achieved when an undertaking is able to provide an appropriate projected return on capital after having covered all its costs including depreciation and financial charges. The restructured undertaking should be able to compete in the marketplace on its own merits.]
- (110) [FIThe alternative scenario should not involve State aid. It may concern, for example: debt reorganisation, asset disposal, private capital raising, sale to a competitor or break-up, in each case either through entry into an insolvency or reorganisation procedure or otherwise.]
- (111) [FICommunication from the Commission on the revision of the method for setting the reference and discount rates (OJ C 14, 19.1.2008, p. 6) or any future communication which might replace it.]
- (112) [FI In line with point 115(e) of the Guidelines, that plan need not contain all the elements set out in points 47 to 52 of the Guidelines, but must, as a minimum, identify the actions that the beneficiary must take to restore its long-term viability without State support.]
- (113) [FIFor example, where the aid to be granted enhances the beneficiary's equity position, the own contribution should similarly include measures that are equity-enhancing, such as raising fresh equity from incumbent shareholders, the write-down of existing debt and capital notes or the conversion of existing debt to equity, or the raising of new external equity on market terms.]
- (114) [F1For example where the State provides grants, injects capital or writes off debt.]
- (115) [FIPlease note where less than 10 years have elapsed since the rescue aid or temporary restructuring support was granted or the restructuring period came to an end or implementation of the restructuring plan was halted (whichever occurred the latest), further rescue aid, restructuring aid or temporary restructuring support can be only granted: (a) where temporary restructuring support follows the granting of rescue aid as part of a single restructuring operation; (b) where restructuring aid follows the granting of rescue aid or temporary restructuring support as part of a single restructuring operation; (c) where rescue aid or temporary restructuring support has been granted in accordance with these guidelines and that aid was not followed by restructuring aid, if: (i) it could reasonably have been believed that the beneficiary would be viable in the long term when the aid pursuant to these guidelines was granted, and (ii) new rescue or restructuring aid or temporary restructuring support becomes necessary after at least five years due to unforeseeable circumstances for which the beneficiary is not responsible; (d) in exceptional and unforeseeable circumstances for which the beneficiary is not responsible.]
- (116) [FI Including any such aid granted before the date from when the Commission applies the Guidelines, i.e. before 1.8.2014.]
- (117) [F1Member States are not obliged to require such measures from small enterprises, except where otherwise provided by rules on State aid in a particular sector. However, small enterprises should not normally increase their capacity during a restructuring period.]
- (118) [FIDivestments, write-offs and closure of loss-making activities which would at any rate be necessary to restore long-term viability will generally not be considered sufficient.]
- (119) [FISuch measures should take place in the market(s) where the beneficiary will have a significant market position after the restructuring, in particular those where there is significant excess capacity.]

- (120) [FIDivestments to limit distortions of competition should take place without undue delay, taking into account the type of asset being divested and any obstacles to its disposal, and in any case within the duration of the restructuring plan.]
- (121) [FIThis could in particular include measures to open up certain markets directly or indirectly linked to the beneficiary's activities to other Union operators, in compliance with Union law. Such initiatives may replace other measures to limit distortions of competition that would normally be required of the beneficiary.]
- (122) [FICommunication from the Commission on State aid for films and other audiovisual works (OJ C 332, 15.11.2013, p. 1).]
- (123) [FIEU Guidelines for the application of State aid rules in relation to the rapid deployment of broadband networks (OJ C 25, 26.1.2013, p. 1).]
- (124) [FIEUROPE 2020 A strategy for smart, sustainable and inclusive growth, COM(2010) 2020.]
- (125) [F1For details see points 56 to 60 of the Broadband Guidelines.]
- (126) [F1For details see points 82 to 85 of the Broadband Guidelines.]
- (127) [FISee for an example the Commission Guide to high speed broadband investment (http://ec.europa.eu/regional_policy/sources/docgener/presenta/broadband2011/broadband2011_en.pdf.]
- (128) [FI Directive 2014/23/EU of the European Parliament and of the Council of 26 February 2014 on the award of concession contracts (OJ L 94, 28.3.2014, p. 1), Directive 2014/24/EU of the European Parliament and of the Council of 26 February 2014 on public procurement and repealing Directive 2004/18/EC (OJ L 94, 28.3.2014, p. 65) and Directive 2014/25/EU of the European Parliament and of the Council of 26 February 2014 on procurement by entities operating in the water, energy, transport and postal services sectors and repealing Directive 2004/17/EC, (OJ L 94, 28.3.2014, p. 243).]
- (129) [FIFor instance, investment and business model selected, size and characteristics of the geographic intervention areas or measures to control projects costs.]
- (130) [F1The list of Commission decisions on State aid to broadband are available on DG Competition's homepage: http://ec.europa.eu/competition/sectors/telecommunications/broadband decisions.pdf.]
- (131) [FIOJ C 200, 28.6.2014, p. 1. For some details concerning the use of this supplementary notification sheet in agriculture and the fishery and aquaculture sectors see point 14 of the EEAG.]
- (132) [F1Commission Regulation (EU) No 651/2014 of 17 June 2014 declaring certain categories of aid compatible with the internal market in application of Articles 107 and 108 of the Treaty (OJ L 187, 26.6.2014, p. 1).]
- (133) [F1Commission Regulation (EU) No 1407/2013 of 18 December 2013 on the application of Articles 107 and 108 of the Treaty on the Functioning of the European Union to *de minimis* aid (OJ L 352, 24.12.2013, p. 1).]
- (134) [FISee section 3.7 of EEAG, aid in the form of reductions in or exemptions from environmental taxes and in the form of reductions in funding support for electricity from renewable sources.]
- (135) [FIWhen introducing a measure co-financed by the European Structural and Investments Funds, Member States may rely on the reasoning in the relevant Operational Programmes in indicating the environmental or energy objectives pursued.]
- (136) [FIThe European Network of Transmission System Operators for Electricity.]
- (137) [FIFor the purpose of demonstrating the appropriateness of schemes, the Member State can also rely on the results of past evaluations as described in Chapter 4 of the EEAG.]
- (138) [F1'Start of works' is defined in point 19(44) of the EEAG.]
- (139) [FIThat requirement is not applicable to SMEs and is without prejudice to the assessment of the incentive effect of State aid for energy-efficiency measure prescribed by or carried out as a result of the energy audit or those resulting from other tools (see points 56 and 57 of the EEAG).]

- (140) [F1For details see points 72 to 76 of the EEAG.]
- (141) [FIThe correct counterfactual is the cost of a technically comparable investment that provides a lower degree of environmental protection (corresponding to mandatory Union standards in force) which could possibly be carried out without aid. See point 73(b) of the EEAG. For instance see list provided in Annex 2 to the EEAG. In case the reference investment is not the one listed in Annex 2, please explain and justify its appropriateness.]
- (142) [F1Please note that this condition does not apply if the intangible asset is technically out of date.]
- (143) [FISee Annex 2 to the EEAG: The extra investment costs consist of the additional investment costs necessary to go beyond the level of environmental protection required by the Union standards.]
- (144) [FIPlease note that the cost of investments needed to reach the level of protection required by the Union standards is not eligible.]
- (145) [F1See point 78(a) of the EEAG.]
- (146) [F1See point 78(b) of the EEAG.]
- (147) [FI See point 78(c) of the EEAG. 'Eco-innovation' is defined in point 19(4) of the EEAG.]
- (148) [FIFor details of the genuinely competitive bidding process required, see the definition in point 19(43) of the EEAG.]
- (149) [FI Directive 2000/60/EC of the European Parliament and of the Council of 23 October 2000 establishing a framework for Community action in the field of water policy (OJ L 327, 22.12.2000, p. 1).]
- (150) [FIDirective 2009/28/EC of the European Parliament and of the Council of 23 April 2009 on the promotion of the use of energy from renewable sources and amending and subsequently repealing Directives 2001/77/EC and 2003/30/EC (OJ L 140, 5.6.2009, p. 16).]
- (151) [FIF or aid schemes the information can be provided in the form of atypical calculation (or several examples).]
- (152) [F1Please note that the Commission can authorise such notified measures for a period of 10 years.]
- (153) [F1The decision as to whether the aid is necessary will take account of the costs and revenue resulting from the production and sale of the electric power or heat.]
- (154) [FIThe production cost may include the plant's normal return on capital, but any gains by the undertaking in terms of heat production must be deducted from production costs.]
- (155) [FIPlease note that any investment aid granted to the undertaking in respect of the new plant must be deducted from production costs.]
- (156) [F1Please note that the duration must be limited to maximum 5 years.]
- (157) [F1See section 3.7.1 of the EEAG.]
- (158) [F1As set out in point 19(16) of the EEAG, 'Union minimum tax level' means the minimum level of taxation provided for in Union legislation. For energy products and electricity, the Union minimum tax level means the minimum level of taxation laid down in Annex I to Council Directive 2003/96/ EC of 27 October 2003 restructuring the Community framework for the taxation of energy products and electricity (OJ L 283, 31.10.2003, p. 51).]
- (159) [FIn this respect, Member States may provide estimations of, inter alia, the product price elasticity of the sector concerned in the relevant geographic market as well as estimates of lost sales and/or reduced profits for the companies in the sector/category concerned.]
- (160) [F1Guidelines on certain State aid measures in the context of the greenhouse gas emission allowance trading scheme post 2012 (OJ C 158, 5.6.2012, p. 4).]
- (161) [FIThe most direct way to demonstrate the causal link is by reference to a charge or levy on top of the electricity price, which is dedicated to the funding of energy from renewable sources. An indirect way to demonstrate the additional costs would be to calculate the impact of higher net costs

- for the electricity suppliers from green certificates and calculate the impact on the electricity price assuming the higher net costs are passed on by the supplier.]
- (162) [F1Guidelines on State aid to promote risk finance investments (OJ C 19, 22.1.2014, p. 4).]
- (163) [FICommission Regulation (EU) No 651/2014 of 17 June 2014 declaring certain categories of aid compatible with the internal market in application of Articles 107 and 108 of the Treaty (OJ L 187, 26.6.2014, p. 1).]
- (164) [F1Commission Regulation (EU) No 1407/2013 of 18 December 2013 on the application of Articles 107 and 108 of the Treaty on the Functioning of the European Union to *de minimis* aid (OJ L 352, 24.12.2013, p. 1).]
- (165) [F1Communication from the Commission on the revision of the method for setting the reference and discount rates (OJ C 14, 19.1.2008, p. 6).]
- (166) [F1Commission Notice on the application of Articles 87 and 88 of the EC Treaty to State aid in the form of guarantees (OJ C 155, 20.6.2008, p. 10).]
- (167) [F1The RFG (paragraphs 46 49) require that an *ex-ante* assessment be carried out and submitted for all notifiable risk finance measures.]
- (168) [F1 Regulation (EU) No 1303/2013 of the European Parliament and of the Council of 17 December 2013 laying down common provisions on the European Regional Development Fund, the European Social Fund, the Cohesion Fund, the European Agricultural Fund for Rural Development and the European Maritime and Fisheries Fund and laying down general provisions on the European Regional Development Fund, the European Social Fund, the Cohesion Fund and the European Maritime and Fisheries Fund and repealing Council Regulation (EC) No 1083/2006 (OJ L 347, 20.12.2013, p. 320).]
- (169) [FICommission Regulation (EU) No 651/2014 of 17 June 2014 declaring certain categories of aid compatible with the internal market in application of Articles 107 and 108 of the Treaty (OJ L 187, 26.6.2014, p. 1).]
- (170) [F1SWD(2014)179 final of 28.5.2014.]
- (171) [F1Beyond providing a general description of the objectives and eligibility rules of the scheme, the aim of this section is to assess how the eligibility and exclusion rules of the scheme may be used to identify the effect of aid. In some cases, the precise eligibility rules may not be known in advance. In those cases the best available expectations should be provided.]
- (172) [FIExamples of negative effects are regional and sectorial biases or crowding out of private investments induced by the aid scheme.]
- (173) [F1Aid schemes defined in Article 1(2)(a) of Regulation (EU) No 651/2014 are excluded from the scope of the Regulation six months after their entry into force. After having assessed the evaluation plan, the Commission may decide to extend the application of the Regulation to such schemes for a longer period. Member States are invited to precisely indicate the intended duration of the scheme.]
- (174) [F1Please make reference to SWD(2014)179 final of 28.5.2014.]
- (175) [F1Please note that the evaluation might require sourcing of both historical data and data that will become progressively available during the deployment of the aid scheme. Please identify the sources for both types of information. Both types of data should preferably be collected from the same source as to guarantee consistency across time.]
- (176) [F6OJ C 204, 1.7.2014, p. 1, as modified by OJ C 390, 24.11.2015, p. 4.]
- (177) [F6However, a requirement to have an establishment or branch in the Member State granting the aid at the time of payment of the aid is permitted.]
- (178) [F6Regulation (EU) No 1305/2013 of the European Parliament and of the Council of 17 December 2013 on support for rural development by the European Agricultural Fund for Rural Development (EAFRD) and repealing Council Regulation (EC) No 1698/2005 (OJ L 347, 20.12.2013, p. 487).]
- (179) [F6Please note that this second requirement does not apply in the case of fiscal successor schemes provided the activity was already covered by the previous schemes in the form of tax advantages.]

- (180) [F6Directive 2000/60/EC of the European Parliament and of the Council of 23 October 2000 establishing a framework for Community action in the field of water policy (OJ L 327, 22.12.2000, p. 1) ('the Water framework directive').]
- (181) [F6Regulation (EU) No 1307/2013 of the European Parliament and of the Council of 17 December 2013 establishing rules for direct payments to farmers under support schemes within the framework of the common agricultural policy and repealing Council Regulation (EC) No 637/2008 and Council Regulation (EC) No 73/2009 (OJ L 347, 20.12.2013, p. 608).]
- (182) [F6This information must be published within six months from the date of granting the aid (or, for aid in the form of tax advantage, within one year from the date of the tax declaration). In the case of unlawful aid, Member States will be required to ensure the publication of this information *ex post*, at least within a period of six months from the date of the Commission decision. The information must be available in a format which allows data to be searched, extracted, and easily published on the internet, for instance in CSV or XML format.]
- (183) [F6Publication of information on aid awards granted before 1 July 2016 and, for fiscal aid, publication for aid claimed or granted before 1 July 2016, will not be required.]
- (184) [F6Regulation (EU) No 1306/2013 of the European Parliament and of the Council of 17 December 2013 on the financing, management and monitoring of the common agricultural policy and repealing Coun cil Regulations (EEC) No 352/78, (EC) No 165/94, (EC) No 2799/98, (EC) No 814/2000, (EC) No 1290/2005 and (EC) No 485/2008 (OJ L 347, 20.12.2013, p. 549).]
- (185) [F6Directive 2009/28/EC of the European Parliament and of the Council of 23 April 2009 on the promotion of the use of energy from renewable sources and amending and subsequently repealing Directives 2001/77/EC and 2003/30/EC (OJ L 140, 5.6.2009, p. 16).]
- (186) [F6OJ C 200, 28.6.2014, p. 1.]
- (187) [F6Commission Regulation (EU) No 651/2014 of 17 June 2014 declaring certain categories of aid compatible with the internal market in application of Articles 107 and 108 of the Treaty (OJ L 187, 26.6.2014, p. 1).]
- (188) [F6Council Directive 91/676/EEC of 12 December 1991 concerning the protection of waters against pollution caused by nitrates from agricultural sources (OJ L 375, 31.12.1991, p. 1).]
- (189) [169] [169] [160]
- (190) [F6c processing of agricultural products' means any operation on an agricultural product resulting in a product which is also an agricultural product, except on-farm activities necessary for preparing an animal or plant product for first sale.]
- (191) [F64 marketing of agricultural products' means the holding or display with a view to sale, offering for sale, delivery or any other manner of placing on the market, except the first sale by a primary producer to resellers or processors and any activity preparing a product for such first sale; a sale by a primary producer to final consumers is considered as marketing of agricultural products if it takes place in separate premises reserved for that purpose.]
- (192) [F6Commission Regulation (EU) No 651/2014 of 17 June 2014 declaring certain categories of aid compatible with the internal market in application of Articles 107 and 108 of the Treaty (OJ L 187, 26.6.2014, p. 1).]
- (193) [F6OJ C 209, 23.7.2013, p. 1.]
- (194) [F6Commission Regulation (EC) No 794/2004 of 21 April 2004 implementing Council Regulation (EU) 2015/1589 laying down detailed rules for the application of Article 108 of the Treaty on the Functioning of the European Union (OJ L 140, 30.4.2004, p. 1).]
- (195) [F6OJ L 302, 1.11.2006, p. 10.]
- (196) [F6Council Regulation (EC) No 1217/2009 of 30 November 2009 setting up a network for the collection of accountancy data on the incomes and business operation of agricultural holdings in the European Union (OJ L 328, 15.12.2009, p. 27).]

- (197) [F6Commission Implementing Regulation (EU) 2015/220 of 3 February 2015 laying down rules for the application of Council Regulation (EC) No 1217/2009 setting up a network for the collection of accountancy data on the incomes and business operation of agricultural holdings in the European Union (OJ L 46, 19.2.2015, p. 1).]
- (198) [F6See the definition of SME's in point (35).13 of the Guidelines.]
- (199) [F6Directive 2000/60/EC of the European Parliament and of the Council of 23 October 2000 establishing a framework for Community action in the field of water policy (OJ L 327, 22.12.2000, p. 1).]
- (200) [F6Council Directive 92/43/EEC of 21 May 1992 on the conservation of natural habitats and of wild fauna and flora (OJ L 206, 22.7.1992, p. 7).]
- (201) [F6Directive 2009/147/EC of the European Parliament and of the Council of 30 November 2009 on the conservation of wild birds (OJ L 20, 26.1.2010, p. 7).]
- (202) [F6Council Regulation (EC) No 1698/2005 of 20 September 2005 on support for rural development by the European Agricultural Fund for Rural Development (EAFRD)(OJ L 277, 21.10.2005, p. 1).]
- (203) [F6Council Regulation (EC) No 834/2007 of 28 June 2007 on organic production and labelling of organic products and repealing Regulation (EEC) No 2092/91 (OJ L 189, 20.7.2007, p. 1).]
- (204) [F6Regulation (EU) No 1308/2013 of the European Parliament and of the Council of 17 December 2013 establishing a common organisation of the markets in agricultural products and repealing Council Regulations (EEC) No 922/72, (EEC) No 234/79, (EC) No 1037/2001 and (EC) No 1234/2007 (OJ L 347, 20.12.2013, p. 671).]
- (205) [F6Regulation (EU) No 1151/2012 of the European Parliament and of the Council of 21 November 2012 on quality schemes for agricultural products and foodstuffs (OJ L 343, 14.12.2012, p. 1).]
- (206) [F6Regulation (EC) No 110/2008 of the European Parliament and of the Council of 15 January 2008 on the definition, description, presentation, labelling and the protection of geographical indications of spirit drinks and repealing Council Regulation (EEC) No 1576/89 (OJ L 39, 13.2.2008, p. 16).]
- (207) [F6Regulation (EU) No 251/2014 of the European Parliament and of the Council of 26 February 2014 on the definition, description, presentation, labelling and the protection of geographical indications of aromatised wine products and repealing Council Regulation (EEC) No 1601/91 (OJ L 84, 20.3.2014, p. 14).]
- (208) [F6OJ C 341, 16.12.2010, p. 5.]
- (209) [F6Regulation (EC) No 1107/2009 of the European Parliament and of the Council of 21 October 2009 concerning the placing of plant protection products on the market and repealing Council Directives 79/117/EEC and 91/414/EEC (OJ L 309, 24.11.2009, p. 1).]
- (210) [F6Directive 2009/128/EC of the European Parliament and of the Council of 21 October 2009 establishing a framework for Community action to achieve the sustainable use of pesticides (OJ L 309, 24.11.2009, p. 71).]
- (211) [F6This applies to cooperation relating to the production of energy from renewable sources or the production of biofuels on holdings, provided that the conditions laid down in Section 1.1.1.1 of Part II of the Guidelines are complied with.]
- (212) [F6Regulation (EU) No 1303/2013 of the European Parliament and of the Council of 17 December 2013 laying down common provisions on the European Regional Development Fund, the European Social Fund, the Cohesion Fund, the European Agricultural Fund for Rural Development and the European Maritime and Fisheries Fund and laying down general provisions on the European Regional Development Fund, the European Social Fund, the Cohesion Fund and the European Maritime and Fisheries Fund and repealing Council Regulation (EC) No 1083/2006 (OJ L 347, 20.12.2013, p. 320).]
- (213) [F6In the case of ex-ante framework aid schemes, questions 2, 3, 4 and 8 are not applicable.]
- (214) [F6Council Directive 2000/29/EC of 8 May 2000 on protective measures against the introduction into the Community of organisms harmful to plants or plant products and against their spread within the Community (OJ L 169, 10.7.2000, p. 1).]

- (215) [F6Directive 2009/128/EC of the European Parliament and of the Council of 21 October 2009 establishing a framework for Community action to achieve the sustainable use of pesticides (OJ L 309, 24.11.2009, p. 71).]
- (216) [F6Regulation (EC) No 1107/2009 of the European Parliament and of the Council of 21 October 2009 concerning the placing of plant protection products on the market and repealing Council Directives 79/117/EEC and 91/414/EEC (OJ L 309, 24.11.2009, p. 1).]
- (217) [F6OJ L 189, 27.6.2014, p. 1.]
- (218) [F6OJ L 189, 27.6.2014, p. 1.]
- (219) [F6OJ L 347, 20.12.2013, p. 549.]
- (220) [F6Directive 2010/75/EU of the European Parliament and of the Council of 24 November 2010 on industrial emissions (integrated pollution prevention and control)(OJ L 334, 17.12.2010, p. 17).]
- (221) [F6Regulation (EU) No 1169/2011 of the European Parliament and of the Council of 25 October 2011 on the provision of food information to consumers, amending Regulations (EC) No 1924/2006 and (EC) No 1925/2006 of the European Parliament and of the Council, and repealing Commission Directive 87/250/EEC, Council Directive 90/496/EEC, Commission Directive 1999/10/EC, Directive 2000/13/EC of the European Parliament and of the Council, Commission Directives 2002/67/EC and 2008/5/EC and Commission Regulation (EC) No 608/2004 (OJ L 304, 22.11.2011, p. 18).]
- (222) [F6The description must reflect how the Member State intends to ensure that the aid is given only in respect of the extra cost of the transport of goods inside national borders, and is calculated on the basis of the most economical form of transport and the shortest route between the place of production of the agricultural products or processing of the agricultural products and commercial outlets, and cannot be given towards the costs of the transport of the agricultural products of businesses with an alternative location.]
- (223) [F6Please note that only measures falling with the definition of State aid and the rules concerning the interpretation of that definition set out in the Commission Notice on the notion of aid must be notified. In the case of uncertainty whether a measure complies with the conditions of State aid, it may be notified for the assessment of the European Commission. Forestry measures in Regulation (EU) No 1305/2013, in principle, are considered to meet all the criteria of State aid.]
- (224) [F6Regulation (EU) No 1305/2013 of 17.12.2013 (OJ L 347, 20.12.2013, p. 487).]
- (225) [F6In accordance with point (495) of the Guidelines, this exception is included in application of Article 5(5)(c), Article 21(1)(e) and Article 26 of Regulation (UE) No 1305/2013.]
- (226) [F6Communication from the Commission Guidelines on State aid for environmental protection and energy 2014-2020 (OJ C 200, 28.6.2014, p. 1).]
- (227) [F6Regulation (EU) No 229/2013 of the European Parliament and of the Council of 13 March 2013 laying down specific measures for agriculture in favour of the smaller Aegean islands and repealing Council Regulation (EC) No 1405/2006 (OJ L 78, 20.3.2013, p. 41–50).]
- (228) [F6Second Ministerial Conference on the protection of Forests in Europe, 1617 June 1993, Helsinki/ Finland, 'Resolution H1- General Guidelines for the Sustainable Management of Forests in Europe'.]
- (229) [F6For aid as to setting up advisory services and training of advisors in the rural areas, please fill in the forms related to Sections 3.3 and 3.6 of Part II of the Guidelines.]
- (230) [F6Regulation (EC) No 1107/2009 of the European Parliament and of the Council of 21 October 2009 concerning the placing of plant protection products on the market and repealing Council Directives 79/117/EEC and 91/414/EEC, (OJ L 309 24.11.2009, p. 1).]
- (231) [F6Directive 2009/128/EC of the European Parliament and of the Council of 21 October 2009 establishing a framework for Community action to achieve the sustainable use of pesticides (OJ L 309, 24.11.2009, p. 71).]
- (232) [F7[X1OJ C 319, 27.12.2006, p. 1.]]

- (233) [F7 XI Council Regulation (EC) No 1698/2005 of 20 September 2005 on support for rural development by the European Agricultural Fund for Rural Development (EAFRD) (OJ L 277, 21.10.2005, p. 1).]]
- (234) [F7[X1OJ L 109, 6.5.2000, p. 29.]]
- (235) [F7]XI Council Regulation (EC) No 2702/1999 of 14 December 1999 on measures to provide information on, and to promote, agricultural products in third countries (OJ L 327, 21.12.1999, p. 7)]]
- (236) [F⁷[X¹Council Directive 2003/96/EC of 27 October 2003 restructuring the Community framework for the taxation of energy products and electricity (OJ L 283, 31.10.2003, p. 51).]]
- (237) [F7[X1OJ C 319, 27.12.2006, p. 1.]]
- (238) [F7[XI] Council Regulation (EC) No 1698/2005 of 20 September 2005 on support for rural development by the European Agricultural Fund for Rural Development (EAFRD) (OJ L 277, 21.10.2005, p. 1.)]]
- (239) [F1Guidelines on State aid to airports and airlines (OJ C 99, 4.4.2014, p. 3).]
- (240) [FIThe costs relating to the investments in airport infrastructure, including planning costs, ground handling infrastructure (such as baggage belts, etc.) and airport equipment can be eligible as investment aid to airports. Investment costs relating to non-aeronautical activities (in particular parking, hotels, restaurants, and offices) are ineligible. The investment costs relating to the provision of ground handling services (such as buses, vehicles, etc.) are ineligible, insofar as they are not part of ground handling infrastructure.]
- (241) [F1*Catchment area of an airport' means a geographic market boundary that is normally set at around 100 kilometres or around 60 minutes travelling time by car, bus, train or high-speed train; however, the catchment area of a given airport may be different and needs to take into account the specificities of each particular airport. The size and shape of the catchment area varies from airport to airport, and depends on various characteristics of the airport, including its business model, location and the destinations it serves.]
- (242) [FIPlease use the following categories of sizes: airports with annual passenger traffic up to 200 000 passengers; airports with annual passenger traffic of between 200 000 and 1 million; airports with annual passenger traffic of 1–3 million.]
- (243) [F1For example, direct grants, exemptions or reductions in taxes, social security or other compulsory charges, or the supply of land, goods or services at favourable prices, etc.]
- (244) [FIFor example, low-interest loans or interest rebates, state guarantees, the purchase of a shareholding or an alternative provision of capital on favourable terms.]
- (245) [F¹The capital cost funding gap is the difference between the positive and negative cash flows including investment costs into fixed capital assets over the lifetime of the investment in net present value terms.]
- (246) [F1Guidelines on State aid to airports and airlines (OJ C 99, 4.4.2014, p. 3).]
- (247) [F1'Operating funding gap' means the operating losses of an airport over the relevant period, discounted to their current value using the cost of capital, that is to say the shortfall (in Net Present Value terms) between airport revenues and operating costs of the airport.]
- (248) [F1Balance sheet, profit and loss statement, statement of chartered accountant or audit firm.]
- (249) [F1Services provided by an airport or any of its subsidiaries, to ensure the handling of aircraft, from landing to take-off, and of passengers and freight, so as to enable airlines to provide air transport services, including the provision of ground handling services and the provision of centralised ground handling infrastructure.]
- (250) [FIUnderlying costs of an airport in respect of the provision of airport services, including cost categories such as cost of personnel, contracted services, communications, waste, energy, maintenance, rent and administration, but excluding the capital costs, marketing support or any other incentives granted to airlines by the airport, and costs falling within a public policy remit.]

- (251) [Fi Catchment area of an airport' means a geographic market boundary that is normally set at around 100 kilometres or around 60 minutes travelling time by car, bus, train or high-speed train; however, the catchment area of a given airport may be different and needs to take into account the specificities of each particular airport. The size and shape of the catchment area varies from airport to airport, and depends on various characteristics of the airport, including its business model, location and the destinations it serves.]
- (252) [F1For example, direct grants, exemptions or reductions in taxes, social security or other compulsory charges, or the supply of land, goods or services at favourable prices, etc.]
- (253) [FIFor example, low-interest loans or interest rebates, State guarantees, the purchase of a shareholding or an alternative provision of capital on favourable terms.]
- (254) [F1'Operating funding gap' means the operating losses of an airport over the relevant period, discounted to their current value using the cost of capital, that is to say the shortfall (in Net Present Value terms) between airport revenues and operating costs of the airport.]
- (255) [FI Guidelines on State aid to airports and airlines (OJ C 99, 4.4.2014, p. 3).]
- (256) [F1'High-speed rail service' is a service provided with trains capable of reaching speeds of over 200 km/h.]
- (257) [F1 Catchment area of an airport' means a geographic market boundary that is normally set at around 100 kilometres or around 60 minutes travelling time by car, bus, train or high-speed train; however, the catchment area of a given airport may be different and needs to take into account the specificities of each particular airport. The size and shape of the catchment area varies from airport to airport, and depends on various characteristics of the airport, including its business model, location and the destinations it serves.]
- (258) [FI Actual average annual passenger traffic during the two financial years preceding that in which the aid is notified or actually granted or paid in the case of non-notified aid. In the case of a newly created passenger airport, the forecasted average annual passenger traffic during the two financial years after the beginning of the operation of commercial passenger air traffic should be considered. These thresholds refer to a one-way count. As a result a passenger flying for example to the airport and back would be counted twice; the count applies to individual routes.]
- (259) [FIDecision 2006/682/EC of the Council and of the Representatives of the Member States of the European Union meeting within the Council on the signature and provisional application of the Multilateral Agreement between the European Community and its Member States, the Republic of Albania, Bosnia and Herzegovina, the Republic of Bulgaria, the Republic of Croatia, the former Yugoslav Republic of Macedonia, the Republic of Iceland, the Republic of Montenegro, the Kingdom of Norway, Romania, the Republic of Serbia and the United Nations Interim Administration Mission in Kosovo on the Establishment of a European Common Aviation Area (ECAA) (OJ L 285, 16.10.2006, p. 1).]
- (260) [FIFor example, low-interest loans or interest rebates, State guarantees, the purchase of a shareholding or an alternative provision of capital on favourable terms.]
- (261) [F1 Eligible costs are the airport charges in respect of the route.]
- (262) [FI Guidelines on State aid to airports and airlines (OJ C 99, 4.4.2014, p. 3).]
- (263) [FIWhere the route concerned links remote regions, such as outermost regions, islands, and sparsely populated areas, the aid could cover the entire population of that region.]
- (264) [FISuch as outermost regions, islands and sparsely populated areas.]
- (265) [F1Community guidelines on State aid to maritime transport (OJ C 13, 17.1.2004, p. 3).]
- (266) [F8(1)* Communication from the Commission Guidelines for the examination of State aid to the fishery and aquaculture sector (OJ C 217, 2.7.2015, p. 1).
- (267) Regulation (EU) No 508/2014 of the European Parliament and of the Council of 15 May 2014 on the European Maritime and Fisheries Fund and repealing Council Regulations (EC) No 2328/2003, (EC) No 861/2006, (EC) No

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1198/2006 and (EC) No 791/2007 and Regulation (EU) No 1255/2011 of the European Parliament and of the Council (OJ L 149, 20.5.2014, p. 1).

- The objectives of the CFP are set out in Article 2 of Regulation (EU) No 1380/2013 of the European Parliament and the Council of 11 December 2013 on the Common Fisheries Policy, amending Council Regulations (EC) No 1954/2003 and (EC) No 1224/2009 and repealing Council Regulations (EC) No 2371/2002 and (EC) No 639/2004 and Council Decision 2004/585/EC (OJ L 354, 28.12.2013, p. 22).
- Point (22)(1) of the Guidelines defines operating aid.
- The information must be published within six months from the date of granting the aid (or, for aid in the form of tax advantage, within one year from the date when the tax declaration is due). In the case of unlawful aid, Member State must publish the information ex post, at least within a period of six months from the date of the Commission decision. The information must be available in a format which allows data to be searched, extracted, and easily published on the internet, for instance in CSV or XML format.
- Publication will not be required for aid awards granted before 1 July 2017 and, for fiscal aid, publication will not be required for aid claimed or granted before 1 July 2017.
- Commission Regulation (EU) No 1388/2014 of 16 December 2014 declaring certain categories of aid to undertakings in the production, processing and marketing of fishery and aquaculture products compatible with the internal market in application of Articles 107 and 108 of the Treaty on the Functioning of the European Union (OJ L 369 24.12.2014, p. 37).
- (273) [F8(8)* Horizontal guidelines and other instruments include, amongst others, Communication from the Commission Criteria for the analysis of the compatibility of State aid for training subject to individual notification (OJ C 188, 11.8.2009, p. 1); Communication from the Commission Guidelines on State aid to promote risk finance investments (OJ C 19, 22.1.2014, p. 4); the Communication from the Commission Framework for State aid for research and development and innovation (OJ C 198, 27.6.2014, p. 1); Communication from the Commission Guidelines on State aid for environmental protection and energy 2014-2020 (OJ C 200, 28.6.2014, p. 1); the Communication from the Commission Guidelines on State aid for rescuing and restructuring non-financial undertakings in difficulty (OJ C 249, 31.7.2014, p. 1).
- (274) [F8(9)* Regulation (EU) No 652/2014 of the European Parliament and of the Council of 15 May 2014 laying down provisions for the management of expenditure relating to the food chain, animal health and animal welfare, and relating to plant health and plant reproductive material, amending Council Directives 98/56/EC, 2000/29/EC and 2008/90/EC, Regulations (EC) No 178/2002, (EC) No 882/2004 and (EC) No 396/2005 of the European Parliament and of the Council, Directive 2009/128/EC of the European Parliament and of the Council and Regulation (EC) No 1107/2009 of the European Parliament and of the Council and repealing Council Decisions 66/399/EEC, 76/894/EEC and 2009/470/EC (OJ L 189, 27.6.2014, p. 1).
- (275) [F8(10)* Council Directive 2006/88/EC of 24 October 2006 on animal health requirements for aquaculture animals and products thereof, and on the prevention and control of certain diseases in aquatic animals (OJ L 328, 24.11.2006, p. 14).
- (276) [F1t is the year in which the data are requested.]
- (277) [FINACE Rev. 2, or subsequent legislation amending or replacing it; NACE is the statistical classification of economic activities in the European Union as laid down in Regulation (EC) No

Changes to legislation: There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004. (See end of Document for details)

1893/2006 of the European Parliament and of the Council of 20 December 2006 establishing the statistical classification of economic activities NACE Revision 2 and amending Council Regulation (EEC) No 3037/90 as well as certain EC Regulations on specific statistical domains (OJ L 393, 30.12.2006, p. 1).]

- (278) [F1t is the year in which the data are requested.]
- (279) [FINACE Rev. 2, or subsequent legislation amending or replacing it; NACE is the statistical classification of economic activities in the European Union as laid down in Regulation (EC) No 1893/2006 of the European Parliament and of the Council of 20 December 2006 establishing the statistical classification of economic activities NACE Revision 2 and amending Council Regulation (EEC) No 3037/90 as well as certain EC Regulations on specific statistical domains (OJ L 393, 30.12.2006, p. 1).]

Editorial Information

X1 Inserted by Corrigendum to Commission Regulation (EC) No 1935/2006 of 20 December 2006 amending Regulation (EC) No 794/2004 implementing Council Regulation (EC) No 659/1999 laying down detailed rules for the application of Article 93 of the EC Treaty (Official Journal of the European Union L 407 of 30 December 2006).

Textual Amendments

- F1 Substituted by Commission Regulation (EU) 2015/2282 of 27 November 2015 amending Regulation (EC) No 794/2004 as regards the notification forms and information sheets (Text with EEA relevance).
- **F6** Substituted by Commission Regulation (EU) 2016/246 of 3 February 2016 amending Annex I to Regulation (EC) No 794/2004 as regards the forms to be used for the notification of State aid in the agricultural and forestry sectors and in rural areas.
- F7 Inserted by Commission Regulation (EC) No 1935/2006 of 20 December 2006 amending Regulation (EC) No 794/2004 implementing Council Regulation (EC) No 659/1999 laying down detailed rules for the application of Article 93 of the EC Treaty.
- **F8** Substituted by Commission Regulation (EU) 2016/2105 of 1 December 2016 amending Annex I to Regulation (EC) No 794/2004 as regards the form to be used for the notification of State aid to the fishery and aquaculture sector.

Status:

Point in time view as at 31/01/2020.

Changes to legislation:

There are currently no known outstanding effects for the Commission Regulation (EC) No 794/2004.